Abstract

Driven by fundamental technological achievements like the digital revolution in communication and computing and the miniaturization of electronic components, a growing interest in the field of networked systems has appeared, in recent years, in different scientific areas like physics, communication and control engineering, economy, and lately, mathematics. Besides differences among the various approaches, the basic model consists of a certain number of agents, namely, systems living in a common environment and which communicate among each other according to some pre-specified communication pattern. Such a pattern may be fixed or varying according to their physical position, to their internal state and can possibly be affected by noises. The evolution law of each system typically depends on the information obtained through its communication links and is, in general, modeled by a difference or differential equation. For this type of models, the interest is, in general, in studying time evolution, asymptotic behaviors, and in formulating and solving related control problem. In particular, a fundamental issue is understanding the emerging of a group behavior from the individual dynamics and the communication pattern. While many models have already been proposed, the mechanisms which determine how the various individual actions get reflect in the group behavior are, in general, hard to understand and, up to now, few results are known.

In this thesis we will focus on the so-called *consensus problem*, where the group of agents has to reach an agreement on key pieces of information or on a common decision (represented by scalar or vector values) that enable them to cooperate in a coordinate fashion. We will consider a standard algorithm proposed in literature to solve this problem. We will provide some theoretical developments by

- characterizing the speed of convergence of this algorithm for particular communication pattern exhibiting symmetries,
- investigating the realistic and practical situation in which the systems can communicate each other only through digital channels and hence can exchange only quantized information,
- proposing the application of this algorithm to a problem of distributed estimation.

Acknowledgements

Completing my graduate studies has been an extremely enriching and rewarding experience both under a scientific and a human point of view. My doctorate is a team achievement, and in the next paragraphs I want to thank the people that contributed to this accomplishment.

First and foremost, I would like to thank my advisor prof. Sandro Zampieri. Sandro introduced me to the present wonders of control theory and managed to keep me focused all these years. His advice and words of wisdom always came at the right time. The constant confidence he placed in me, his encouragement and his support at all levels have been invaluable.

I particularly wish to express my sincere gratitude to my co-workers: to prof. Fabio Fagnani for his contagious creativity and enthusiasm, to Paolo Frasca, Alberto Speranzon and Jean Charles Delvenne for the days spent together in Padua, to prof. Alessandro Chiuso and prof. Luca Schenato for the uncountable stimulating and thought-provoking discussions I had with them. It has truly been a pleasure to colloborate with them over the years.

I would like to express my sincere gratitude to Francesco Bullo for having extended to me an invitation and an opportunity to work side by side at Santa Barbara University. With him I had the pleasure of sharing inspiring conversations dealing with subjects as motion coordination algorithms and robotic networks. In fact I want to emphasize how instrumental our collaborations have been.

I have been honored to share the lab with great researches and wonderful people: their support, acceptance, help and friendship have been fundamental part of this experience. Ale, Paolo, Eugenio, Luca, Maura, Francesco, Giuseppe, Federico, Stefano, Simone, Martina, Mirko, thank you all and to everybody else who has been a part of our research group. I also want to thank prof. Giovanni Marchesini, prof. Alessandro Beghi, prof. Elena Valcher, prof. Augusto Ferrante who made my life as a grad student much easier and smooth.

During these years I shared countless wonderful moments and enriching experiences outside the lab with people that eventually became my "extended family". I would like to take a moment to mention my great group of friends. Paolo and Romi, Ilaria and Michele, Laura, Monica, Francesco, Valeria, Gregory, Davide, Elena, Selene and everyone whose name may have slipped the controls of my keyboard. Your friendship has been invaluable during the first twenty-nine years of my life. I am forever indebted to my brothers Giovanni and Alberto for their continuous support and encouragement. Thanks to my grandmothers Virginia and Maria, to my grandfathers Giovanni and Gio' Maria for having been always present in my life.

My last thought goes to my parents, Giacomo and Luigia. To your teachings, guidance, patience and support to ensure I could have the best possible education. You have always had a firm belief in me. *Vi ringrazio con tutto il cuore.*

Contents

| 1 | Intr | oduction | 9 |
|----------|-----------------------|--|-----------|
| | 1.1 | Overview and contributions of the Thesis | 16 |
| 2 | The | e Consensus Problem | 21 |
| | 2.1 | Introduction | 21 |
| | 2.2 | Problem Formulation | 21 |
| | | 2.2.1 Convergence Analysis of Consensus Algorithms | 25 |
| | | 2.2.2 Consensus Synthesis | 29 |
| | 2.3 | Conclusion | 32 |
| 3 | The | symmetries in the consensus problem | 33 |
| | 3.1 | Introduction | 33 |
| | 3.2 | Problem formulation | 34 |
| | | 3.2.1 Circulant symmetries | 35 |
| | 3.3 | Cayley symmetries | 40 |
| | | 3.3.1 Cayley stochastic matrices on finite Abelian groups | 42 |
| | | 3.3.2 The essential spectral radius of Cayley matrices | 43 |
| | 3.4 | Conclusions | 51 |
| 4 | Ran | adomized Consensus Algorithms | 53 |
| | 4.1 | Introduction | 53 |
| | 4.2 | Problem formulation | 54 |
| | 4.3 | Conditions for the probabilistic consensus | 56 |
| | | 4.3.1 Time-varying strategy with bounded in-degree | 57 |
| | | 4.3.2 Time-varying Cayley graphs strategy | 58 |
| | 4.4 | The measures of the performance | 59 |
| | | 4.4.1 Mean square performance for time-varying with bounded | |
| | | in-degree strategy | 60 |
| | | 4.4.2 Mean square performance for time-varying Cayley graphs | |
| | | strategy | 65 |
| | 4.5 | Symmetric Gossip algorithms | 69 |
| | | | |

| | | 4.5.1 Analysis of the Symmetric Gossip Algorithm | 72 |
|----------|------------|---|------------|
| | | 4.5.2 Mean-square Analysis for the Cayley Graphs | 76 |
| | 4.6 | Conclusions | 79 |
| 5 | Qua | antized Consensus: Time-Invariant case | 81 |
| | 5.1 | Introduction | 81 |
| | 5.2 | Problem formulation | 83 |
| | 5.3 | Time Invariant case- Deterministic quantizers | 91 |
| | | 5.3.1 Bounded error model-Worst case analysis | 96 |
| | | 5.3.2 Probabilistic model \ldots | 109 |
| | 5.4 | Time Invariant case- Probabilistic quantizers | 115 |
| | | 5.4.1 An alternative solution to the consensus problem | 121 |
| | 5.5 | Conclusions | 121 |
| 6 | Qua | antized Consensus: Gossip Algorithms | 23 |
| | 6.1 | Introduction | 123 |
| | 6.2 | Quantized Symmetric Gossip Algorithms | 124 |
| | 6.3 | Quantized symmetric gossip algorithms via deterministic quan- | 100 |
| | | | 126 |
| | | 6.3.1 Partially quantized strategy | 127 |
| | | 6.3.2 Globally quantized strategy | 133 |
| | | 0.3.3 Some considerations on the speed of convergence of the | |
| | | giobany quantized strategy and of the partiany quan- | 126 |
| | 64 | Quantized symmetric gossin algorithms via probabilistic guan | 100 |
| | 0.4 | tizers | 138 |
| | | 6.4.1 Partially quantized strategy | 139 |
| | | 6.4.2 Globally quantized strategy | 144 |
| | 6.5 | Conclusions | 148 |
| _ | 0.0 | | 1 10 |
| 7 | Qua | antized average consensus via dynamic coding/decoding | 1 1 1 |
| | 5CH0 | Introduction | 151 |
| | 1.1 7.9 | Problem Formulation | 152 |
| | 7.2 | Coder/decoder pairs for digital shappols | 154 |
| | 1.5 | 7.3.1 Zoom in zoom out uniform order | 154 |
| | | 7.3.2 Logarithmic coder | 155 157 |
| | 7 / | Consensus algorithm with exchange of quantized information | 159 158 |
| | 1.4 | 7.4.1 Convergence analysis | 163 163 |
| | 7.5 | Exponential convergence | 167 |
| | 7.6 | Numerical simulations | 174 |
| | 1.0 | | - I I |

| | 7.7 | Conclusions | 177 |
|------------------|--|--|---|
| 8 | Distributed Kalman filtering | | |
| | 8.1 | Introduction | 179 |
| | 8.2 | Problem formulation | 183 |
| | 8.3 | Optimal consensus matrix P for fixed gain ℓ | 189 |
| | | 8.3.1 Undirected communication graph \mathcal{G} | 190 |
| | | 8.3.2 Fast communication $(m \to \infty)$ | 192 |
| | | 8.3.3 Large gain $(\ell \to 1)$ | 193 |
| | 8.4 | Optimal gain ℓ for fixed consensus matrix P | 194 |
| | 8.5 | Joint optimization of P and ℓ : special cases | 199 |
| | | 8.5.1 Fast communication $(m \to \infty)$ | 199 |
| | | 8.5.2 Small measurement noise $(r/q \rightarrow 0)$ | 201 |
| | | 8.5.3 High measurement noise $(q/r \rightarrow 0)$ | 204 |
| | 8.6 | Illustrative examples | 206 |
| | 8.7 | Conclusions | 213 |
| | | | |
| 9 | Con | clusions | 215 |
| 9 A | Con Har | clusions monic analysis on finite group | $\frac{215}{219}$ |
| 9 A | Con Har A.1 | nclusions monic analysis on finite group Introduction | 215 219 |
| 9 A | Con Har A.1 A.2 | monic analysis on finite group Introduction | 215 219 219 219 |
| 9 A | Con Har A.1 A.2 A.3 | monic analysis on finite group Introduction | 215 219 219 219 222 |
| 9 A | Con Har A.1 A.2 A.3 A.4 | monic analysis on finite group Introduction Finite Abelian groups Group characters Fourier transform on groups | 215 219 219 219 222 226 |
| 9 A B | Con Har A.1 A.2 A.3 A.4 Gra | monic analysis on finite group Introduction Finite Abelian groups Group characters Fourier transform on groups ph Theory | 215 219 219 219 222 226 229 |
| 9 A B | Con Har A.1 A.2 A.3 A.4 Gra B.1 | monic analysis on finite group Introduction Finite Abelian groups Group characters Fourier transform on groups ph Theory Graph Theory | 215 219 219 222 226 229 |
| 9 A B C | Con Har A.1 A.2 A.3 A.4 Gra B.1 Som | monic analysis on finite group Introduction Finite Abelian groups Group characters Group characters Fourier transform on groups ph Theory Graph Theory Graph Theory And Algebraic results | 215 219 219 222 226 229 229 231 |
| 9 A B C | Con Har A.1 A.2 A.3 A.4 Gra B.1 Som C.1 | monic analysis on finite group Introduction . Finite Abelian groups . Group characters . Fourier transform on groups . ph Theory . Graph Theory . Stability of discrete time linear parameter varying (LPV) sys- | 215 219 219 222 226 229 231 |
| 9 A B C | Con Har A.1 A.2 A.3 A.4 Gra B.1 Som C.1 | monic analysis on finite group Introduction | 215 219 219 222 226 229 229 231 231 |
| 9 A B C | Con Har A.1 A.2 A.3 A.4 Gra B.1 C.1 C.2 | monic analysis on finite group Introduction | 215 219 219 222 226 229 229 231 233 |

7

Chapter 1

Introduction

In recent years there has been an increasing interest in studying dynamical systems which can be modeled as an interconnection of a large number of identical subsystems [98, 141, 2, 50]. Typically the subsystems exhibit simple dynamics while the overall behavior of the aggregated dynamical units has the complexity features depending on the way the interconnection is built up. The scientific fields where these models appear are manifold.

On the one hand, interesting dynamical systems arise in biological networks at multiple levels of resolution, from interactions among molecules and cells [87] to the behavioral ecology of animal groups [95]. The brain is a network of nerve cells connected by axons, while cells are networks of molecules connected by biochemical reactions. Flocks of birds and schools of fish can travel in formation and act as one unit (see [105] and Figures 1.2 and 1.1), allowing these animals to defend themselves against predator and protect their territories. Wildebeest and other animals exhibit complex collective behaviors when migrating, such as obstacle avoiding, leader election, and formation keeping (see [126, 66] and Figure 1.3). Certain foraging behaviors include animals partitioning their environment into nonverlapping zones (see [12] and Figure 4). Honey bees [122], gorillas [133], and whitefaced capuchins [17] exhibit synchronized group activities such as initiation of motion and change of travel direction.

All these social animal behavior has been investigated since many years: see [95, 105] and [126, 66, 12, 122, 133, 17] for specific examples of animal species and [38] and [41] for general studies. Remarkably, it has been noticed that all the above coherent behaviors (individuals typically maintain a fixed distance from neighbors) are obtained by individuals following simple decentralized rules and often with clear bio-mechanical limits to locomotion and



Figure 1.1: School of fish. Groups of animals can act as one unit apparently without following a group leader.



Figure 1.2: Flock of snow geese. Self-organized behaviors emerge in biological groups, even though no individual has global knowledge of the group state. Snow geese fly in formation during migration.

information exchange (an individual changes its trajectory only on the basis of the behavior of its closest neighbors). In other words, these coordination behaviors emerge despite the fact that each individual lacks global knowledge of the network state and can plan its motion by observing only its closest neighbors.



Figure 1.3: Herd of wildebeest in the Serengeti National Park, Tanzania. Wildebeest and other animals exhibit complex coordinated behaviors when migrating, such as obstacle avoiding, leader election, and formation keeping.

At the same time, recently, technological advances in communication and computation have paved the way for the development of large scale multiagent systems, in particular groups of embedded systems, such as multivehicle and sensor networks. The ambitious goal in this context is to determine local individual rules which under a certain communication exchange pattern make it possible to reach certain global goals. In particular in few years, groups of autonomous agents with computing, communication, and mobility capabilities are expected to become economically feasible and perform a variety of spatially distributed sensing tasks, such as search and rescue, surveillance, environmental monitoring, and exploration.

In general, there are technological limitations on the amount of computation and communication that each unit can perform and the solution will have to take into consideration these constraints. This has stimulated the growth of a new area in control theory known has cooperative control: many identical input/output subsystems are connected through a communication network and each individual has at disposal, at every instant, the information on the state of only a limited number of other units and must implement a local feedback rule only based of this information. Different models are possible depending on the choice of the communication pattern: the set of neighbor systems with which each system exchange information can be fixed (a typically wired situation) or may be time-varying depending on the mutual positions (a typically wireless situation). Moreover, this information can be exact, but often instead quantized if we are using a digital channel with limited capacity and also possibly corrupted by noise. Moreover, there may be exhogeneous disturbances in the system. The control goals can be of classical type (as stabilization, rendezvous, tracking of a reference trajectory) or of new type (as shape formation), often the goal can be multi-objective: maintaining formation while moving in a certain direction and avoiding obstacles.

A fundamental issue both on the 'natural' and 'artificial' side is to quantitatively explain the emerging of group behavior from the individual dynamics and communication pattern (experimentally deduced in biology or chosen in artificial systems). Though many models have already been proposed, the mechanisms explaining how the group behavior arises from the various individual actions are, in general, hard to understand and, up to now, few results are known. In the cooperative control context however other issues show up. We want in this case to be able to explicitly synthesize control laws for which provable desired behaviors can be achieved and also understand which are the fundamental limitations: indeed a trade-off between performance of the overall system measured through some suitable cost functional and the amount of exchanged information allowed (capacity of the digital channels, number of individual connections, etc.) is expected.

A first intermediate goal toward a comprehensive understanding of how the dynamics of the individual systems can give rise to a group behavior, is to study in full depth some simple problem for N units whose state evolution is described by a simple difference or differential equation (of degree 1 or 2 typically). In this thesis, we will focus on the so-called *consensus problem* where a group of systems must communicate with its neighbors to reach agreement (consensus) on key pieces of information or on a common decision (represented by scalar or vector values) that enable them to cooperate in a coordinate fashion. In general, the systems start with some different initial decisions and communicate between them locally, under some constraints on connectivity and inter-agent information exchange. The averaging problem is a special case in which the goal is to compute the exact average of the initial values of the systems. A natural and widely studied consensus algorithm, proposed and analyzed by Tsitsiklis [138] and Tsitsiklis and al. [139], involves, at each time step, every system taking a weighted average of its own value with values received from some of the other systems. To be more precise, we consider a set V of N agents, numbered 1 through N, which will henceforth be referred to as *nodes* (we will see later that the nodes can represent different objects, autonomous agents, sensors, processors, vehicles, clocks according to the different application we are considering). Each node

i starts with a scalar value $x_i(0)$. At each nonnegative integer time *t*, node *i* receives from some of the other nodes *j* a message with the value of $x_j(t)$, and updates its value according to:

$$x_i(t+1) = \sum_{j=1}^{N} P_{ij}(t) x_j(t)$$

where the $P_{ij}(t)$ are weights with the property that $P_{ij}(t) \neq 0$ only if node i receives information from node j at time t. We use the notation P(t) to denote the weighted matrix $\{P_{ij}(t)\}_{i,j=1,\dots,N}$. Given a matrix P, let $\mathcal{E}(P)$ denote the set of directed edges (j, i), including self-edges (i, i), such that $P_{ij} \neq 0$. At each time t, the nodes' connectivity can be represented by the directed graph $\mathcal{G}(t) = (V, \mathcal{E}(P(t)))$. The goal is to study the convergence of the iterates $x_i(t)$ to some common value \bar{x} , as t approaches infinity. In the case of the average consensus problem this value \bar{x} is the average of the initial values, $\bar{x} = x_{ave} := \frac{1}{N} \sum_{i=1}^{N} x_i(0)$. In this thesis we will focus particularly on the average consensus algorithm.

The consensus problem has been massively addressed in the last years by many researchers. Indeed the consensus problem arises in a number of applications including distributed algorithms in Multi-vehicle Cooperative Control, information processing in sensor networks, load balancing, and distributed optimization (e.g. agreeing on the estimates of some unknown parameters). In the following, we will briefly review the applications of the algorithms of consensus.

Distributed Algorithms in Multi-Vehicle Cooperative control

The recent advances in miniaturizing of computing, communication, sensing, and actuation, that we have already cited in the Abstract and along the Introduction, have made it feasible to envision large numbers of autonomous vehicles (air, ground and water) working cooperatively to accomplish an objective. Compared to autonomous vehicles that perform solo missions, greater efficiency and operational capability can be realized from teams of autonomous vehicles operating in a coordinated fashion. Cooperative control of multivehicle systems has potential impact in numerous civilian, homeland security, and military applications. Potential civilian applications include monitoring forest fires, oil fields, pipelines, and tracking wildlife. Potential homeland security applications include border patrol and monitoring the perimeter of nuclear power plants. For the military, applications include surveillance, reconnaissance, and battle damage assessment. To enable these applications, various cooperative control capabilities need to be developed, including formation control, rendezvous, attitude alignment, flocking, foraging, task and role assignment, payload transport, air traffic control and cooperative search. Many of these capabilities require that the team of autonomous vehicles reach the agreement on key pieces of information or on a common decision in order to operate in a coordinating fashion. This agreement can be obtain by the standard consensus algorithm, briefly described above.

Load Balancing

In load balancing [44] the nodes are processors, computers, and the edges as physical connections among them. The corresponding communication graph presents in general some nice symmetry (e.g. a line, a ring, a torus, a hypercube, etc) and also symmetry with respect to communication exchange (if *i* and *j* are connected by an edge, it means that *i* can send data to *j* and viceversa). In most situations the communication graph is fixed. The measure x_i at each node is in this case the number of tasks which the processor *i* has to accomplish. The idea is that, in order to speed up the whole computation, processors should exchange tasks along the available edges in order to balance as much as possible the tasks among the various processors. The natural goal is that each processor will have at the end the same quantity of tasks to work on, namely a quantity of task close to the average.

Clock Synchronization

The recent advances in technology have also made low cost, low power wireless sensors a reality. For several applications of a wireless sensor networks, such as mobile target tracking, event detection, efficient TDMA scheduling, and sleep scheduling with very low duty cycle, it is essential that the nodes act in a coordinated and synchronized fashion. All these applications required global clock synchronization, that is, all the nodes of the network need to refer to a common notion of time. For instance, consider the problem of tracking a moving target using proximity sensors, where some nodes are deployed in the environment and their proximity sensors detect when the moving object passes in their vicinity [94]. Assuming that the position of the sensors is known, it is essential that the instants of detection are precisely timestamped for determining the trajectory (direction and speed) of the moving object. Clearly, the precision of the tracking algorithm based on this system is limited by the accuracy of the clock synchronization. Other interesting applications, which need a time-synchronization service, are habitat monitoring [136], power scheduling and TDMA communication schemes in wireless sensor networks [72], and rapid synchronized coordination of powerlines nodes in electric power distribution networks for catastrophic power-outage prevention [6]. Recently novel consensus-based protocol, for synchronizing a wireless has been elaborated, see for instance [120, 125, 32].

Decentralized estimation

Wireless sensor networks (WSN) have broad applications in surveillance and environmental monitoring, collaborative processing and information, and gathering scientific data from spatially distributed sources for environmental modeling and protection. Dealing with sensor networks, distributed estimation and tracking is one of the most fundamental collaborative information processing problem. Multi-sensor data fusion and tracking problems have a long history in signal processing, control theory, and robotics [1]. Moreover, estimation issues in wireless networks with fully distributed protocol of communication and elaboration data, i.e., without any hierarchical structure, have been the center of much attention only lately [96, 130]. Due to the huge number of devices that many applications could involve, collecting measurements from distributed wireless sensors nodes at a single location for on-line data processing may not be feasible due to several reasons among which long packet delay (e.g. due to multi-hop transmission) and/or limited bandwidth of the wireless network (e.g. due to energy consumption requirements). This problem is particularly relevant in wireless ad-hoc sensor networks where information needs to be multi-hopped from one to another using closer neighbors.

It turns out that a fundamental problem in sensor networks is to solve detection and estimation problem using *scalable algorithms*. This requires development of novel distributed algorithms for estimation and in particular for Kalman filtering. Recently, new scalable sensor fusion schemes requiring fusion of sensor measurements combined with local Kalman filtering have been proposed [130]. The key component of this approach is the introduction of a distributed filter that allows the nodes of a sensor network to track the average of N sensor measurements using an average consensus based distributed filter called *consensus filter* [102].

1.1 Overview and contributions of the Thesis

The dissertation is organized as follows:

- Chapter 2. In this chapter we review the fundamental features of the *consensus algorithm* and of the *average consensus algorithm*, both for the time-varying case and for the time-invariant case. In particular, we summarize the main results concerning the convergence and we introduce the concept of speed of convergence toward the consensus.
- Chapter 3. In this chapter we derive bounds on the convergence rate to the average consensus for set of systems that exchange information over time-invariant communication networks with symmetries (the Cayley symmetries). We show that, in time-invariant networks, symmetries yield rather slow convergence to the average consensus. In particular for such networks we have computed a tight bound for the convergence rate.
- Chapter 4. In this chapter we focus on the time-varying consensus problem, by addressing, in particular, the analysis of the so-called *randomized consensus algorithms*. Precisely, in this chapter, we assume that the consensus matrices constitute a sequence of i.i.d. matrix valued random variables. We review the concepts of *probabilistic consensus* and *average probabilistic consensus*. In the first part of the chapter, we introduce two random strategies, that illustrate a remarkable property of the randomized consensus algorithms, i.e., they allow to achieve better performance than deterministic ones with comparable complexity. In the second part of the chapter we review a well-known random consensus algorithm: the symmetric gossip. In particular, we will provide an interesting characterization of the symmetric gossip algorithm over Cayley graphs.
- Chapter 5. In this chapter we consider the more realistic and practical situation in which the communication network between the systems is constituted of only rate-constrained digital links. This, in general, prevents the nodes from having a precise knowledge about the state of the other nodes. Indeed, through digital channels the nodes can exchange only symbolic data in a finite alphabet and using this information they can build at most an estimate of their neighbors. Here, we assume that the nodes quantize their information before transmitting it. In particular we introduce two quantizers, well-known in the literature: the *deterministic* uniform quantizer and the *probabilistic* uniform

quantizer. Moreover, we restrict to the time-invariant average consensus problem. Since in the literature, remarkable results regarding the time-invariant quantized average consensus problem via probabilistic quantizers are already present, we focus mainly to the study of the time-invariant quantized average consensus problem via deterministic quantizers. The main contribution is the introduction of a simple and effective adaptation of the standard average consensus algorithm which does not converge to an asymptotic agreement, but is able to preserve the average of states and to drive the systems reasonably near to the consensus. We analyze this scheme by means of a worst-case model and a probabilistic model showing favorable convergence properties and providing performance bounds for the limit points of the iterates generated.

- Chapter 6. In this chapter we analyze the effects due to the presence of a communication network constituted of only digital channels on the symmetric gossip algorithm introduced in Chapter 4. We introduce two particular strategies, the *partially quantized* strategy and the globally quantized strategy, depending whether the systems use exact information regarding their own state or not to update their states. We will analyze these strategies both via the deterministic quantizer and via the probabilistic quantizer. We show that the *globally quantized* strategy both via the deterministic quantizer and via the probabilistic quantizer ensures that, almost surely, the consensus is reached. The drawback of this strategy is that it does not preserve the average of the initial conditions. On the other hand, the *partially quantized* strategy maintains the initial average at each iteration of the algorithm, but does not guarantee that the consensus is reached in general. However we show that the *partially quantized* drives asymptotically all the states very close to the initial average (we quantify how close).
- Chapter 7. Also in this chapter we assume that the nodes can communicate between them only through digital channels. As in Chapter 5, we consider, here, the time-invariant case. In the previous two chapters, in order to face with the effects due to the forced quantization, we have elaborated strategies which either preserves the average of the state but do not converge to a consensus or do not preserve the average but converge to a consensus which, in general, does not coincide with the initial average. The main contribution of this chapter is to introduce a novel quantized strategy that permits both to maintain the initial average and to reach it asymptotically. More precisely we adapt

coding/decoding strategies, that were proposed for centralized control and communication problems, to the distributed consensus problem. In particular, we present two coding/decoding strategies, one based on the exchange of logarithmically quantized information, the other on a zoom in - zoom out strategy (this latter involves the use of uniform quantizers). We provide analytical and simulative results illustrating the convergence properties of these strategies. In particular we show that the convergence factors depend smoothly on the accuracy parameter of the quantizers used and that, remarkably, that the critical quantizer accuracy sufficient to guarantee convergence is independent from the network dimension.

- Chapter 8. In this chapter we deal with a possible application of the consensus ideas to the wide field of the distributed estimation. In particular we study a prototypical problem of distributed estimation for sensor networks; the state of a scalar linear system is estimated via a two stage procedure which consists in (i) a standard (and decentralized) Kalman-like update and (ii) information propagation using consensus strategies. To this purpose, two design parameters, i.e. the Kalman gain and the consensus matrix have to be designed. This choice is made by optimizing the steady state prediction (or estimation) error. We discusse, under specific circumstances, the behavior of the "optimal" parameters.
- Chapter 9. In this chapter we summarize the result found and we gather out our conclusions.

In the appendices, we include some reference material for the reader, in particular

- Appendix A. In this appendix, we review some concepts on harmonic analysis on finite groups which are useful for the analysis, illustrated in Chapter 3, of the spectral properties of the Cayley matrices
- Appendix B. In this appendix, we recall some notation and concepts on directed and undirected graphs which are useful throughout all the thesis.
- Appendix C. In this appendix, we collect some algebraic results regarding the stability of discrete time linear parameter varying (LPV) systems and the solvability of a particular Lyapunov equation. The proofs of some theorem stated in this thesis, are based on these algebraic results.

Notation

We let $\prod_{i \in \{1,\ldots,N\}} S_i$ denote the Cartesian product of sets S_1, \ldots, S_N .

We let \mathbb{C} , \mathbb{R} denote the set of complex numbers and the set of real numbers, respectively. The set of natural numbers is denoted by \mathbb{N} .

For $x \in \mathbb{R}^d$, we denote by ||x|| (or $||x||_2$), by $||x||_{\infty}$ and by $||x||_F$ the Euclidean norm, the ∞ -norm and the Frobenius norm of x, respectively.

Given a vector $x \in \mathbb{C}^d$ we denote by x^* its conjugate transpose. We use the same notation even if $x \in \mathbb{R}^d$.

We define the vectors $\mathbf{0} = [0, \dots, 0]^*$ and $\mathbb{1} = [1, \dots, 1]^*$ in \mathbb{R}^d .

Given a vector $x = [x_1, \ldots, x_d]^*$ belonging to \mathbb{R}^d or to \mathbb{C}^d , diag $\{x\}$ or diag $\{x_1, \ldots, x_N\}$ mean a diagonal matrix having the components of x as diagonal elements.

Given a matrix $M \in \mathbb{R}^{d \times d}$, with $\sigma(M)$ we denote the spectrum of M, i.e., the set of its eigenvalues.

Let M any matrix belonging to $\mathbb{R}^{d \times d}$. With tr M we denote the trace of M, i.e., the sum of the diagonal entries. With M^* we denote the transpose of M.

For $f, g : \mathbb{N} \to \mathbb{R}$, we say that $f \in O(g)$ if there exist $N_0 \in \mathbb{N}$ and k > 0such that $|f(N)| \leq k|g(N)|$ for all $N \geq N_0$. For $f, g : \mathbb{N} \to \mathbb{R}$, we say that $f \in o(g)$ if $\lim_{n \to \infty} \frac{f(n)}{g(n)} = 0$.

Let Ω be a random variable. Then, $\mathbb{E}[\Omega]$ denotes the expectation of Ω .

For notations related to graphs, we refer the reader to the Appendix A.

Chapter 2

The Consensus Problem

2.1 Introduction

This chapter overviews the fundamental discrete-time consensus algorithm in which a scalar information state is updated by each system using a first-order difference equation [101, 74, 112, 88, 139]. The information flow between the systems is modeled by a direct graph. No constraints of bandwidth are considered. Both fixed and dynamically changing interaction topologies are analyzed. The main results regarding the convergence of the consensus algorithm, are summarized. The concept of speed of convergence toward the consensus is introduced and a optimization problem is formulated.

In this chapter we will refer to the notions of graph theory collected in Appendix B.

2.2 Problem Formulation

Consider N > 1 systems whose dynamics are coupled by the following discrete time state equations

$$x_i(t+1) = \sum_{j=1}^{N} P_{ij}(t) x_j(t) \qquad i = 1, \dots, N, \qquad (2.1)$$

where $x_i(t) \in \mathbb{R}$ is the state of the *i*-th system at time *t* and $P_{ij}(t) \in \mathbb{R}$ are coefficients which vary with the time *t*. More compactly we can write

$$x(t+1) = P(t)x(t),$$
(2.2)

where $x(t) \in \mathbb{R}^N$ and $P(t) \in \mathbb{R}^{N \times N}$. The sequence P(t) is said to achieve the *consensus* if the following conditions are satisfied:

- (a) If $x(0) \in \mathcal{I}$ then x(t) = x(0) for every $t \in \mathbb{N}$, where \mathcal{I} is the subspace generated by the N- dimensional vector $\mathbb{1} := [1, \ldots, 1]^*$.
- (b) For any $x(0) \in \mathbb{R}^N$, there exists a scalar α such that

$$\lim_{t\to\infty} x(t) = \alpha \mathbb{1}$$

Moreover, if $\alpha = N^{-1} \mathbb{1}^* x(0)$, we say that *average consensus* is achieved.

It is worth noting that condition (a) is imposed in order to avoid meaningless updating rules. Indeed if $x(\bar{t}) = \alpha \mathbb{1}$ at some time instant \bar{t} , then condition (a) guarantees that $x(t) = \alpha \mathbb{1}$ for all $t \ge \bar{t}$. Moreover it is possible to see that condition (a) implies also that

$$P(t)\mathbb{1} = \mathbb{1} \tag{2.3}$$

for every $t \in \mathbb{N}$. A matrix satisfying (2.3) is called *quasi-stochastic* [128].

From now on, we assume that this condition holds for the sequence of matrices P(t).

Assumption 2.1 The matrix P(t) in (2.2) is a quasi-stochastic matrix for each time instant $t \in \mathbb{N}$.

It is worth providing now other three definitions that will be useful throughout all this thesis and that specify the concept of *quasi-stochastic* matrix for some particular case. If the *quasi-stochastic* matrix P(t) satisfies also the condition $\mathbb{1}^*P(t) = \mathbb{1}^*$ then it is called *quasi-doubly stochastic*. Moreover if we restrict the *quasi-stochastic* (respectively the *quasi-doubly stochastic*) matrix P(t) to have all the elements nonnegative, namely $P_{ij}(t) \ge 0$ for all pair i, j, then P(t) is said to be a *stochastic* (respectively a *doubly stochastic*) matrix [123].

Note now that, the fact that in the matrix P(t) the element in position i, jis different from zero, means that the system i needs the state of the system j in order to update its state. This implies that we need to communicate the state $x_j(t)$ from the system j to the system i. In this context, a good description of the information flow required by a specific matrix P(t) is given by the directed graph $\mathcal{G}_{P(t)}$ with set of vertices $\{1, \ldots, N\}$ in which there is an arc from j to i whenever in the matrix P(t) the element $P_{ij}(t) \neq 0$. The graph $\mathcal{G}_{P(t)}$ is said to be the communication graph associated with P(t). Conversely, given any directed graph \mathcal{G} with set of vertices $\{1, \ldots, N\}$, we say that a matrix P(t) is compatible with \mathcal{G} if $\mathcal{G}_{P(t)}$ is a subgraph of \mathcal{G} (we will use the notation $\mathcal{G}_{P(t)} \subseteq \mathcal{G}$). For the sake of the clarity, it is worth noting that if a graph \mathcal{G} contains the self loop (i, i) it means that the *i*-th agent has access to its own state.

We say that the consensus problem is solvable on a sequence of graphs $\mathcal{G}(t)$ if there exists a sequence of matrices P(t) compatible with the sequence $\mathcal{G}(t)$ and satisfying conditions (a) and (b).

Observe now that, if both $P_{ij}(t)$ and $P_{ji}(t)$ are different from zero, then both the edges (j, i) and (i, j) belongs to $\mathcal{G}_{P(t)}$. In such case the interaction topology between the systems can be modeled by an undirected graph. However, in general, an undirected graph can be viewed as a special case of a direct graph. A direct graph is useful because it takes into account the general case where information flow may be unidirectional.

Note that consensus algorithm (2.1) is distributed in the sense that each system needs only the information from its neighbors. Intuitively, the information state of each system is updated as the weighted average (with possible negative weights) of its current state and the current states of its neighbors. Also note that the weights $P_{ij}(t)$ in (2.1) may be time-varying to represent both the time-varying relative confidence of each'system information state and the fact that the interaction topology may be changing dynamically due to unreliable transmission or a limited communication/sensing range.

Remark 2.2 The consensus problem first appeared in [141]. It is worth noting that the Vicsek model can be viewed as a special case of (2.1) by letting

$$P_{ij} = \begin{cases} \frac{1}{1+|\mathcal{N}_i|} & \text{if } (j,i) \in \mathcal{G}(t) \text{ or } i=j\\ 0 & \text{otherwise} \end{cases}$$

where $\mathcal{N}_i(t) = \{j \in V : j \neq i, (j, i) \in \mathcal{G}(t)\}$ represents the set of neighbors of system *i* at time *t* and $|\mathcal{N}_i(t)|$ its cardinality, that is, each system simply averages its own information state with those that are communicated to it. Note that the matrix P(t) built in this way is a stochastic matrix but not in general a doubly stochastic matrix, even if \mathcal{G} is an undirected graph. Instead the simplified Vicsek model used in [74] which is obtained by letting $P_{ij}(t) = 1/k$ if $(j,i) \in \mathcal{G}(t)$ and $P_{ij}(t) = 0$ otherwise, $\forall j \neq i$, and $P_{ii}(t) =$ $1 - \sum_{j\neq i} P_{ij}(t)$, where k > N is a constant, leads to a doubly stochastic matrix in the case \mathcal{G} is undirected.

Remark 2.3 When considering a team of vehicles cooperating together, a

nice interpretation of the consensus problem (called in this case also rendezvous problem), closer to the classical control theory, is the following. Let us rewrite (2.1) as a discrete time single-integrator

$$x_i(t+1) = x_i(t) + u_i(t)$$
 $i = 1, \dots, N$, (2.4)

where $u_i(t) \in \mathbb{R}$ is the control input that the *i*-th vehicle applies at time *t*. More compactly, we can write that

$$x(t+1) = x(t) + u(t), \qquad (2.5)$$

where $x(t), u(t) \in \mathbb{R}^N$. In the rendezvous problem, the goal will be to design a sequence of feedback control laws

$$u(t) = K(t)x(t), \qquad K(t) \in \mathbb{R}^{N \times N}$$

yielding the consensus of the states, namely a sequence of feedback control laws control such that the closed loop system

$$x(t+1) = (I + K(t))x(t)$$
(2.6)

yields

$$\lim_{t \to \infty} x(t) = \alpha \mathbb{1},\tag{2.7}$$

where α is a scalar depending only on x(0) and on the sequences K(t). Note that if we define

$$P(t) := I + K(t)$$

then we obtain (2.2). Conversely, we can rewrite (2.1) as

$$x_i(t+1) = x_i(t) + \sum_{j=1}^N P_{ij}(t) (x_j(t) - x_i(t)),$$

and defining

$$u_i(t) = \sum_{j=1}^{N} P_{ij}(t) \left(x_j(t) - x_i(t) \right)$$

or, more compactly,

$$u(t) = (P(t) - I)x(t)$$
(2.8)

we obtain (2.4) and (2.5).

We conclude this section by formulating a first preliminary algebraic result on the convergence of the consensus algorithm. The proof can be found in [112]. **Theorem 2.4** Discrete-time algorithm (2.1) achieves consensus if and only if

$$P(t)P(t-1)\cdots P(2)P(1)P(0) \to \mathbb{1}c^*,$$
 (2.9)

as $t \to \infty$, where c is a N-dimensional vector.

2.2.1 Convergence Analysis of Consensus Algorithms

In this subsection we summarize the results known about the conditions ensuring the solvability and the convergence of the consensus algorithm. First we will consider the general case of Time-varying Communication Topologies, then we will specialize to the case of Time-invariant Communication Topologies and weights P_{ij} constant.

Convergence Analysis for Time-varying Communication Topologies

We start by considering the case of direct switching interaction topologies. The next two theorems, whose proof can be found in [88, 25], rely mainly on the notion of connectivity of a node to an another node, provided at the end of Appendix B.

Theorem 2.5 Let P(t), $t \in \mathbb{N}$, be a sequence of stochastic matrices, such that $P_{ii}(t) \neq 0$, i = 1, ..., N, for all $t \in \mathbb{N}$. Consider the sequence of directed graphs $\mathcal{G}_{P(t)}$, associated to the sequence P(t). Assume the existence of real numbers $0 < m \leq M$ such that $m \leq P_{ij}(t) \leq M$ for all $t \in \mathbb{N}$ and $(j,i) \in \mathcal{G}_{P(t)}$. If there exists a duration $T \geq 0$ such that for all $t_0 \in \mathbb{N}$ the graph

$$\mathcal{G}_{P(t_0+1)} \cup \cdots \cup \mathcal{G}_{P(t_0+T)}$$

contains a node connected to all the other nodes, then

$$\lim_{t \to \infty} x(t) = c \mathbb{1}$$

where $c \in \mathbb{R}$ is a constant depending only on the initial condition x(0) and on the sequence P(t).

Theorem 2.5 gives a general result about stochastic matrices that are not necessarily symmetric. The following theorem presents a convergence result for the case of symmetric matrices (i.e., undirected graphs) under connectivity requirements that are weaker than the ones stated in the previous Theorem 2.5. **Theorem 2.6** Let P(t), $t \in \mathbb{N}$, be a sequence of symmetric stochastic matrices, such that $P_{ii}(t) \neq 0$, i = 1, ..., N, for all $t \in \mathbb{N}$. Consider the sequence of undirected graphs $\mathcal{G}_{P(t)}$, associated to the sequence P(t), $t \in \mathbb{N}$. Assume the existence of real numbers $0 < m \leq M$ such that $m \leq P_{ij}(t) \leq M$ for all $t \in \mathbb{N}$ and $(j, i) \in \mathcal{G}_{P(t)}$. If for all $t_0 \in \mathbb{N}$ the graph

$$\bigcup_{t \ge t_0} \mathcal{G}_{P(t)}$$

is connected, then

 $\lim_{t \to \infty} x(t) = c\mathbf{1},$

where $c \in \mathbb{R}$ is a constant depending only on the initial condition x(0) and on the sequence P(t).

Remark 2.7 Note that the above two Theorems provide conditions ensuring the convergence of the consensus algorithms, only for sequence of stochastic matrices. To the best of our knowledge, there are no similar result for the more general case of quasi-stochastic matrices.

Remark 2.8 For the sake of the completeness it is worth noting that the authors in [113] and [112] provide very similar results. In particular, given a sequence of stochastic matrices graphs (*respectively* a sequence of symmetric stochastic matrices) P(t), they relate the convergence of the consensus problem to the following two facts:

- (i) $P_{ij}(t) \in \overline{\mathcal{P}}$, for all $t \in \mathbb{N}$, where $\overline{\mathcal{P}}$ is a finite set of nonnegative numbers that are no larger than one;
- (ii) there exists an infinite sequence of contiguous, nonempty, uniformly bounded time intervals $[t_j, t_{j+1}), j = 1, 2, ...$ starting at $t_1 = 0$, with the property that the union of the directed graphs associated to the matrices (*respectively* undirected graphs) across each such interval has a directed spanning tree (*respectively* is connected).

If the above two facts yield true, then

$$\lim_{t \to \infty} x(t) = c\mathbb{1},$$

where $c \in \mathbb{R}$ is a constant depending only on the initial condition x(0) and on the sequence P(t).

Convergence Analysis for Time-Invariant Communication Topologies

In this subsection, we investigate conditions under which the information state of consensus algorithm converges when the communication topology is time invariant and the gains P_{ij} are constant. In this case (2.1) and (2.2) become

$$x_i(t+1) = \sum_{j=1}^{N} P_{ij} x_j$$
 $i = 1, \dots, N$, (2.10)

and

$$x(t+1) = Px(t)$$
(2.11)

where $P \in \mathbb{R}^{N \times N}$.

We state immediately the following proposition which provides an algebraic characterization of the quasi-stochastic matrices achieving the consensus (see [35]).

Proposition 2.9 Let P be any quasi-stochastic matrix, namely P1 = 1. Then P yields asymptotically the consensus if and only if the following three conditions hold :

- (A) 1 is the only eigenvalue of P on the unit circle centered in 0;
- (B) the eigenvalue 1 has algebraic multiplicity one (namely it is a simple root of the characteristic polynomial of P) and 1 is its eigenvector;
- (C) all the other eigenvalues are strictly inside the unit disk centered in 0.

Now, it is worth noting that a matrix P satisfying the above three properties achieves the average consensus if and only if $\mathbb{1}^*P = \mathbb{1}^*$, that is if and only if P is a quasi- doubly stochastic matrix. To the best of our knowledge, there are no other conditions regarding the quasi-stochastic matrices (or quasidoubly stochastic matrices) characterizing the convergence of the consensus algorithm (2.11). Something more can be said if we restrict to nonnegative matrices, i.e., to stochastic (or doubly stochastic) matrices. The following result is the straightforward consequence of standard results on stochastic matrices [63, pag. 88 and pag. 95].

Theorem 2.10 Let \mathcal{G} be a directed graph and assume that \mathcal{G} contains all loops (i, i). The following conditions are equivalent:

- (i) the consensus problem is solvable on \mathcal{G} ;
- (ii) \mathcal{G} contains a node which is connected to all the other node.

Moreover, if the above conditions are satisfied, any stochastic matrix P such that $\mathcal{G}_P = \mathcal{G}$ and $P_{ii} \neq 0$ for every $i = 1, \ldots, n$ solves the consensus problem.

We observe that requiring that a graph \mathcal{G} has a node connected to all the other nodes is equivalent to requiring that the graph \mathcal{G} has a spanning tree. Moreover it is worth also noting that, as shown in [115], if (2.11) converges asymptotically, then the consensus equilibrium will be equal to a weighted average of the initial conditions of those systems that have a direct path to all other systems. Hence we can hope that, if \mathcal{G} is strongly connected, adding probably some further condition, also the average consensus problem can be solved. Indeed the next proposition, shows that, differently from the time-varying case, for the time-invariant consensus algorithm it is also possible provide conditions guaranteeing the solvability of the average consensus problem (see [101]). Also the next theorem is the straightforward of standard results on stochastic matrices.

Theorem 2.11 Let \mathcal{G} be a directed graph and assume that \mathcal{G} contains all loops (i, i). The following conditions are equivalent:

- (i) the average consensus problem is solvable on \mathcal{G} ;
- (ii) \mathcal{G} is strongly connected.

Moreover, if the above conditions are satisfied, any doubly stochastic matrix P such that $\mathcal{G}_P = \mathcal{G}$ and $P_{ii} \neq 0$ for every $i = 1, \ldots, n$ solves the average consensus problem.

We conclude this subsection by stating the following result which is a direct consequence of the above Theorem and characterizes the undirected communication graphs and symmetric matrices P. Moreover we provide two interesting and simple rules for choosing the weights of the symmetric matrix P, often used in the literature: the *Maximum-degree weights* and the *Metropolis weights* (see [149]).

Theorem 2.12 Let \mathcal{G} be an undirected graph. Assume that \mathcal{G} is connected and it contains all loops (i, i). Then any symmetric stochastic matrix P such that $\mathcal{G}_P = \mathcal{G}$ and $P_{ii} \neq 0$ for every i = 1, ..., N achieves asymptotically the average consensus. **Example 2.13** [Maximum-degree weights] Given an undirected graph $\mathcal{G} = (V, \mathcal{E})$, let $d \in \mathbb{R}$ be such that

$$d \ge \max_{1 \le i \le N} \left\{ d_i \right\}$$

where $d_i = |\mathcal{N}_i|$, with $\mathcal{N}_i = \{j \in V | j \neq i, (j, i) \in \mathcal{E}\}$. Here we use the constant weight 1/(d+1) on all the edges, and choose the self-weights so that the sum of weights at each node is 1:

$$P_{ij} = \begin{cases} \frac{1}{d+1} & \text{if } (j,i) \in \mathcal{E} \text{ and } i \neq j \\ 1 - \frac{d_i}{d+1} & \text{if } i = j \\ 0 & \text{otherwise} \end{cases}$$

Example 2.14 [Metropolis weights]

Let \mathcal{G} be as in the previous example. The Metropolis weights are defined as

$$P_{ij} = \begin{cases} \frac{1}{1 + \max\{d_i, d_j\}} & \text{if } (j, i) \in \mathcal{E} \\ 1 - \sum_{(i,k) \in \mathcal{E} \setminus \{(i,i)\}} P_{ik} & \text{if } i = j \\ 0 & \text{otherwise} \end{cases}$$

where d_i is as in the above example. With Metropolis weights, the weight on each edge is one over one plus the larger degree at its two incident vertices, and the self-weights are chosen so the sum of weights at each node is 1.

2.2.2 Consensus Synthesis

In some applications, consensus algorithms must satisfy given requirements or optimize a suitable performance index. The simplest control performance index is the rate of convergence toward the consensus equilibrium. We follow the approach proposed in [147]. Given a sequence of matrices $\mathbb{P}(t)$, $t \in \mathbb{N}$, yielding the consensus, i.e satisfying (2.9), we define the *asymptotic conver*gence factor as

$$r_{asym} = \sup_{x(0)} \limsup_{t \to \infty} \left(\|x(t) - x_{ave}(t)\|_2 \right)^{\frac{1}{t}}.$$

where $x_{ave}(t) = 1/N\mathbb{1}^* x(t)$.

From this choice of performance, the problem one would like to solve is the following.

Problem: Given a sequence of graphs $\mathcal{G}(t)$, $t \in \mathbb{N}$, find a sequence of matrices P(t), $t \in \mathbb{N}$, such that $\mathcal{G}_{P(t)} \subseteq \mathcal{G}(t)$ and minimizing r_{asym} .

The solution of the above problem is very hard in general. Some interesting results can be found in the literature, if we restrict to the time-invariant average consensus. In order to illustrate them, we need first to introduce the concept of *essential spectral radius*. Given any quasi-stochastic matrix P, we define the quantity

$$\rho_{ess}(P) = \begin{cases} 1 & \text{if } \dim \ker(P-I) > 1\\ \max\{|\lambda| : \lambda \in \sigma(P) \setminus \{1\}\} \text{ if } \dim \ker(P-I) = 1 \,, \end{cases}$$
(2.12)

which is called the *essential spectral radius* of the matrix P.

It is easy to see that a quasi-stochastic matrix P yields asymptotically the consensus, namely satisfies the conditions (A), (B), (C) stated in Proposition 2.9, if and only if $\rho_{ess}(P) < 1$.

Consider now the time-invariant average consensus algorithm

$$x(t+1) = Px(t),$$

where P is a quasi-doubly stochastic matrix such that $\rho_{ess}(P) < 1$. The following result relates the asymptotic convergence to the essential spectral radius of P (see [25]).

Theorem 2.15

$$\rho_{ess}(P) = r_{asym}(P).$$

Using this result the above problem can be formulated as the following minimization problem:

$$\begin{array}{ll}\text{minimize} & \rho_{ess}(P)\\ \text{subject to} & P \in \mathcal{P} \end{array} \tag{2.13}$$

where $\mathcal{P} = \{\mathcal{P} : \mathcal{G}_{\mathcal{P}} \subseteq \mathcal{G}, \mathcal{P}\mathbb{1} = \mathbb{1}, \mathbb{1}^*\mathcal{P} = \mathbb{1}^*\}$. The authors in [147] refer to this problem as the *fastest distributed linear averaging problem (FDLA)*. It is worth noting that $\rho_{ess}(P) = \rho(P - 1/N\mathbb{1}\mathbb{1}^*)$, where given a matrix $M \in \mathbb{R}^{N \times N}$, $\rho(M)$ denotes the spectral radius of M. Hence the above problem can be seen as a spectral radius minimization problem. Even though the constraint in this problem are linear equalities, the solution of the problem in general is very hard. The main reason is that the objective function, i.e. the spectral radius of a matrix, is not a convex function.

Now suppose we add the additional constraint that weights are symmetric, i.e., $P_{ij} = P_{ji}$. In this case the problem can be posed as

minimize
$$\rho_{ess}(P)$$

subject to $P \in \mathcal{P}, P = P^*$ (2.14)

which is a convex problem. This represent the symmetric version of the FDLA problem. In [147] it is shown that this problem can be cast as a semidefinite program, and therefore efficiently and globally solved.

Remark 2.16 When we are dealing with average consensus algorithm it is meaningful to consider the displacement from the average of the initial conditions

$$\Delta(t) := x(t) - \left(\frac{1}{N} \mathbb{1}^T x(0)\right) \mathbb{1}.$$
(2.15)

It is immediate to check that, if P satisfies condition $P\mathbb{1} = \mathbb{1}$ then $\Delta(t)$ satisfies the recursive equation

$$\Delta(t+1) = P\Delta(t). \qquad (2.16)$$

Moreover, if P also satisfies condition $\mathbb{1}^*P = \mathbb{1}^*$, then

$$\Delta(t) = x(t) - \left(\frac{1}{N} \mathbb{1}^T x(t)\right) \mathbb{1}.$$

Notice finally that the initial conditions $\Delta(0)$ are such that

$$\mathbb{1}^T \Delta(0) = 0. \tag{2.17}$$

Hence the asymptotic behavior of our average consensus problem can equivalently be studied by looking at the evolution (2.16) on the hyperplane characterized by the condition (2.17).

We conclude this section by noting that the FDLA problem (2.13) is closely related to the problem of finding the fastest mixing Markov chain on a graph [20]. This last problem can be posed as

minimize
$$\rho_{ess}(P)$$

subject to $P \in \tilde{\mathcal{P}}$, (2.18)

where $\tilde{\mathcal{P}} = \{P : \mathcal{G}_P \subseteq \mathcal{G}, \mathcal{P} \text{ is doubly stochastic}\}$. Note that minimizing the essential spectral radius $\rho_{ess}(P)$ it is equivalently to maximizing $1 - \rho_{ess}(P)$ (which is called the spectral gap of the associated Markov chain). Recently some very effective algorithms have been proposed for this maximization limited, however, to the case in which P is a symmetric matrix [20]. The only difference in the two problem formulations is that in the FDLA problem, the weights can be (and the optimal ones often are) negative, hence faster convergence could be achieved compared with the fastest mixing Markov chains on the same graph.

2.3 Conclusion

In this chapter we have overviewed the standard discrete-time consensus algorithm. In particular we have summarized the main results regarding the convergence. It is worth now ending by mentioning the numerous recent directions of research on consensus and averaging: continuous-time consensus algorithms [101, 88, 80, 81], characterization of the convergence rates and time complexity [103, 104, 77, 35, 27, 110], consensus over random networks [70, 145, 109, 137, 108, 106, 107, 58, 57], consensus in finite-time [40, 135], consensus algorithms for general functions [13, 39, 84, 135], quantized consensus problems [29, 30, 75, 152, 150, 148, 33, 92, 8, 61], connections with the heat equation and partial difference equation [60], spatially-decaying interactions [42], convergence in time- delayed and asynchronous settings [16, 7, 59], consensus on manifolds [118, 119, 73], applications to distributed signal processing [151, 31, 130, 149, 100]. Numerous interesting results are reported in the recent PhD theses [79, 83, 26, 71]. Finally, we point out two recent surveys [99, 114] and the texts [113, 25].

Chapter 3

The symmetries in the consensus problem

3.1 Introduction

As explained in the previous chapter, the connectivity properties of the *com*munication graph influence the convergence properties of the consensus algorithm (2.2). A natural question now arises: " How much does the amount of information exchanged by the systems in the consensus algorithm influence the rate of convergence towards the asymptotic agreement?"

This chapter aims to provide an answer to the above question. Intuitively one should expect that the larger is the communication effort between the systems, the better are the performance achievable by the consensus algorithms. The main result of the chapter is a mathematical characterization of this fact for a particular class of graphs and matrices exhibiting symmetries.

We have pointed out in the previous chapter how quasi-doubly stochastic matrices, whose associated communication graphs satisfy some assumptions, described in Theorem 2.5, in Theorem 2.6, in Theorem 2.10, in Theorem 2.11 and in Theorem 2.12, are matrices that guarantee the solvability of the average consensus, with a degree of efficiency which is related to the spectral properties of such matrices (see Theorem 2.15). Moreover we recall that, if we restrict to the case in which the weights are nonnegative number, then the consensus matrices are doubly stochastic and hence regarding them as Markov chains allows to relate the consensus convergence rate to the mixing rate of the chain [14]. The problem of bounding the mixing rate of a Markov chain is a large research area: see for instance the survey [68] and references there in.

Spectral properties of doubly stochastic matrices can be characterized in a easier way if we impose some symmetries on the matrices themselves, and, as consequence, on the associated communication graph. Both Markov chains and graphs satisfying symmetries, which are called Cayley graphs, are widely studied in the literature [10, 89, 140]. It is known that symmetries described by Abelian groups yield rather poor convergence rates [4].

In this chapter, by modelling the communication networks as Cayley graphs defined on Abelian groups we are able to extend the available bounds on the mixing time of Markov chains defined on such groups [48, 14, 117]. In particular, in Section 3.3, we derive a bound that is a function of the number of agents and the incoming arcs in each vertex. The main result shows that, if we impose symmetries in the communication network and we keep the number of incoming arcs in each vertex bounded, then the convergence rate degrades as the number of agents increases. Moreover, we show that the proposed bound is tight. It is worth noting that the idea of imposing symmetries on the communication graph is not new [45, 111, 127]. One of the reasons to impose symmetries in the communication graph derives from the fact that graphs with symmetries allow a much more compact representation, which can be relevant when a very large number of agents is considered. In particular in [127] the authors show, for particular symmetries, how it is possible to obtain better performance by increasing the number of incoming arcs on each vertex. In this chapter we extend the result to a broader class of graphs with symmetries.

The chapter is organized as follows. In Section 3.2 we will formulate the problem. In particular our goal will be to minimize the *essential spectral radius*, quantity that we have defined in (2.12). We will characterize the solution for a first class of matrices and graphs exhibiting circulant symmetries. In Section 3.3 we will extend our results to the broader class of Cayley graphs and Cayley matrices. Finally we will gather out our conclusions in Section 3.4.

3.2 Problem formulation

In this section we consider the minimization problem (2.18) formulated in Section 2.2.2. For the sake of the clarity we briefly reformulate it. Suppose that there are N system connected in a network. The network's communication topology is given by the direct graph $\mathcal{G} = (V, \mathcal{E})$, where $V = \{1, \ldots, N\}$ is the nodes set and $\mathcal{E} \subseteq V \times V$ is the edges set. Let $x_i(t)$ denote the state information of the *i*-th system at time *t*. We define $x(t) = [x_1(t), \ldots, x_N(t)]^*$ to be the global information state. We have that x(t) is updated according to the following recursive equation

$$x(t+1) = Px(t),$$

where $P \in \mathbb{R}^{N \times N}$ is a doubly stochastic matrix, such that $\mathcal{G}_P \subseteq \mathcal{G}$. Let $\rho(P)$ be the *essential spectral radius* of P^1 . Moreover, given a graph \mathcal{G} let

$$\rho_{\mathcal{G}} = \min \{ \rho(P) \mid P \text{ is doubly stochastic}, \mathcal{G}_P \subseteq \mathcal{G} \}$$

Given a graph \mathcal{G} , an interesting issue should be that of understand how the connectivity of \mathcal{G} influences the best performance achievable by the consensus algorithm on \mathcal{G} , which is represented by $\rho_{\mathcal{G}}$. Intuitively we expect $\rho_{\mathcal{G}}$ to be sensitive to the communication effort. Precisely we conjecture that if $\mathcal{G}_1 \subset \mathcal{G}_2$ then $\rho_{\mathcal{G}_1} > \rho_{\mathcal{G}_2}$. However we have been not able to prove this so far. This analysis becomes more treatable if we limits our considerations to graphs \mathcal{G} and matrices P exhibiting symmetries. We will show how these symmetries limit the achievable performance in terms of convergence rate.

We start our analysis in the next subsection by limiting ourselves to the cyclic symmetry. Then we will extend our results to a broader class of symmetries: the Cayley symmetries.

3.2.1 Circulant symmetries

In order to introduce cyclic symmetry, we define the following map

$$p: \{1, \dots, N\} \to \{1, \dots, N\}: i \mapsto i+1 \mod N$$

The matrix P is said to be symmetric with respect to p if

$$P_{i,j} = P_{p(i),p(j)} \quad \forall \ i,j \in \{1,\ldots,N\}.$$

This condition is equivalent to impose that

$$P = \sum_{i=0}^{N} p_i \Pi^i$$

¹Since there is no risk of confusion, for the sake of the notational convenience, in this chapter we denote the *essential spectral radius* of P by $\rho(P)$ instead of $\rho_{ess}(P)$ as in the previous chapter.

where

$$\Pi := \begin{pmatrix} 0 \ 1 \ 0 \ \cdots \ 0 \ 0 \\ 0 \ 0 \ 1 \ \cdots \ 0 \ 0 \\ 0 \ 0 \ 0 \ \cdots \ 0 \ 0 \\ \vdots \ \vdots \ \vdots \ \ddots \ \vdots \ \vdots \\ 0 \ 0 \ 0 \ \cdots \ 0 \ 1 \\ 1 \ 0 \ 0 \ \cdots \ 0 \ 0 \end{pmatrix}$$

These matrices are called circulant [46]. Notice that, in this case, since $\Pi \mathbb{1} = \mathbb{1}$ and $\mathbb{1}^*\Pi = \mathbb{1}^*$, then circulant matrices satisfy condition $P\mathbb{1} = \mathbb{1}$ if and only if $\sum_i p_i = 1$. Moreover condition $\mathbb{1}^*P = \mathbb{1}^*$ is also satisfied, and thus such matrices drive the state to the average of the initial conditions. Consequently, if we choose P to be nonnegative, then P is always doubly stochastic. The spectral properties of circulant matrices are particularly simple. Indeed, it can be shown that

$$\sigma(P) = \left\{ P\left(e^{j\frac{2\pi}{N}h}\right) : h = 0, 1, \dots, N-1 \right\}$$

where $P(z) := \sum_{i=0}^{N} p_i z^i$. Notice that $P(e^{j0}) = 1$. Hence,

$$\rho(P) = \max\left\{ \left| P\left(e^{j\frac{2\pi}{N}h}\right) \right| : h = 1, \dots, N-1 \right\}.$$

Moreover, the corresponding eigenvectors v_h 's form an orthonormal basis and $v_0 = (1/N)\mathbb{1}$. Notice that, in order to have consensus stability in this context, it is sufficient to impose that

$$\left| P\left(e^{j\theta} \right) \right| < 1 \quad \forall \ \theta \neq 0 \,. \tag{3.1}$$

This condition is slightly stronger than consensus stability, however it provides a stability condition independent of the number of systems N.

Circulant solutions to the consensus problem exist if the graph \mathcal{G} admits an analogous symmetry. Indeed, consider a strongly connected graph \mathcal{G} on $\{1, \ldots, N\}$ containing all the self loops (i, i) and that is symmetric with respect to p, in the sense that if there is an arc from i to j, there is also an arc from p(i) to p(j). Then, it is immediate to find a circulant matrix P such that $\mathcal{G}_P = \mathcal{G}$ and which solves the consensus problem. Indeed if j_1, \ldots, j_{μ} are the incoming arcs of vertex 1 in \mathcal{G} , it is sufficient to choose weights p_0, \ldots, p_{μ} such that $p_i > 0$ for $i = 0, \ldots, \mu$, and $\sum_i p_i = 1$. If we consider $P = \sum p_i \Pi^i$, then it is clear that condition (3.1) is satisfied.

The particular spectral structure of circulant matrices allows to obtain asymptotic results on the behavior of the essential spectral radius $\rho(P)$ and
therefore on the rate of convergence of the corresponding consensus scheme. Define

$$\rho_{\mathcal{G}}^{\text{circ}} = \inf\{\rho(P) \mid P \text{ circulant and stochastic}, \mathcal{G}_P \subseteq \mathcal{G}\}.$$

Let us start from some examples.

Example 3.1 Suppose \mathcal{G} is described by the arcs $i \leftarrow i + 1 \pmod{N}$ and all the self-loops (i, i). We can choose therefore $P(z) = p_0 + p_1 z$, where $p_0, p_1 \in \mathbb{R}$. In this case we have that

$$x(t+1) = \{p_0 I + p_1 \Pi\} x(t) \,.$$

The condition $P(1) = p_0 + p_1 = 1$ implies that P(z) = p + (1 - p)z for some $p \in \mathbb{R}$. In this case it can be shown that we have consensus stability if and only if 0 and that the rate of convergence is

$$\rho(P) = \left((1-p)^2 + p^2 - 2p(1-p)\cos\left(\frac{2\pi}{N}\right) \right)^{\frac{1}{2}}.$$

The p that minimizes $\rho(P)$ is p = 1/2 and yields

$$\rho_{\mathcal{G}}^{\text{circ}} = \left(\frac{1}{2} + \frac{1}{2}\cos\left(\frac{2\pi}{N}\right)\right)^{\frac{1}{2}} \simeq 1 - \frac{\pi^2}{2}\frac{1}{N^2}$$

where the last approximation is meant for $N \to \infty$.

Example 3.2 Suppose \mathcal{G} is described by the arcs $i \leftarrow i - 1$ and $i \leftarrow i + 1$ (mod N) and all the self-loops (i, i). For the sake of simplicity we assume that N is even; very similar results can be obtained for odd N. We can choose in this case $P(z) = p_0 + p_1 z + p_{-1} z^{-1}$, where $p_0, p_1, p_{-1} \in \mathbb{R}$. In this case we have that

$$x(t+1) = \{I + p_0 I + p_1 \Pi + p_{-1} \Pi^{-1}\} x(t).$$

The condition P(1) = 1 becomes in this case $p_0 + p_1 + p_{-1} = 1$. Symmetry and convexity arguments [23, 19] allow to say that a minimum of $\rho(P)$ is for sure of the type $p_1 = p_{-1}$. With this assumption the cost functional reduces to

$$\rho(P) = \max\left\{ \left| 1 - 2p_1 \left(1 - \cos\left(\frac{2\pi}{N}\right) \right) \right|, \left| p_0 - 2p_1 \right| \right\}.$$

The minimum is achieved for

$$p_0 = 1 - \frac{2}{3 - \cos\left(\frac{2\pi}{N}\right)}, \quad p_1 = p_{-1} = \frac{1}{3 - \cos\left(\frac{2\pi}{N}\right)}$$

and we have

$$\rho_{\mathcal{G}}^{\text{circ}} = \frac{1 + \cos\left(\frac{2\pi}{N}\right)}{3 - \cos\left(\frac{2\pi}{N}\right)} \simeq 1 - 2\pi^2 \frac{1}{N^2}$$

where the last approximation is meant for $N \to \infty$.

Notice that in the first example the optimality is obtained when all the nonzero elements of π are equal. This is not a general feature since the same does not happen in the second example. Notice moreover that in this example, as N tends to infinity, the optimal solution tends to $p_0 = 0, p_1 = p_{-1} = 1/2$.

The case of communication exchange with two neighbors (Example 3.2) offers a better performance compare to the case with one neighbor (Example 3.1). However, in both cases $\rho_{\mathcal{G}}^{\text{circ}} \to 1$ for $N \to +\infty$. This fact is more general: if we keep bounded the number of incoming edges in a vertex, the essential spectral radius will always converge to 1. This is very easy to see in the case when there is only one incoming edge. Indeed, in this case, by repeating the arguments of Example 3.1 we have the following result.

Proposition 3.3 Consider a strongly connected graph \mathcal{G} on $\{1, \ldots, N\}$, containing all the self-loops (i, i). Assume that \mathcal{G} is symmetric with respect to p and assume there is only one incoming edge in any vertex. Then,

$$\rho_{\mathcal{G}}^{\text{circ}} \ge 1 - \frac{\pi^2}{2} \frac{1}{N^2} \,.$$

In the general situation a much more careful analysis permits to obtain the following bound.

Theorem 3.4 Consider a strongly connected graph \mathcal{G} on $\{1, \ldots, N\}$, containing all the self-loops (i, i). Assume that \mathcal{G} is symmetric with respect to p and let ν be the number of incoming edges in any vertex. Then,

$$\rho_{\mathcal{G}}^{\rm circ} \ge 1 - C \frac{1}{N^{2/\nu}} \,,$$

where C is a constant independent of the chosen graph.

We omit the proofs of the above two results, since they are particular cases of Theorem 3.7, that we will state and prove in the next Section. Moreover, in order to avoid confusion, we remark that, given a node i, we do not encounter the self-loop (i, i) in the set of the incoming arcs to this node i. Notice that in Example 3.2 we have that $\nu = 2$ and we have an asymptotic behavior $\rho_{\mathcal{G}}^{\text{circ}} \simeq 1 - 2\pi^2 N^{-2}$, while the lower bound of Theorem 3.4 is, in this case, $1 - 2\pi^2 N^{-1}$. Now we can wonder whether it is possible to achieve the bound performance. In other words, we would like to understand whether the lower bound we have just found is tight or not. In the following example we will show that this is the case.

Example 3.5 Suppose that $N = M^{\nu}$ and that

$$P = \frac{1}{\nu + 1} \sum_{i=0}^{\nu - 1} \Pi^{M^i}$$

The matrix P has eigenvalues

$$\lambda_h = p_{\nu} \left(e^{j\frac{2\pi}{N}h} \right) \qquad h = 1, \dots, N-1$$

where

$$p_{\nu}(z) := \frac{1}{\nu+1} \left(1 + \sum_{i=0}^{\nu-1} z^{M^{i}} \right)$$

We will show that, for all h = 1, ..., N - 1 we have that

$$\left| p_{\nu}(e^{j\frac{2\pi}{M^{\nu}}h}) \right| \le 1 - \frac{1}{\nu+1} \frac{1}{M^2}$$

This fact will be shown by induction on ν . The fact that the assertion holds for $\nu = 1$ follows from Example 3.1. Assume now that the assertion holds for $\nu - 1$. Let h_0, h_1 such that $0 \le h_0 \le M - 1$, $0 \le h_1 \le M^{\nu-1} - 1$ and $h = h_0 + Mh_1$. If $h_0 \ne 0$ then

$$\begin{split} \left| p_{\nu} \left(e^{j\frac{2\pi}{N}h} \right) \right| &\leq \frac{1}{\nu+1} \left| 1 + e^{j\frac{2\pi}{M}\nu M^{\nu-1}h} \right| + \frac{1}{\nu+1} \left| \sum_{i=0}^{\nu-2} e^{j\frac{2\pi}{M}\nu M^{i}h} \right| \\ &\leq \frac{1}{\nu+1} \left| 1 + e^{j\frac{2\pi}{M}h_{0}} \right| + \frac{\nu-1}{\nu+1} \\ &= \frac{2}{\nu+1} \left| p_{1}(e^{j\frac{2\pi}{M}h}) \right| + \frac{\nu-1}{\nu+1} \\ &\leq \frac{2}{\nu+1} \left(1 - \frac{1}{2}\frac{1}{M^{2}} \right) + \frac{\nu-1}{\nu+1} \\ &\leq 1 - \frac{1}{\nu+1}\frac{1}{M^{2}} \end{split}$$

If $h_0 = 0$, then $h = Mh_1$ and so

$$p_{\nu}(e^{j\frac{2\pi}{N}h})\Big| = \frac{1}{\nu+1} \left| 1 + \sum_{i=0}^{\nu-1} e^{j\frac{2\pi}{M^{\nu-1}}M^{i}h_{1}} \right|$$
$$= \frac{1}{\nu+1} \left| 2 + \sum_{i=0}^{\nu-2} e^{j\frac{2\pi}{M^{\nu-1}}M^{i}h_{1}} \right|$$
$$= \frac{\nu}{\nu+1} \left| p_{\nu-1}(e^{j\frac{2\pi}{M}h_{1}}) \right| + \frac{1}{\nu+1}$$
$$\leq \frac{\nu}{\nu+1} \left(1 - \frac{1}{\nu}\frac{1}{M^{2}} \right) + \frac{1}{\nu+1}$$
$$\leq 1 - \frac{1}{\nu+1}\frac{1}{M^{2}}$$

This bound proves that there exists a circulant graph \mathcal{G} with ν incoming edges in any vertex such that

$$\rho_{\mathcal{G}}^{\text{circ}} \le 1 - \frac{1}{\nu + 1} \frac{1}{N^{2/\nu}}.$$

proving in this way that the bound proposed by the previous theorem is tight.

3.3 Cayley symmetries

The analysis carried out in the previous section can be extended to graphs \mathcal{G} and matrices P exhibiting more general symmetries.

In order to treat symmetries on a graph \mathcal{G} in a general setting, we need to introduce the concept of Cayley graphs defined on Abelian groups [10, 4]. Let G be any finite Abelian group of order |G| = N, and let S be a subset of G containing zero. The Cayley graph $\mathcal{G}(G, S)$ is the directed graph with vertex set G and arc set

$$\mathcal{E} = \{(g,h) : h - g \in S\}.$$

Note that the fact that S contains zero implies that all the self loops (g, g) belong to \mathcal{E} . From now on, along this section, when dealing with a graph \mathcal{G} we will assume that it contains all the self-loops.

Notice that a Cayley graph is always in-regular, namely the in-degree of each vertex is equal to |S|. Notice also that strongly connectivity can be checked

algebraically. Indeed, it can be seen that a Cayley graph $\mathcal{G}(G, S)$ is strongly connected if and only if the set S generates the group G, which means that any element in G can be expressed as a finite sum of (not necessarily distinct) elements in S. If S is such that -S = S we say that S is inverse-closed. In this case the graph obtained is undirected.

Symmetries can be introduced also on matrices. Let G be any finite Abelian group of order |G| = N. A matrix $P \in \mathbb{R}^{G \times G}$ is said to be a Cayley matrix over the group G if

$$P_{i,j} = P_{i+h,j+h} \qquad \forall \ i, j, h \in G$$
.

It is clear that for a Cayley matrix P there exists a $\pi : G \to \mathbb{R}$ such that $P_{i,j} = \pi(i-j)$. The function π is called the generator of the Cayley matrix P. Notice that, if π and π' are generators of the Cayley matrices P and P' respectively, then $\pi + \pi'$ is the generator of P + P' and $\pi * \pi'$ is the generator of PP', where $(\pi * \pi')(i) := \sum_{j \in G} \pi(j)\pi'(i-j)$ for all $i \in G$. This shows that P and P' commute. Notice finally that, if P is a Cayley matrix generated by π , then \mathcal{G}_P is a Cayley graph with $S = \{h \in G : \pi(h) \neq 0\}$.

It is clear that the graph \mathcal{G}_P supporting a Cayley matrix P is a Cayley graph with

$$S := \{i : \pi(i) \neq 0\} \cup \{0\}.$$

Moreover it is easy to see that for any Cayley matrix P we have that $P\mathbb{1} = \mathbb{1}$ if and only if $\mathbb{1}^T P = \mathbb{1}^T$. This implies that a Cayley stochastic matrix is automatically doubly stochastic. In this case the function π associated with the matrix P is a probability distribution on the group G. Among the multiple possible choices of the probability distribution π , there is one which is particularly simple, namely $\pi(g) = 1/|S|$ for every $g \in S$.

Example 3.6 Let us consider the group \mathbb{Z}_N of integers modulo N and the Cayley graph $\mathcal{G}(\mathbb{Z}_N, S)$ where $S = \{-1, 0, 1\}$. Notice that in this case S is inverse-closed. Consider the uniform probability distribution

$$\pi(0) = \pi(1) = \pi(-1) = 1/3$$

The corresponding Cayley stochastic matrix is given by

$$P = \begin{pmatrix} 1/3 & 1/3 & 0 & 0 & \cdots & 0 & 0 & 1/3 \\ 1/3 & 1/3 & 1/3 & 0 & \cdots & 0 & 0 & 0 \\ 0 & 1/3 & 1/3 & 1/3 & \cdots & 0 & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \cdots & \vdots & \vdots & \vdots \\ 1/3 & 0 & 0 & 0 & \cdots & 0 & 1/3 & 1/3 \end{pmatrix}$$

Notice that in this case we have two symmetries. The first is that the graph is undirected and the second that the graph is circulant. These symmetries can be seen in the structure of the transition matrix P that, indeed, turns out to be both symmetric and circulant.

Given a Cayley graph \mathcal{G} we can define

$$\rho_{\mathcal{G}}^{\text{Cayley}} = \min\{\rho(P) \mid P \text{ Cayley stochastic, } \mathcal{G}_P \subseteq \mathcal{G}\}.$$

It will turn out that $\rho_{\mathcal{G}}^{\text{Cayley}}$ can be evaluated or estimated in many cases. Moreover, it clearly holds that $\rho_{\mathcal{G}}^{\text{Cayley}} \geq \rho_{\mathcal{G}}$. Before continuing we give a short analysis of the spectral properties of the Cayley stochastic matrices on finite Abelian groups, which is the basis of our main results.

3.3.1 Cayley stochastic matrices on finite Abelian groups

In this subsection we will refer to notions on group characters and on harmonic analysis on groups which are illustrated in Appendix A.

Fix now a Cayley matrix P on the Abelian group G generated by the function $\pi: G \to \mathbb{R}$. The spectral structure of P is very simple. To see this, first notice that P can be interpreted as a linear function from \mathbb{C}^G to itself simply by considering, for $f \in \mathbb{C}^G$, $(Pf)(g) := \sum_h P_{gh}f(h)$. Notice that the trivial character χ_0 corresponds to the vector $\mathbb{1}$ having all components equal to 1. For every $\chi \in \hat{G}$, it holds

$$(P\chi)(g) = \sum_{h \in G} P_{gh}\chi(h) = \sum_{h \in G} \pi(g-h)\chi(h) = \sum_{h \in G} \pi(h)\chi(g-h) = \hat{\pi}(\chi)\chi(g).$$

Hence, χ is an eigenfunction of P with eigenvalue $\hat{\pi}(\chi)$. Since the characters form an orthonormal basis it follows that P is diagonalizable and its spectrum is given by

$$\sigma(P) = \{\hat{\pi}(\chi) \mid \chi \in \hat{G}\}$$

We can interpret a character as a linear function from \mathbb{C} to \mathbb{C}^{G} as follows

$$\chi: \mathbb{C} \to \mathbb{C}^G : z \mapsto z\chi.$$

Its adjoint is the linear functional

$$\chi^*: \mathbb{C}^G \to \mathbb{C} : f \mapsto < f, \chi > .$$

With this notation, $N^{-1}\chi\chi^*$ is a linear function from \mathbb{C}^G to itself, projecting \mathbb{C}^G on the eigenspace generated by χ . In this way, P can be represented as

$$P = \sum_{\chi \in \hat{G}} \hat{\pi}(\chi) N^{-1} \chi \chi^*.$$

Conversely, it can easily be shown that, given any $\hat{\theta} : \hat{G} \to \mathbb{C}$, the matrix

$$P = \sum_{\chi \in \hat{G}} \hat{\theta}(\chi) N^{-1} \chi \chi^* \,,$$

is a Cayley matrix generated by the Fourier transform θ .

Suppose now that P is the matrix of the system (2.11). The displacement from the $\Delta(t)$, defined in (2.15), can be represented as

$$\Delta(t) = (I - N^{-1}\chi_0\chi_0^*)x(t)$$

As we had already remarked, Δ is governed by the dynamics $\Delta^+ = P\Delta$ and the initial condition $\Delta(0)$ is characterized by $\langle \Delta(0), \chi_0 \rangle = 0$. Notice that

$$\Delta(t) = P^t \Delta(0) = \frac{1}{N} \sum_{\chi \in \hat{G}} \hat{\pi}(\chi)^t \chi < \Delta(0), \chi > = \frac{1}{N} \sum_{\chi \neq \chi_0} \hat{\pi}(\chi)^t \chi < \Delta(0), \chi > .$$

Hence,

$$||\Delta(t)||^{2} = \frac{1}{N} \sum_{\chi \neq \chi_{0}} |\hat{\pi}(\chi)|^{2t}| < \Delta(0), \chi > |^{2}.$$

This shows in a very simple way, in this case, the role of $\rho(P) = \max_{\chi \neq \chi_0} |\hat{\pi}(\chi)|$ in the rate of convergence.

3.3.2 The essential spectral radius of Cayley matrices.

The particular spectral structure of Cayley matrices allows to obtain asymptotic results on the behavior of the essential spectral radius $\rho(P)$ and therefore on the rate of convergence of the corresponding consensus algorithm. In Section 3.2.1 we have proved the following fact for circulant graphs: if we keep bounded the number of incoming arcs in a vertex, the essential spectral radius for stochastic circulant matrices always converges to 1, for $N \to \infty$. The next result provides a bound which proves that this negative behavior is a general feature also of the broader class of the Abelian stochastic Cayley matrices. This slow convergence rate has already been noticed, for some specific case, in the literature [19, 127, 97, 24]. **Theorem 3.7** Let G be any finite Abelian group of order N and let $S \subseteq G$ be a subset containing zero. Moreover let \mathcal{G} be the Cayley graph associated with G and S. If $|S| = \nu + 1$, then

$$\rho_{\mathcal{G}}^{\text{Cayley}} \ge 1 - CN^{-2/\nu} \,, \tag{3.2}$$

where C > 0 is a constant independent of G and S.

In order to prove Theorem 3.7 we need the following technical lemma.

Lemma 3.8 Let $\mathbb{T} = \mathbb{R}/\mathbb{Z} \cong [-1/2, 1/2[$. Let $0 \leq \delta < 1/2$ and consider the hypercube $V = [-\delta, \delta]^k \subseteq \mathbb{T}^k$. For every finite set $\Lambda \subseteq \mathbb{T}^k$ such that $|\Lambda| \geq \delta^{-k}$, there exist $\bar{x}_1, \bar{x}_2 \in \Lambda$ with $\bar{x}_1 \neq \bar{x}_2$ such that $\bar{x}_1 - \bar{x}_2 \in V$.

Proof: For any $x \in \mathbb{T}$ and $\delta > 0$, define the following set

$$L(x,\delta) = [x, x+\delta] + \mathbb{Z} \subseteq \mathbb{T}.$$

Observe that for all $y \in \mathbb{T}$, $L(x,\delta) + y = L(x + y,\delta)$. Now let $\bar{x} = (\bar{x}_1, \ldots, \bar{x}_k) \in \mathbb{T}^k$ and define

$$L(\bar{x},\delta) = \prod_{i=1}^{k} L(\bar{x}_i,\delta) \,.$$

Also in this case we observe that $L(\bar{x}, \delta) + \bar{y} = L(\bar{x} + \bar{y}, \delta)$ for every $\bar{y} \in \mathbb{T}^k$. Consider now the family of subsets

$$\{L(\bar{x},\delta), \ \bar{x}\in\Lambda\}.$$

We claim that there exist \bar{x}_1 and \bar{x}_2 in Λ such that $\bar{x}_1 \neq \bar{x}_2$ and such that $L(\bar{x}_1, \delta) \cap L(\bar{x}_2, \delta) \neq \emptyset$. Indeed, if not, we would have that

$$1 \ge m\left(\bigcup_{\bar{x} \in \Lambda} L(\bar{x}, \delta)\right) = \sum_{\bar{x} \in \Lambda} m\left(L(\bar{x}, \delta)\right) = |\Lambda| \delta^k \ge 1$$

where $m(\cdot)$ is the Lebesgue measure on \mathbb{T}^k and where we used the hypothesis $|\Lambda| \geq \delta^{-k}$. However, since all $L(\bar{x}_1, \delta)$ are closed, it is not possible that $m\left(\bigcup_{\bar{x}\in\Lambda}L(\bar{x}_1,\delta)\right) = 1$. Notice finally that

$$L(\bar{x}_1, \delta) \cap L(\bar{x}_2, \delta) \neq \emptyset \iff L(0, \delta) \cap L(\bar{x}_2 - \bar{x}_1, \delta) \neq \emptyset \iff \bar{x}_2 - \bar{x}_1 \in V$$

We are able now to provide the proof of Theorem 3.7. *Proof:* From Theorem A.9 stated in Appendix A, we can assume, with no loss of generality, that

$$G = \mathbb{Z}_{N_1} \oplus \ldots \oplus \mathbb{Z}_{N_r}$$
.

Assume we have fixed a probability distribution π supported on S. Let P be the corresponding stochastic Cayley matrix. It follows from previous considerations that the spectrum of P is given by

$$\sigma(P) = \begin{cases} \sum_{k_1=0}^{N_1-1} \sum_{k_2=0}^{N_2-1} \dots \sum_{k_r=0}^{N_r-1} \pi(k_1, \dots, k_r) e^{i\frac{2\pi}{N_1}k_1\ell_1} e^{i\frac{2\pi}{N_2}k_2\ell_2} \dots e^{i\frac{2\pi}{N_r}k_r\ell_r} :\\ :\ell_1 \in \mathbb{Z}_{N_1}, \dots, \ell_r \in \mathbb{Z}_{N_r} \end{cases}$$

Denote by $\bar{k}^j = (k_1^j, \ldots, k_r^j)$, for $j = 1, \ldots, \nu$, the non-zero elements in S, and consider the subset

$$\Lambda = \left\{ \left(\sum_{i=1}^r \frac{k_i^1 \ell_i}{N_i}, \dots, \sum_{i=1}^r \frac{k_i^\nu \ell_i}{N_i} \right) + \mathbb{Z}^\nu : \ell_1 \in \mathbb{Z}_{N_1}, \dots, \ell_r \in \mathbb{Z}_{N_r} \right\} \subseteq \mathbb{T}^\nu.$$

Let $\delta = (\prod_i N_i)^{-1/\nu}$ and let V be the corresponding hypercube in \mathbb{T}^{ν} defined as in Lemma 3.8. We want to show that there exists $\bar{\ell} = (\ell_1, \ldots, \ell_r) \in \mathbb{Z}_{N_1} \times \cdots \times \mathbb{Z}_{N_r}, \ \bar{\ell} \neq 0$ such that

$$\left(\sum_{i=1}^r \frac{k_i^1 \ell_i}{N_i}, \dots, \sum_{i=1}^r \frac{k_i^\nu \ell_i}{N_i}\right) + \mathbb{Z}^\nu \in V.$$

We consider two cases.

(i) If there exists $\bar{\ell} = (\ell_1, \dots, \ell_r) \in \mathbb{Z}_{N_1} \times \dots \times \mathbb{Z}_{N_r}, \ \bar{\ell} \neq 0$ such that

$$\left(\sum_{i=1}^{r} \frac{k_i^1 \ell_i}{N_i}, \dots, \sum_{i=1}^{r} \frac{k_i^{\nu} \ell_i}{N_i}\right) + \mathbb{Z}^{\nu} = 0 \in V$$
(3.3)

then clearly we are done.

(ii) Assume now there are no $\bar{\ell} = (\ell_1, \dots, \ell_r) \in \mathbb{Z}_{N_1} \times \dots \times \mathbb{Z}_{N_r}, \ \bar{\ell} \neq 0$ satisfying condition (3.3). In this case it can be shown that two different $\bar{\ell}', \bar{\ell}'' \in \mathbb{Z}_{N_1} \times \dots \times \mathbb{Z}_{N_r}$ yield

$$\left(\sum_{i=1}^{r} \frac{k_i^1 \ell_i'}{N_i}, \dots, \sum_{i=1}^{r} \frac{k_i^{\nu} \ell_i'}{N_i}\right) + \mathbb{Z}^{\nu} \neq \left(\sum_{i=1}^{r} \frac{k_i^1 \ell_i''}{N_i}, \dots, \sum_{i=1}^{r} \frac{k_i^{\nu} \ell_i''}{N_i}\right) + \mathbb{Z}^{\nu},$$

namely different elements in $\mathbb{Z}_{N_1} \times \cdots \times \mathbb{Z}_{N_r}$ always lead do distinct elements in Λ . This implies that $|\Lambda| = \prod_i N_i = \delta^{-\nu}$ and so we are in a position to apply Lemma 3.8 and conclude that there exist two different $\bar{\ell}', \bar{\ell}'' \in \mathbb{Z}_{N_1} \times \cdots \times \mathbb{Z}_{N_r}$ such that

$$\left[\left(\sum_{i=1}^r \frac{k_i^1 \ell_i'}{N_i}, \dots, \sum_{i=1}^r \frac{k_i^\nu \ell_i'}{N_i}\right) + \mathbb{Z}^\nu\right] - \left[\left(\sum_{i=1}^r \frac{k_i^1 \ell_i''}{N_i}, \dots, \sum_{i=1}^r \frac{k_i^\nu \ell_i''}{N_i}\right) + \mathbb{Z}^\nu\right] \in V$$

and hence

$$\left(\sum_{i=1}^r \frac{k_i^1 \ell_i}{N_i}, \dots, \sum_{i=1}^r \frac{k_i^{\nu} \ell_i}{N_i}\right) + \mathbb{Z}^{\nu} \in V,$$

where $\bar{\ell} = \bar{\ell}' - \bar{\ell}'' \neq 0$.

Consider now the eigenvalue

$$\lambda = \sum_{k_1=0}^{N_1-1} \sum_{k_2=0}^{N_2-1} \dots \sum_{k_r=0}^{N_r-1} \pi(k_1, \dots, k_r) e^{i(\frac{2\pi}{N_1}k_1\ell_1 + \frac{2\pi}{N_2}k_2\ell_2 + \dots + \frac{2\pi}{N_r}k_r\ell_r)}$$

= $\pi(0, \dots, 0) + \sum_{j=1}^{\nu} \pi(k_1^j, \dots, k_r^j) e^{i(\frac{2\pi}{N_1}k_1^j\ell_1 + \frac{2\pi}{N_2}k_2^j\ell_2 + \dots + \frac{2\pi}{N_r}k_r^j\ell_r)}.$

Its norm can be estimated as follows

$$\begin{aligned} |\lambda| &\geq \pi(0, \dots 0) + \sum_{j=1}^{\nu} \pi(k_1^j, \dots, k_r^j) \cos\left(\frac{2\pi}{N_1} k_1^j \ell_1 + \frac{2\pi}{N_2} k_2^j \ell_2 + \dots + \frac{2\pi}{N_r} k_r^j \ell_r\right) \\ &\geq \pi(0, \dots 0) + \sum_{j=1}^{\nu} \pi(k_1^j, \dots, k_r^j) \left(1 - 2\pi^2 \left(\frac{k_1^j}{N_1} \ell_1 + \frac{k_2^j}{N_2} \ell_2 + \dots + \frac{k_r^j}{N_r} \ell_r\right)^2\right) \\ &\geq \pi(0, \dots 0) + \sum_{j=1}^{\nu} \pi(k_1^j, \dots, k_r^j) - \sum_{j=1}^{\nu} \pi(k_1^j, \dots, k_r^j) 2\pi^2 \frac{1}{N^{2/\nu}} \\ &\geq 1 - 2\pi^2 \frac{1}{N^{2/\nu}} \end{aligned}$$

and so we can conclude.

Theorem 3.7 in particular implies that, if we consider a sequence of Abelian Cayley graphs $\mathcal{G}(G_N, S_N)$ such that $|G_N| = N$ and $|S_N|$ grows less then logarithmically in N and we consider a sequence of Cayley stochastic matrices P_N compatible with $\mathcal{G}(G_N, S_N)$, then, necessarily, $\rho(P_N)$ converges to 1. This had already been shown, for adjacency matrices, in [4].

Notice that Example 3.5 shows that also the bound we have just found is tight. However we provide now another interesting example confirming this.

Consider the group $(\mathbb{Z}_N^r, +)$, with $r \in \mathbb{N}$ and $r \geq 1$, and the Cayley graph $\mathcal{G}(\mathbb{Z}_N^r, S)$, where $S = \{0, e_1, \ldots, e_r\}$, where e_j is the vector with all elements equal to 0 except a 1 in position j. Consider the probability distribution π on S described by

$$\pi(0) = k_0, \ \pi(e_j) = k_j, \ \forall j = 1, \dots, r$$

with $k_j \ge 0$ and $\sum_{j=0}^r k_j = 1$. The Fourier transform of π is

$$\hat{\pi}(\chi_{\ell_1}, \dots, \chi_{\ell_r}) = \sum_{g \in S} \chi(-g)\pi(g) = k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j}$$

for $\ell_j = 0, 1, \dots, N-1$, $j = 1, \dots, r$. We thus have

$$\rho_{\mathcal{G}}^{\text{Cayley}} = \min_{\substack{k_j \ge 0 \\ \sum k_j = 1 \ (\ell_1, \dots, \ell_r) \neq (0, \dots, 0)}} \max_{\substack{k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j}} \\ k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j} \\ k_0 + k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j} \\ k_0 + k_$$

We have the following result.

Proposition 3.9 The above min-max is reached by $(\ell_1, ..., \ell_r) = (1, ..., 1)$ or by $(\ell_1, ..., \ell_r) = (0, ..., 1, ..., 0)$ and $k_0 = k_1 = ... = k_r = 1/(r+1)$, which yield

$$\rho_{\mathcal{G}}^{\text{Cayley}} = \left(1 - \frac{2r}{(r+1)^2} \left(1 - \cos\frac{2\pi}{N}\right)\right)^{\frac{1}{2}} \simeq 1 - \frac{4\pi^2 r}{(r+1)^2} \frac{1}{N^2}$$

In order to prove the above proposition we need the following technical Lemma.

Lemma 3.10 Suppose that k_0, \ldots, k_r are fixed. Then

$$\max_{\substack{0 \le \ell_j \le N-1 \\ (\ell_1, \dots, \ell_r) \ne (0, \dots, 0)}} \left| k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j} \right| = \max_{\substack{\ell_j \in \{0,1\} \\ (\ell_1, \dots, \ell_r) \ne (0, \dots, 0)}} \left| k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j} \right|$$

Proof: Consider a *r*-tuple $(\ell_1, \ldots, \ell_r) \neq (0, \ldots, 0)$ and let $\ell \neq 0$ be a value taken by some of the ℓ_j . Let $J \subseteq \{1, \ldots, r\}$ be the nonempty index set such that $\ell_j = \ell$ if and only if $j \in J$. Consider the new *r*-tuple $(\ell'_1, \ldots, \ell'_r)$ defined as follows

$$\ell'_j = \begin{cases} 1 \text{ if } j \in J\\ 0 \text{ otherwise} \end{cases}$$

We want to show that

$$\left|k_{0} + \sum_{j=1}^{r} k_{j} e^{-i\frac{2\pi}{N}\ell_{j}}\right| \leq \left|k_{0} + \sum_{j=1}^{r} k_{j} e^{-i\frac{2\pi}{N}\ell_{j}'}\right|$$
(3.4)

Indeed, we have that

$$\begin{aligned} \left| k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi\ell_j}{N}} \right|^2 &= \\ &= \left| k_0 + \sum_{j=1}^r k_j \cos \frac{2\pi\ell_j}{N} - i \left(\sum_{j=1}^r k_j \sin \frac{2\pi\ell_j}{N} \right) \right|^2 = \\ &= \left(k_0 + \sum_{j=1}^r k_j \cos \frac{2\pi\ell_j}{N} \right)^2 + \left(\sum_{j=1}^r k_j \sin \frac{2\pi\ell_j}{N} \right)^2 \\ &= \sum_{j=0}^r k_j^2 + 2k_0 \sum_{j=1}^r k_j \cos \left(\frac{2\pi\ell_i}{N} \right) + \\ &+ 2\sum_{i=1}^{r-1} \sum_{j=i+1}^r k_i k_j \cos \left(\frac{2\pi\ell_i}{N} \right) \cos \left(\frac{2\pi\ell_j}{N} \right) + \\ &+ 2\sum_{i=1}^{r-1} \sum_{j=i+1}^r k_i k_j \sin \left(\frac{2\pi\ell_i}{N} \right) \sin \left(\frac{2\pi\ell_j}{N} \right) \\ &= \sum_{j=0}^r k_j^2 + 2k_0 \sum_{j=1}^r k_j \cos \left(\frac{2\pi\ell_i}{N} \right) + 2\sum_{i=1}^{r-1} \sum_{j=i+1}^r k_i k_j \cos \left(\frac{2\pi\ell_i}{N} \right) \end{aligned}$$

Observe now that

$$\cos\left(\frac{2\pi\ell_i}{N}\right) \le \cos\left(\frac{2\pi\ell_i'}{N}\right)$$

for all j, and that

$$\cos\left(\frac{2\pi(\ell_j-\ell_i)}{N}\right) \le \cos\left(\frac{2\pi(\ell_j'-\ell_i')}{N}\right)$$

for all i, j. This yields (3.4) and proves the lemma.

We can now prove Proposition 3.9.

Proof:

We start by observing that, if $\ell_i \in \{0, 1\}$, then

$$\left| k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j} \right|^2 = 1 - 2\left(\sum_{j\in J} k_j\right) \left(1 - \sum_{j\in J} k_j\right) \left(1 - \cos\frac{2\pi}{N}\right)$$

where $J \subseteq \{1, \ldots, r\}$ is such that $j \in J$ if and only if $\ell_j = 1$. Maximizing this quantity over the non identically zero vectors $(\ell_1, \ldots, \ell_r) \in \{0, 1\}^r$ is equivalent to maximize over all the possible nonempty sets J. From this it is not difficult to see that

$$\max_{\substack{0 \le \ell_j \le N-1\\ (\ell_1,\dots,\ell_r) \ne (0,\dots,0)}} \left| k_0 + \sum_{j=1}^r k_j e^{-i\frac{2\pi}{N}\ell_j} \right|^2 = 1 - 2m(1-m)\left(1 - \cos\frac{2\pi}{N}\right)$$

where $m := \min\{k_0, k_1, \ldots, k_r\}$. Since $m \le 1/2$, then in order to minimize $1 - 2m(1-m)\left(1 - \cos\frac{2\pi}{N}\right)$ over the possible k_0, k_1, \ldots, k_r , we need to maximize m. This is obtained by choosing $k_0 = k_1 = \ldots = k_r = 1/(r+1)$, which yield m = 1/(r+1) and

$$\rho_{\mathcal{G}}^{\text{Cayley}} = \left(1 - \frac{2r}{(r+1)^2} \left(1 - \cos\frac{2\pi}{N}\right)\right)^{\frac{1}{2}} \simeq 1 - \frac{4\pi^2 r}{(r+1)^2} \frac{1}{N^2}.$$

From previous result we see that, by keeping N fixed and by varying r we obtain a sequence of matrices for which the rate of convergence tends to 1 logarithmically in the number of systems and for which the degree of the associated graph grows logarithmically in the number of systems.

The previous example is based on a Cayley graph over the group \mathbb{Z}_N^r . The same behavior exhibited in this example can also be obtained starting from a Cayley graph over the cyclic group \mathbb{Z}_N . Indeed if we take \mathbb{Z}_{N^r} and the subset

$$S := \{0, 1, N, N^2, \dots, N^{r-1}\},\$$

we can construct a Cayley stochastic matrix with an essential spectral radius that is asymptotically equivalent to the one obtained in the previous example. More in general, it can be proved that, if in the family of groups \mathbb{Z}_N^r we maintain N fixed (prime) and we vary r, there exists a constant c < 1 such that for every r there exists $S_r \subseteq \mathbb{Z}_N^r$ such that $|S_r| \simeq cr$ and such that

$$\rho(P_r) \le d < 1 \, .$$

where $P_r = |S_r|^{-1}A_r$ and where A_r is the adjacency matrices of the corresponding Cayley graph. Such Cayley graphs are constructed using the theory of channel codes over finite fields [4]. Extensions to non-prime N are likely to be possible considering the theory of group codes.

The question at this point is the following: Is the Cayley structure on the matrix or the Cayley structure on the graph that prevents to obtain good performance? In other words, do there exist stochastic matrices supported by Abelian Cayley graphs that exhibit better performance than what imposed by the bound (3.2)? Notice that, in order to make fair comparisons, we need to limit to doubly stochastic matrices. We conjecture that for doubly stochastic matrices supported on Abelian Cayley graphs the bound (3.2) continues to hold.

What about other graphs? An easy way to restrict to doubly stochastic matrices is by imposing that they are symmetric and so that the corresponding graphs are undirected. If A is the adjacency matrix of a ν -regular undirected graph, then, $P = \nu^{-1}A$ is doubly stochastic. For these graphs, we recall a basic asymptotic lower bound by Alon and Boppana [3] on the second eigenvalue

$$\liminf_{N \to +\infty} \rho(P) \ge \frac{2\sqrt{\nu - 1}}{\nu},$$

where the lim inf is intended to be performed along the family of all ν -regular undirected graphs having N vertices.

Ramanujan graphs (see [89] and references therein) are those ν -regular undirected graphs achieving the previous bound, namely such that $\rho(P) = 2\nu^{-1}\sqrt{\nu-1}$. Hence, through these graphs, it would be possible to keep the essential spectral radius bounded away from 1, while keeping the degree fixed. In fact, there are plenty of Ramanujan graphs (for instance any complete graph), but it is still an open problem if for any N and ν there exists a Ramanujan graph with N vertices and degree ν . There are only partial results in this direction. For example it is possible to prove that, if ν is such that $\nu - 1$ is the power of a prime, then there exist a sequence of Ramanujan graphs with a growing number of vertices and of fixed degree ν . Moreover, when available, these constructions are quite complicated and the fact that they strictly depend on the choice of particular number of vertices makes them not so interesting from our point of view. However, it is interesting to notice that graphs behaving similarly to the Ramanujan ones are not so unlikely. Indeed Friedman [62] showed that, by averaging the essential spectral radius of the adjacency matrices A of all undirected ν -regular Cayley graphs having N vertices, we obtain for $P = \nu^{-1}A$

$$\mathbb{E}(\rho(P)) \le \frac{2\sqrt{\nu-1}}{\nu} \left(1 + \frac{\ln\nu/2}{\sqrt{\nu}} + O\left(\frac{1}{\sqrt{\nu}}\right)\right) + O\left(\frac{\nu^{1/2}\ln\ln N}{\ln N}\right) + O\left(\frac$$

Of course the previous bound has to be interpreted in an asymptotic sense for $N \to +\infty$ and $\nu \to +\infty$. As a consequence we have in particular that, if we fix ν sufficiently large, in the average, $\rho(P)$ will remain bounded away from 1 as $N \to +\infty$.

3.4 Conclusions

We have derived bounds on the convergence rate to the average consensus for a set of systems that exchange information over time-invariant communication networks with symmetries. We have showed that, in time-invariant networks, symmetries yield rather slow convergence to the average consensus. In particular for such networks we have computed a tight bound for the convergence rate.

Chapter 4

Randomized Consensus Algorithms

4.1 Introduction

In the previous chapter we have considered the time-invariant average consensus algorithm (2.10). In this chapter we will focus on the more general time-varying case by addressing, in particular, the analysis of the so-called *randomized consensus algorithms*. Roughly speaking a randomized algorithm works as follows. Assume we have a family \mathcal{P} of quasi-stochastic matrices. Then, at each iteration of the consensus algorithm, the consensus matrix P(t) is chosen randomly inside this set of matrices, according to a certain probability distribution probability. For a randomized algorithm, convergence is considered in a probabilistic sense and performance are studied in mean square sense.

The use of the *random algorithms* is quite appealing, since they take into consideration some fundamental limitations arising in the realistic implementation of the consensus algorithms. Indeed, it should be pointed out that, in many practical applications, a node can not simultaneously receive data from two different neighbor nodes (for instance collision can delete messages in wireless environment) and in some applications it cannot simultaneously transmit to more than a node (this happens for instance for processors nets).

In this chapter, we consider three interesting examples of randomized consensus algorithms. The first two emphasize a remarkable feature of the randomized consensus algorithms, i.e., that they allow to achieve better performance than the deterministic ones with comparable complexity. This comparison is done by considering the consensus algorithms having as communication graphs the Abelian Cayley graphs presented in the previous chapter. The last example reviews the symmetric gossip algorithms introduced in [22] and [24] (other random linear schemes under the name of gossip algorithms have been studied for instance in [51] and [76]). As underlined in [24], the gossip algorithms are mainly motivated by applications to sensors, peer-topeer and *ad hoc* networks. In general, these networks are constrained by the following operational characteristics: i) they may not have a centralized entity for facilitating computation, communication, and time-synchronization, ii) the network topology may not be completely known to the nodes of the network, iii) nodes may join or leave the network (even expire), so that the network topology itself may change, and iv) in the case of sensor networks, the computational power and energy resources may be very limited. These constraints have motivated the design of gossip algorithms: schemes which distribute the computational burden and in which a node communicates with a randomly chosen neighbor.

The Chapter is organized as follows. In Section 4.2, we will introduce the randomized consensus algorithms and we will adopt, from [58], the concept of probabilistic consensus. In Section 4.3, following [58], we will review some algebraic conditions ensuring the achievement of the probabilistic consensus. Moreover, we will introduce two particular randomized strategies: the time-varying strategy with bounded in-degree and the time-varying Cayley graphs strategy. In Section 4.4, we will provide some criteria in order to evaluate the performance of randomized consensus algorithms. We will analyze, from this point of view, the time-varying strategy with bounded in-degree and the time-varying Cayley graphs strategy. In Section 4.5, we will introduce the symmetric gossip algorithms, particularizing them for the class of Cayley graphs. Finally, in Section 4.6, we will summarize the main conclusions.

4.2 Problem formulation

Consider the time varying consensus problem, as defined in Section 2.2. Namely, we have N coupled linear dynamical systems

$$x_i(t+1) = \sum_{j=1}^{N} P_{ij}(t) x_j(t) \qquad i = 1, \dots, N$$
(4.1)

where $x_i(t) \in \mathbb{R}$ is the state of the *i*-th system at time *t* and $P_{ij}(t) \in \mathbb{R}$ coefficients which vary with the time *t*. More compactly, we can write

$$x(t+1) = P(t)x(t),$$

where $x(t) \in \mathbb{R}^N$ and $P(t) \in \mathbb{R}^{N \times N}$. We recall that the sequence P(t) is said to achieve consensus if conditions (a) and (b) stated in Section 2.2 are satisfied.

In this chapter, we will assume to have statistical information on the matrices P(t) and we will adopt the probabilistic approach introduced in [58], instead of a worst case analysis considered in [138, 74, 88, 112, 16]. More precisely, in this chapter we will assume that P(t) is a sequence of i.i.d. matrix valued random variables and x(t) is the stochastic process which is the solution of the equation (4.1). We say that the sequence P(t) achieves the *probabilistic consensus* if condition (a) holds while (b) is replaced by

(b) For any $x(0) \in \mathbb{R}^N$, there exists a scalar random variable α such that

$$\lim_{t \to \infty} x(t) = \alpha \mathbb{1} \qquad \text{almost surely.} \tag{4.2}$$

If $\alpha = N^{-1} \mathbb{1}^* x(0)$ almost surely, we talk about *probabilistic average consensus*.

In this chapter, we will restrict to cases in which P(t) are stochastic matrices. Notice that condition (a) is then clearly automatically satisfied. If, moreover, P(t) is doubly stochastic, then the average is invariant, namely $N^{-1}\mathbb{1}^*x(t) = N^{-1}\mathbb{1}^*x(0)$ for every t and hence in this case, consensus implies average consensus. Let now

$$Q(t) = P(t-1)\cdots P(0),$$

so that we can write x(t) = Q(t)x(0). The random variable α in (4.2) is a linear function of the initial condition x(0) so that we can write $\alpha = \rho^* x(0)$ for some random variable ρ taking values in \mathbb{R}^N and such that $\mathbb{1}^* \rho = 1$. Therefore probabilistic consensus can be equivalently expressed by saying that there exists a random variable ρ taking values in \mathbb{R}^N such that

$$\lim_{t \to \infty} Q(t) = \mathbb{1}\rho^* \tag{4.3}$$

almost surely. Moreover notice that $1\rho^*$ is a matrix whose rows are all equal to ρ^* . Notice that $x(\infty) = \rho^* x(0)$. We have probabilistic average consensus exactly when $\rho = N^{-1}1$ almost surely.

4.3 Conditions for the probabilistic consensus

In this section we briefly review some conditions which ensures the probabilistic consensus. In doing this, we follow the approach adopted in [58], where the authors show that the probabilistic consensus turns out to be an easily checkable property, namely as easily checkable as the deterministic consensus in the time-invariant case.

We immediately state the following result which appears in [37].

Theorem 4.1 The algorithm P(t) achieves probabilistic consensus if and only if for every pair of systems i, j we have that

$$\mathbb{P}[\mathcal{E}_{ij}] = 1$$

where

$$\mathcal{E}_{ij} = \{ \exists k, \exists t | Q_{ik}(t)Q_{jk}(t) > 0 \}.$$

To obtain a more handy condition, the authors in [58] impose a hypothesis which is however always satisfied in all the cases which are commonly considered in literature. In the following proposition they assume that all the diagonal elements of P(t) are nonzero with probability 1. The simple condition, ensuring probabilistic consensus, that they have found is a direct consequence of the above Theorem and is based on the expected value of P(t) which is denoted as $\bar{P} = \mathbb{E}[P(t)]$. We have the following result, whose proof is in [58].

Proposition 4.2 Assume that for any *i* we have that $P(t)_{ii} > 0$ almost surely. If $\mathcal{G}_{\bar{P}}$ is strongly connected, then P(t) achieves probabilistic consensus.

Remark 4.3 Notice, as remarked in [58], that there is a sort of weak converse to previous result. Indeed, assume that the sequence P(t) achieves probabilistic consensus. From the almost sure convergence $Q(t) \to \mathbb{1}\rho^*$ and Lebesgue dominated convergence theorem it follows that $\mathbb{E}[Q(t)] \to \mathbb{1}\mathbb{E}[\rho]^*$ and so $\bar{P}^t \to \mathbb{1}\mathbb{E}[\rho]^*$. In other words, \bar{P} achieves consensus. It can very well happen that $\mathbb{E}\rho = \mathbb{1}$. even if ρ is not equal to $\mathbb{1}$ almost surely. In other terms, even if \bar{P} achieves average consensus, not necessarily P(t) will also achieve average probabilistic consensus. This will appear in the two cases we will consider in the next section.

We proceed now by introducing two interesting examples of randomized consensus algorithms. We will prove that in both these cases the probabilistic consensus occurs.

4.3.1 Time-varying strategy with bounded in-degree

In this example we consider a time-varying consensus algorithm in which we assume that each system receives the state of ν systems chosen randomly and independently between all the other system. Because of this it can happen that the resulting communication graph has multiple arcs connecting the same pair of nodes.

Fix $k_0, k_1, \ldots, k_{\nu} \ge 0$ such that $\sum_j k_j = 1$. The consensus matrix is

$$P(t) = k_0 I + \sum_{i=1}^{\nu} k_i E_i(t)$$
(4.4)

where $E_i(t)$, $i = 1, ..., \nu$, are ν independent sequences of independent random variables taking values on the set of matrices

$$\mathcal{E} := \{ E \in \{0, 1\}^{N \times N} : E\mathbb{1} = \mathbb{1} \}$$

and uniformly distributed in such a set. Notice that the set \mathcal{E} is constituted by all matrices with entries 0 or 1 which have exactly one 1 in each row. The state x(t) becomes a random variable which evolves according to

$$x(t) = \prod_{s=1}^{t} P(s)x(0) \,,$$

where x(0) is a random variable independent of the processes $E_i(t)$. We have that

$$\bar{P} = \mathbb{E}\left[k_0I + \sum_{i=1}^{\nu} k_iE_i(t)\right]$$
$$= k_0I + \sum_{i=1}^{\nu} k_i\mathbb{E}\left[E_i(t)\right]$$
$$= k_0I + \frac{\sum_{i=1}^{\nu} k_i}{N}\mathbb{1}\mathbb{1}^*,$$

where in the last equality we used the fact that $\mathbb{E}[E_i] = 1/N\mathbb{1}\mathbb{1}^*$. Note that $\mathcal{G}_{\bar{P}}$ is the complete graph and hence it is trivially strongly connected. Moreover in this case, if $k_0 > 0$, all the diagonal elements of P(t) are nonzero with probability 1. Applying Proposition 4.2 we can conclude that this algorithm yields the probabilistic consensus.

Finally notice that, in this example the matrices P(t) are in general not doubly stochastic and so the average probabilistic consensus is not achieved. This, even if \overline{P} is doubly stochastic.

4.3.2 Time-varying Cayley graphs strategy

In this example we consider a time-varying strategy similar to the one presented in the previous section. The difference is that here we impose the time-varying matrices to be Cayley.

Fix an Abelian group G and a number $\nu < |G|$. We consider a sequence of subsets $S_t \subseteq G$ that are randomly generated in the following way.

Let $\alpha_i(t)$, $i = 1, ..., \nu$, be ν independent sequences of independent random variables taking value on G and uniformly distributed in such a set. We put

$$S_t = \{ \alpha_0(t) = 0, \alpha_1(t), \dots, \alpha_{\nu}(t) \}.$$

Notice that in S_t there might be repetitions and so its cardinality may be less than $\nu + 1$.

Fix $k_0, k_1, \ldots, k_{\nu} \ge 0$ such that $\sum_j k_j = 1$, and consider the sequence of probability distributions π_t on G supported on the sequence of sets S_t defined as

$$\pi_t(g) = \begin{cases} k_j \text{ if } g = \alpha_j(t) \\ 0 \text{ otherwise.} \end{cases}$$

Let P(t) be the stochastic Cayley matrix associated with π_t . The state x(t) becomes a random variable that evolves according to

$$x(t) = \prod_{s=1}^{t} P(s)x(0),$$

where x(0) is a random variable independent of the processes $\alpha_i(t)$. We know from Section 3.3.1 that we can represent

$$P(t) = \sum_{\chi \in \hat{G}} \hat{\pi}_t(\chi) N^{-1} \chi \chi^*$$

where

$$\hat{\pi}_t(\chi) = k_0 + \sum_{j=1}^{\nu} k_j \chi(-\alpha_j(t)),$$

It is immediate to verify that $\mathbb{E}[\chi(\alpha_j(t))] = 0$ when $\chi \neq \chi_0$. Therefore we have that

$$\mathbb{E}\left[\hat{\pi}_t(\chi)\right] = \begin{cases} k_0 & \text{if} \quad \chi \neq \chi_0\\ \sum_{j=0}^{\nu} k_j & \text{if} \quad \chi = \chi_0 \end{cases}$$

and hence that

$$\bar{P} = \mathbb{E}\left[\sum_{\chi \in \hat{G}} \hat{\pi}_t(\chi) N^{-1} \chi \chi^*\right]$$

$$= \mathbb{E}\left[\hat{\pi}_t(\chi_0) \frac{1}{N} \chi_0 \chi_0^*\right] + \mathbb{E}\left[\sum_{\chi \neq \chi_0} \hat{\pi}_t(\chi) \frac{1}{N} \chi \chi^*\right]$$

$$= \frac{\sum_{j=0}^{\nu} k_j}{N} \mathbb{1} \mathbb{1}^* + k_0 \sum_{\chi \neq \chi_0} \frac{1}{N} \chi \chi^*$$

$$= k_0 \sum_{\chi} \frac{1}{N} \chi \chi^* + \frac{\sum_{j=1}^{\nu} k_j}{N} \mathbb{1} \mathbb{1}^*$$

$$= k_0 I + \frac{\sum_{j=1}^{\nu} k_j}{N} \mathbb{1} \mathbb{1}^*$$

where in the last equality we have used the fact that $\sum_{\chi} \frac{1}{N} \chi \chi^* = I$. Observe that \overline{P} is the same of the previous example. Therefore, if $k_0 > 0$, from Proposition 4.2 we can conclude that also this algorithm yields the probabilistic consensus. Moreover, since each P(t) is a Cayley matrix and hence it is doubly stochastic, we have that in this case the average probabilistic consensus is achieved.

4.4 The measures of the performance

Assume now that we have a fixed random algorithm P(t) achieving probabilistic consensus so that (4.3) is satisfied with a suitable ρ . The authors in [58] proposes of measuring the performance of the algorithm P(t) by considering two figures. The first figure they consider is a normalized version of the *distance from the consensus*

$$d(t) = \frac{1}{N} \|x(t) - \mathbb{1}x_A(t)\|^2 = \frac{1}{N} \sum_{i=1}^N |x_i(t) - x_A(t)|^2$$
(4.5)

where $x_A(t) = N^{-1} \mathbb{1}^* x(t)$ is the average of the components of x(t). The second one is the *average displacement* from its initial value

$$\beta(t) = |x_A(t) - x_A(0)|^2.$$
(4.6)

Obviously in those situations where P(t) is always doubly stochastic, we have that $\beta(t) = 0$ for every t. Notice moreover that

$$\frac{1}{N} \|x(t) - \mathbb{1}x_A(0)\|^2 = d(t) + \beta(t)$$

which shows that the evolution of d(t) and $\beta(t)$ determines the evolution of $\frac{1}{N} ||x(t) - \mathbb{1}x_A(0)||^2$. This coincides with the average distance from $x_i(t)$ to $x_A(0)$ and so it is the most important error parameter that typically one has to minimize.

We know that d(t) converges to 0 almost surely. The main point is to estimate the speed of this convergence. In what follows, in order to evaluate the performance of the random strategies illustrated in the previous section, we will study the expectations of the two variables d(t) and $\beta(t)$, i.e. $\mathbb{E}[d(t)]$ and $\mathbb{E}[\beta(t)]$, so providing a mean-square analysis of (4.1) in the case of *time*varying with bounded in-degree and time-varying Cayley graphs strategies.

4.4.1 Mean square performance for time-varying with bounded in-degree strategy

We can prove the following result.

Theorem 4.4 The quantity $\mathbb{E}[d(t)]$ satisfies the following recursive equation

$$\mathbb{E}[d(t+1)] = \rho \mathbb{E}[d(t)] \tag{4.7}$$

where

$$\rho = k_0^2 + \frac{N-1}{N} \sum_{i=1}^{\nu} k_i^2.$$

Proof: We start by introducing the following variable

$$y(t) = x(t) - \frac{1}{N} (\mathbb{1}^* x(t)) \,\mathbb{1} = Y x(t)$$

where $Y = I - \frac{1}{N} \mathbb{1}\mathbb{1}^*$. A straightforward calculation shows that $YE_i = YE_iY$. Hence from (4.4) it is easy to see that y(t) satisfies the following recursive equation

$$y(t+1) = k_o y(t) + Y\left(\sum_{i=1}^{\nu} k_i E_i(t)\right) y(t).$$
(4.8)

Let now $W(t) = \mathbb{E}[y(t)y^*(t)]$. Notice that $\mathbb{E}[d(t)] = \text{tr } \{W(t)\}$. From (4.8) it results that

$$\begin{split} W(t+1) &= \\ &= \mathbb{E}\left[\left(k_{0}y(t) + Y\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)y(t)\right)\left(k_{0}y(t) + Y\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)y(t)z\right)^{*}\right] \\ &= k_{0}^{2}W(t) + k_{0}\mathbb{E}\left[y(t)y^{*}(t)\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)^{*}\right]Y + \\ &+ k_{0}\mathbb{E}\left[Y\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)y(t)y^{*}(t)\left(\sum_{j=1}^{\nu}k_{j}E_{j}(t)\right)^{*}\right]Y \\ &+ Y\mathbb{E}\left[\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)y(t)y^{*}(t)\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)^{*}\left|E_{1},\ldots,E_{\nu}\right]\right]Y + \\ &+ k_{0}\mathbb{E}\left[\mathbb{E}\left[Y\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)y(t)y^{*}(t)\left|E_{1},\ldots,E_{\nu}\right]\right] + \\ &+ Y\mathbb{E}\left[\mathbb{E}\left[\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)y(t)y^{*}(t)\left(\sum_{j=1}^{\nu}k_{j}E_{j}(t)\right)^{*}\left|E_{1},\ldots,E_{\nu}\right]\right]Y \\ &= k_{0}^{2}W(t) + k_{0}W(t)\mathbb{E}\left[\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)^{*}\right]Y + k_{0}Y\mathbb{E}\left[\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right]W(t) + \\ &+ Y\mathbb{E}\left[\left(\sum_{i=1}^{\nu}k_{i}E_{i}(t)\right)W(t)\left(\sum_{j=1}^{\nu}k_{j}E_{j}(t)\right)^{*}\right]Y \end{split}$$

Since $\mathbb{E}[E_i] = \mathbb{1}\mathbb{1}^*/N$ and since $Y \mathbb{1}\mathbb{1}^* = \mathbb{1}\mathbb{1}^* Y = 0$ we have that the first two expectations in the previous equation are equal to zero. To compute the last expectation we need to distinguish two cases:

 $i \neq j$: then $E_i(t)$, $E_j^*(t)$ and W(t) are all independent and thus the expectation factorizes. Recall now that $\mathbb{E}[E_i] = \frac{1}{N} \mathbb{1}\mathbb{1}^*$. Two terms of the type $Y \mathbb{1}\mathbb{1}^*$ appear and thus for $i \neq j$ the expectation is zero, i = j: then we have that

$$Y \mathbb{E} \left[k_i E_i(t) W(t) k_i E_i^*(t) \right] Y = k_i^2 Y \mathbb{E} \left[E_i(t) W(t) E_i^*(t) \right] Y$$

By straightforward calculations, it is possible to see that, for any $M \in \mathbb{R}^{N \times N}$ it holds

$$\mathbb{E}[E_i^T M E_i] = \frac{1}{N} \operatorname{tr} \{M\} I + \frac{1}{N^2} \mathbb{1}^* M \mathbb{1}(\mathbb{1}\mathbb{1}^* - I).$$
(4.9)

Hence

$$k_i^2 Y \mathbb{E} \left[E_i(t) W(t) E_i^*(t) \right] Y = \frac{k_i^2}{N} \operatorname{tr} \left(W(t) \right) Y - \frac{k_i^2}{N^2} Y \mathbb{1}^* W(t) \, \mathbb{1} Y.$$

We thus obtain that

$$W(t+1) = k_0^2 W(t) + \frac{1}{N} \sum_{i=1}^{\nu} k_i^2 \operatorname{tr} (W(t)) Y - \frac{1}{N^2} \sum_{i=1}^{\nu} k_i^2 Y \mathbb{1}^* W(t) \, \mathbb{1} Y.$$

Let w(t) = tr(W(t)), then we have

$$w(t+1) = k_0^2 w(t) + \frac{1}{N} \sum_{i=1}^{\nu} k_i^2 \operatorname{tr} \left(\operatorname{tr} \left(W(t) \right) Y \right) - \frac{1}{N^2} \sum_{i=1}^{\nu} k_i^2 \operatorname{tr} \left(\mathbb{1}^* W(t) \, \mathbb{1} Y \right).$$

The term tr (tr(W(t))Y) = (N-1)tr(W(t)) since tr (Y) = N-1 and the last term is zero since

$$\mathbb{1}^* W(t) \,\mathbb{1} = \sum_{i=1}^N \sum_{j=1}^N (W(t))_{ij} = 0 \,.$$

We thus have the following difference equation

$$w(t+1) = \left(k_0^2 + \frac{N-1}{N}\sum_{i=1}^{\nu}k_i^2\right)w(t).$$
(4.10)

By observing that $\mathbb{E}[d(t)] = w(t)$ we have proved the claim of the theorem.

Notice that the strongest exponential rate of convergence in (4.10) is given by

$$\min\left\{k_0^2 + \frac{N-1}{N}\sum_{i=1}^{\nu}k_i^2 \mid k_0, k_1, \dots, k_{\nu} \ge 0, \ k_0 + \sum_{i=1}^{\nu}k_i = 0\right\} = \frac{N-1}{N(\nu+1)-1},$$

obtained by choosing

$$k_0 = \frac{N-1}{N(\nu+1)-1} \tag{4.11}$$

and

$$k_i = \frac{N}{N(\nu+1) - 1}$$
 $\forall i = 1, \dots, \nu.$ (4.12)

As outlined previously, this random strategy will not reach the consensus at the average of the initial conditions. Therefore, in such case it is interesting to study also the behavior of $\mathbb{E}[\beta(t)]$ in particular to evaluate $\lim_{t\to\infty} \mathbb{E}[\beta(t)]$. We have the following result.

Proposition 4.5

$$\mathbb{E}[\beta(\infty)] = \alpha \mathbb{E}||\left(I - N^{-1}\mathbb{1}\mathbb{1}^T\right) x(0)||^2,$$

where

$$\alpha = \frac{\sum_{i=1}^{\nu} k_i^2}{N[N(1-k_0^2) + (1-N)\sum_{i=1}^{\nu} k_i^2]}$$

Proof: Consider $\Delta(t) := x(t) - N^{-1} \mathbb{1} \mathbb{1}^T x(0)$. One can show that the dynamics of $\Delta(t)$ is described by the equation $\Delta(t+1) = P(t)\Delta(t)$ where P(t) is given in (4.4). By defining $W(t) := \mathbb{E}[\Delta(t)\Delta(t)^*], w(t) = \operatorname{tr}(W(t)) = \mathbb{E}||\Delta(t)||^2$ and $s(t) = N^{-1} \mathbb{1}^* W(t) \mathbb{1}$, we obtain that

$$\begin{split} W(t+1) &= \\ &= \mathbb{E}\left[\left(k_0I + \sum_{i=1}^{\nu} k_iE_i(t)\right)\Delta(t)\Delta^*(t)\left(k_0I + \sum_{i=1}^{\nu} k_iE_i(t)\right)^*\right] \\ &= k_0^2W(t) + \mathbb{E}\left[\sum_{i=1}^{\nu} k_iE_i(t)\Delta(t)\Delta^*(t)k_0\right] + \mathbb{E}\left[k_0\Delta(t)\Delta^*(t)\sum_{i=1}^{\nu} k_iE_i^*(t)\right] + \\ &+ \mathbb{E}\left[\sum_{i=1}^{\nu} k_iE_i(t)\Delta(t)\Delta^*(t)\sum_{j=1}^{\nu} k_jE_j^*(t)\right] \\ &= k_0^2W(t) + k_0\sum_{i=1}^{\nu} k_i\left(W(t)\frac{1}{N}\mathbb{1}\mathbb{1}^* + \frac{1}{N}\mathbb{1}\mathbb{1}^*W(t)\right) \\ &+ \sum_{i=1}^{\nu} k_i^2\left(\frac{1}{N}(\mathbb{1}^*W(t)\mathbb{1})\frac{1}{N}\mathbb{1}\mathbb{1}^* + \left(\frac{1}{N}\mathrm{tr}\,W(t) - \frac{1}{N^2}\mathbb{1}^*W(t)\mathbb{1}\right)I\right) + \\ &+ \sum_{i=1}^{\nu}\sum_{j=1}^{\nu} k_ik_j\frac{1}{N}(\mathbb{1}^*W(t)\mathbb{1})\frac{1}{N}\mathbb{1}\mathbb{1}^* \end{split}$$

where in the last equality we used again the fact that $\mathbb{E}[E_i] = \frac{1}{N} \mathbb{1}\mathbb{1}^*$ and the property (4.9). Let us now define the following variables

$$w(t) = \operatorname{tr} W(t)$$
$$s(t) = \frac{1}{N} \mathbb{1}^* W(t) \mathbb{1}$$

then we have that

$$w(t+1) = \left(k_0^2 + \sum_{i=1}^{\nu} k_i^2\right) w(t) + \left(2k_0 \sum_{i=1}^{\nu} k_i + \sum_{\substack{i=1\\i\neq j}}^{\nu} \sum_{j=1}^{\nu} k_i k_j\right) s(t)$$
$$s(t+1) = \left((k_0^2 + 2k_0 \sum_{i=1}^{\nu} k_i + \frac{N-1}{N} \sum_{\substack{i=1\\i\neq j}}^{\nu} k_i^2 + \sum_{\substack{i=1\\i\neq j}}^{\nu} \sum_{j=1}^{\nu} k_i k_j\right) s(t) + \frac{1}{N} \sum_{\substack{i=1\\i\neq j}}^{\nu} k_i^2 w(t) \cdot \frac{1}{N} \sum_{i=1}^{\nu} k_i^2 w(t) \cdot \frac{1}{$$

Using the fact that $1 - k_0 = \sum_{i=1}^{\nu} k_i$ we have the following discrete-time system

$$\begin{pmatrix} w(t+1) \\ s(t+1) \end{pmatrix} = \underbrace{\begin{pmatrix} \sum_{i=0}^{\nu} k_i^2 & 1 - \sum_{i=0}^{\nu} k_i^2 \\ \frac{1}{N} \sum_{i=1}^{\nu} k_i^2 & 1 - \frac{1}{N} \sum_{i=1}^{\nu} k_i^2 \end{pmatrix}}_{A} \begin{pmatrix} w(t) \\ s(t) \end{pmatrix} .$$

In order to prove the theorem we need to compute $w(\infty) := \lim_{t\to\infty} w(t)$. For t that tends to infinity, the state $(w(t), s(t))^*$ will be aligned to the dominant eigenvector of A. The eigenvectors of A are

$$\lambda_1 = 1$$

 $\lambda_2 = k_0^2 + \frac{N-1}{N} \sum_{i=1}^{\nu} k_i^2.$

Corresponding to the eigenvalues λ_1 and λ_2 we have the eigenvectors

$$a_{1} = \begin{pmatrix} 1 \\ 1 \end{pmatrix}$$

$$a_{2} = \begin{pmatrix} -\frac{N\left(1 - \sum_{i=1}^{\nu} k_{i}^{2}\right)}{\sum_{i=1}^{\nu} k_{i}^{2}} \\ 1 \end{pmatrix}$$

and thus we have that

$$\binom{w(t)}{s(t)} = c_1 \lambda_1^t a_1 + c_2 \lambda_2^t a_2$$

from which we have $w(\infty) \approx c_1 a_1$. The constant c_1 is computed from the initial conditions

$$\begin{pmatrix} w(0)\\ s(0) \end{pmatrix} = c_1 a_1 + c_2 a_2 \,.$$

Note that

$$w(0) = \operatorname{tr} \{W(0)\} = \mathbb{E} || (I - N^{-1} \mathbb{1} \mathbb{1}^T) x(0) ||^2$$

and

$$s(0) = \frac{1}{N} \mathbb{1}^* W(0) \mathbb{1} = \frac{1}{N} \mathbb{E} \left[\sum_{i=1}^N \left(x_i(0) - \frac{1}{N} \mathbb{1}^* x(0) \mathbb{1} \right) \right]^2 = 0.$$

Simple calculations yields to

$$c_1 = \frac{\sum_{i=1}^{\nu} k_i^2}{N \left[N(1-k_0^2) + (1-N) \sum_{i=1}^{\nu} k_i^2 \right]} w(0)$$

This concludes the proof.

If we use the control gains $k_0, k_1, \ldots, k_{\nu}$ as in (4.11) and (4.12), which yield the fastest convergence rate, then we have

$$\mathbb{E}[\beta(t)] = \frac{1}{N(N(1+\nu)-1)} \mathbb{E}\|(I-N^{-1}\mathbb{1}\mathbb{1}^*)x(0)\|^2.$$

Notice that, if the initial states $x_i(0)$ of the systems are independent and $\mathbb{E}[x_i(0)^2]$ is the same for all *i*, then as $N \to \infty$ the mean square distance to the average of the initial conditions tends to zero.

4.4.2 Mean square performance for time-varying Cayley graphs strategy

Since in this case the matrix P(t) is doubly stochastic for every $t \ge 0$, the only quantity we want to study is $\mathbb{E}[d(t)]$. In order to do so, we consider

the displacement from the average of the initial conditions $\Delta(t) := x(t) - \frac{1}{N} \mathbb{1} \mathbb{1}^T x(0) = (I - \frac{1}{N} \chi_0 \chi_0^*) x(t)$, which is governed by

$$\Delta(t) = \prod_{s=1}^{t} P(s)\Delta(0) \,,$$

where $\Delta(0)$ is now a random variable taking values on \mathbb{R}^G such that $< \Delta(0), \chi_0 >= 0$ and independent of the set of variables $\{\alpha_i(t)\}$. Clearly $\mathbb{E}[d(t)] = \mathbb{E}||\Delta(t)||^2$. For the sake of the clarity, we recall that χ_0 denotes the trivial character, as explained in Subsection 3.3.1. We have the following result.

Proposition 4.6

$$\mathbb{E}||\Delta(t)||^{2} = \left(\sum_{j=0}^{\nu} k_{j}^{2}\right)^{t} \mathbb{E}||\Delta(0)||^{2}$$

Proof: We know we can represent

$$P(t) = \sum_{\chi \in \hat{G}} \hat{\pi}_t(\chi) N^{-1} \chi \chi^* \,.$$

Hence,

$$\prod_{s=1}^{t} P(s) = \sum_{\chi \in \hat{G}} \left[\prod_{s=1}^{t} \hat{\pi}_s(\chi) \right] N^{-1} \chi \chi^*.$$

Let us study the average of the squared norm of the various eigenvalues.

$$\mathbb{E}\left[\left|\prod_{s=1}^{t} \hat{\pi}_{s}(\chi)\right|^{2}\right] = \prod_{s=1}^{t} \mathbb{E}\left[\left|\hat{\pi}_{s}(\chi)\right|^{2}\right].$$

Since

$$\hat{\pi}_t(\chi) = k_0 + \sum_{j=1}^{\nu} k_j \chi(-\alpha_j(t)),$$

we obtain

$$\mathbb{E}\left[|\hat{\pi}_{t}(\chi)|^{2}\right] = k_{0}^{2} + \sum_{j=1}^{\nu} k_{0}k_{j}\left[\mathbb{E}\left[\chi(\alpha_{j}(t))\right] + \mathbb{E}\left[\chi(\alpha_{j}(t))^{*}\right]\right] + \sum_{j=1}^{\nu} \sum_{\ell=1}^{\nu} k_{j}k_{\ell}\mathbb{E}\left[\chi(\alpha_{j}(t))\chi(\alpha_{\ell}(t))^{*}\right].$$
 (4.13)

It is immediate to verify that $\mathbb{E}[\chi(\alpha_j(t))] = 0$ when $\chi \neq \chi_0$, that $\mathbb{E}[\chi(\alpha_j(t))\chi(\alpha_\ell(t))^*] = 0$ when $j \neq \ell$, and that $\mathbb{E}[|\chi(\alpha_j(t))|^2] = 1$. Substituting in (4.13) we then obtain

$$\mathbb{E}\left[|\hat{\pi}_t(\chi)|^2\right] = k_0^2 + \sum_{j=1}^{\nu} k_j^2 = \sum_{j=0}^{\nu} k_j^2, \quad \forall \chi \neq 0.$$

We finally have

$$\mathbb{E}||\Delta(t)||^{2} = \sum_{\chi \neq \chi_{0}} \mathbb{E}\left[\left|\prod_{s=1}^{t} \hat{\pi}(\chi)\right|^{2}\right] \frac{1}{N} \mathbb{E}| < \Delta(0), \chi > |^{2} \\ = \left(\sum_{j=0}^{\nu} k_{j}^{2}\right)^{t} \frac{1}{N} \sum_{\chi \neq \chi_{0}} \mathbb{E}| < \Delta(0), \chi > |^{2} = \left(\sum_{j=0}^{\nu} k_{j}^{2}\right)^{t} \mathbb{E}||\Delta(0)||^{2}.$$

Notice that

$$\min\left\{\sum_{j=0}^{\nu} k_j^2 \; \middle|\; k_j \ge 0 \; , \; \sum_{j=1}^{\nu} k_j = 1\right\} = \frac{1}{\nu+1}$$

which is obtained by choosing $k_j = 1/(\nu + 1)$ for all j. With such a choice we have thus obtained the following mean convergence result

$$\mathbb{E}||\Delta(t)||^2 = \left(\frac{1}{1+\nu}\right)^t \mathbb{E}||\Delta(0)||^2$$

We noticed in the previous section that strategies with symmetries behave quite poorly. As suggested in the last part of the previous section, a way to overcome this difficulty is to resort to Ramanujan graphs or to undirected regular graphs generated randomly.

The results found in this subsection show that an alternative way to solve this problem while maintaining the symmetry of the matrices is by a time-varying strategy in which at every time instant the communication graph is chosen randomly in a set of Cayley graphs. Indeed such strategy yields a mean square convergence rate that is higher and, more importantly, independent of the number of systems.

Moreover notice that $\frac{1}{1+\nu} \geq \frac{N-1}{N(\nu+1)-1}$, which is the rate obtained through the time-varying strategy with bounded degree discussed before. However, for $N \to +\infty$, the two strategies yield the same rate.

Remark 4.7 From an implementation point of view this strategy has an evident drawback: the same random choice done at every time instance has to be done by all systems. This seems to require a supervised communication of this information to every system. A possible way to overcome this limitation is by imposing that each agent uses the same pseudorandom number generator starting from the same seed.

Remark 4.8 As any average result, it is not immediately evident how the average computation above reflects on the behavior of the system when we consider a generic sequence S_t of subsets chosen at random. A simple standard probabilistic argument however allows us to show that such a convergence rate is indeed achieved by almost every sequence S_t . Fix any c > 1 and notice that for every $\chi \in \hat{G}$,

$$P\left(\prod_{s=1}^{t} |\hat{\pi}_s(\chi)| \ge \left(\frac{c}{\nu+1}\right)^t\right) = P\left(\sum_{s=1}^{t} \ln |\hat{\pi}_s(\chi)| \ge t \ln \frac{c}{\nu+1}\right). \quad (4.14)$$

Notice that $Y_s = \ln |\hat{\pi}_s(\chi)|$ is a sequence of independent identically distributed random variables taking values on $[-\infty, 0]$. The idea is to apply Chebyschev inequality. To overcome the problem of possible unboundedness of Y_s , we consider two different cases. Suppose that there exist a $\chi \in \hat{G}$ such that $\hat{\pi}_s(\chi)$ assumes the value 0 with probability p > 0. In this case we can simply estimate

$$P\left(\prod_{s=1}^{t} |\hat{\pi}_s(\chi)| \ge \left(\frac{c}{\nu+1}\right)^t\right) \le (1-p)^t.$$

$$(4.15)$$

If instead the event $\{\hat{\pi}_s(\chi) = 0\}$ has probability zero, then the random variable Y_s is bounded and can be estimated as follows. First notice that, using Jensen inequality, we have

$$\mathbb{E}[Y_s] = \mathbb{E}[\ln |\hat{\pi}_s(\chi)|] \le \ln \mathbb{E}[|\hat{\pi}_s(\chi)|] = \ln \frac{1}{\nu+1}.$$

Let

$$\delta := \ln \frac{c}{\nu + 1} - \mathbb{E}[Y_s] = \ln c + \ln \frac{1}{\nu + 1} - \mathbb{E}[Y_s] \ge \ln c > 0.$$

We can now estimate

$$P\left(\sum_{s=1}^{t} \ln |\hat{\pi}_{s}(\chi)| \ge t \ln \frac{c}{\nu+1}\right) = P\left(\sum_{s=1}^{t} Y_{s} \ge t \ln \frac{c}{\nu+1}\right)$$
$$= P\left(\sum_{s=1}^{t} (Y_{s} - \mathbb{E}[Y_{s}]) \ge t\delta\right) \qquad (4.16)$$
$$\le \frac{\operatorname{Var}[Y_{s}]}{\delta^{2}t^{2}}$$

A straightforward application of Borel-Cantelli lemma now allows to conclude from relations (4.15) and (4.16) that, for almost every sequence S_t of subsets,

$$\prod_{s=1}^{t} |\hat{\pi}_{s}(\chi)| < \left(\frac{c}{\nu+1}\right)^{t} \quad \text{for } t \text{ sufficiently large and} \quad \text{for every } \chi \in \hat{G} \,.$$

From this we also obtain that, for almost every sequence S_t ,

$$||\Delta(t)||^2 \le \left(\frac{c}{\nu+1}\right)^t ||\Delta(0)||^2$$
 for t sufficiently large.

Considering that c can be chosen arbitrarily close to 1, this proves our claim. A similar performance result on generic random samples from the average behavior can be proved also for the previous strategy. The probabilistic tools needed become however a bit more refined: matrices are not simultaneously diagonalizable and we have to use Oseledec ergodic theorem for products of random matrices.

4.5 Symmetric Gossip algorithms

In this section we review the symmetric gossip algorithm as proposed in [24].

We start from an a priori fixed communication skeleton, namely a fixed underlying undirected graph $\mathcal{G} = (V, \mathcal{E})$, establishing which are the feasible communications among the agents. We assume that at every time instant a node among the N possible is chosen randomly. This node then chooses also randomly one of its neighbors, it establishes a bidirectional link with it and they average their quantity. Precisely, consider the instant time t. Assume i is the node chosen randomly and assume that j is the node chosen by the node *i*. Then the updating rule is given by

$$x_{i}(t+1) = \frac{1}{2}x_{i}(t) + \frac{1}{2}x_{j}(t)$$

$$x_{j}(t+1) = \frac{1}{2}x_{j}(t) + \frac{1}{2}x_{i}(t)$$
(4.17)

and

$$x_h(t+1) = x_h(t),$$
 if $h \neq i$ and $h \neq j.$ (4.18)

Now, let, for every $(i, j) \in \mathcal{E}$,

$$E^{(i,j)} = (e_i - e_j)(e_i - e_j)^*$$
(4.19)

where $e_i = [0, ..., 0, 1, 0, ..., 0]^*$ is a $N \times 1$ unit vector with the *i*-th component equal to 1 and let

$$R^{(i,j)} = I - \frac{1}{2}E^{(i,j)}.$$
(4.20)

Then we can write the Equation (4.17) and Equation (4.18) in vector form as

$$x(t+1) = P(t)x(t)$$
 (4.21)

where $P(t) = R^{(i,j)}$.

If we assume that the node *i* is chosen with probability 1/N and that this node chooses randomly a node *j* among its ν_i neighbors with probability $1/\nu_i$, we have that

$$\mathbb{P}\left(P(t) = R^{(i,j)}\right) = \frac{1}{N}\left[\frac{1}{\nu_i} + \frac{1}{\nu_j}\right]$$

and hence that

$$\bar{P} = \sum_{(i,j) \in \mathcal{E}} \frac{1}{N} \left[\frac{1}{\nu_i} + \frac{1}{\nu_j} \right] \left[I - \frac{1}{2} (e_i - e_j) (e_i - e_j)^* \right].$$

Notice that, if $(h, k) \in \mathcal{E}$ and $h \neq k$,

$$\bar{P}_{hk} = \frac{1}{2N} \left[\frac{1}{\nu_h} + \frac{1}{\nu_k} \right].$$

where \bar{P}_{hk} denotes the element in the *h*-th row and in the *k*-th column of the matrix \bar{P} . Since each $R^{(i,j)}$ has all the diagonal terms different from 0 and since $\mathcal{G}_{\bar{P}} = \mathcal{G}$ we have, from Proposition 4.2, that the symmetric gossip consensus algorithm reaches, almost surely, the average consensus if and only if the graph \mathcal{G} is connected.

We proceed now by providing an alternative way of introducing the symmetric consensus gossip algorithm that generalizes the case above described. Assume again that a graph $\mathcal{G} = (V, \mathcal{E})$ is given and assume that, at each time instant an edge is selected inside the set \mathcal{E} with a probability $W^{(i,j)}$. The updating rule is again

$$x(t+1) = P(t)x(t)$$

where $\mathbb{P}\left[P(t) = R^{(i,j)}\right] = W^{(i,j)}$. Clearly $\sum_{(i,j) \in \mathcal{E}} W^{(i,j)} = 1$. Observe that this approach recovers the above case of the symmetric gossip algorithm by letting $W^{(i,j)} = \frac{1}{N} \left[\frac{1}{\nu_i} + \frac{1}{\nu_j}\right]$.

We have that

$$\bar{P} = \sum_{(i,j) \in \mathcal{E}} W^{(i,j)} R^{(ij)}$$

Clearly if $\mathcal{G} = (V, \mathcal{E})$ is connected and each edge $(i, j) \in \mathcal{E}$ can be selected with a strictly positive probability $W_{(i,j)}$, then $\mathcal{G}_{\bar{P}}$ is automatically connected. Moreover in this case all the diagonal elements of P(t) are nonzero with probability 1. Applying Proposition 4.2 we can conclude that this algorithm yields the probabilistic consensus. Moreover, since the P(t) are all symmetric we can conclude that in this case probabilistic average consensus is achieved.

It is worth noting that the fact that each $R^{(i,j)}$ has all the diagonal elements different from 0 implies also the following interesting property. Given $\mathcal{G} = (V, \mathcal{E})$ let $\tilde{\mathcal{G}} = (V, \tilde{\mathcal{E}})$ be the graph that we obtain from \mathcal{G} where

$$\mathcal{E} = \{(i,j) : (i,j) \in \mathcal{E}\} \setminus \{(i,i) : (i,i) \in \mathcal{E}\},\$$

namely $\tilde{\mathcal{E}}$ is the set \mathcal{E} without all the possible self-loops. Assume that each edge $(i, j) \in \tilde{\mathcal{E}}$ can be selected with a strictly positive probability \tilde{W}^{ij} , where $\sum_{(i,j)\in\tilde{\mathcal{E}}}\tilde{W}^{ij} = 1$. Let \bar{P} be the expected value of P(t) constructed on the graph \mathcal{G} and let \tilde{P} be the expected value of P(t) constructed on the graph $\tilde{\mathcal{G}}$. Then $\mathcal{G}_{\bar{P}} = \mathcal{G}_{\tilde{P}}$. Observe moreover that, if, at time instant t a self loop is selected then P(t) = I and hence x(t+1) = x(t), i.e. the state remains inalterated. In other words the self-loops are useless in order to reach the consensus. From now on, we will assume the following.

Assumption 4.9 The graph $\mathcal{G} = (V, \mathcal{E})$ is a undirected connected graph without any self-loop and, at every time instant t, each edge $(i, j) \in \mathcal{E}$ can be selected with a strictly positive probability $W^{(i,j)}$.

We end this subsection by defining the matrix $W \in \mathbb{R}^{N \times N}$ as the matrix having W^{ij} as element in the *i*-th row and *j*-th column and in the *j*-th row and *i*-th column, namely

$$W^{ij} = W^{ji} := W^{(i,j)}.$$
(4.22)

Note that, since from Assumption 4.9 we are considering graphs without self-loops, we have that $W_{ii} = 0$.

4.5.1 Analysis of the Symmetric Gossip Algorithm

Assume that an undirected graph \mathcal{G} satisfying Assumption 4.9 is given and that W is the matrix defined in (4.22) at the end of Section 4.5. The objective of this subsection is to provide a mean-square analysis of the symmetric gossip algorithm as previously described, i.e.,

$$x(t+1) = P(t)x(t)$$
(4.23)

where $\mathbb{P}\left(P(t) = R^{(i,j)}\right) = W^{ij}$.

We start by assuming the following on the initial condition x(0).

Assumption 4.10 The initial condition x(0) is a random vector variable such that $\mathbb{E}[x(0)] = 0$ and $\mathbb{E}[x(0)x^*(0)] = \sigma^2 I$ for some $\sigma^2 > 0$.

As first step, we introduce the following variable

$$y(t) := \left(I - \frac{1}{N} \mathbb{1}\mathbb{1}^*\right) x(t)$$

Notice that $y(t) = x(t) - x_{ave}(t)\mathbb{1}$, where we recall that $x_{ave}(t)$ denotes the current average of the state x(t), i.e., $x_{ave}(t) = 1/N\mathbb{1}^*x(t)$. Since $R^{(i,j)}$ is a doubly stochastic matrix for each pair (i, j), it follows that $x_{ave}(t) = x_{ave}(0)$ for all t and hence we can write also $y(t) = x(t) - x_{ave}(0)\mathbb{1}$. Moreover, this implies also that the variable $\beta(t) = 0$, defined in (4.6), is equal to 0 for all $t \ge 0$.

By observing that $R^{(i,j)}(I-1/N\mathbb{1}\mathbb{1}^*) = (I-1/N\mathbb{1}\mathbb{1}^*)R^{(i,j)}$ for each $R^{(i,j)}$, we obtain that the variable y(t) satisfies the same recursive equation of x(t), i.e.,

$$y(t+1) = P(t)y(t).$$
 (4.24)

Let now

$$\Sigma(t) = \mathbb{E}\left[y(t)y^*(t)\right].$$
From Assumption 4.10 we have that

$$\Sigma(0) = \sigma^2 \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right).$$

Now note that the quantity d(t) defined in (4.5) is related to Σ and hence to y by

$$d(t) = \frac{1}{N} \mathrm{tr} \left\{ \Sigma(t) \right\}.$$

We see now for a recursive expression for $\Sigma(t)$. From (4.24) we have that

$$\Sigma(t+1) = \mathbb{E} \left[y(t+1)y^*(t+1) \right]$$

= $\mathbb{E} \left[P(t)y(t)y^*(t)P(t) \right]$
= $\mathbb{E} \left[\mathbb{E} \left[P(t)y(t)y^*(t)P(t) | P(t) \right] \right]$
= $\mathbb{E} \left[P(t) \mathbb{E} \left[y(t)y^*(t) \right] P(t) \right]$
= $\mathbb{E} \left[P(t)\Sigma(t)P(t) \right]$ (4.25)

The calculation of the expectation $\mathbb{E}\left[P(t)\Sigma(t)P(t)\right]$ is not immediate. The next Theorem characterizes it. Before stating it, we provide some notational definitions that will be useful throughout the subsection. Given $X_1, X_2 \in \mathbb{R}^{N \times N}$, $X_1 \odot X_2$ denotes the component-wise product (Hadamard product) between X_1 and X_2 , i.e. $(X_1 \odot X_2)_{hk} = (X_1)_{hk} (X_2)_{hk}$. Given a matrix $X \in \mathbb{R}^{N \times N}$ we have that diag $\{X\}$ means a diagonal matrix with the same diagonal elements of the matrix X. Given a N-dimensional column vector $v = [v_1, \ldots, v_N]^*$, either the symbol diag $\{v_1, \ldots, v_N\}$ or the symbol diag $\{v\}$ mean a diagonal matrix with $(\underline{\text{diag}}\{v_1, \ldots, v_N\})_{ii} = v_i$. Finally, given the graph \mathcal{G} , the matrix A denotes its adjacency matrix, i.e

$$A_{ij} = A_{ji} = \begin{cases} 1 & \text{if } (j,i) \in \mathcal{E} \\ 0 & \text{otherwise} \end{cases}$$

Since in our work we are assuming that \mathcal{G} is without self-loops, then $A_{ii} = 0, i = 1, \ldots, N$.

We can now state the following result.

Theorem 4.11 Let $\mathcal{G} = (V, \mathcal{E})$ be a given graph satisfying Assumption 4.9 and let W be the matrix defined as in (4.22). Consider the equation (4.25). Then we have that

$$\Sigma(t+1) = \Sigma(t) - \frac{1}{2} \underline{\text{diag}} \{W1\} \Sigma(t) + \frac{1}{2} W\Sigma(t) - \frac{1}{2} \Sigma(t) \underline{\text{diag}} \{W1\} \\ + \frac{1}{2} \Sigma(t) W + \frac{1}{4} \underline{\text{diag}} \{W1\} \underline{\text{diag}} \{\Sigma(t)\} + \frac{1}{2} W \odot \Sigma(t) + \\ + \frac{1}{4} \underline{\text{diag}} \{(W \underline{\text{diag}} \{\Sigma(t)\})1\} - \frac{1}{4} \underline{\text{diag}} \{\Sigma(t)\} W \\ - \frac{1}{4} W \underline{\text{diag}} \{\Sigma(t)\} - \frac{1}{2} \underline{\text{diag}} \{A(W \odot \Sigma(t))\}$$
(4.26)

In order to prove the above Theorem we need the following technical Lemma.

Lemma 4.12 Let $\mathcal{G} = (V, \mathcal{E})$ be a given graph satisfying Assumption 4.9 and let W be the matrix defined as in (4.22). Let Q any symmetric matrix. Then

$$\mathbb{E}\left[E^{(i,j)}Q\right] = \underline{\operatorname{diag}}\left\{W\mathbb{1}\right\}Q - WQ,\tag{4.27}$$

and

$$\mathbb{E}\left[E^{(i,j)}QE^{(i,j)}\right] = \underline{\operatorname{diag}}\left\{W\mathbb{1}\right\}\operatorname{diag}\left\{Q\right\} + 2W \odot Q + \underline{\operatorname{diag}}\left\{(W\operatorname{diag}\left\{Q\right\})\mathbb{1}\right\} \\ - \left[\operatorname{diag}\left\{Q\right\}W + W\operatorname{diag}\left\{Q\right\}\right] \\ - 2\operatorname{diag}\left\{A(W \odot Q)\right\}$$
(4.28)

Proof: We have that

$$\mathbb{E}\left[E^{(i,j)}Q\right] = \sum_{(i,j)\in\mathcal{E}} W_{ij}(e_i - e_j)(e_i - e_j)^*Q$$
$$= \sum_{(i,j)\in\mathcal{E}} (e_i e_i^* + e_j e_j^*)Q - \sum_{(i,j)\in\mathcal{E}} (e_i e_j^* + e_j e_i^*)Q$$
$$= \underline{\operatorname{diag}}\left\{W1\right\}Q - WQ.$$

Then we have that

$$\begin{split} \mathbb{E}\left[E^{(i,j)}Q\right] &= \\ &= \sum_{(i,j) \in \mathcal{E}} W_{ij}(e_i - e_j)(e_i - e_j)^* Q(e_i - e_j)(e_i - e_j)^* \\ &= \sum_{(i,j) \in \mathcal{E}} W_{ij}(e_i e_i^* + e_j e_j^*) Q(e_i e_i^* + e_j e_j^*) \\ &- \sum_{(i,j) \in \mathcal{E}} W_{ij}\left\{(e_i e_j^* + e_j e_i^*) Q(e_i e_i^* + e_j e_j^*) + (e_i e_i^* + e_j e_j^*) Q(e_i e_j^* + e_j e_i^*)\right\} \\ &+ \sum_{(i,j) \in \mathcal{E}} W_{ij}(e_i e_j^* + e_j e_i^*) Q(e_i e_j^* + e_j e_i^*) \end{split}$$

Consider now $\sum_{(i,j)\in\mathcal{E}} W_{ij}(e_i e_i^* + e_j e_j^*) Q(e_i e_i^* + e_j e_j^*)$. It follows that

$$\sum_{(i,j)\in\mathcal{E}} W_{ij}(e_i e_i^* + e_j e_j^*)Q(e_i e_i^* + e_j e_j^*) =$$

$$= \sum_{(i,j)\in\mathcal{E}} W_{ij}\left(e_i e_i^* Q e_i e_i^* + e_j e_j^* Q e_j e_j^*\right) + \sum_{(i,j)\in\mathcal{E}} W_{ij}\left(e_j e_j^* Q e_i e_i^* + e_i e_i^* Q e_j e_j^*\right)$$

$$= \sum_{(i,j)\in\mathcal{E}} \left(W_{ij}Q_{ii}e_i e_i^* + W_{ij}Q_{jj}e_j e_j^*\right) + \sum_{(i,j)\in\mathcal{E}} \left(W_{ij}Q_{ji}e_j e_i^* + W_{ij}Q_{ij}e_i e_j^*\right)$$

$$= \sum_{(i,j)\in\mathcal{E}} \left(W_{ij}Q_{ii}e_i e_i^* + W_{ij}Q_{jj}e_j e_j^*\right) + \sum_{(i,j)\in\mathcal{E}} \left(W_{ji}Q_{ji}e_j e_i^* + W_{ij}Q_{ij}e_i e_j^*\right)$$

$$= \underline{\operatorname{diag}} \left\{W1\right\} \operatorname{diag} \left\{Q\right\} + W \odot Q$$

In a similar way it can be shown that

$$\sum_{(i,j)\in\mathcal{E}} W_{ij} \left\{ (e_i e_j^* + e_j e_i^*) Q(e_i e_i^* + e_j e_j^*) + (e_i e_i^* + e_j e_j^*) Q(e_i e_j^* + e_j e_i^*) \right\} =$$

= [diag {Q} W + W diag {Q}] + 2diag {A(W \cdots Q)}

and that

$$\sum_{(i,j)\in\mathcal{E}} W_{ij}(e_i e_j^* + e_j e_i^*) Q(e_i e_j^* + e_j e_i^*) = W \odot Q + \underline{\operatorname{diag}} \left\{ (W \operatorname{diag} \{Q\}) \mathbb{1} \right\}.$$

Plugging together all the contributions we obtain (4.28).

We are able now to provide the proof of Theorem 4.11. *Proof:* Consider the Equation (4.26). We have that

$$\mathbb{E}\left[P(t)\Sigma(t)P(t)\right] = \mathbb{E}\left[\left(I - \frac{1}{2}E_{ij}\right)\Sigma(t)\left(I - \frac{1}{2}E_{ij}\right)\right]$$
$$= \Sigma_{yy} - \frac{1}{2}\mathbb{E}\left[E_{ij}\Sigma(t)\right] - \frac{1}{2}\mathbb{E}\left[\Sigma(t)E_{ij}\right] + \frac{1}{4}\mathbb{E}\left[E_{ij}\Sigma(t)E_{ij}\right]$$

Substituting in the above expectations the expressions given in Lemma 4.12 we obtain the recursive equation given in the statement of the Theorem. \blacksquare

Since the graph \mathcal{G} satisfies Assumption 4.9 which guarantees, as previously observed, that the symmetric gossip algorithm (4.23) converges almost surely, we have that $\lim_{t\to\infty} \Sigma(t) = 0$ and hence also $\lim_{t\to\infty} d(t) = 0$. Unfortunately providing further theoretical analysis on the symmetric gossip algorithms, for instance on the speed of convergence toward zero of d(t), is quite hard in general. In [24] and in [58] the authors provide some bounds on the rate of convergence of d(t) but these bound could be not tight.

However, further and interesting analytical developments in this direction can be made if we restrict the graph \mathcal{G} to possess some symmetries. In the sequel we will work with the Cayley Graphs.

4.5.2 Mean-square Analysis for the Cayley Graphs

In this section we will show that when \mathcal{G} is a Cayley graph, the analysis proposed above simplifies considerably. Let G be a finite Abelian group of order N and let S be a subset of G which does not contain zero. Let $\mathcal{G}(G, S)$ be the Cayley graph associated to the group G and the subset S as described in Section 3.3. Observe the fact that S does not contain zero implies that the graph \mathcal{G} does not contain self-loops. In the following we will assume that $|S| = \nu$, namely that the degree of \mathcal{G} is ν . Moreover we will assume that each edge of the edge set \mathcal{E} of the Cayley graph $\mathcal{G}(G, S)$ can be selected with probability $1/|\mathcal{E}|$, where it is easy to see that $|\mathcal{E}| = \frac{2}{\nu N}$. Moreover observe that it follows that $W = \frac{2}{\nu N}A$. We start our analysis with the following simple result.

Lemma 4.13 Consider (4.26). Assume that $\Sigma(0) = \sigma^2 (I - 1/N \mathbb{1}\mathbb{1}^*)$. Then $\Sigma(t)$ is a Cayley matrix for all $t \ge 0$.

Proof: We prove the statement of the theorem by induction on t. Let t = 0. $\Sigma_{yy}(0)$ is Cayley since $(I - 1/N \mathbb{11}^*)$ is Cayley. Let now $\Sigma(t)$ be Cayley for a generic t. Consider t + 1. Since A and W are Cayley it follows that all the matrices on the right-side of (4.26) are Cayley. Therefore also $\Sigma(t + 1)$ is Cayley.

By using the fact that $W = \frac{2}{\nu N}A$ and the properties of the Cayley matrices we have (4.26) can be rewritten in a simpler way as

$$\Sigma(t+1) = \left(1 - \frac{2}{N}\right)\Sigma(t) + \frac{2}{\nu N}A\Sigma(t) + \frac{1}{N}\text{diag}\left\{\Sigma(t)\right\} + \frac{1}{\nu N}A\odot\Sigma(t) - \frac{1}{\nu N}\left[A\text{diag}\left\{\Sigma(t)\right\} + \text{diag}\left\{A\Sigma(t)\right\}\right]$$
(4.29)

Note now that a Cayley matrix is completely determined by the values that its generator π assume on the elements of the group G, equivalently by the values of one of its row or column. In order to simplify further the problem we assume that the elements of G are ordered in some way, more precisely we assume that $G = \{g_0, g_1, \ldots, g_{N-1}\}$, where by convention $g_0 = 0$. Define the column vector in \mathbb{R}^N as

$$\Pi(t) = \begin{bmatrix} \pi_{\Sigma(t)}(g_0) \\ \pi_{\Sigma(t)}(g_1) \\ \vdots \\ \pi_{\Sigma(t)}(g_{N-1}) \end{bmatrix}$$
(4.30)

In order to find a recursive relation for $\Pi(t)$, it is useful to introduce the $N \times N$ matrices B, C, D defined as follows

$$(B)_{ij} := \begin{cases} 1 & \text{if } i = j \text{ and } \pi_A(g_i) = 1\\ 0 & \text{otherwise} \end{cases}$$
(4.31)

$$(C)_{ij} := \begin{cases} 1 & \text{if } i = j = 1\\ 0 & \text{otherwise} \end{cases}$$
(4.32)

and

$$(D)_{ij} := \begin{cases} 1 & \text{if } i = 1 & \text{and } \pi_A(g_j) = 1\\ 1 & \text{if } j = 1 & \text{and } \pi_A(g_i) = 1\\ 0 & \text{otherwise} \end{cases}$$
(4.33)

Observe that $B = \underline{\text{diag}} \{e_1^*A\}, C = e_1e_1^*$ and $D = e_1e_1^*A + Ae_1e_1^*$ where $e_1 = [1, 0, \dots, 0]^*$. We have the following result.

Theorem 4.14 Let $\Pi(t)$ be as above defined. Then we have

$$\Pi(t+1) = \left[\left(1 - \frac{2}{N} \right) I + \frac{2}{\nu N} A + \frac{1}{N} C + \frac{1}{\nu N} B - \frac{1}{\nu N} D \right] \Pi(t).$$
(4.34)

Proof: We start by observing that from the definition of A, B, C, D we have that

$$\begin{bmatrix} \pi_{A\Sigma(t)}(g_0) \\ \pi_{A\Sigma(t)}(g_1) \\ \vdots \\ \pi_{A\Sigma(t)}(g_{N-1}) \end{bmatrix} = A \begin{bmatrix} \pi_{\Sigma(t)}(g_0) \\ \pi_{\Sigma(t)}(g_1) \\ \vdots \\ \pi_{\Sigma(t)}(g_{N-1}) \end{bmatrix}, \quad (4.35)$$

$$\begin{bmatrix} \pi_{\text{diag}}\{\Sigma(t)\}(g_0) \\ \pi_{\text{diag}}\{\Sigma(t)\}(g_1) \\ \vdots \\ \pi_{\Sigma(t)}(g_1) \end{bmatrix} = C \begin{bmatrix} \pi_{\Sigma(t)}(g_0) \\ \pi_{\Sigma(t)}(g_1) \\ \vdots \\ \pi_{\Sigma(t)}(g_1) \end{bmatrix}, \quad (4.36)$$

$$\begin{bmatrix} \pi_{\operatorname{diag}\left\{\Sigma(t)\right\}}(g_1) \\ \vdots \\ \pi_{\operatorname{diag}\left\{\Sigma(t)\right\}}(g_{N-1}) \end{bmatrix} = C \begin{bmatrix} \pi_{\Sigma(t)}(g_1) \\ \vdots \\ \pi_{\Sigma(t)}(g_{N-1}) \end{bmatrix},$$
(4.36)

$$\begin{bmatrix} \pi_{A \odot \Sigma(t)}(g_0) \\ \pi_{A \odot \Sigma(t)}(g_1) \\ \vdots \\ \pi_{A \odot \Sigma(t)}(g_{N-1}) \end{bmatrix} = B \begin{bmatrix} \pi_{\Sigma(t)}(g_0) \\ \pi_{\Sigma(t)}(g_1) \\ \vdots \\ \pi_{\Sigma(t)}(g_{N-1}) \end{bmatrix}, \qquad (4.37)$$

and

$$\begin{bmatrix} \pi_{A \operatorname{diag} \{\Sigma(t)\} + \operatorname{diag} \{A\Sigma(t)\}}(g_{0}) \\ \pi_{A \operatorname{diag} \{\Sigma(t)\} + \operatorname{diag} \{A\Sigma(t)\}}(g_{1}) \\ \vdots \\ \pi_{A \operatorname{diag} \{\Sigma(t)\} + \operatorname{diag} \{A\Sigma(t)\}}(g_{N-1}) \end{bmatrix} = D \begin{bmatrix} \pi_{\Sigma(t)}(g_{0}) \\ \pi_{\Sigma(t)}(g_{1}) \\ \vdots \\ \pi_{\Sigma(t)}(g_{N-1}) \end{bmatrix}.$$
(4.38)

From (4.29) and from the above expressions we obtain (4.34).

For the sake of the notational convenience let

$$F := \left(1 - \frac{2}{N}\right)I + \frac{2}{\nu N}A + \frac{1}{N}C + \frac{1}{\nu N}B - \frac{1}{\nu N}D.$$
(4.39)

Notice that F is a doubly stochastic matrix and that $\mathbb{1}^*\Pi(t) = 0$ for all t. Clearly, since $\lim_{t\to\infty} \Sigma(t) = 0$ we have that also $\lim_{t\to\infty} \Pi(t) = 0$ and the speed of convergence of $\Pi(t)$ toward zero is given by the essential spectral radius of the matrix F that, accordingly to the definition provided in (2.12) we denote by $\rho_{ess}(F)$. Let now $\Pi_1(t)$ denote the first component of $\Pi(t)$. Observe that

$$d(t) = \Pi_1(t).$$

which implies that also the speed of convergence of d(t) toward zero is determined by $\rho_{ess}(F)$. We have not been able to provide a theoretical characterization of $\rho_{ess}(F)$. However, supported by several numerical results, we conjecture the following.

Conjecture 4.15 Let \mathcal{G} be a graph satisfying Assumption 4.9, and let F be defined as in (4.39). Moreover let ν denote the degree of \mathcal{G} . Then

$$\rho_{ess}(F) = 1 - \frac{C}{N^{\frac{4+\nu}{\nu}}} + o\left(\frac{1}{N^{\frac{4+\nu}{\nu}}}\right).$$

as $N \to \infty$, where C is a constant independent of the graph \mathcal{G} .

A possible way of proving the above Conjecture, could be by considering the matrix $\frac{1}{N}C + \frac{1}{\nu N}B - \frac{1}{\nu N}D$ as a perturbation of the matrix $(1 - \frac{2}{N})I + \frac{2}{\nu N}A$. Indeed, we notice that the matrix $\frac{1}{N}C + \frac{1}{\nu N}B - \frac{1}{\nu N}D$ has only $3\nu + 1$ elements different from 0, which should become less and less significant as $N \to \infty$. Moreover note that, since $(1 - \frac{2}{N})I + \frac{2}{\nu N}A$ is a Cayley doubly stochastic matrices, we could characterize precisely its essential spectral radius by the arguments illustrated in Section 3.3.1.

4.6 Conclusions

In this Chapter we have introduced the randomized consensus algorithms. We have reviewed the concept of probabilistic consensus and some algebraic conditions ensuring the achievement of this probabilistic consensus. We have then analyzed two particular random strategies, the time-varying strategy with bounded in-degree and the time-varying Cayley graphs strategy, proving that they allow to achieve better performance than deterministic strategies with comparable complexity. Finally we have considered the symmetric gossip algorithm, whose study is motivated by applications to sensors, peer-to-peer and *ad hoc* networks. In particular we have provided an interesting characterization of the symmetric gossip algorithm over the Cayley graphs.

Some important questions remain open. In particular, a detailed analysis of the convergence rate of the symmetric gossip algorithms needs to be carried out. We believe that quite rich and complete results can be obtained for the broad class of graphs possessing symmetries the Cayley graphs. Another interesting family of graphs to work on would be the random geometric graphs.

Chapter 5

Quantized Consensus: Time-Invariant case

5.1 Introduction

We have seen in the Introduction, in Chapter 2 and in Chapter 4 that different algorithms for average consensus have been proposed in the literature. They can be distinguished on the basis of the amount of communication and computation they require, of their scalability properties with respect to the number of nodes, of their adaptability to time-varying graphs, and, finally, on the basis of their deterministic or randomized operating protocol.

However there is a feature common to most of the literature on the consensus problem: the assumption that the communication channel between the nodes allows to transfer real numbers with no errors. In many practical applications this is not a realistic assumption: if we think, for instance, of sensor networks communicating in a wireless fashion, it is evident that energy and bandwidth limitation yield a finite capacity channel. This suggests that in many situations the communication channel should be rather considered as digital, accepting messages taking values in some finite alphabet. This clearly forces a quantization on the real numbers that agents have to transmit.

The effects of quantization in feedback control problems have been widely studied in the past [91], mainly in the stabilization problem. Moreover granularity effects different from quantization in the consensus problems have been tackled in few papers especially in the load balancing applications [54]. Our setting is however different: this chapter studies the consensus problem under the assumption that transmissions are quantized, in particular we assume here that the communications network is constituted of ideal noiseless digital channels.

We review now the literature on the quantized consensus. The investigation of consensus under quantized communication started with [75]. In this paper the authors study systems having (and transmitting) integer-valued states and propose a class of gossip algorithms which preserve the average of states and are guaranteed to converge up to one quantization bin. Besides the fact there is not precise consensus, since the algorithm requires the use of a single link per iteration, the convergence is slow.

The authors in [150] analyzed the impact of the quantization noise through modification of the consensus algorithm proposed in [148], where the case of noisy communication links is addressed. Precisely, the authors in [148] consider the case in which the information transmitted by each node is noisecorrupted, being the noise additive and zero-mean. In [150] it is noted that the noise component can be considered as the quantization noise and by simulations, it is shown for small N that, if the increasing correlation among the node states is taken into account, the variance of the quantization noise diminishes and nodes converge to a consensus.

In [33] the authors propose a distributed algorithm that uses quantized values and preserves the average at each iteration. They showed favorable convergence properties using simulations on some static topologies, and provided performance bounds for the limit points of the generated iterates. In [92] the more general case of time-varying topologies is considered. A scheme reaching the consensus is proposed even if not at the initial average. However polynomial bounds on both the convergence time and the discrepancy from the initial average are provided in terms of the number of quantization levels.

Of note is that all the paper mentioned above considered quantized strategy based on deterministic uniform quantizers. Reference [8] proposes a simple distributed and iterative scheme to compute the average at each sensor node utilizing only quantized information communication. The authors in [8] adopt the probabilistic quantization (PQ) scheme (see [146]) to quantize the information before transmitting to the neighboring sensors. They show that the node states reach consensus to a quantized level; only in expectation do they converge to the desired average. Moreover if the quantization step size is large this approach will lead to large residual errors.

In this chapter and in the next two, the main goal is to provide an analysis identifying what are the effects due to the presence of the digital transmission and trying to overcome these effects. We will do this by following the analysis presented in [61] which extends the algorithm appeared in [33, 34].

The first problem we will address is to understand what happens when in the equation (2.1), ruling the average consensus algorithm, in place of the exact values of the states of the systems we substitute a quantized version of these obtained through a deterministic uniform quantizer or a probabilistic quantizer (we will define rigorously these quantizers in Section (5.2)). Two questions mainly are in order:

- (i) is it still possible to reach the average consensus, and
- (ii) if not, how far is it the asymptotic behavior of the states of the systems from the average consensus?

Answering the above questions is quite challenging. We will try to provide some insights on the above issues in this and the next two chapters.

We postpone the explanation of the organization of this chapter inside the next Section, where we will formulate precisely the problem we are dealing with.

5.2 Problem formulation

In this chapter we consider the consensus algorithm (2.10) restricting to the case in which the consensus matrix P is doubly stochastic. For the sake of the clarity we briefly review it. Assume that we have a set V of systems and a graph \mathcal{G} on V describing the feasible communications among the systems. For each system $i \in V$ let $x_i(t)$ denote the estimate of the average of agent i at time t. We assume that at every times t agent i updates its estimate according to

$$x_i(t+1) = \sum_{j=1}^{N} P_{ij} x_j(t).$$
(5.1)

where P_{ij} are the elements of a doubly stochastic $P \in \mathbb{R}^{N \times N}$ matrix compatible with \mathcal{G} . More compactly we can write

$$x(t+1) = Px(t),$$
 (5.2)

where x(t) is the column vector entries $x_i(t)$ represent the agents states. We have seen in Section 2.2.1 that, if P is a doubly stochastic matrix with positive diagonal entries and such that \mathcal{G}_P is strongly connected, then the algorithm (5.2) solves asymptotically the *average consensus problem*, namely

$$\lim_{t \to +\infty} x(t) = \left(\frac{1}{N} \sum_{i=1}^{N} x_i(0)\right) \mathbb{1},$$
(5.3)

where 1 is the column vector of all ones. From now on we will assume the following.

Assumption 5.1 *P* is a doubly stochastic stochastic matrix such that $P_{ii} > 0$, for $i \in \{1, ..., N\}$, and such that \mathcal{G}_P is strongly connected.

Note that the algorithm (5.2) relies upon a crucial assumption: each agent transmits to its neighboring agents the precise value of its state. This implies the exchange of perfect information through the communication network.

In what follows, we consider a more realistic case, i.e., we assume that the communication network is constituted only of rate-constrained digital links. This prevents the agents from having a precise knowledge about the state of the other agents. In fact, through a digital channel, the *i*-th agent can only send to its neighbors symbolic data in a finite alphabet; using only this data, the neighbors of the *i*-th agent can build at most an estimate of the *i*-th agent's state. In the following we will denote this estimate by $\hat{x}_i(t)$. Let

$$\hat{x}(t) = \begin{bmatrix} \hat{x}_1(t) \\ \vdots \\ \hat{x}_N(t) \end{bmatrix}$$

By remarking that (7.1) and (7.2) can be written respectively as

$$x_i(t+1) = x_i(t) + \sum_{j=1}^{N} P_{ij} \left(x_j(t) - x_i(t) \right)$$

and

$$x(t+1) = x(t) + (P - I)x(t),$$

in this chapter we consider the following updating rule

$$x_i(t+1) = x_i(t) + \sum_{j=1}^{N} P_{ij}\left(\hat{x}_j(t) - \hat{x}_i(t)\right)$$
(5.4)

or more compactly

$$x(t+1) = x(t) + (P - I)\hat{x}(t).$$
(5.5)

It is easy to see that this law has the remarkable property of maintaining the initial state average, whatever $\hat{x}(t)$ is. This fact is stated in the following proposition.

Proposition 5.2 Consider (5.5). Let

$$x_{ave}(t) = \frac{1}{N} \mathbb{1}^* x(t).$$

Then

$$x_{ave}(t) = x_{ave}(0)$$

for all $t \geq 0$.

Proof: We have that

$$\mathbf{1}^* x(t+1) = \mathbf{1}^* x(t) + \mathbf{1}^* (P - I) \hat{x}(t)
= \mathbf{1}^* x(t)$$

where the last inequality follows from the fact, since P is doubly stochastic, it holds $\mathbb{1}^*(P-I) = 0$.

We proceed now by illustrating two types of quantizers which has been introduced in literature in order to transmit information throughout a digital channel. In [33], the authors analyze the case in which

$$\hat{x}_i(t) = q_d(x_i(t)),$$
(5.6)

where, given a real number $z, q_d : \mathbb{R} \to \mathbb{Z}$ is the mapping sending z to its nearest integer, namely,

$$q_d(z) = n \in \mathbb{Z} \quad \Leftrightarrow \quad \begin{array}{l} z \in [n - 1/2, n + 1/2], & \text{if } z \ge 0\\ z \in [n - 1/2, n + 1/2], & \text{if } z < 0. \end{array}$$
(5.7)

We will refer to this quantizer as *deterministic quantizer*¹.

Instead in [8, 146], the so-called *probabilistic quantizer* is introduced. This quantizer woks as follows. Let $x \in \mathbb{R}$ and let $q_p(\cdot)$ denote the *probabilistic quantizer*². Assume, as for the *deterministic quantizer* above described, that the set of quantization levels is composed by the integer numbers, then $q_p(x)$ is the binary random variable defined as follows

$$q_p(x) = \begin{cases} \lfloor x \rfloor \text{ with probability } \lceil x \rceil - x \\ \lceil x \rceil \text{ with probability } x - \lfloor x \rfloor \end{cases}$$
(5.8)

¹the subscript d means deterministic

²The subscript p means *probabilistic*

The following lemma discusses two important properties of the *probabilistic* quantizer (for the proof see [146]).

Lemma 5.3 Let $q_p(x)$ be a probabilistic quantization of $x \in \mathbb{R}$. Then $q_p(x)$ is an unbiased representation of x, i.e.,

$$\mathbb{E}\left[q_p(x)\right] = x.\tag{5.9}$$

Moreover

$$\mathbb{E}\left[\left(x-q_p(x)\right)^2\right] \le \frac{1}{4}.$$
(5.10)

From now on, given a vector $x \in \mathbb{R}^N$, with a slight abuse of notation, we will use the notation $q_d(x) \in \mathbb{R}^N$ (respectively $q_p(x) \in \mathbb{R}^N$) to denote the vector such that $q_d(x) = [q_d(x_1), \ldots, q_d(x_N)]^*$ (respectively $q_p(x) = [q_p(x_1), \ldots, q_p(x_N)]^*$).

The main goal of this chapter will be to analyze the following systems

$$x(t+1) = x(t) + (P - I)q_d(x(t))$$
(5.11)

and

$$x(t+1) = x(t) + (P - I)q_p(x(t)).$$
(5.12)

Some remarks are now in order.

Remark 5.4 In this paper we consider quantizers having quantization step equal to 1. More general quantizers, with quantization step a generic positive real number ϵ , can be obtained from q_d and q_p by defining $q_d^{(\epsilon)}(x) = \epsilon q_d(x/\epsilon)$ and $q_p^{(\epsilon)}(x) = \epsilon q_p(x/\epsilon)$. Namely, the general case can be simply recovered by a suitable scaling. For the sake of the completeness, it is worth noting that in this case

$$\mathbb{E}\left[(x-q_p^{(\epsilon)}(x))^2\right] \le \frac{\epsilon^2}{4}.$$

Remark 5.5 We could define two different state update equations. By assuming that each node *i* uses the exact knowledge of its own state x_i instead of the estimate \hat{x}_i we could adopt

$$x_{i}(t+1) = x_{i}(t) + \sum_{j=1}^{N} P_{ij} \left(\hat{x}_{j}(t) - x_{i}(t) \right),$$

$$= P_{ii}x_{i}(t) + \sum_{j \neq i} P_{ij}\hat{x}_{j}(t),$$

(5.13)

instead of equation (5.4). Note that (7.9) can be written, in vector form, as

$$x(t+1) = \operatorname{diag} \{P\} x(t) + (P - \operatorname{diag} \{P\}) \hat{x}(t),$$

where diag $\{P\}$ means a diagonal matrix with the same diagonal elements of the matrix P. Instead, by assuming that each node i uses only quantized information also related to its own state, we could adopt

$$x_{i}(t+1) = \hat{x}_{i}(t) + \sum_{j=1}^{N} P_{ij} \left(\hat{x}_{j}(t) - \hat{x}_{i}(t) \right),$$

$$= \sum_{j=1}^{N} P_{ij} \hat{x}_{j}(t),$$

(5.14)

or in vector form

$$x(t+1) = P\hat{x}(t). \tag{5.15}$$

The drawback of these choices is that, differently from (5.5), they do not preserve the average of the initial condition. However, we will come back on these laws at the end of this chapter.

Remark 5.6 The authors in [8] analyze the rule (5.14), when the estimates are given by the *probabilistic quantizer*, namely the algorithm

$$x(t+1) = Pq_p(x(t)).$$
(5.16)

They prove that (5.16) reaches, almost surely, consensus on a quantization level. However the distance of the consensus point from the initial average depends heavily on the matrix P. We will review later the main features of this strategy.

It is worth noting that both the algorithm (5.11) and the algorithm (5.12), because of the quantization effects, are not expected to converge in the sense (5.3). What we can hope is for the agents to reach states which are close to each other and close to the average $x_a(0)$. To measure this asymptotic disagreement, we introduce the following quantity $\Delta_i(t) := x_i(t) - x_a(t) =$ $x_i(t) - x_a(0)$ which represents the distance, at time t, of the *i*-th agent from the average of the initials conditions. Let now $Y = I - N^{-1}\mathbb{1}\mathbb{1}^*$ and $\Delta(t) =$ $[\Delta_1(t), \ldots, \Delta_N(t)]^*$. Then, $\Delta(t) = Y x(t)$. Finally we define the performance index

$$d_{\infty}(P, x(0)) = \limsup_{t \to \infty} \frac{1}{\sqrt{N}} ||\Delta(t)||.$$
(5.17)

Note that the factor $\frac{1}{\sqrt{N}}$ is introduced in order to scale $d_{\infty}(P, x(0))$ with the dimension of the network. Moreover we can get rid of the initial condition by considering

$$d_{\infty}(P) = \sup_{x(0)} d(P, x(0)).$$
(5.18)

The problem we would like to address is, given a matrix P, to evaluate how big d_{∞} is. In particular, we are interested in evaluating how, for sequences of graphs of increasing size, this quantity depends on the number of the nodes N.

In order to obtain a first insight in the problem, we provide immediately some simulative results regarding two particular communication topologies: the random geometric graph and the hypercube graph.

The random geometric graph is commonly used to model wireless networks [67]. It is constructed by randomly placing N nodes in the unit square, and joining them with edges whenever their distance is below a threshold $R = \Theta(\sqrt{\log N/N})$ for $N \to \infty$. Moreover we assume the weights of the matrix P are chosen by using the Metropolis weights, illustrated in Example 2.14.

The *n*-dimensional hypercube graph is the graph obtained drawing the edges of a *n*-dimensional hypercube. It has $N = 2^n$ nodes which can be identified with the binary words of length *n*. Two nodes are neighbors if the corresponding binary words differ in only one component. Thus every node exchanges information with other *n* nodes. In this case a matrix *P* can be constructed by setting equal every non-zero entry of *P*. We will be more precise in Section 5.3.1.

The behavior of $d_{\infty}(P)$, which is obtained as the average over several realizations on the initial conditions, for sequences of graphs of increasing size of both topologies is depicted in Figures 5.1 and 5.2 for the strategy using the deterministic quantizer, and Figures 5.3 and 5.5 for the strategy using the probabilistic quantizer. It is worth noting that remarkably in all the cases, $d_{\infty}(P)$ appears to be bounded on N, the size of the communication graph. Moreover the two different quantizers behave quite similarly for both the topologies.

We conclude this section with some considerations and explaining how the rest of the chapter is organized.

Obviously we expect that $d_{\infty}(P)$ depend on the particular communication topology we are considering and on the weights assigned to the matrix P. In general, assigned the communication topology, what one would like to do is



Figure 5.1: Performance of the random geometric graph when the deterministic quantizer is used. Since the graph itself is random, the plot of $d_{\infty}(P)$ comes from averaging over realizations of both the graph and the initial condition.



Figure 5.2: Performance of the n-dimensional hypercube graph (of order $N = 2^n$) when the deterministic quantizer is used.

finding the matrix of weights P minimizing $d_{\infty}(P)$. In order to do so, it is important to provide some tools which permit to estimate $d_{\infty}(P)$.

In spite of the fact that the proposed algorithms are still intrinsically linear, the quantization effects introduce nonlinearities which make the exact asymptotic analysis of the algorithm, quite hard. Indeed, we are able to carry



Figure 5.3: Performance of the random geometric graph when the probabilistic quantizer is used. Since the graph itself is random, the plot of $d_{\infty}(P)$ comes from averaging over realizations of both the graph and the initial condition.



Figure 5.4: Performance of the n-dimensional hypercube graph when the probabilistic quantizer is used.

on such an analysis only in very specific examples.

In Section 5.3 we undertake a twofold analysis of the algorithm using deterministic quantizers: worst case and probabilistic. The basic idea is to study the system considering the communication errors induced by quantization as (unknown) bounded disturbances. The worst case analysis is obtained by introducing a bounded error model and maximizing $d_{\infty}(P)$ with respect to all possible realizations of the communication errors: of course in this way we obtain an upper bound to the performance of our algorithm. This is done in Section 5.3.1 where we prove in general the convergence to a neighborhood of the average, obtaining bounds on its size. These bounds are independent of the initial condition but depend on the diffusion matrix. However the worst-case analysis is intrinsically conservative. In fact we will show, for the hypercube example cited above, that the worst-case displacement grows logarithmically in N, behavior which is in disagreement with the experimental evidence displayed by Figure 5.2.

For this reason, in Section 5.3.2 we propose an alternative method and develop a probabilistic analysis, modeling the quantization error as additive random noise affecting the received data. A classical mean squared analysis for the asymptotic error can be carried on in this case. A similar analysis is done in [148]. It comes out that, under mild assumptions on the diffusion matrix, the expected behavior depends only on the assumed distribution of the errors and on the spectrum of the evolution matrix. Of course, this probabilistic analysis, in principle, does not offer any rigorous bound on our system. However, simulations clearly show that the probabilistic analysis is very close to the experimental evidence, contrarily to the worst case analysis which instead is quite conservative.

In spite to the fact that there is no theoretical reason explaining why the probabilistic model should be a suitable model for analyzing the algorithm (5.11), we will show in Section 5.4 that the probabilistic model, due to structure of the probabilistic quantizer, represent the correct way to analyze (5.12). In particular we will compare (5.12) with another scheme proposed in literature based on the use of probabilistic quantizers, and we will show how plugging together these two schemes it is possible to solve, almost surely, the consensus problem, very close to the average of the initial conditions.

5.3 Time Invariant case- Deterministic quantizers

Consider (5.11), i.e. the updating law using the deterministic quantizer. We start with a remark about the best achievable performance. It is clear that, when all states lie in the same quantization interval, namely $q(x_i(t)) = Q$ for all *i*, differences are not perceivable and states do not evolve. Therefore the best the algorithm can assure is that the system reaches such an equilibrium in which $q(x_i(t)) = Q$ for all *i* and for all $t \ge T$. In this case we would obtain that $|\Delta_i(t)| \le 1$ for all $t \ge T$. This implies that the best we can obtain is $d_{\infty} \le 1/2$. Unfortunately simulations show that in many cases the

error from the agreement can be bigger. However, it is worth noting that the states of the agents subject to (5.11) are bounded. In particular one can see, by convexity arguments, that for any node $i \in V$ and for all $t \geq 0$,

$$x_i(t) \in \left[\min_{j \in V} \{q(x_j(0))\} - \frac{1}{2}, \max_{j \in V} \{q(x_j(0))\} + \frac{1}{2}\right].$$
 (5.19)

Of course (5.19) is a very weak result since we can hope that in general the disagreement will decrease as time goes on and that the asymptotic disagreement will not depend on the initial conditions, but just possibly on P. We consider now two examples in which the evolution of the system can be studied explicitly, the complete graph and the directed circuit graph.

Example 5.7 (Complete graph) If the communication graph is complete and the communication is exact, i.e. not quantized, the average consensus problem can be solved in one step taking $P = \frac{1}{N} \mathbb{1}\mathbb{1}^*$. We now compute the exact performance degradation due to quantization. The system is in this case

$$x(t+1) = x(t) - q(x(t)) + N^{-1} \mathbb{1} \mathbb{1}^* q(x(t)).$$
(5.20)

We have that, for $t \ge 1$,

$$\begin{aligned} |\Delta_i(t)| &= \left| x_i(t) - \frac{1}{N} \sum_{j=1}^N x_j(0) \right| \\ &= \left| x_i(t) - \frac{1}{N} \sum_{j=1}^N x_j(t-1) \right| \\ &= \left| x_i(t-1) - q\left(x_i(t-1) \right) + \frac{1}{N} \sum_{j=1}^N q\left(x_j(t-1) - \frac{1}{N} \sum_{j=1}^N x_j(t-1) \right) \right| \\ &\leq |x_i(t-1) - q\left(x_i(t-1) \right)| + \frac{1}{N} \sum_{j=1}^N |q\left(x_j(t-1) - x_j(t-1) \right)| \\ &= \frac{1}{2} + \frac{1}{N} \frac{N}{2} = 1. \end{aligned}$$

This implies that

$$d_{\infty}(N^{-1}\mathbb{1}\mathbb{1}^*) \le 1. \tag{5.21}$$

Example 5.8 (Directed circuit) Now we consider a more interesting example, the directed circuit graph, which is described by the Cayley graph

 $\mathcal{G}(\mathbb{Z}_N, \{0, 1\})$ (see Section 3.3). In this case each agent communicates with only one neighbor and evolves following

$$x_i(t+1) = x_i(t) + \frac{1}{2}[q(x_{i+1}(t)) - q(x_i(t))] \quad i = 1, \dots, N,$$
 (5.22)

where summation of the indexes is to be intended mod N. The evolution of (5.22) can be studied exactly by means of a *symbolic dynamics* approach. This analysis is definitively not trivial, but permits us to characterize precisely the evolution of (5.22) and to obtain a strong result.

To start, we need the following technical lemma. Let $\lfloor \cdot \rfloor$ and $\lceil \cdot \rceil$ denote the floor and ceiling operators from \mathbb{R} to \mathbb{Z} .

Lemma 5.9 Given $\alpha, \beta \in \mathbb{N}$ and $x \in \mathbb{R}$, it holds

$$\lfloor x \rfloor = \left\lfloor \frac{\lfloor \alpha x \rfloor}{\alpha} \right\rfloor \tag{5.23}$$

$$q(x) = \lfloor x + 1/2 \rfloor = \left\lceil \frac{1}{2} \left\lfloor \frac{\lfloor 2\beta x \rfloor}{\beta} \right\rfloor \right\rceil.$$
 (5.24)

Proof: We first prove (5.23). Let $m = \lfloor x \rfloor$. So

$$m \le x < m+1$$

$$\alpha m \le \alpha x < \alpha m + \alpha.$$

Hence, we can find $s \in \mathbb{N}, 0 \le s \le \alpha - 1$ such that $\alpha m + s \le \alpha x < \alpha m + s + 1$. This yields $\lfloor \alpha x \rfloor = \alpha m + s$ and $\lfloor \frac{\lfloor \alpha x \rfloor}{\alpha} \rfloor = m$.

Then we prove equation (5.24). The equality $q(x) = \lfloor x + 1/2 \rfloor$ is clear from the definition of q(x). To prove the second equality, let $h = \lfloor 2x \rfloor$. Then $h \leq 2x < h + 1$, from which follows that

$$\frac{h}{2} + \frac{1}{2} = \frac{h+1}{2} \le x + 1/2 < \frac{h+2}{2} = \frac{h}{2} + 1$$

From this inequality it follows that $\lfloor x+1/2 \rfloor = \lceil \frac{h}{2} \rceil$. This, with (5.23), implies (5.24).

We define now $n_i(t) = \lfloor 2x_i(t) \rfloor$. Simple properties of floor and ceiling operators, together with the above lemma, allow us to remark that $q(x_i(t)) = \left\lceil \frac{n_i(t)}{2} \right\rceil$ and to derive from (5.22) that

$$\lfloor 2x_{i}(t+1) \rfloor = \lfloor 2x_{i}(t) \rfloor + q(x_{i+1}(t)) - q(x_{i}(t))$$

and hence

$$n_i(t+1) = n_i(t) + \left\lceil \frac{n_{i+1}(t)}{2} \right\rceil - \left\lceil \frac{n_i(t)}{2} \right\rceil$$
$$= \left\lfloor \frac{n_i(t)}{2} \right\rfloor + \left\lceil \frac{n_{i+1}(t)}{2} \right\rceil.$$

We have thus found an iterative system involving only the symbolic signals $n_i(t)$:

$$n_i(t+1) = g(n_i(t), n_{i+1}(t))$$
(5.25)

where

$$g(h,k) = \left\lfloor \frac{h}{2} \right\rfloor + \left\lceil \frac{k}{2} \right\rceil.$$
 (5.26)

The asymptotic analysis of (5.25) will then allow us to obtain information about the asymptotics of $x_i(t)$ (since $n_i(t) = \lfloor 2x_i(t) \rfloor$) up to quantization errors equal to 1.

We now start the analysis of system (5.25). Define the following quantities: $m(t) = \min_{1 \le i \le N} n_i(t)$, $M(t) = \max_{1 \le i \le N} n_i(t)$, D(t) = M(t) - m(t). From the form of (5.25) one can easily remark that m(t) can not decrease and M(t) can not increase. Hence D(t) is not increasing. A much stronger result about the monotonicity of D(t) is the content of the following lemma.

Lemma 5.10 If $D(t_0) \ge 2$, there exists $T \in \mathbb{N}$ such that $D(t_0 + T) < D(t_0)$.

Proof: Let $I_m(t) = \{j \in \mathbb{Z}_N \text{ s.t. } n_j(t) = m(t)\}$. The idea of the proof will be to show that the set $I_m(t)$ eventually decreases if we are in the range $D(t) \geq 2$.

Notice first that, for $h, k \in \mathbb{Z}$, g(h+2, k+2) = g(h, k) + 2. Hence, by an appropriate translation of the initial condition, we can always restrict ourselves to the case $m(0) \in \{0, 1\}$.

Case $m(t_0) = 0$. Notice that

$$g(h,k) > 0 \ \forall h \ge 0, \ k > 0, \ g(h,0) > 0 \ \forall h \ge 2.$$
 (5.27)

This easily implies that $I_m(t)$ is not increasing. Now, since $D(t_0) \ge 2$, we can find $j_0 \in I_m(t_0)$ and two integers U > 0 and $W \ge 0$ such that

$$n_{j_0-W-1}(t_0) > 1$$

$$n_{j_0-v}(t_0) = 1 \quad 0 < v \le W$$

$$n_{j_0+u}(t_0) = 0 \quad 0 \le u < U$$

$$n_{j_0+U}(t_0) > 0.$$

After W instants step, we then obtain,

$$n_{j_0-W-1}(t_0+W) > 1$$

$$n_{j_0-u}(t_0+W) = 0 \quad W - U + 1 \le u \le W$$

$$n_{j_0+W-U}(t_0+W) > 0.$$

The time after one 0 will then disappear

$$n_{j_0-W-1}(t_0+W+1) > 0$$

$$n_{j_0-u}(t_0+W+1) = 0 \quad W-U+2 \le u \le W$$

$$n_{j_0+W-U-1}(t_0+W+1) > 0.$$

This implies that $|I_m(t_0 + W + 1)| < |I_m(t_0)|$.

Case $m(t_0) = 1$. Notice that

$$g(h,k) > 1 \ \forall h \ge 2, \ k \ge 1, \quad g(1,k) > 1 \ \forall k \ge 3.$$
 (5.28)

This easily implies that $I_m(t)$ is again not increasing. Now, since $D(t_0) \ge 2$, we can find $j_0 \in I_m(t_0)$ and an integer $W \ge 0$ such that

$$n_{j_0}(t_0) = 1$$

$$n_{j_0+w}(t_0) = 2 \quad 1 \le w \le W$$

$$n_{j_0+W+1}(t_0) > 2.$$

The evolution of the above configuration yields, after W instant steps

$$n_{j_0}(t_0 + W) = 1$$

 $n_{j_0+1}(t_0 + W) > 2.$

The next step, we obtain $n_{j_0}(t_0 + W + 1) > 1$. Therefore, $|I_m(t_0 + W + 1)| < |I_m(t_0)|$.

In both cases we have proven that $|I_m(t)|$ strictly decreases in finite number of steps. A straightforward induction principle then implies that a finite $T \in \mathbb{N}$ exists such that $m(t_0 + T) > m(t_0)$. This proves the result.

The interesting consequence of this lemma is the following result which characterizes the asymptotic behavior of the variable n_i .

Theorem 5.11 There exist $T \in \mathbb{N}$ and $h \in \mathbb{Z}$ such that, for all t > T, D(t) < 2 and, moreover, one of the following condition holds

- i) $n_i(t) = h, \forall i;$
- ii) $\{n_i(t) \mid i = 0, \dots, N\} = \{h, h+1\}$ and each $n_i(t)$ is constant in time;
- iii) $\{n_i(t) \mid i = 0, \dots, N\} = \{h, h+1\}$ and each $n_i(t)$ is periodic in time of period N.

Proof: From Lemma 5.10, it follows that a finite $T \in \mathbb{N}$ can be found, such that D(t) < 2 for all t > T. Once we reach his condition, there are two possibilities: either the $n_i(t)$ are all equal or they differ by 1. In the first case, the system remains constant (case i). In the second case, it follows from the way g is defined that if the lowest state is odd, the evolution is constant (case ii), while if the lowest one is even, the state evolution is a leftward shift (case iii). This is periodic of period N (and possibly also of some divisor of N).

We can now go back to the original system. The following result follows directly from Theorem 5.11.

Corollary 5.12 For system (5.22), there exists $T \in \mathbb{N}$ such that

$$|x_i(t) - x_j(t)| \le 1 \quad \forall i, j \quad \forall t > T, \tag{5.29}$$

and hence

$$d_{\infty}(P) \le 1/2.$$

Proof: Immediate consequence of Theorem 5.11, considering the relation $n_i(t) = \lfloor 2x_i(t) \rfloor$.

Unfortunately an exact analysis of the dynamics of system (5.11), as we did in the previous two examples, it is not feasible for general graphs. In the following subsections we will try to introduce some auxiliary models in order to provide some bounds on d_{∞} and some further considerations.

5.3.1 Bounded error model-Worst case analysis

In this section we undertake a worst case analysis which can be applied general graphs. We start by observing that (5.11) can be rewritten in the following way

$$x(t+1) = Px(t) + (P - I)(q_d(x(t)) - x(t)),$$
(5.30)

where $q_d(x(t)) - x(t)$ is such that $||q_d(x(t)) - x(t)||_{\infty} \leq 1/2$. In order to carry out a worst-case analysis of (5.30), we introduce the following bounded error model

$$\begin{cases} x_w(t+1) = Px_w(t) + (I-P)e(t), & x_w(0) = x(0) \\ \Delta_w(t) = Yx_w(t) \end{cases}$$
(5.31)

where $e(t) \in \mathbb{R}^N$ is such that $||e(t)||_{\infty} \leq 1/2$ for all $t \geq 0$ and where we again $Y = I - \frac{1}{N} \mathbb{1} \mathbb{1}^*$. Notice that in this case e(t) is no more a quantization error, but instead represents an unknown bounded disturbance. Clearly, when e(t) = q(x(t)) - x(t) it turns out that $x_w(t) = x(t)$ and $\Delta_w(t) = \Delta(t)$ for all $t \geq 0$.

We define now a performance index for (5.31), considering the worst asymptotic disagreement, worst with respect to all the possible choices of the time sequence of the vectors e(t). To be more precise, let us introduce $\mathcal{E}^{\infty} = \{\{e(\cdot)\}_{t=0}^{\infty} | ||e(t)||_{\infty} \leq \frac{1}{2}, \forall t \geq 0\}$, namely the set of all the sequences of *N*-dimensional vectors having sup norm less than 1/2. Then, for the system (5.31), we define

$$d_{\infty}^{w}(P, x_{w}(0)) = \sup_{\mathcal{E}^{\infty}} \limsup_{t \to \infty} \frac{1}{\sqrt{N}} ||\Delta_{w}(t)||.$$
(5.32)

Note that $\lim_{t\to\infty} YP^t = 0$. This implies that the asymptotic behavior of $\Delta_w(t)$ is independent of the initial condition $x_w(0)$ and hence this is the case also for the quantity $d_{\infty}^w(P, x_w(0))$. Thus, from now on we will denote $d_{\infty}^w(P, x_w(0))$ simply by $d_{\infty}^w(P)$. As a preliminary remark, note that

$$d_{\infty}(P) \le d_{\infty}^{w}(P). \tag{5.33}$$

We start our analysis of $d_{\infty}^{w}(P)$ by the following example.

Example 5.13 In this example we consider the hypercube graph. Precisely, we consider the group \mathbb{Z}_2^n where $2^n = N$ and the Cayley graph $\mathcal{G}(\mathbb{Z}_2^n, S)$, where $S = \{e_0, e_1, \ldots, e_n\}$, with $e_0 = [0, \ldots, 0]^*$ and e_j the vector with all elements equal to 0 except a 1 in position j if $j \neq 0$. Clearly |S| = n + 1. We assume that the matrix P has the following structure

$$P_{ij} = \begin{cases} \frac{1}{n+1} & \text{if } i-j = e_h \ \exists \ h : \ 0 \le h \le n, \\ 0 & \text{otherwise} \end{cases}$$
(5.34)

for all *i* and *j* belonging to \mathbb{Z}_2^n . In other words we have that $P = \frac{1}{n+1}A$ where A is the adjacency matrix of the Cayley graph $\mathcal{G}(\mathbb{Z}_2^n, S)$. For the sake of the

clarity we recall that, according to our convention, in the adjacency matrix we consider also the presence of self-loops (see Appendix B). We have the following result.

Theorem 5.14 Let P be as above. Then

$$d^w_{\infty}(P) = \frac{n}{2} = \frac{\log_2 N}{2}$$

where given a matrix $M \in \mathbb{R}^{N \times N}$, $\rho(M)$ denotes the spectral radius of M.

In order to prove the above Theorem we need of the following preliminary result.

Lemma 5.15 Let P be as above. Then

$$\sum_{s=0}^{\infty} \rho(P^s(I-P)) = n = \log_2 N, \tag{5.35}$$

where we recall that, according to definition (2.12), given a matrix $M \in \mathbb{R}^{N \times N} \rho_{ess}(M)$ denotes the essential spectral radius of M.

Proof: The eigenvalues of P are $\lambda_k = 1 - \frac{2k}{n+1}$ $k = 0 \dots n$, with multiplicities $p_k = \binom{n}{k}$ (see [43]). Then,

$$\sum_{s=0}^{\infty} \rho(P^s(I-P)) = \sum_{s=0}^{\infty} \rho_{ess}(P^s)\rho(I-P) = \sum_{s=0}^{\infty} (1 - \frac{2}{n+1})^s (2 - \frac{2}{n+1}) = n,$$

We are able now to provide the proof of Theorem 5.14.

Proof: First we rewrite the expression of $d_{\infty}^{w}(P)$. Since our P is symmetric, P is diagonalizable by an orthogonal matrix. We can write that $P = \sum_{h=0}^{N-1} \lambda_h q_h q_h^*$ where q_h are orthonormal. These facts are true also for $P^s(I-P)$. Moreover we have that $\rho(P^s(I-P)) = ||P^s(I-P)||$. Let

$$\Delta_w^{(f)}(t) := \sum_{s=0}^{t-1} P^s (I-P) e(t-s-1).$$

Then

$$\begin{split} \|\Delta_w^{(f)}(t)\|^2 &= \|\sum_{s=0}^{t-1} P^s (I-P) e(t-s-1)\|^2 \\ &= \|\sum_{s=0}^{t-1} \sum_{h=0}^{N-1} \lambda_h^s (1-\lambda_h) q_h q_h^* e(t-s-1)\|^2 \\ &= \|\sum_{h=0}^{N-1} q_h (1-\lambda_h) \sum_{s=0}^{t-1} \lambda_h^s q_h^* e(t-s-1)\|^2 \\ &= \sum_{h=0}^{N-1} \left[(1-\lambda_h) \sum_{s=0}^{t-1} \lambda_h^s q_h^* e(t-s-1) \right]^2. \end{split}$$

Hence, $[d_{\infty}^{w}(P)]^{2} = \max_{\mathcal{E}^{\infty}} \limsup_{t\to\infty} \frac{1}{N} \|\Delta_{w}^{(f)}(t)\|^{2}$. Now we start using combinatorial tools. Indeed the vertices of the hypercube, as well as the eigenvalues and eigenvectors of P, can be indexed by the subsets of $\{1 \dots, n\}$ (see [69]). With this indexing, for each $I \subseteq \{1 \dots n\}$ the corresponding eigenvalue is $\lambda_{I} = 1 - \frac{2|I|}{n+1}$ and the eigenvector is the 2^{n} -dimensional vector $q^{(I)}$, such that its J-th component is equal to $q_{J}^{(I)} = 2^{-n/2}(-1)^{|I \cap J|}$. Let T be any positive integer and consider the sequence of vectors

$$e(0), e(1), \ldots, e(t), \ldots$$

such that J-th component of the vector e(t) is equal to $\frac{1}{2}(-1)^{T-1-r}(-1)^{|J|}$, where r is the remainder in the euclidean division of t over T. Observe that e(t+T) = e(t) for all $t \ge 0$. Observe, moreover, that e(t) is an eigenvector of P corresponding to the eigenvalue $\frac{1-n}{1+n}$ for all $t \ge 0$. Hence we have that

$$\begin{split} \frac{1}{N} \|\Delta_w^{(f)}(T)\|^2 &= \\ &= \frac{1}{N} \sum_{h=0}^{N-1} \left[(1-\lambda_h) \sum_{s=0}^{T-1} \lambda_h^s q_h^* e(T-s-1) \right]^2 \\ &= \frac{1}{2^n} \left[(1-\frac{1-n}{n+1}) \sum_{s=0}^{T-1} \left(\frac{1-n}{n+1} \right)^s 2^{-\frac{n}{2}} \sum_{J \subseteq \{1,\dots,n\}} (-1)^{|J|} \frac{1}{2} (-1)^s (-1)^{|J|} \right]^2 \end{split}$$

$$= \frac{1}{4^n} \left[\frac{n}{n+1} \sum_{s=0}^{T-1} \left(\frac{n-1}{n+1} \right)^s 2^n \right]^2$$
$$= \frac{n^2}{(n+1)^2} \left[\frac{1 - \left(\frac{n-1}{n+1} \right)^T}{1 - \frac{n-1}{n+1}} \right]^2$$
$$= \frac{n^2}{4} \left[1 - \left(\frac{n-1}{n+1} \right)^T \right]^2.$$

Assume now that T is an even positive integer. By recalling that e(t+T) = e(t) for all $t \ge 0$, for t = kT where $k \in \mathbb{N}$ it turns out that

$$\begin{split} \frac{1}{N} \|\Delta_w^{(f)}(kT)\|^2 &= \frac{n^2}{4} \left[1 - \left(\frac{n-1}{n+1}\right)^T \right]^2 \sum_{u=0}^{k-1} \left(\frac{1-n}{n+1}\right)^{uT} \\ &= \frac{n^2}{4} \left[1 - \left(\frac{n-1}{n+1}\right)^T \right]^2 \left[\frac{1 - \left(\frac{n-1}{n+1}\right)^{kT}}{1 - \left(\frac{n-1}{n+1}\right)^T} \right]^2 \\ &= = \frac{n^2}{4} \left[1 - \left(\frac{n-1}{n+1}\right)^{kT} \right]^2. \end{split}$$

Letting $k \to \infty$ we obtain that, for the particular sequence considered

$$\lim_{k \to \infty} \frac{1}{N} \|\Delta_w^{(f)}(kT)\|^2 = \frac{n^2}{4}$$
(5.36)

Therefore we have proved that

$$\limsup_{t\to\infty}\frac{1}{N}\|\Delta^{(f)}_w(kT)\|^2\geq \frac{n^2}{4}$$

and hence $[d_{\infty}^{w}(P)]^{2} \geq \frac{n^{2}}{4}$. Now, Lemma 5.15 implies that $[d_{\infty}^{w}(P)]^{2} \leq \frac{n^{2}}{4}$, and then the claim follows.

The above example shows immediately that there exists a discrepancy between the simulative evidence on d_{∞} and the estimate d_{∞}^{w} . Indeed, while d_{∞} seems to be uniformly bounded on N (see Figure 5.2), $d_{\infty}^{w} \to \infty$ as $N \to \infty$.

Nevertheless, in the literature of the quantized control, the bounded error model is the only model which permits us to infer some theoretical analysis on (5.11) and thus to provide some bound on d_{∞} . There is an another model, the probabilistic model that we will propose in the next section, that seems to be more in accordance with the experimental results. But the fact that the probabilistic model seems to capture the main features of (5.11) comes only from the experimental evidence: there is no theoretical justification motivating this agreement. For this reason we proceed in the analysis of d_{∞}^w . We start from the following result that provides a general bound for d_{∞}^w .

Proposition 5.16 Let P be a matrix satisfying Assumption 5.1. Then

$$\|PY\| < 1 \tag{5.37}$$

and

$$d_{\infty}^{w}(P) \le \frac{1}{1 - \|PY\|}.$$
(5.38)

Proof: We have that

$$\|PY\| = \sqrt{\rho\left((PY)^*PY\right)}.$$

Since PY = YP and $Y^2 = Y$ we can write that $(PY)^*PY = P^*PY$. Notice that the fact that P satisfies Assumption 5.1 implies both that $(P^*P)_{ii} > 0$ and \mathcal{G}_{P^*P} is strongly connected. Therefore we can write that $\sigma(P^*P) =$ $\{1, \lambda_1, \ldots, \lambda_{N-1}\}$, where $|\lambda_i| < 1$, $1 \le i \le N-1$. Observe that $\sigma(P^*PY) =$ $\{\sigma(P^*P) - \{1\}\} \cup \{0\}$. Hence ||PY|| < 1.

Consider now $\Delta_w(t)$. From standard algebraic tools we have that

$$\Delta_w(t) = YP^t x(0) + Y \sum_{s=0}^{t-1} P^s (I-P) e(t-s-1)$$
$$= (PY)^t \Delta(0) + \sum_{s=0}^{t-1} (PY)^s (I-P) e(t-s-1)$$

where in the last equality we have used again the facts that PY = YP and that $Y^k = Y$ for all k > 0. Now we have that

$$\begin{aligned} \|\Delta_w(t)\| &= \|(PY)^t \Delta_w(0) + \sum_{s=0}^{t-1} (PY)^s (I-P) e(t-s-1)\| \\ &\leq \|(PY)^t\| \|\Delta_w(0)\| + \|I-P\| \sum_{s=0}^{t-1} \|(PY)\|^s \|e(t-s-1)\| \\ &= \|(PY)^t\| \|\Delta_w(0)\| + \sqrt{N} \frac{1-\|PY\|^t}{1-\|PY\|} \end{aligned}$$

where in the last inequality we used the facts that $||I - P|| \le 2$ and $||e(t)|| \le \sqrt{N}/2$ for all $t \ge 0$. By letting $t \to \infty$ we obtain (5.38).

Note that, if P is normal we have that $||PY|| = \rho_{ess}(P)$ and hence (5.38) becomes

$$d_{\infty}^{w}(P) \le \frac{1}{1 - \rho_{ess}(P)}.$$

However, when P is a normal matrix the bound on $d_{\infty}^{w}(P)$ can be improved as stated in the next proposition.

Proposition 5.17 If P is normal, then

$$d_{\infty}^{w}(P) \le \frac{1}{2} \sum_{s=0}^{\infty} \rho(P^{s}(I-P)).$$
 (5.39)

Proof: Starting from the expression of $\Delta_w(t)$ provided along the proof of Proposition 5.16 we can write that

$$\begin{aligned} \|\Delta_w(t)\| &\leq \|(PY)^t \Delta_w(0)\| + \|\sum_{s=0}^{t-1} (PY)^s (I-P) e(t-s-1)\| \\ &\leq \|(PY)^t \Delta_w(0)\| + \frac{\sqrt{N}}{2} \|(PY)^s (I-P)\| \end{aligned}$$

Since P is normal we have that $||(PY)^s(I - P)|| = \rho((PY)^s(I - P)) = \rho(P^s(I - P))$. By letting $t \to \infty$ in the last inequality, we obtain (5.39).

Remark 5.18 It is worth noting that, from the sub-multiplicative inequality $||(PY)^s(I-P)|| \leq ||PY||^s ||I-P||$, it follows immediately that $\frac{1}{2} \sum_{s=0}^{\infty} \rho(P^s(I-P)) \leq \frac{1}{1-\rho_{ess}(P)}$ which shows that the bound (5.38) is indeed an improvement of the bound (5.39).

Example 5.19 (Complete graph) We recall that, in this case, $P = \frac{1}{N} \mathbb{1}\mathbb{1}^*$. Hence PY = 0. Thus we have that $\frac{1}{1-\|PY\|} = 1$. This is an alternative way to prove (5.21). However, since $(\frac{1}{N}\mathbb{1}\mathbb{1}^*)^k = \frac{1}{N}\mathbb{1}\mathbb{1}^*$ for all k > 0, it follows immediately that $\frac{1}{2}\sum_{s=0}^{\infty} \rho(P^s(I-P)) = \frac{1}{2}$ and this represents a refinement of (5.21). In general it is quite hard to evaluate (5.39). We provide two results which permit us to approximate (5.39) under some mild assumptions. First a notational definition. Given $c \in \mathbb{C}$ and $r \in \mathbb{R}$ such that $r \geq 0$, we denote

$$B_{c,r} := \{ z \in \mathbb{C} \mid ||z - c|| \le r \},\$$

namely the closed ball of complex numbers of radius r and centered in c.

Proposition 5.20 Let P be a normal matrix satisfying the Assumption 5.1. Let R be such that 0 < R < 1 and $\sigma(P) \subseteq B_{1-R,R}$ and let $\bar{\rho} = \rho_{ess}(P)$ denote the essential spectral radius of P. Then

$$\sum_{s=0}^{\infty} \rho(P^s(I-P)) \le \frac{1}{1-R} + \sqrt{\frac{8R}{(1-R)(1-\bar{\rho})}}.$$
 (5.40)

Proof: We want to upper bound $\rho(P^s(I-P)) = \max_{k=1}^{N-1} |\lambda_k^s(1-\lambda_k)|$. In order to do so we consider the function $f : \mathbb{C} \to \mathbb{R}$ defined as $f(z) = z^s(1-z)$. Let us consider the closed balls $B_{1-R,R}$ and $B_{0,\bar{\rho}}$. By Gershgorin's Theorem we have that $\sigma(P) \subseteq B_{1-R,R}$. By the definition of essential spectral radius it holds that $\sigma(P) \setminus \{1\} \subseteq B_{0,\bar{\rho}}$. Hence $\sigma(P) \setminus \{1\} \subseteq B_{0,\bar{\rho}} \cap B_{1-R,R}$. Let $A := B_{1-R,R} \cap B_{0,\bar{\rho}}$. Clearly

$$\max_{k=1}^{N-1} |\lambda_k^s (1 - \lambda_k)| \le \max_{z \in A} |f(z)|.$$

Since f is an analytic function and A is a compact set, from the Maximum Modulus Principle it follows that

$$\max_{k=1}^{N-1} |\lambda_k^s(1-\lambda_k)| \le \max_{z \in \partial A} |f(z)|,$$

where ∂A denotes the boundary of A. Consider now the curves $\gamma : [0, 2\pi] \to \mathbb{C}$,

$$\gamma(t) = 1 - R + Re^{jt},$$

and $\theta : [0, 2\pi] \to \mathbb{C}$,

$$\theta(t) = \bar{\rho}e^{jt},$$

which represent, respectively, the boundaries of $B_{1-R,R}$ and of $B_{0,\bar{\rho}}$. In the following, since $|f(z)| = |f(z^*)|$, we will consider γ and θ only on the interval $[0, \pi]$.

By calculating the intersection between γ and θ one can see that $\partial A = \tilde{\gamma} \cup \tilde{\theta}$ where

$$\tilde{\gamma} := \left\{ z = z_x + i z_y \in \gamma : z_x \le \frac{1 - 2R + \bar{\rho}^2}{2(1 - R)} \right\}$$

and

$$\tilde{\theta} := \left\{ z = z_x + i z_y \in \theta \, : \, z_x \ge \frac{1 - 2R + \bar{\rho}^2}{2(1 - R)} \right\}$$

We consider now |f(z)| on $\tilde{\gamma}$. By straightforward calculations one can show that

$$|f(\gamma(t))|^2 = 2R^2(1 - \cos t) \left[1 - 2R(1 - R)(1 - \cos t)\right]^s$$

Now let $x = R \cos t + 1 - R$. In order to analyze the behavior of $|f(\gamma(t))|^2$ we introduce the following auxiliary function

$$F(x) = 2R(1-x) \left[1 - 2(1-R)(1-x)\right]^{s}$$

A straightforward calculation shows that studying $|f(z)|^2$ on $\tilde{\gamma}$ is equivalent to study F on $\left[1-2R, \frac{1-2R+\bar{\rho}^2}{2(1-R)}\right]$. By taking the first derivative of F we obtain

$$\frac{\partial F}{\partial x} = 2R \left[1 - 2(1-R)(1-x) \right]^{s-1} \left[-1 + 2(1-R)(s+1)(1-x) \right]$$

We have that $\frac{\partial F}{\partial x} = 0$ for $x := x_1 = 1 - \frac{1}{2(1-R)}$ and $x := x_2 = 1 - \frac{1}{2(1-R)(s+1)}$. Note that $1 - \frac{1}{2(1-R)} \leq 1 - 2R$ for all R > 0. Moreover note that F is monotone increasing in $[x_1, x_2]$ and monotone decreasing for $[x_2, +\infty)$. Hence F reached its maximum value inside the interval $\left[1 - 2R, \frac{1-2R+\bar{\rho}^2}{2(1-R)}\right]$ on 1 - 2R if $x_2 \leq 1 - 2R$, on x_2 if $1 - 2R \leq x_2 \leq \frac{1-2R+\bar{\rho}^2}{2(1-R)}$, on $\frac{1-\bar{\rho}^2}{2R(1-R)}$ if $x_2 \geq \frac{1-\bar{\rho}^2}{2R(1-R)}$. We have that $x_2 \leq 1 - 2R \Leftrightarrow s \leq \frac{(1-2R)^2}{4R(1-R)}$, $1 - 2R \leq x_2 \leq \frac{1-2R+\bar{\rho}^2}{2(1-R)} \Leftrightarrow \frac{(1-2R)^2}{4R(1-R)} < s < \frac{\bar{\rho}^2}{1-\bar{\rho}^2}$. Let $\bar{s} = \lfloor \frac{(1-2R)^2}{4R(1-R)} \rfloor$ and $s^* = \lfloor \frac{\bar{\rho}^2}{1-\bar{\rho}^2} \rfloor$. Therefore

$$\max_{1-2R \le x \le \frac{1-2R+\bar{\rho}^2}{2(1-R)}} F(x) = \begin{cases} 4R^2 \left(1-2R\right)^{2s} & \text{if } s \le \bar{s} \\ \frac{R}{1-R} \frac{s^s}{(s+1)^{s+1}} & \text{if } \bar{s}+1 \le s \le s^* \\ \frac{R}{1-R} \bar{\rho}^{2s} \left(1-\bar{\rho}^2\right) & \text{if } s \ge s^*+1 \end{cases}$$

Consider now $|f(z)|^2$ on $\tilde{\theta}$. By simple algebraic manipulations one can see that

$$|f(\theta(t))|^{2} = \bar{\rho}^{2s} \left(1 + \bar{\rho}^{2} - 2\bar{\rho}\cos t\right)$$

Note that $|f(\theta(t))|^2$ is monotone increasing for $t \in [0, \pi]$ and hence it reaches its maximum on $\tilde{\theta}$ when $\cos t = \frac{1-2R+\bar{\rho}^2}{2(1-R)}$ Therefore we can conclude that

$$\max_{k=1}^{N-1} |\lambda_k^s (1-\lambda_k)| \leq \begin{cases} 2R |1-2R|^s & \text{if } s \leq \bar{s} \\ \sqrt{\frac{R}{1-R} \frac{s^s}{(s+1)^{s+1}}} & \text{if } \bar{s}+1 \leq s \leq s^* \\ \sqrt{\frac{R}{1-R} \bar{\rho}^{2s} (1-\bar{\rho}^2)} & \text{if } s \geq s^*+1. \end{cases}$$

Hence we can write

$$\begin{split} \sum_{s=0}^{t-1} \rho(P^s(I-P)) &\leq \sum_{s=0}^{\bar{s}} 2R \left| 1 - 2R \right|^s + \sum_{s=\bar{s}+1}^{s^*} \sqrt{\frac{R}{1-R} \frac{s^s}{(s+1)^{s+1}}} + \\ &+ \sum_{s=s^*+1}^{t-1} \sqrt{\frac{R}{1-R} \bar{\rho}^{2s} \left(1 - \bar{\rho}^2\right)}. \end{split}$$

Notice now that

$$\sum_{s=0}^{s} 2R |1 - 2R|^s \le \frac{2R}{1 - |1 - 2R|} \le \frac{1}{1 - R}.$$

and that

$$\begin{split} \sum_{s=s^*+1}^{t-1} \sqrt{\frac{R}{1-R}} \bar{\rho}^{2s} \left(1-\bar{\rho}^2\right) &\leq \sqrt{\frac{R}{1-R}} \sqrt{1-\bar{\rho}^2} \sum_{s=0}^{\infty} \bar{\rho}^s \\ &= \sqrt{\frac{R(1-\bar{\rho}^2)}{(1-R)(1-\bar{\rho})^2}} \\ &\leq \sqrt{\frac{2R}{(1-R)(1-\bar{\rho})}}. \end{split}$$

Notice finally that, since $\sum_{i=1}^{m} \sqrt{\frac{1}{i+1}} \leq 2\sqrt{m+1}$, we can argue that

$$\sum_{s=\bar{s}+1}^{s^*} \sqrt{\frac{R}{1-R}} \frac{s^s}{(s+1)^{s+1}} \le \sqrt{\frac{R}{1-R}} \sum_{s=1}^{s^*} \sqrt{\frac{1}{2}} \sqrt{\frac{1}{1+s}}$$
$$\le \sqrt{\frac{4R}{2(1-R)(1-\bar{\rho}^2)}}$$
$$\le \sqrt{\frac{2R}{(1-R)(1-\bar{\rho})}}.$$

Putting together these three inequalities we obtain (5.40).

Example 5.21 (Direct circuit) Consider the direct circuit graph introduced in Example 5.8 and consider the evolution law given by (5.22). Note that in this case P is the circulant matrix

$$P = \begin{pmatrix} \frac{1}{2} & \frac{1}{2} & 0 & 0 & \cdots & 0 & 0 & 0 \\ 0 & \frac{1}{2} & \frac{1}{2} & 0 & \cdots & 0 & 0 & 0 \\ 0 & 0 & \frac{1}{2} & \frac{1}{2} & \cdots & 0 & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \cdots & \vdots & \vdots & \vdots \\ \frac{1}{2} & 0 & 0 & 0 & \cdots & 0 & 0 & \frac{1}{2} \end{pmatrix}$$

We have theoretically proved (see Corollary 5.12) that $d_{\infty} \leq \frac{1}{2}$. Instead we could not evaluate d_{∞}^w in this case. Consider the bound introduced in Proposition 5.16. Since $\rho_{ess} = 1 - \frac{\pi^2}{2} \frac{1}{N^2} + o\left(\frac{1}{N^2}\right)$ (see [35]) and since each circulant matrix is a normal matrix we have that $\frac{1}{1-||PY||} = \frac{1}{1-\rho_{ess}(P)} = \Theta(N^2)$. Observe now that all the eigenvalues of P are inside the ball $B_{\frac{1}{2},\frac{1}{2}}$. Hence we obtain that $\frac{1}{1-R} + \sqrt{\frac{8R}{(1-R)(1-\bar{\rho})}} = \Theta(N)$. This means that the bound (5.40) improves the bound proposed in (5.38). Moreover, by numerical experiments one can see that $\frac{1}{2}\sum_{s=0}^{\infty} \rho(P^s(I-P)) = \Theta(N)$ meaning, that for $N \to \infty$, (5.40) behaves as (5.39).

If P is symmetric we can provide a stronger result.

Proposition 5.22 Let P be a symmetric stochastic matrix satisfying Assumption 5.1. Let R be such that 0 < R < 1 and $\sigma(P) \subseteq B_{1-R,R}$ and let $\bar{\rho} = \rho_{ess}(P)$ denote the essential spectral radius of P. Then,

$$\sum_{s=0}^{+\infty} \rho\left(P^s(I-P)\right) \le \frac{3}{2} + \frac{1}{1-R} + \frac{1}{2}\log\left(\frac{1}{1-\bar{\rho}}\right).$$
(5.41)

Proof: Assume that $\sigma(P) = \{\lambda_0 = 1, \lambda_1, \dots, \lambda_{N-1}\}$. Note that $\sigma(P) \setminus \{1\} \subseteq [1-2R, \bar{\rho}]$. We want to upper bound $\rho(P^s(I-P)) = \max_{k=1}^{N-1} |\lambda_k^s(1-\lambda_k)|$. To do this, consider the function $f(x) = |x^s(1-x)|$. It is continuous in [-1, 1], positive and decreasing in [-1, 0], it vanishes in x = 0 and in x = 1 and has a local maximum in $x = x_M = \frac{s}{1+s}$, with $f(x_M) = (\frac{s}{1+s})^s(\frac{1}{1+s})$. We need to evaluate $\max_{x \in [1-2R, \bar{\rho}]} f(x)$.

First observe that there exists \bar{s} , only depending on the value 1 - 2R, such that for all $s > \bar{s}$ we have $f(1 - 2R) < (\frac{s}{1+s})^s(\frac{1}{1+s})$ and then the global

maximum of f(x) is assumed at $x = x_M$ if $x_M \leq \bar{\rho}$. Since x_M tends to 1 as s goes to infinity, it happens that, when $s > \frac{\bar{\rho}}{1-\bar{\rho}}$, $x_M(s) > \bar{\rho}$. In conclusion we have that

$$\max_{x \in [1-2R,\bar{\rho}]} f(x) = \begin{cases} 2R|1-2R|^s, \text{ if } 0 \le s \le \bar{s};\\ (\frac{s}{1+s})^s(\frac{1}{1+s}), & \text{ if } \bar{s} < s \le s^*;\\ \bar{\rho}^s(1-\bar{\rho}), & \text{ if } s^* < s < \infty \end{cases}$$

where $s^* := \lfloor \frac{\bar{\rho}}{1-\bar{\rho}} \rfloor$. Hence we can write

$$\begin{split} \sum_{s=0}^{t-1} \rho(P^s(I-P)) &\leq \sum_{s=0}^{\bar{s}} 2R |1-2R|^s + \sum_{s=\bar{s}+1}^{s^*} \left(\frac{s}{1+s}\right)^s \left(\frac{1}{1+s}\right) + \\ &+ \sum_{s=s^*+1}^{t-1} \bar{\rho}^s (1-\bar{\rho}). \end{split}$$

Notice now that

$$\sum_{s=0}^{\bar{s}} 2R|1 - 2R|^s \le \frac{1}{1 - R},$$

and that

$$\sum_{s=s^*+1}^{t-1} \bar{\rho}^s (1-\bar{\rho}) = \bar{\rho}^{s^*+1} - \bar{\rho}^t \le 1.$$

Notice finally that, since $\sum_{i=1}^{m} 1/i \le 1 + \ln m$ we can argue that

$$\sum_{s=\bar{s}+1}^{s^*} \left(\frac{s}{1+s}\right)^s \left(\frac{1}{1+s}\right) \le \sum_{s=0}^{s^*} \left(\frac{1}{2}\right) \left(\frac{1}{1+s}\right)$$
$$\le \frac{1}{2}(1+\log(s^*+1))$$
$$\le \frac{1}{2} + \frac{1}{2}\log\left(\frac{\bar{\rho}}{1-\bar{\rho}} + 1\right)$$
$$= \frac{1}{2} + \frac{1}{2}\log\left(\frac{1}{1-\bar{\rho}}\right).$$

Putting together these three inequalities we obtain (5.41).

107

Example 5.23 (Undirect circuit) In this example we consider the circulant matrix

$$P = \begin{pmatrix} \frac{1}{3} & \frac{1}{3} & 0 & 0 & \cdots & 0 & 0 & \frac{1}{3} \\ \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & 0 & \cdots & 0 & 0 & 0 \\ 0 & \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & \cdots & 0 & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \cdots & \vdots & \vdots & \vdots \\ \frac{1}{3} & 0 & 0 & 0 & \cdots & 0 & \frac{1}{3} & \frac{1}{3} \end{pmatrix}$$

By simulations one can see that d_{∞} seems to be uniformly bounded on N. Also in this case we are not able to evaluate d_{∞}^w . By considering the bound (5.38), since $\rho_{ess}(P) \cong 1 - \frac{4}{3} \frac{\pi^2}{N^2}$ (see [35]) we obtain that $\frac{1}{1-\|PY\|} = \frac{1}{1-\rho_{ess}} = \Theta(N^2)$. Observe that all the eigenvalues of P are greater than $-\frac{1}{3}$. Hence it results, letting R = 2/3, that $\frac{3}{2} + \frac{1}{1-R} + \frac{1}{2} \log \left(\frac{1}{1-\bar{\rho}}\right) = \Theta(\log N)$. Moreover, numerically it is possible to observe that also $\frac{1}{2} \sum_{s=0}^{\infty} \rho(P^s(I-P))$ grows logarithmically, meaning that asymptotically in N, we have that (5.41) behaves as (5.39).

Remark 5.24 It is worth noting that this improvement is more general. Consider a sequence of symmetric Cayley matrices P, having the elements on the diagonal uniformly lower bounded on N and supported on Cayley graphs having in-degree uniformly upper bounded on N. Then the above arguments can be applied to argue that, for this class of graphs, we can not obtain from (5.39) a bound for $d^w(P)$ stronger than a logarithmic dependence on N. In the next example we will show that a logarithmic bound can be proved to be tight, that is $d^w(P) = \Theta(\log N)$.

Example 5.25 (Hypercube) Consider the hypercube graph and the matrix P compatible with the hypercube graph as defined in Example 5.13. We have already seen, that, by simulations, $d_{\infty}(P)$ seems to be a uniformly bounded quantity on N, while we have analytically proved that $d_{\infty}^w = \frac{\log_2 N}{2}$ (see Example 5.13). It is possible to see that also $\frac{1}{2} \sum_{s=0}^{\infty} \rho(P^s(I-P)) = \frac{\log_2 N}{2}$. This fact is stated in Lemma 5.15. Moreover, since the eigenvalues of P are $1 - \frac{2k}{n+1}$ for $k = 0, \ldots, n$ we have that all the eigenvalues of P different from 1 are inside the interval $\left[\frac{1-n}{n+1}, \frac{n-1}{n+1}\right]$. This implies that $\frac{3}{2} + \frac{1}{1-R} + \frac{1}{2} \log\left(\frac{1}{1-\bar{\rho}}\right) = \Theta(\log_2(N))$.
5.3.2 Probabilistic model

In the previous section we have shown that the bounded error model does not seem to really capture the behavior of the quantized model. In particular, the upper bound to the performance we have found, seems to be quite conservative. In this section we undertake a probabilistic approach, modeling the quantization error as a random variable. We will carry on a classical mean square analysis and we will show that it gets quite close to simulations of the real quantized model. This suggests that the probabilistic approach is more appropriate to describe quantization errors even though we have no theoretical evidence of this fact. Therefore we have to point out that the probabilistic model, differently from the worst case analysis, does not provide any bound to the performance.

For $i \in V$ and $t \in \mathbb{N}$, let $n_i(t)$ be random variables of zero mean and variance σ^2 , which have their supports inside [-1/2, 1/2] and are uncorrelated and identically distributed in both i and t, i.e., $\mathbb{E}[n_i(t)n_j(\tau)] = 0$ if $i \neq j$ or $t \neq \tau$. Define n(t) as the random vector whose components are $n_i(t)$ and consider the stochastic model

$$\begin{cases} x_r(t+1) = Px_r(t) + (I-P)n(t), & x_r(0) = x(0) \\ \Delta_r(t) = Yx_r(t) \end{cases}$$
(5.42)

where $Y = I - \frac{1}{N} \mathbb{1}\mathbb{1}^*$. We define

$$d_{\infty}^{r}(P, x_{r}(0)) = \limsup_{t \to \infty} \sqrt{\frac{1}{N} \mathbb{E}[\|\Delta_{r}(t)\|^{2}]}$$

Since $\lim_{t\to\infty} YP^t = 0$ also in this case, we have that $d_{\infty}^r(P, x_r(0))$ is independent of the initial condition $x_r(0)$. Hence, in the sequel we will denote $d_{\infty}^r(P, x_r(0))$ with the symbol $d_{\infty}^r(P)$. It is worth to point out that this index, as the previously defined $d^w(P)$, captures the asymptotic error induced by quantization, but no information about diverse issues like speed of convergence or finite time behavior.

We start our analysis of the probabilistic model with the following result.

Theorem 5.26 Let P be a matrix satisfying Assumption 5.1. Then

$$[d_{\infty}^{r}(P)]^{2} = \frac{\sigma^{2}}{N} \operatorname{tr} \left[(I - P)(I - \tilde{P}\tilde{P}^{*})^{-1}(I - P)^{*} \right].$$
 (5.43)

where $\tilde{P} = PY$. In particular, if P is normal, and $\sigma(P) = \{1, \lambda_1, \ldots, \lambda_{N-1}\}$ denotes the spectrum of P, we have that

$$[d_{\infty}^{r}(P)]^{2} = \frac{\sigma^{2}}{N} \sum_{i=1}^{N-1} \frac{|1-\lambda_{i}|^{2}}{1-|\lambda_{i}|^{2}}.$$
(5.44)

Proof: Define $Q(t) = \mathbb{E}[\Delta_r(t)\Delta_r(t)^*]$, and remark that $\frac{1}{N}\mathbb{E}[||\Delta_r(t)||^2] = \frac{1}{N}\operatorname{tr} Q(t)$. Using the facts that $Y^k = Y$ for all positive integer k and Y(P - I) = P - I, it is easy to see that Δ_r satisfies the following recursive equation

$$\Delta_r(t+1) = \dot{P}\Delta_r(t) + (P-I)n(t).$$

Now, thanks to the hypotheses on $n_i(t)$,

$$Q(t+1) = \mathbb{E}[\Delta_r(t+1)\Delta_r(t+1)^*] = \mathbb{E}[\tilde{P}\Delta_r(t)\Delta_r(t)^*\tilde{P}^*] + (I-P)\mathbb{E}[n(t)n(t)^*](I-P)^* = \tilde{P}Q(t)\tilde{P}^* + \sigma^2(I-P)(I-P)^*,$$

and by a simple recursion

$$Q(t) = \tilde{P}^t Q(0) \, (\tilde{P}^*)^t + \sigma^2 \sum_{s=0}^{t-1} \tilde{P}^s (I-P) (I-P)^* (\tilde{P}^*)^s.$$

Recall now from the proof of Proposition 5.16 that, since P satisfies Assumption 1, then $\rho_{ess}(P^*P) < 1$. Moreover we have that $\rho(\tilde{P}) = \rho_{ess}(P) < 1$ and $\rho(\tilde{P}^*\tilde{P}) = \rho_{ess}(P^*P) < 1$. Using the linearity and properties of the trace,

$$\operatorname{tr} Q(t) = \operatorname{tr} \left[\tilde{P}^{t} Q(0) (\tilde{P}^{*})^{t} \right] + \operatorname{tr} \left[\sigma^{2} \sum_{s=0}^{t-1} (PP^{*} - (P + P^{*}) + I) (\tilde{P}\tilde{P}^{*})^{s} \right]$$
$$= \operatorname{tr} \left[\tilde{P}^{t} Q(0) (\tilde{P}^{*})^{t} \right] + \sigma^{2} \operatorname{tr} \left[(PP^{*} - (P + P^{*}) + I) (I - (\tilde{P}\tilde{P}^{*})^{t}) (I - \tilde{P}\tilde{P}^{*})^{-1} \right]$$

and hence

$$\lim_{t \to \infty} \operatorname{tr} Q(t) = \sigma^2 \operatorname{tr} \left[(PP^* - (P + P^*) + I)(I - \tilde{P}\tilde{P}^*)^{-1} \right]$$

If moreover P is normal, we can find a unitary matrix O of eigenvectors and a diagonal matrix of eigenvalues Λ , such that $P = O\Lambda O^*$. This implies

tr
$$\left[(PP^* - (P + P^*) + I)(1 - \tilde{P}\tilde{P}^*)^{-1} \right] = \sum_{i=1}^{N-1} \frac{|1 - \lambda_i|^2}{1 - |\lambda_i|^2}.$$

From now on we restrict to the case in which P is normal. Note that the expression for the mean square error of formula (5.44) is the product of two terms, $[d_{\infty}^{r}(P)]^{2} = \sigma^{2} \Phi(P)$ where

$$\Phi(P) := \frac{1}{N} \sum_{i=1}^{N-1} \frac{|1 - \lambda_i|^2}{1 - |\lambda_i|^2},$$

is a functional³ of the matrix P, depending only on its spectral structure.

As in the previous section, we are mainly interested in sequences of matrices of increasing size. We will see that the above functional scales well with N in the following examples.

Example 5.27 (Complete graph) In this case it is easy to compute $\Phi(P)$. We have that $\Phi(P) = \frac{N-1}{N}$.

Example 5.28 (Direct graph) Consider the matrix P defines in Example 5.21. In this case $\Phi(P)$ can be exactly computed. We have

$$\Phi(P_N) = \frac{1}{N} \sum_{h=1}^{N-1} \frac{(1-\lambda_h)(1-\lambda_h^*)}{1-\lambda_h \lambda_h^*}$$

= $\frac{1}{N} \sum_{h=1}^{N-1} \frac{(1-(1/2+1/2e^{i\frac{2\pi}{N}h}))(1-(1/2+1/2e^{-i\frac{2\pi}{N}h}))}{1-(1/2+1/2e^{i\frac{2\pi}{N}h})(1/2+1/2e^{-i\frac{2\pi}{N}h})}$
= $\frac{1}{N} \sum_{h=1}^{N-1} \frac{1/2(1-\cos(\frac{2\pi}{N}h))}{1/2(1-\cos(\frac{2\pi}{N}h))} = \frac{N-1}{N}.$

Example 5.29 (Undirected graph) Consider the undirected undirected circuit graph and the matrix P introduced in Example 5.23. The eigenvalues of P are

$$\lambda_h = \frac{1}{3} + \frac{2}{3}\cos\left(\frac{2\pi}{N}h\right) \quad h = 0, \dots, N-1,$$

and we have

$$\Phi(P_N) = \frac{1}{N} \sum_{h=1}^{N-1} \frac{1-\lambda_h}{1+\lambda_h} = \frac{1}{N} \sum_{h=1}^{N-1} \frac{(1-\cos\left(\frac{2\pi}{N}h\right))}{2+\cos\left(\frac{2\pi}{N}h\right)}.$$

³Remarkably, the functional $\Phi(P)$ also arises, with a rather different meaning, in [47], as a cost functional describing the transient of the diffusion methods for average consensus over graphs with ideal communication.

In this case it is difficult to work out the computation explicitly. However, it is possible to compute the limit for $N \to \infty$, since the summation can be interpreted as Riemann sum relative to the function $f(x) = \frac{1 - \cos(x)}{2 + \cos(x)}$. We thus obtain

$$\lim_{N \to \infty} \Phi(P_N) = \frac{1}{2\pi} \int_0^{2\pi} \frac{1 - \cos(x)}{2 + \cos(x)} dx = \sqrt{3} - 1.$$
 (5.45)

Example 5.30 (Hypercube) Consider the hypercube graph and the matrix P defined in (5.34). We have that

$$\Phi(P) = \frac{1}{N} \sum_{i=1}^{N-1} \frac{|1-\lambda_i|^2}{1-|\lambda_i|^2}$$
$$= \frac{1}{2^n} \sum_{k=1}^n \frac{\left(\frac{2k}{n+1}\right)^2}{1-\left(\frac{n+1-2k}{n+1}\right)^2} \binom{n}{k}$$
$$= \frac{1}{2^n} \sum_{k=1}^n \frac{k}{n+1-k} \binom{n}{k}$$
$$= \frac{1}{2^n} \sum_{k=1}^n \binom{n}{k-1}$$
$$= \frac{N-1}{N}.$$

Then

$$\Phi(P) = \frac{N-1}{N}.$$

While in the previous section the hypercube provided the negative example for the worst case behavior, the probabilistic analysis is in agreement with the evidence showed in the simulations. This highlights the differences between the bounded error model and the probabilistic model: with the same assumptions on P the two worst-case analysis and the mean-square analysis give different results.

Example 5.31 (Random geometric graph) For the random geometric graph, as defined in the introduction, we have no explicit formula for the eigenvalues. However, d_{∞}^r can be numerically evaluated, and compared with d_{∞} . The results are shown in Figure 5.5 where d_{∞}^r seems to describe well the behavior of d_{∞} .



Figure 5.5: Comparison of d_{∞}^r and d_{∞} for random geometric graphs, averaged over 25 realizations both of the initial condition and of the graphs, for each N. The plot assumes $\sigma^2 = 1/12$.

In all the above examples we have that $\Phi(P)$ scales well with N. We provide now some general conditions ensuring that the functional P for a given sequence of matrices of increasing size is uniformly bounded on N. We start by observing that, if $\rho_{ess}(P) \leq B < 1$ then $\Phi(P) \leq \frac{N-1}{N} \frac{4}{1-B^2}$. This implies that, given a sequence of matrices of increasing size, if the essential spectral radius of the sequence is uniformly bounded away from 1 then the functional cost Φ is uniformly bounded in N. This fact is true also for d_{∞}^w as can be easily seen by recalling the expression of the bound provided by Proposition 5.16. The interesting fact is that the performance index d_{∞}^r can exhibit the same behavior even when the essential spectral radius is not bounded away from 1 as the following proposition shows.

Proposition 5.32 Let $B_{c,r} \subset \mathbb{C}$ denote the closed ball of complex numbers with center the point c and radius r. If there exists 0 < R < 1 such that $\sigma(P) \subseteq B_{1-R,R}$ then

$$\Phi(P) \le \frac{R}{1-R}.\tag{5.46}$$

Proof: The inequality 0 < R < 1 is clear from Assumption 5.1. It means that the spectrum is contained in a disc of radius R internally tangent in 1

to the unit disc of the complex plane. Then, we need to prove (5.46). For all i, the eigenvalue $\lambda_i \in B_{1-R,R}$, so

$$\lambda_i = (1 - r) + r \, e^{i\theta}$$

with $\theta \in [0, 2\pi[$ and $0 \le r \le R$. Moreover, if $i \ge 1$, then $\theta > 0$. Hence,

$$\frac{|1 - \lambda_i|^2}{1 - |\lambda_i|^2} = \frac{|r - re^{i\theta}|^2}{1 - |1 - r + re^{i\theta}|^2}$$
$$= \frac{r^2 |1 - e^{i\theta}|^2}{1 - (1 - r)^2 - 2r(1 - r)\cos\theta - r^2}$$
$$= \frac{r^2 2(1 - \cos\theta)}{2r(1 - r)(1 - \cos\theta)}$$
$$= \frac{r}{1 - r}$$
$$\leq \frac{R}{1 - R} \quad \forall i.$$

This yields the result.

Note that the above bound depends only on R and does not depend on the essential spectral radius of the matrix P, while the worst case bounds provided in Proposition 5.40 and in Proposition 5.22 did. This is why (5.46) is bounded in all the cases we considered.

Many of them are covered by the following corollary.

Corollary 5.33 Let $p = \min_i P_{ii}$ and R as above. Then, $R \leq 1 - p$, and

$$\Phi(P) \le \frac{1-p}{p}.\tag{5.47}$$

Proof: By Gershgorin theorem, $\sigma(P) \subseteq \bigcup_i B_{P_{ii},1-P_{ii}} \subseteq B_{p,1-p}$.

The above result has the following interpretation. If in a family of matrices P_N we have that $\min_i (P_N)_{ii}$ is lower bounded uniformly in N, then (5.47) gives a finite bound, uniform in N, on the asymptotic displacement. This is a useful hint to construct sequences of matrices whose performance scales well with N. It would be enough to prescribe that the agents assign a minimum weight to their own values.

5.4 Time Invariant case- Probabilistic quantizers

Consider (5.12), i.e., the updating rule using the probabilistic quantizer. In this section we will show that a probabilistic model, similar to the one used in Section 5.3.2 for the deterministic quantizer-model, is a theoretically based way to analyze (5.12).

We start our analysis from the following assumption on the initial condition x(0).

Assumption 5.34 The initial condition x(0) is a random variable such that $\mathbb{E}[x(0)] = 0$ and $\mathbb{E}[x(0)x^*(0)] = \sigma_0^2 I$ for some $\sigma_0^2 > 0$.

Observe now that (5.12) can be rewritten in the following way

$$x(t+1) = Px(t) + (P - I) (q_p(x(t)) - x(t)).$$
(5.48)

We introduce the variables

$$y(t) := \left(I - \frac{1}{N} \mathbb{1}\mathbb{1}^*\right) x(t) = x(t) - \left(\frac{1}{N} \mathbb{1}^* x(t)\right) \mathbb{1} = x(t) - \left(\frac{1}{N} \mathbb{1}^* x(0)\right) \mathbb{1}$$

and

$$w(t) = q(x(t)) - x(t).$$

Note that the variable $y(t) = [y_1(t), \ldots, y_N(t)]^*$ represents the distance from the initial average and the variable $w(t) = [w_1(t), \ldots, w_N(t)]^*$ represents the quantization error. Moreover note that $\mathbb{1}^* y(t) = 0$ for all $t \ge 0$. From (5.48) have that

$$\left(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*\right)x(t+1) = \\ = \left(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*\right)Px(t) + \left(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*\right)(P - I)\left(q_p(x(t)) - x(t)\right).$$

Since $\left(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*\right)P = P\left(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*\right)$ and $\left(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*\right)(P - I) = P - I$ we obtain the following recursive equation

$$y(t+1) = Py(t) + (P-I)w(t).$$
(5.49)

In order to perform an asymptotic analysis of (5.49) we introduce the following matrices. Let

$$\Sigma_{yy}(t) := \mathbb{E}\left[y(t)y^*(t)\right],$$

$$\Sigma_{ww}(t) := \mathbb{E}\left[w(t)w^*(t)\right],$$

and

$$\Sigma_{yw}(t) := \mathbb{E}\left[y(t)w^*(t)\right].$$

Equation (5.49) leads to the following recursive equation in terms of the above matrices

$$\Sigma_{yy}(t+1) = P\Sigma_{yy}(t)P^* + P\Sigma_{yw}(t)(P-I)^* + (P-I)\Sigma_{yw}^*P^* + (P-I)\Sigma_{ww}(t)(P-I)^*.$$
(5.50)

Notice that, since x(0) is a random variable satisfying Assumption 5.34, one can show that

$$\Sigma_{yy}(0) = \sigma_0^2 \left(I - 1/N \, \mathbb{1} \, \mathbb{1}^* \right). \tag{5.51}$$

The following proposition states some remarkable properties of the variables y and w.

Proposition 5.35 Consider the variables y(t) and w(t) above defined. Then

$$\mathbb{E}[w(t)] = 0 \quad and \quad \mathbb{E}[w(t)w^*(t)] = \text{diag}\left\{\sigma_1^2(t), \dots, \sigma_N^2(t)\right\} \quad (5.52)$$

where $\sigma_i^2(t) := \mathbb{E}[w_i^2(t)]$ is such that $\sigma_i^2(t) \le 1/4$ for all $1 \le i \le N$ and for all $t \ge 0$. Moreover

$$\Sigma_{yw}(t) = 0, \tag{5.53}$$

for all $t \geq 0$.

Proof: Observe that

$$\mathbb{E}[w_i(t)] = \mathbb{E}\left[\mathbb{E}\left[q_p(x_i(t)) - x_i(t)|x_i(t)\right]\right]$$

= $\mathbb{E}\left[\mathbb{E}\left[q_p(x_i(t))|x_i(t)\right] - x_i(t)\right]$
= $\mathbb{E}\left[x_i(t) - x_i(t)\right]$
= 0, (5.54)

and that, for $i \neq j$,

$$\mathbb{E} [w_i(t)w_j(t)] = \mathbb{E} [w_i(t)(q(x_j(t)) - x_j(t))] = \mathbb{E} [w_i(t)\mathbb{E} [q(x_j(t)) - x_j(t)|x_i(t), x_j(t)]] = \mathbb{E} [w_i(t)\mathbb{E} [q(x_j(t)) - x_j(t)|x_j(t)]] = \mathbb{E} [w_i(t) (\mathbb{E} [q(x_j(t))|x_j(t)] - x_j(t))] = \mathbb{E} [w_i(t) (x_j(t) - x_j(t))] = 0$$
(5.55)

where, both in (5.54) and in (5.55), we have used the fact that by Lemma 5.3, $\mathbb{E}[q(x_j(t))|x_j(t)] = x_j(t)$. If i = j we have that

$$\mathbb{E}\left[w_i^2(t)\right] = \mathbb{E}\left[\left(q(x_i(t)) - x_i(t)\right)^2\right]$$
$$= \mathbb{E}\left[\mathbb{E}\left[\left(q(x_i(t)) - x_i(t)\right)^2 | x_i(t)\right]\right]$$
$$\leq \mathbb{E}\left[\frac{1}{4}\right]$$
$$= \frac{1}{4}$$
(5.56)

where again by Lemma 5.3, we used the fact $\mathbb{E}\left[\left(q(x_i(t)) - x_i(t)\right)^2 | x_i(t)\right] \le 1/4$. (5.54), (5.55) and (5.56) establish the validity of (5.52). Observe now that, given any pair of indexes i, j it holds also that

$$\mathbb{E} [x_i(t)w_j(t)] = \mathbb{E} [x_i(t)(q(x_j(t)) - x_j(t))] = \mathbb{E} [x_i(t)\mathbb{E} [q(x_j(t)) - x_j(t)|x_i(t), x_j(t)]] = \mathbb{E} [x_i(t)\mathbb{E} [q(x_j(t)) - x_j(t)|x_j(t)]] = \mathbb{E} [x_i(t) (\mathbb{E} [q(x_j(t))|x_j(t)] - x_j(t))] = \mathbb{E} [x_i(t) (x_j(t) - x_j(t))] = 0$$

It is easy to see that the fact that $\mathbb{E}[x_i(t)w_j(t)] = 0$ implies also that $\mathbb{E}[y_i(t)w_j(t)] = 0$ and hence that $\Sigma_{yw}(t) = 0$ for all $t \ge 0$.

From (5.53), it follows that (5.50) can be rewritten as

$$\Sigma_{yy}(t+1) = P\Sigma_{yy}(t)P^* + (P-I)\Sigma_{ww}(t)(P-I)^*.$$
 (5.57)

In order to evaluate the asymptotic distance from the initial average, a suitable functional cost is

$$d^{p}_{\infty}(P) := \limsup_{t \to \infty} \sqrt{\frac{1}{N} \operatorname{tr} \{\Sigma_{yy}(t)\}}.$$
(5.58)

Note that $d^p_{\infty}(P)$ has a similar meaning to $d^r_{\infty}(P)$ defined in the previous Section, when we introduced the probabilistic model to study the evolution of (5.11). By straightforward calculations one can show that,

$$d_{\infty}^{p}(P) = \lim_{t \to \infty} \left(P^{t} \Sigma_{yy}(0) \left(P^{*}\right)^{t} + \sum_{i=0}^{t-1} P^{t-1-i} \Sigma_{ww}(i) \left(P^{*}\right)^{t-1-i} \right)$$
$$= \lim_{t \to \infty} \left(\sum_{i=0}^{t-1} P^{t-1-i} \Sigma_{ww}(i) \left(P^{*}\right)^{t-1-i} \right), \tag{5.59}$$

where the last equality derives from the fact that

$$\lim_{t \to \infty} \left(P^t \Sigma_{yy}(0) \left(P^* \right)^t \right) = \frac{1}{N} \mathbb{1} \mathbb{1}^* \Sigma_{yy}(0) \mathbb{1} \mathbb{1}^*$$
$$= \frac{1}{N} \mathbb{1} \mathbb{1}^* \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right) \mathbb{1} \mathbb{1}^*$$
$$= 0.$$

In general, inferring some theoretical analysis on (5.59) is quite hard. We try now to simplify the problem. Let us introduce the matrix

$$\bar{W} = \frac{1}{4}I,$$

and the auxiliary system

$$\bar{\Sigma}_{yy}(t+1) = P\bar{\Sigma}_{yy}(t)P^* + (P-I)\bar{W}(P-I)^*$$
$$= P\bar{\Sigma}_{yy}(t)P^* + \frac{1}{4}(P-I)(P-I)^*.$$
(5.60)

where $\bar{\Sigma}_{yy}(0) = \Sigma_{yy}(0)$. Moreover let

$$\bar{d}^p_{\infty}(P) := \limsup_{t \to \infty} \sqrt{\frac{1}{N} \operatorname{tr} \left\{ \bar{\Sigma}_{yy}(t) \right\}}.$$

We have the following proposition.

Proposition 5.36 Let $d^p_{\infty}(P)$ and $\bar{d}^p_{\infty}(P)$ be as previously defined. Then

$$d^p_{\infty}(P) \le \bar{d}^p_{\infty}(P).$$

Proof: In order to prove the statement of the Theorem we prove that

$$\bar{\Sigma}(t) \ge \Sigma(t) \tag{5.61}$$

for all $t \ge 0$, where the inequality is meant in the matricial sense, namely $\bar{\Sigma}(t) - \Sigma(t) \ge 0$. We prove (5.61) by induction on t. Let t = 0, then (5.61) is trivially true since $\bar{\Sigma}_{yy}(0) = \Sigma_{yy}(0)$. Assume now that (5.61) is verified for a generic t and consider t + 1. We have that

$$\bar{\Sigma}(t+1) - \Sigma(t+1) =
= P\bar{\Sigma}(t)P^* + (P-I)\bar{W}(P-I)^* - (P\Sigma(t)P^* + (P-I)W(t)(P-I)^*)
= P\left(\bar{\Sigma}(t) - \Sigma(t)\right)P^* + (P-I)\left(\bar{W} - W(t)\right)(P-I)^*$$
(5.62)

Since, by inductive hypothesis we have that $\overline{\Sigma}(t) - \Sigma(t) \ge 0$ and since $\overline{W} - W(t) \ge 0$ for all t, it follows from (5.62) that also $\overline{\Sigma}(t+1) - \Sigma(t+1) \ge 0$ holds true.

Now note that, from (5.60), by simple algebraic tools one can show that

$$\left[\bar{d}_{\infty}^{p}(P)\right]^{2} = \frac{1}{4N} \operatorname{tr} \left\{ \lim_{t \to \infty} \sum_{i=0}^{t-1} P^{i}(P-I)(P-I)^{*}(P^{*})^{i} \right\}$$
$$= \frac{1}{4N} \operatorname{tr} \left\{ (P-I)(I-\tilde{P}\tilde{P})^{(-1)}(P-I)^{*} \right\}$$

where $\tilde{P} = PY$ with $Y = I - \frac{1}{N}\mathbb{11}^*$. Note that, if in (5.43) $\sigma^2 = \frac{1}{4}$ then $\bar{d}^p_{\infty}(P) = d^r_{\infty}(P)$. Moreover, if P is normal, also \bar{d}^p_{∞} can be expressed by (5.44). These observations imply that all the considerations and the results stated in Example 5.28, in Example 5.29, in Example 5.30, in Example 5.31, in Lemma 5.32 and in Corollary 5.33 hold true also for the Equation (5.60). In particular this shows that \bar{d}^p_{∞} is uniformly bounded on N for many interesting families of graphs of increasing size and that asymptotically reaches a value less than $\frac{1}{4}$.

Remark 5.37 As pointed out in Remark 5.6, the authors in [8] proposed the following updating scheme

$$x(t+1) = Pq_p(x(t)).$$
(5.63)

They proved the following theorem

Theorem 5.38 Consider (5.63). Then, almost surely, it holds that

$$\lim_{t \to \infty} x(t) = c\mathbb{1}$$

for some integer c.

Note that the scheme (5.63) in general do not preserve the initial average. In order to quantify the displacement from the initial average, the authors in [9] introduced the following variable $\tilde{y}(t) := q_p(x(t)) - (1/N\mathbb{1}^*x(0))\mathbb{1}$ and the following functional cost

$$\tilde{J} := \limsup_{t \to \infty} \sqrt{\frac{1}{N}} \mathrm{tr} \, \left\{ \mathbb{E} \left[\tilde{y}(t) \tilde{y}(t)^* \right] \right\}.$$

Note that the meaning of the variables $\tilde{y}(t)$ and \tilde{J} is very close to the meaning of the variable y(t) and J, respectively. The following result, stated in [9], characterizes \tilde{J} .

Theorem 5.39 Let $\tilde{y}(t)$ and \tilde{J} as above defined. Then

$$\lim_{N \to \infty} \tilde{J} \le \frac{1}{2} \frac{1}{1 - \rho_{ess}(P)}.$$
(5.64)

Consider now the sequence of circulant matrices $P_N \in \mathbb{R}^{N \times N}$ defined by

$$P_N = \begin{pmatrix} \frac{1}{3} & \frac{1}{3} & 0 & 0 & \cdots & 0 & 0 & \frac{1}{3} \\ \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & 0 & \cdots & 0 & 0 & 0 \\ 0 & \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & \cdots & 0 & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots & \vdots & \vdots \\ \frac{1}{3} & 0 & 0 & 0 & \cdots & 0 & \frac{1}{3} & \frac{1}{3} \end{pmatrix}$$
(5.65)

For this sequence of symmetric stochastic matrices we have that (see Section 3.3.1)

$$\rho_{ess}(P_N) = \frac{1}{3} + \frac{2}{3}\cos\left(\frac{2\pi}{N}\right).$$

Hence, for $N \to \infty$ we have that

$$\rho_{ess}(P_N) = 1 - \frac{4}{3} \frac{\pi^2}{N^2} + o\left(\frac{1}{N^2}\right)$$

implying that

$$\lim_{N \to \infty} \frac{1}{2} \frac{1}{1 - \rho_{ess}(P_N)} = +\infty.$$

Hence, even if this strategy ensures, that, almost surely, the consensus is reached, the distance from the average of the initial conditions could increase drastically for those consensus matrices P, whose essential spectral radius tends to 1.

Remark 5.40 For the sake of the completeness we briefly consider again the laws

$$x(t+1) = Pq_d(x(t))$$
(5.66)

and

$$x(t+1) = \operatorname{diag} \{P\} x(t) + (P - \operatorname{diag} \{P\}) \hat{x}(t)$$
(5.67)

introduced in Remark 5.5. Besides the fact that these strategies do not preserve the average of the state, the authors in [33], supported by several simulations, stated that (5.66) and (5.67) exhibit also poor performance in terms of reaching a consensus. In particular for matrices possessing symmetries, as (5.65), they asymptotically lead the overall system to periodic orbits whose amplitude seems to be increasing as $N \to \infty$.

5.4.1 An alternative solution to the consensus problem

Consider (5.12), i.e.,

$$x(t+1) = x(t) + (P - I)q_p(x(t)),$$
(5.68)

and (5.63), i.e.,

$$x(t+1) = Pq_p(x(t)).$$
(5.69)

Since (5.69) uses only quantized information we refer to it as globally quantized, while since in (5.68) the systems use also perfect information related their own state we refer to it as *partially quantized*.

We have seen in Remark 5.37 that the distance from the average of the initial condition of the consensus point toward which the globally quantized strategy leads the systems could be not negligible. On the other hand the partially quantized strategy does not assure a consensus in a strict sense. If, depending on the application, one can not relax the convergence requirement we could suggest the following heuristic solution to the consensus problem, which combines the positive features of both strategies

$$x(t+1) = Pq_p(x(t)) + \epsilon(t)(x(t) - q_p(x(t))),$$

where $\epsilon(t)$, $t \ge 0$, is a nonnegative sequence such that $\epsilon(t) \le 1$, $\forall t \ge 0$ and $\lim_{t\to\infty} \epsilon(t) = 0$.

An example, could be the case in which, given $T \in \mathbb{N}$, $\epsilon(t) = 1$ for $t \leq T$ and $\epsilon(t) = 0$ for t > T. In other words, the overall system uses as updating rule (5.68), i.e., the *partially quantized* strategy, for $t \leq T$, and (5.69), i.e., the *globally quantized* strategy, for t > T. It is clear that this combined strategy reaches almost surely the consensus, and the deviation from the initial average will be smaller than using only (5.69).

5.5 Conclusions

In this chapter we studied the effects of a uniform quantization (both deterministic and probabilistic) on the average consensus problem, and, starting from the standard average consensus algorithm, we proposed a simple and effective adaptation which is able to preserve the average of states and to drive the system reasonably near to the consensus.

An exact analysis of this adaptation is, in general, very hard and we have been able to perform it only in one special case. The main features of this new law have been investigated by a worst case analysis and by a probabilistic analysis. A special attention has been given to the scalability in N of the performance, which is a crucial issue for applications in which the number of agents is huge. In this direction we obtained several favorable results, which we applied to sequences of graphs with symmetries.

A potential future development is be the extension to random geometric graphs of the analysis we performed on Cayley examples. Moreover an interesting question remain open: why and in which sense the probabilistic analysis seems to be closer to experimental results also when the information is quantized by means of deterministic quantizers.

Chapter 6

Quantized Consensus: Gossip Algorithms

6.1 Introduction

In the previous chapter we have studied the average consensus algorithm, when the consensus matrix P is time-invariant, by assuming that the communication network is constituted of only digital links and hence the information exchanged between the systems is quantized. Also in this chapter, we assume that the systems exchange information through digital channels but, differently from the previous chapter, we consider the case in which the consensus matrix is time-varying. Precisely, the main goal of this chapter is to analyze the effects of the quantization on the symmetric gossip algorithm described in Section 4.5.

Similarly to the previous chapter, we will introduce two particular strategies, the *partially quantized* strategy and the *globally quantized* strategy, depending on whether the systems use exact information regarding their own state, or not, to update their states. We will analyze these strategies both via the deterministic quantizer and via the probabilistic quantizer. We will show that the *globally quantized* strategy, via both the deterministic quantizer and the probabilistic quantizer, ensures that, almost surely, the consensus is reached. The drawback of this strategy is that it does not preserve the average of the initial conditions. On the other hand, the *partially quantized* strategy maintains the initial average at each iteration of the algorithm, but does not guarantee that the consensus is reached in general. However, we will see that the *partially quantized* drives asymptotically all the states very close to the initial average (we will quantify how close).

The chapter is organized as follows. In Section 6.2 we formulate the problem. In particular we introduce the *partially* quantized strategy and the *globally* quantized strategy. In section 6.3 and in Section 6.4, we analyze these two strategies assuming, respectively, that the systems quantize the information by means of deterministic quantizers and by means of probabilistic quantizers. Finally in Section 6.5 we gather out our conclusions.

6.2 Quantized Symmetric Gossip Algorithms

For the sake of the clarity, we start by briefly reviewing the symmetric gossip consensus algorithm as described in Section 4.5, where the systems communicate each other the exact value of their states. Assume we are given an undirected graph $\mathcal{G} = (V, \mathcal{E})$. At each time step, one edge (i, j) is randomly selected in \mathcal{E} with probability $W^{(i,j)}$ such that $\sum_{(i,j) \in \mathcal{E}} W^{(i,j)} = 1$. The two agents connected by that edge average their states according to

$$x_i(t+1) = \frac{1}{2}x_i(t) + \frac{1}{2}x_j(t)$$

$$x_j(t+1) = \frac{1}{2}x_j(t) + \frac{1}{2}x_i(t)$$
(6.1)

and

$$x_h(t+1) = x_h(t)$$
(6.2)

if $h \neq i, j$. Let

$$E_{ij} = (e_i - e_j)(e_i - e_j)$$

and

$$P(t) = I - \frac{1}{2}E_{ij}$$

where $e_i = [0, \ldots, 0, 1, 0, \ldots, 0]^*$ is a $N \times 1$ unit vector with the *i*-th component equal to 1, then (6.1) and (6.2) can be written in a vector form as

$$x(t+1) = P(t)x(t)$$
(6.3)

where $x(t) = [x_1(t), \ldots, x_N(t)]^*$ denotes the state of the overall system. Note that P(t) is a doubly stochastic matrix. We have seen in in Section 4.5 that, if the graph \mathcal{G} is connected and each edge $(i, j) \in \mathcal{E}$ can be selected with a probability $W^{(i,j)}$ strictly positive, then (6.3) reaches, almost surely, the *average consensus*, namely

$$\lim_{t \to \infty} x(t) = x_{ave} \mathbb{1},$$

where $x_{ave} = \frac{1}{N} \mathbb{1}^* x(0)$. In the sequel, as in Section 4.5, we will make the following assumption.

Assumption 6.1 The graph $\mathcal{G} = (V, \mathcal{E})$ is a undirected connected graph and, at every time instant $t \in \mathbb{N}$, each edge $(i, j) \in \mathcal{E}$ can be selected with a strictly positive probability $W^{(i,j)}$.

As in the previous chapter, here we assume that the comunication network is constituted only of digital channels through which the systems can exchange only quantized information. We have already emphasized that this implies that the systems can not have access to the exact value of the state x of the systems with which they are communicating, but only to an estimate \hat{x} of it. In this chapter we will introduce two updating rules of the state which represent the gossip-version of the two updating rules (5.5) and (5.15), considered in the previous chapter. Precisely, if (i, j) is the edge selected at the *t*-th iteration, in the first strategy we assume that *i* and *j*, in order to update its state, use only the estimates of their states, i.e.,

$$x_i(t+1) = \frac{1}{2}\hat{x}_i(t) + \frac{1}{2}\hat{x}_j(t)$$

$$x_j(t+1) = \frac{1}{2}\hat{x}_j(t) + \frac{1}{2}\hat{x}_i(t),$$
 (6.4)

or, equivalently in vector form, by recalling the definition of P(t),

$$x(t+1) = P(t)\hat{x}(t),$$
(6.5)

where $\hat{x}(t) = [\hat{x}_1(t), \dots, \hat{x}_N(t)]$. In the second strategy, by remarking that (6.1) can be written as

$$x_{i}(t+1) = x_{i}(t) - \frac{1}{2}x_{i}(t) + \frac{1}{2}x_{j}(t)$$

$$x_{j}(t+1) = x_{j}(t) - \frac{1}{2}x_{j}(t) + \frac{1}{2}x_{i}(t)$$
(6.6)

we propose the following updating rule, where the systems use also perfect information regarding their own state,

$$x_{i}(t+1) = x_{i}(t) - \frac{1}{2}\hat{x}_{i}(t) + \frac{1}{2}\hat{x}_{j}(t)$$

$$x_{j}(t+1) = x_{j}(t) - \frac{1}{2}\hat{x}_{j}(t) + \frac{1}{2}\hat{x}_{i}(t),$$
(6.7)

or, equivalently in vector form,

$$x(t+1) = x(t) + (P(t) - I)\hat{x}(t).$$
(6.8)

Accordingly to Section 5.4.1 of the previous chapter, we call the law (6.4) globally quantized and the law (6.7) partially quantized. It is easy to see that the partially quantized law (6.7), as the law (6.1) and the time-invariant partially quantized rule (5.5), maintains the initial state average. This fact is stated in the following proposition.

Proposition 6.2 Consider (6.7). Let

$$x_{ave}(t) = \frac{1}{N} \mathbb{1}^* x(t).$$

Then

$$x_{ave}(t) = x_{ave}(0)$$

for all $t \geq 0$.

Proof: We have that

$$\mathbb{1}^* x(t+1) = \mathbb{1}^* x(t) + \mathbb{1}^* (P(t) - I) \hat{x}(t)$$

= $\mathbb{1}^* x(t)$

where the last inequality follows from the fact that, since P(t) is doubly stochastic for all $t \ge 0$, it holds $\mathbb{1}^*(P(t) - I) = 0$ for all $t \ge 0$.

We proceed our analysis of these two rules by assuming first the $\hat{x}_i(t) = q_d(x_i(t))$, i.e., the information transmitted is quantized by means of deterministic quantizer, and then by assuming that $\hat{x}_i(t) = q_p(x_i(t))$, i.e., the information transmitted is quantized by means of probabilistic quantizer, introduced in Section 5.2.

Remark 6.3 As in the previous chapter, we assume here that the set of quantization levels is constituted of only integers numbers. As underlined in Remark 5.4, the more general case can be treated by a suitable scaling.

6.3 Quantized symmetric gossip algorithms via deterministic quantizers

In this section we assume that the information exchanged between the systems is quantized by means of the deterministic quantizer q_d described in (5.7), namely $\hat{x}_i(t) = q_d(x_i(t))$. Hence (6.4) and (6.7) become

$$x_{i}(t+1) = \frac{1}{2}q_{d}[x_{i}(t)] + \frac{1}{2}q_{d}[x_{j}(t)]$$

$$x_{j}(t+1) = \frac{1}{2}q_{d}[x_{j}(t)] + \frac{1}{2}q_{d}[x_{i}(t)],$$
(6.9)

and

$$x_{i}(t+1) = x_{i}(t) - \frac{1}{2}q_{d}[x_{i}(t)] + \frac{1}{2}q_{d}[x_{j}(t)]$$

$$x_{j}(t+1) = x_{j}(t) - \frac{1}{2}q_{d}[x_{j}(t)] + \frac{1}{2}q_{d}[x_{i}(t)],$$
(6.10)

In the following we will analyze the two strategies separately, starting from the latter.

6.3.1 Partially quantized strategy

Consider the partially quantized strategy (6.10). Let us define

$$d(t) = \frac{1}{\sqrt{N}} \|y(t)\|_2, \tag{6.11}$$

where

$$y(t) = \left(I - \frac{1}{N} \mathbb{1}\mathbb{1}^*\right) x(t).$$
 (6.12)

Such quantity represents the distance of the state x(t) from the current average of the state. Since the *partially quantized strategy* is an average-preserving law, y(t) coincides with the distance of the state from the initial average.

As an example we report in Figure 6.1 the result of simulations relative to a connected random geometric graph. Such graph has been drawn placing N = 50 nodes uniformly at random inside the unit square and connecting two nodes whenever the distance between them is less that R = 0.3. The initial condition $x_i(0)$ is randomly chosen inside the interval [-100, 100] for all $1 \le i \le N$. Note that d(t) does not converge to 0, meaning that the average consensus is not reached, but the values get very close to the initial average.

We now prove this fact in general, quantifying how close asymptotically the systems get to the initial average of their states. To this aim, we will take advantage again of the *symbolic dynamics* which lies under the real



Figure 6.1: Behavior of d for a connected random graph with N = 50 in case of deterministic quantizers and of partially quantized strategy.

states dynamics and that we have introduced in Example 5.8. To the idea of this construction we will adapt the results presented in [75].

Let us start our analysis. We define $n_i(t) = \lfloor 2x_i(t) \rfloor$ for all $i \in V$. Simple properties of floor and ceiling operators, together with the Lemma 5.9 stated in Example 5.8, allow us to remark that $q[x_i(t)] = \lfloor \frac{n_i(t)}{2} \rfloor$ and that

$$x_i(t+1) = x_i(t) - \frac{1}{2}q[x_i(t)] + \frac{1}{2}q[x_j(t)]$$
$$\lfloor 2x_i(t+1) \rfloor = \lfloor 2x_i(t) \rfloor - q[x_i(t)] + q[x_j(t)],$$

from which we can obtain that

$$n_i(t+1) = n_i(t) - \left\lceil \frac{n_i(t)}{2} \right\rceil + \left\lceil \frac{n_j(t)}{2} \right\rceil$$
$$= \left\lfloor \frac{n_i(t)}{2} \right\rfloor + \left\lceil \frac{n_j(t)}{2} \right\rceil.$$

We have thus found an iterative system involving only the symbolic signals $n_i(t)$. When the edge (i, j) is selected, i and j adjourn their states following the pair dynamics

$$(n_i(t+1), n_j(t+1)) = g(n_i(t), n_j(t))$$
(6.13)

where

$$g(h,k) = \left(\left\lfloor \frac{h}{2} \right\rfloor + \left\lceil \frac{k}{2} \right\rceil, \left\lfloor \frac{k}{2} \right\rfloor + \left\lceil \frac{h}{2} \right\rceil \right).$$

It is clear that g is symmetric in the arguments, in the sense that if $g(h,k) = (\eta, \chi)$, then $g(k,h) = (\chi, \eta)$.

The analysis of the evolution of (6.13) will then allow us to obtain information about the asymptotics of $x_i(t)$, since $n_i(t) = \lfloor 2x_i(t) \rfloor$.

Before stating the main result regarding the convergence properties of (6.13), we define the following quantities

$$m(t) = \min_{1 \le i \le N} n_i(t) \tag{6.14}$$

and

$$M(t) = \max_{1 \le i \le N} n_i(t),$$
(6.15)

and, finally,

$$D(t) = M(t) - m(t)$$

Moreover we provide the following two notational definitions. Let

$$\mathcal{R} = \left\{ r \in \mathbb{Z}^N : r - \alpha \mathbb{1} \in \{0, 1\}^N, \exists \alpha \in \mathbb{Z} \right\}.$$
 (6.16)

and let $n(t) = [n_1(t), \dots, n_N(t)]^*$.

We have the following result.

Theorem 6.4 Almost surely there exists $T_{con} \in \mathbb{N}$ such that $n(t) \in \mathcal{R}$ for all $t \geq T_{con}$.

Proof: The proof is based on verifying the following three facts:

- (i) the evolution of n(t) is a Markov process with a finite number of states;
- (ii) the set \mathcal{R} , defined in (6.16), is an invariant subset for the evolution described by (6.13);
- (iii) there is a positive probability of reaching a state belonging to the invariant subset \mathcal{R} , starting from any initial condition.

Standard results in the Markov chains literature (see [93]), ensure that, if the above three facts yield true, then any trajectory starting from a state $\bar{n} \notin \mathcal{R}$, will reach with probability 1 in a finite time, a state belonging to \mathcal{R} .

Let us now check them in order.

(i) We start by observing that the Markov property of the process follows from Assumption 6.1. We prove now that the states are finite. To this aim let

(h', k') = g(h, k).

By the structure of g, it is easy to see that

$$\max\left\{h',k'\right\} \le \max\left\{h,k\right\}$$

and

$$\min\left\{h,k\right\} \le \min\left\{h',k'\right\}.$$

Therefore we have that $m(n(t)) \ge m(n(0))$ and $M(n(t)) \le M(n(0))$ and hence that the cardinality of the set of the states is upper bounded by $(M(n(0)) - m(n(0)) + 1)^N$.

(ii) Let $h \in \mathbb{Z}$. Observe that

$$g(h, h+1) = \begin{cases} (h+1, h) & \text{if } h \text{ is even} \\ (h, h+1) & \text{if } h \text{ is odd} \end{cases}$$

This implies that \mathcal{R} is an invariant subset for the dynamics described by (6.13).

(iii) The proof of this fact is based on the following strong result about the monotonicity of D(t): if $D(t) \ge 2$, then there exists $\tau \in \mathbb{N}$ such that

$$\mathbb{P}[D(t+\tau) < D(t)] > 0.$$
(6.17)

Now we prove (6.17).

Let $\mathcal{I}(t) = \{j \in \mathcal{V} \text{ s.t. } n_j(t) = m(t)\}$. We start by proving that $|\mathcal{I}(t)|$, i.e., the cardinality of $\mathcal{I}(t)$, does not increase and that, if $D(t) \geq 2$, then there is a positive probability that it decreases within a finite number of time steps. Notice first that, for $h, k \in \mathbb{Z}, g(h+2, k+2) = g(h, k)+2$. Hence, by an appropriate translation of the initial condition, we can always restrict ourselves to the case $m(t) \in \{0, 1\}$, which of course is easier to handle.

Case m(t) = 0. In this case it is possible for a nonzero state to decrease to 0, but only in the case of a swap between 0 and 1. This assures that $|\mathcal{I}(t)|$ is nonincreasing. Let $\mathcal{S}(t)$ denote the set of nodes which have value m(t) + 2 or more. Since $D(t) \ge 2$ then $\mathcal{S}(t)$ is non empty at time t. Now let $(v_1, v_2, \ldots, v_{p-1}, v_p)$ be a shortest path between $\mathcal{I}(t)$ and $\mathcal{S}(t)$. Such a path exists since \mathcal{G} is connected. Note that $v_1 \in \mathcal{I}(t)$ and $v_p \in \mathcal{S}(t)$ and that $\{v_2, \ldots, v_{p-1}\}$ could be an empty set; in this case a shortest path between $\mathcal{I}(t)$ and $\mathcal{S}(t)$ has length 1. Moreover note also that all the nodes in the path except v_1 and v_p have value 1 at time t, otherwise $(v_1, v_2, \ldots, v_{p-1}, v_p)$ is not a shortest path. Since each edge of the communication graph has a positive probability of being selected in any time, there is also a positive probability that in the p-1time units following t the edges of this path are selected sequentially, starting with the edge (v_1, v_2) . At the last step of this sequence we have that the values of v_{p-1} and v_p are updated. By observing again, that the pair of value (0, 1) is transformed by (6.13) into the pair (1, 0)we have that the value of v_{p-1} , when the edge (v_{p-1}, v_p) is selected, is equal to 0. This update, for the form of (6.13), will cause the value of both nodes to be strictly greater than 0. Therefore, this proves that $|\mathcal{I}(t+p-1)| < |\mathcal{I}|$ with positive probability. Clearly, if $|\mathcal{I}(t)| = 1$ then we have also that D(t + p - 1) < D(t) with positive probability. Case m(t) = 1. In this case no state can decrease to 1, and then $|\mathcal{I}(t)|$ is not increasing. Let $\mathcal{I}(t)$, $\mathcal{S}(t)$ and $(v_1, v_2, \ldots, v_{p-1}, v_p)$ be defined as in the previous case. Obviously in this case all the nodes v_2, \ldots, v_{p-1} in the path have value equal to 2. Moreover observe that also the sequence of edges $(v_{p-1}, v_p), (v_{p-2}, v_{p-1}), \dots, (v_2, v_3), (v_1, v_2)$ has positive probability of being selected in the p-1 time units following t. At the last step of this sequence of edges, the values of v_1 and v_2 are updated. Clearly the value of v_1 is equal to 1. Since the value of v_p at time t is greater or equal to 3, and since the pair (2,3) is transformed by (6.13) into (3,2), we have that the value of v_2 when the edge (v_1, v_2) is selected, is greater or equal to 3. This update, for (6.13), will cause the value of both nodes to be strictly greater than 1. Hence $|\mathcal{I}(t+p-1)| < |\mathcal{I}|$ with positive probability. Again, if $|\mathcal{I}(t)| = 1$ then we have also that D(t+p-1) < D(t) with positive probability.

Consider now the following sequence of times $t_0 = t, t_1, t_2, \ldots$ For each $i \ge 0$, if $|\mathcal{I}(t)| > 1$, then we let t_{i+1} to be the first time for which there is a positive probability that $|\mathcal{I}(t_{i+1})| < |\mathcal{I}(t_i)|$. Let now $k \in \mathbb{N}$ be such that $|\mathcal{I}(t_k)| = 1$. Then we have that $D(t_{k+1}) < D(t_k)$. This ensures the validity of (6.17).

The proof of the fact (iii) follows directly from (6.17). Indeed, let $\bar{n} \notin \mathcal{R}$, then, from a repeated application of (6.17) it follows that, there exists a path connecting \bar{n} to a state $\bar{n}' = [\bar{n}'_1, \ldots, \bar{n}'_N]$, such that

$$\max\{\bar{n}'_1, \dots, \bar{n}'_N\} - \min\{\bar{n}'_1, \dots, \bar{n}'_N\} < 2,$$

i.e, $\bar{n}' \in \mathcal{R}$.

This proves the thesis.

We can go back to the original system, and prove the following result.

Corollary 6.5 Consider the algorithm (6.10). Then, almost surely, there exists $T_{con} \in \mathbb{N}$ such that

$$|x_i(t) - x_j(t)| \le 1 \quad \forall i, j \quad \forall t \ge T_{con}, \tag{6.18}$$

and hence

$$\|x(t) - x_{ave}\mathbb{1}\|_{\infty} \le 1$$

where $x_{ave} = \frac{1}{N} \mathbb{1}^* x(0)$.

Proof: The proof is an immediate consequence of Theorem 6.4 and of the relation $n_i(t) = \lfloor 2x_i(t) \rfloor$.

Remark 6.6 It is worth noting that Theorem 6.4 is an extension of Lemma 3 and Theorem 1 in [75]. In [75] the authors introduced a class of quantized gossip algorithms, satisfying the following assumptions. Assume that (i, j) is the edge selected at time t and that $n_i(t)$ and $n_j(t)$ are respectively the values of node i and of node j at time t. If $n_i(t) = n_j(t)$ then $n_i(t+1) = n_i(t)$ and $n_j(t+1) = n_j(t)$. Otherwise, defined $D_{ij} = |n_i(t) - n_j(t)|$, the method used to update the values has to satisfy the following three properties:

- (P1) $n_i(t+1) + n_j(t+1) = n_i(t) + n_j(t),$
- (P2) if $D_{ij}(t) > 1$ then $D_{ij}(t+1) < D_{ij}(t)$, and
- (P3) if $D_{ij}(t) = 1$ and (without loss of generality) $n_i(t) < n_j(t)$, then $n_i(t + 1) = n_j(t)$ and $n_j(t+1) = n_i(t)$. This update is called swap.

Now we substitute the property (P3) either with the property

(P3') if $D_{ij}(t) = 1$ and (without loss of generality) $n_i(t) < n_j(t)$, then, if $n_i(t)$ is odd, then $n_i(t+1) = n_j(t)$ and $n_j(t+1) = n_i(t)$, otherwise if $n_i(t)$ is even then $n_i(t+1) = n_i(t)$ and $n_j(t+1) = n_j(t)$

or with the property

(P3") if $D_{ij}(t) = 1$ and (without loss of generality) $n_i(t) < n_j(t)$, then, if $n_i(t)$ is even then $n_i(t+1) = n_j(t)$ and $n_j(t+1) = n_i(t)$, otherwise if $n_i(t)$ is odd then $n_i(t+1) = n_i(t)$ and $n_j(t+1) = n_j(t)$.

If we consider the class of algorithms satisfying (P1), (P2), (P3') or satisfying (P1), (P2), (P3"), it is possible to prove that Lemma 3 and Theorem 1 stated in [75] hold true also for this class. The proofs are analogous to that of Lemma ?? and Theorem 6.4 provided in this paper. Moreover it is easy to see that the algorithm (6.13) satisfies the properties (P1), (P2), (P3'). This represents an alternative way to prove Theorem 6.4.

6.3.2 Globally quantized strategy

In this subsection we consider the globally quantized strategy (6.9). We underline immediately that the fact that (6.9) uses only quantized information and not perfect information combined with quantized information as in (6.10) makes the analysis of (6.10) slightly easier than the analysis of (6.9).

Remarkably, we will show in this subsection that the law (6.10) drives, almost surely, the systems to exact consensus at an integer value. Unfortunately, the initial average of states is not preserved in general. Again, the analysis of this algorithm can be performed efficiently by means of the *symbolic dynamics* previously introduced.

Let again $n_i(t) = \lfloor 2x_i(t) \rfloor$ for all $i \in V$. By recalling that $q_d(x_i(t)) = \left\lceil \frac{n_i(t)}{2} \right\rceil$ and by observing that

$$\lfloor 2x_i(t+1) \rfloor = q_d(x_i(t)) + q_d(x_j(t)),$$

we obtain

$$n_i(t+1) = \left\lceil \frac{n_i(t)}{2} \right\rceil + \left\lceil \frac{n_j(t)}{2} \right\rceil.$$
(6.19)

Hence also for (6.9), we have found an iterative system involving only the symbolic signals $n_i(t)$. When the edge (i, j) is selected, i and j adjourn their states following the pair dynamics

$$(n_i(t+1), n_j(t+1)) = \left(\left\lceil \frac{n_i(t)}{2} \right\rceil + \left\lceil \frac{n_j(t)}{2} \right\rceil, \left\lceil \frac{n_i(t)}{2} \right\rceil + \left\lceil \frac{n_j(t)}{2} \right\rceil \right).$$
(6.20)

Let now $n(t) = [n_1(t), \ldots, n_N(t)]^*$ and let $g_1 : \mathbb{Z} \times \mathbb{Z} \to \mathbb{Z}$ be defined as

$$g_1(h,k) := \left\lceil \frac{h}{2} \right\rceil + \left\lceil \frac{k}{2} \right\rceil.$$

Note that

$$(n_i(t+1), n_j(t+1)) = (g_1(n_i(t), n_j(t)), g_1(n_i(t), n_j(t))).$$
(6.21)

Then we have the following result.

Theorem 6.7 Let n(t) be as defined above and consider (6.21). Then almost surely there exists $T_{con} \in \mathbb{N}$ and $\alpha \in \mathbb{Z}$ such that $n(t) = 2\alpha \mathbb{1}$ for all $t \geq T_{con}$.

Proof: The proof is based on verifying the following three facts:

- (i) the evolution process is a Markov process with a finite number of states;
- (ii) the process (6.21) has absorbing states;
- (iii) there is a positive probability of reaching an absorbing state in a finite time, starting for any initial state.

These facts, similarly to Theorem 6.4, ensures the validity of the thesis.

Let us now check them in order. Given n(t) let m(t) and M(t) be defined as in (6.14) and (6.15).

- (i) The Markov property of the process follows from Assumption 6.1. We show now that the states are finite. Let $p, q \in \mathbb{Z}$, where, without loss of generality, we assume that $p \leq q$. Then, from the structure of g_1 we have that
 - $p \leq g_1(p,q);$
 - if q is even, $g_1(p,q) \leq q$;
 - if q is odd, $g_1(p,q) \leq q + r_q$ where r_q denotes the remainder in the euclidean division of q over 2.

It follows that $m(0) \leq n_i(t) \leq M(0) + r_{M(0)}$ for all $i \in \mathcal{V}$ and for all $t \geq 0$, where $r_{M(0)}$ denotes the remainder in the euclidean division of M(0) over 2. Hence the cardinality of the set of the states is upper bounded by

$$(M(0) + r_{M(0)} + 1 - m(0))^{N}$$
.

- (ii) We denote as \mathcal{A} the set of absorbing states for (6.21). The form of g_1 implies that if $n_i(t) \neq n_j(t)$, then $n_i(t+1) = n_j(t+1)$, and if $n_i(t) = n_j(t) = a$ then $n_i(t+1) = n_j(t+1) = a$ if and only if a is even. Thus $\mathcal{A} = \{y \in \mathbb{Z}^N : \exists \alpha \in \mathbb{Z} \text{ such that } y = 2\alpha \mathbb{1}\}$. Thus the absorbing states are consensus states.
- (iii) Let us fix $t = t_0$, and assume that $n(t_0) \notin \mathcal{A}$. We will prove that there exists $\tau \in \mathbb{N}$ such that $\mathbb{P}[n(t_0 + \tau) \in \mathcal{A}] > 0$. We start by observing that, from the assumption of having a connected graph, there exists $(i, j) \in \mathcal{E}$ such that $n_i(t_0) = m(t_0), n_j(t_0) = q$ and $g(m(t_0), q) > m(t_0)$. Indeed, two cases are given when $n(t_0) \notin \mathcal{A}$.

- If $m(t_0) < M(t_0)$, then it suffices to consider there an edge (i, j)such that $n_i(t_0) = m(t_0)$ and $n_j(t_0) = q > m(t_0)$ which gives $g_1(m(t_0), p) > m(t_0)$. Note that this edge there exists from the hypothesis of having a connected graph;
- if $m(t_0) = M(t_0)$, necessarily we have that $m(t_0)$ and $M(t_0)$ are odd; then $g(m(t_0), m(t_0)) > m(t_0)$.

We define now $\mathcal{I}_a(t) = \{i \in \mathcal{V} : n_i(t) = a\}$. The above discussion implies that $|\mathcal{I}_{m(t_0)}(t_0 + 1)| < |\mathcal{I}_{m(t_0)}(t_0)|$ with the positive probability of choosing the edge (i, j) and hence that there is also a positive probability that at some finite time $t' > t_0$, $|\mathcal{I}_{m(t_0)}(t_0)| = 0$, that is $m(t') > m(t_0)$. Iterating this argument and, recalling that $M(t) \leq M(t_0) + r_{M(t_0)}$ for all $t \geq t_0$, show that there exists $\tau \in \mathbb{N}$ such that $\mathbb{P}[n(t_0 + \tau) \in \mathcal{A}] > 0$.

This proves the thesis.

We can go back to the original system. The following corollary follows immediately form the definition of n(t).

Corollary 6.8 Let x(t) evolve according to (6.9). Then almost surely there exists $T_{con} \in \mathbb{N}$ and $\alpha \in \mathbb{Z}$ such that $x_i(t) = \alpha$ for all $i \in \mathcal{V}$ and for all $t \geq T_{con}$.

Proof: It follows directly from the definition of $n_i(t) = \lfloor 2x_i(t) \rfloor$.

We have already underlined the fact that this strategy does not preserve the initial average, in general. Providing some probabilistic estimation of the distance of the consensus point from the initial average is a challenging problem. Unfortunately we have not been able to obtain any theoretical result so far. We limits our analysis to the following simulation. In Figure 6.2 we plotted the variable z that is defined as follows. In the globally quantized strategy we have that, almost surely $\lim_{t\to\infty} = \alpha \mathbb{1}$ for some random integer α . Let $z = |\alpha - 1/N\mathbb{1}^* x(0)|$. In other words z represents the distance from the consensus point to which the globally quantized strategy leads the systems and the average of the initial condition. We have depicted the value of z for a family of random geometric graphs of increasing size from N = 10 up to N = 80. The initial condition $x_i(0)$ is chosen randomly inside the interval [-100, 100] for all $1 \leq i \leq N$. Moreover for each N, z is computed as the mean of 100 trials. We can see that the value of z is increasing in N and



Figure 6.2: Behavior of z for a family of random geometric graphs in case of deterministic quantizers and of globally quantized strategy.

assumes values that are not negligible with respect to the quantization step size.

6.3.3 Some considerations on the speed of convergence of the globally quantized strategy and of the partially quantized strategy

Providing some insights on the speed of convergence of (6.10) and of (6.9) is quite hard in general. In Figure 6.3 and Figure 6.4 we report, respectively, a comparison between the partially quantized strategy (6.10) and the gossip algorithm with exchange of perfect information (6.1) and between the globally quantized strategy (6.9) and again the gossip algorithm with exchange of perfect information (6.1). The simulations are made on the same random geometric graph considered in Figure 6.1.

For all strategies we plotted the behavior of the variable d(t) defined in (6.11). For both the simulations the initial conditions are randomly chosen inside the interval [-100, 100]. It is worth noting that the variable y(t), introduced in (6.12) and whose normalized two-norm defines d(t), represents the distance of the state x(t) from its current average, that, only for (6.10) and (6.1), coincides also with the distance of the state x(t).

From the Figure 6.3 and Figure 6.4 we can infer that the speed of convergence toward the steady state of the quantized strategies (6.9) and (6.10) is

similar to the one of the gossip algorithm with perfect exchange of information. Unfortunately we have not been able to prove this numerical evidence so far.



Figure 6.3: Behavior of d, when using the *partially quantized* strategy, for a connected random geometric graph with N = 50. Note that since the *partially quantized* strategy does not converge to a consensus, d(t) does not go to 0.

Remark 6.9 We have observed in Figure 6.2, that the distance from the average of the initial state of the consensus point toward which the globally quantized strategy leads the systems is not negligible. On the other hand, the partially quantized strategy does not assure a consensus in a strict sense.

If, depending on the application, one can not relax the convergence requirement, inspired by the last subsection of the previous chapter, we could suggest the following heuristic solution to the consensus problem, which combines the positive features of both strategies,

$$x(t+1) = Pq_d(x(t)) + \epsilon(t)(x(t) - q_d(x(t))),$$

where $\epsilon(t)$, $t \ge 0$, is a nonnegative sequence such that $\epsilon(t) \le 1$, $\forall t \ge 0$ and $\lim_{t\to\infty} \epsilon(t) = 0$.

Remark 6.10 Some more insights on the speed of convergence of the quantized strategies and on the combined strategy presented in the above remark,



Figure 6.4: Behavior of d, when using the globally quantized strategy, for a connected random geometric graph with N = 50. In this case, accordingly to the theoretical result stated in Corollary 6.8, d(t) tends to 0.

could be provided by introducing a probabilistic model in which the quantization error is regarded as a white noise uncorrelated with the state of the system. As we have widely underlined in the previous chapter, there is no theoretical reason justifying the validity of this model when using deterministic quantizers. Instead the probabilistic model is a suitable model to explain the quantized algorithms via probabilistic quantizers: we will see this in the next Section.

6.4 Quantized symmetric gossip algorithms via probabilistic quantizers

In this section we assume that the information exchanged between the systems is quantized by means of the probabilistic quantizer q_p described in (5.8), namely $\hat{x}_i(t) = q_p(x_i(t))$. Hence (6.4) and (6.7) become

$$x_{i}(t+1) = \frac{1}{2}q_{p}[x_{i}(t)] + \frac{1}{2}q_{p}[x_{j}(t)]$$

$$x_{j}(t+1) = \frac{1}{2}q_{p}[x_{j}(t)] + \frac{1}{2}q_{p}[x_{i}(t)],$$
(6.22)

and

$$x_{i}(t+1) = x_{i}(t) - \frac{1}{2}q_{p}[x_{i}(t)] + \frac{1}{2}q_{p}[x_{j}(t)]$$

$$x_{j}(t+1) = x_{j}(t) - \frac{1}{2}q_{p}[x_{j}(t)] + \frac{1}{2}q_{p}[x_{i}(t)],$$
(6.23)

From now on we will assume that the initial condition x(0) satisfies the following.

Assumption 6.11 The initial condition x(0) is a random variable such that $\mathbb{E}[x(0)] = 0$ and $\mathbb{E}[x(0)x^*(0)] = \sigma_0^2 I$ for some $\sigma_0^2 > 0$

In the remainder of the section we will analyze the two strategies separately starting from the latter.

6.4.1 Partially quantized strategy

Consider the partially quantized strategy (6.23). Similarly to the partially quantized strategy via deterministic quantizers (6.10), also (6.23) does not reach the consensus in general. Again we report a simulation showing this fact. In Figure 6.5 the behavior of the quantity d(t), defined in (6.11), is depicted for the same connected random geometric graph considered in Figure 6.1. Note that the quantity d(t) does not converge a 0, meaning that the average consensus is not reached.

We observe immediately that the analysis of (6.23) is more complicate that the corresponding law (6.10). This is mainly due to the lack of some convexity arguments which we used in the analysis of (6.10). We will be more explicit in the following example.

Example 6.12 Consider (6.10) and assume that the edge (i, j) has been selected at time t. Without loss of generality assume that $x_i(t) \leq x_j(t)$. Then, by convexity arguments, we have that $\lfloor x_i(t) \rfloor \leq x_i(t+1), x_j(t+1) \leq \lfloor x_j(t) \rfloor$. This is not true for (6.23) anymore. As a numerical example assume that $x_i(t) = 3.4$ and $x_j(t) = 3.6$ and that $q_p(x_i(t)) = 4$ and $q_p(x_j(t)) = 3$. Then by (6.23) we have that $x_i(t+1) = 2.9$ and that $x_j(t+1) = 4.1$, namely $x_i(t+1), x_j(t+1)$ do not belong to the interval $\lfloor x_i(t) \rfloor, \lceil x_j(t) \rceil$.

By simulations we can see that (6.23) does not drive the states of the systems inside the same bin of quantization, as the corresponding strategy (6.10)



Figure 6.5: Behavior of d for a connected random geometric graph with N = 50.

using deterministic quantizers. In Figure 6.6, we depict the behavior of the quantity

$$s(t) = \max_{1 \le i,j \le N} |x_i(t) - x_j(t)|.$$

for the same random geometric graph considered in Figure 6.5. In this simulation we assume that the initial condition $x_i(0)$ is randomly chosen inside the interval [-10, 10]. Note that s asymptotically oscillates around 2. This



Figure 6.6: Behavior of s for a connected random geometric graph with N = 50.

suggests that (6.23) seems to work a little bit worse than (6.10). However, of note, is that also (6.23) behaves quite well.

Interesting results on (6.23), in terms of both the asymptotic distance from the initial average and the speed of convergence, can be provided by a mean-square analysis that we will carry out in the sequel of this subsection. We start by observing that (6.23) can be rewritten as

$$x(t+1) := P(t)x(t) + (P(t) - I)(q(x(t)) - x(t))$$
(6.24)

We consider again the variable

$$y(t) := \left(I - \frac{1}{N} \mathbb{1}\mathbb{1}^*\right) x(t) = x(t) - \left(\frac{1}{N} \mathbb{1}^* x(t)\right) \mathbb{1} = x(t) - \left(\frac{1}{N} \mathbb{1}^* x(0)\right) \mathbb{1}$$

where the last equality follows from the fact that the partially quantized strategy is an average-preserving law and we introduce the new variable

$$e(t) = q(x(t)) - x(t).$$

which represents the quantization error. Note that $\mathbb{1}^* y(t) = 0$ for all $t \ge 0$.

Now, from (6.24), we can write

$$\left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right) x(t+1) = \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right) P(t) x(t) + \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right) \left(P(t) - I \right) \left(q(x(t)) - x(t) \right).$$

Since $(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*) P(t) = P(t) (I - \frac{1}{N}\mathbb{1}\mathbb{1}^*)$ and $(I - \frac{1}{N}\mathbb{1}\mathbb{1}^*) (P(t) - I) = P(t) - I$, from the above expression we obtain the following recursive equation in terms of the variables y, e,

$$y(t+1) = P(t)y(t) + (P(t) - I)e(t).$$
(6.25)

In order to perform an asymptotic analysis of (6.25) it is convenient to introduce the following matrices. Let

$$\Sigma_{yy}(t) := \mathbb{E} \left[y(t)y^*(t) \right],$$
$$\Sigma_{ee}(t) := \mathbb{E} \left[e(t)e(t)^* \right],$$

and

$$\Sigma_{ye}(t) := \mathbb{E}\left[y(t)e(t)^*\right]$$

Equation (6.25) leads to the following recursive equation in terms of the above matrices

$$\Sigma_{yy}(t+1) = \mathbb{E} \left[P(t) \Sigma_{yy}(t) P(t) \right] + \mathbb{E} \left[P(t) \Sigma_{ye}(t) (P(t) - I) \right] + \\ + \mathbb{E} \left[(P(t) - I) \Sigma_{ye}^* P(t) \right] + (P(t) - I) \Sigma_{ee}(t) (P(t) - I) .$$
(6.26)

Notice that, since x(0) is a random variable satisfying Assumption 6.11, one can show that

$$\Sigma_{yy}(0) = \sigma_0^2 \left(I - 1/N \, \mathbb{1} \, \mathbb{1}^* \right). \tag{6.27}$$

We recall, from Proposition 5.35 stated in Section 5.4, that the variables y and e satisfy the following remarkable properties

(i) $\mathbb{E}[e(t)] = 0$ and $\mathbb{E}[e(t)e^*(t)] = \text{diag} \{\sigma_1^2(t), \dots, \sigma_N^2(t)\}$ where $\sigma_i^2(t) := \mathbb{E}[e_i^2(t)]$ is such that $\sigma_i^2(t) \le 1/4$ for all $1 \le i \le N$ and for all $t \ge 0$.

(ii)
$$\Sigma_{ye}(t) = 0$$
, for all $t \ge 0$.

From the above properties we have that (6.26) can be rewritten as

$$\Sigma_{yy}(t+1) = \mathbb{E}\left[P(t)\Sigma_{yy}(t)P(t)\right] + \mathbb{E}\left[\left(P(t)-I\right)\Sigma_{ee}(t)\left(P(t)-I\right)\right].$$
 (6.28)

To estimate the asymptotic distance from the initial average, we introduce the following cost function

$$J(W) := \limsup_{t \to \infty} \sqrt{\frac{1}{N} \operatorname{tr} \{\Sigma_{yy}(t)\}}.$$
(6.29)

We can rewrite the above evolution law as

$$\Sigma_{yy}(t+1) = \mathcal{A}(\Sigma_{yy}(t)) + \mathcal{B}(\Sigma_{ee}(t)),$$

where \mathcal{A} and \mathcal{B} are linear operators from $\mathbb{R}^{N \times N}$ to itself. Namely, given a matrix M, $\mathcal{A}(M) = \mathbb{E}[P(t)MP(t)]$ and $\mathcal{B}(M) = \mathbb{E}[(P(t) - I)M(P(t) - I)]$. It is immediate to remark that \mathcal{A} induces the natural evolution of the dynamical system, in the absence of quantization error, while \mathcal{B} can be regarded as a disturbance due to the quantization error.

From [58], we know that in the case of no quantization the systems converge almost surely to consensus. This implies that \mathcal{A} is an asymptotically stable operator when restricted to the subspace $\mathcal{S} = \{M \in \mathbb{R}^{N \times N} : \mathbb{1}^* M \mathbb{1} = 0\}$. Since $\mathbb{1}^* \mathcal{B}(M)\mathbb{1} = 0$ for any matrix M, we have that $\Sigma_{yy}(t) \in \mathcal{S}$ for all $t \geq 0$.

It is interesting to remark that these observations show how, during the transient, the symmetric gossip algorithm, with exchange of information quantized by means of probabilistic quantizers, behaves like the symmetric gossip algorithm with exchange of exact information.

Providing an expression for J(W) is quite hard in general. We try now to simplify the problem by introducing the following auxiliary system

$$\bar{\Sigma}(t+1) = \mathbb{E}\left[P(t)\bar{\Sigma}(t)P(t)\right] + \frac{1}{4}\mathbb{E}\left[\left(P(t) - I\right)^2\right],\tag{6.30}$$

where $\bar{\Sigma}(0) = \Sigma_{yy}(0)$, and the following cost function

$$\bar{J}(W) := \limsup_{t \to \infty} \sqrt{\frac{1}{N} \operatorname{tr} \{ \bar{\Sigma}(t) \}}.$$

We have the following comparison result.

Proposition 6.13 Consider the functional costs J(W) and $\overline{J}(W)$. We have that

$$J(W) \le \bar{J}(W).$$

Proof: To prove the statement of the Proposition we will show, by induction on t, that $\bar{\Sigma}(t) \geq \Sigma_{yy}(t)$ for all $t \geq 0$, where the inequality is meant in matricial sense, that is, $\bar{\Sigma}(t) - \Sigma_{yy}(t)$ is a semidefinite positive matrix.

Since $\bar{\Sigma}(0) = \Sigma_{yy}(0)$ the assertion holds true for t = 0. Assume now that $\bar{\Sigma}(t) \geq \Sigma_{yy}(t)$ holds true for a generic t. We have that

$$\begin{split} \bar{\Sigma}(t+1) &- \Sigma_{yy}(t+1) = \\ &= \mathbb{E}\left[P(t)\bar{\Sigma}(t)P(t)\right] + \frac{1}{4}\mathbb{E}\left[(P(t)-I)^2\right] \\ &- \left(\mathbb{E}\left[P(t)\Sigma_{yy}(t)P(t)\right] + \mathbb{E}\left[(P(t)-I)\sum_{ee}(t)\left(P(t)-I\right)\right]\right) \\ &= \mathbb{E}\left[P(t)(\bar{\Sigma}(t) - \Sigma_{yy}(t))P(t)\right] \\ &+ \mathbb{E}\left[\left(P(t)-I\right)\left(\frac{1}{4}I - \Sigma_{ee}(t)\right)\left(P(t)-I\right)\right]. \end{split}$$

Since by inductive hypothesis $\overline{\Sigma}(t) \geq \Sigma_{yy}(t)$ and since by Proposition 5.35 we know that $\Sigma_{ee}(t) \leq \frac{1}{4}I$ for all $t \geq 0$, we have that $\overline{\Sigma}(t+1) - \Sigma_{yy}(t+1) \geq 0$.

Observe now that, since $P(t)^2 = P(t)$ we obtain that $\mathbb{E}[(I - P(t))^2] = I - \mathbb{E}[P(t)]$. From this fact we obtain the following result.

Proposition 6.14 Given the above definitions,

$$\lim_{t \to \infty} \bar{\Sigma}(t) = \frac{1}{4} \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right).$$

Proof: Since $\overline{\Sigma}_{yy}(0) \in S$, and \mathcal{A} is asymptotically stable if restricted to the subspace \mathcal{S} , then

$$\lim_{t \to \infty} \bar{\Sigma}_{yy}(t) = \sum_{t=0}^{+\infty} \mathcal{A}^{(t)}(\bar{\mathcal{B}}),$$

where $\bar{\mathcal{B}} := \mathbb{E}[(I - P(t))^2]$. This is the only fixed point of the iteration law (6.30). Thus we are left to prove that $\Sigma^* = \frac{1}{4} \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right)$ is a fixed point, that is $\Sigma^* = \mathcal{A}(\Sigma^*) + \bar{\mathcal{B}}$. Indeed observe that

$$\mathcal{A}(\Sigma^*) + \bar{\mathcal{B}} = \frac{1}{4} \mathbb{E} \left[P(t) \left(I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right) P(t) \right] + \frac{1}{4} (I - \mathbb{E}[P(t)]) \\ = \frac{1}{4} \left\{ \mathbb{E} \left[P(t)^2 \right] - \frac{1}{N} \mathbb{1} \mathbb{1}^* + I - \mathbb{E}[P(t)] \right\} = \frac{1}{4} \left\{ I - \frac{1}{N} \mathbb{1} \mathbb{1}^* \right\}$$

Corollary 6.15 For all probability matrix W we have that $J(W) \leq \frac{1}{2}$.

Proof: From the above theorem we can argue that $\overline{J}(W) = \frac{1}{2}\sqrt{\frac{N-1}{N}}$, and since $J(W) \leq \overline{J}(W)$, we can conclude.

From these theorems we draw a strong conclusion about the convergence of the algorithm. In spite of missing a consensus in the strict sense, the asymptotical mean squared error of the algorithm is smaller than the size of the quantization bin, and has a bound which does not depend on the number of the agents nor on the topology of the graph.

6.4.2 Globally quantized strategy

In this subsection we consider the globally quantized strategy (6.22). We will prove that the law (6.22), as the law (6.9), drives almost surely the systems to exact consensus at an integer value. Moreover, we will show by simulations, that the consensus point, even if (6.22) does not preserve the average of the
state, is rather close to the average of the initial condition. This represents a significant improvement with respect to the strategy (6.9), that, as seen in Figure 6.2, leads to a consensus point whose distance from the average of the initial condition, is not negligible in general.

We observe immediately that, with the globally quantized strategy (6.22), we have to deal with two sorts of randomness, since the interacting pair is randomly selected, and the quantization map is itself random. This makes the analysis of (6.22) more complicate than the analysis of (6.9). However, again, we prove the convergence by a *symbolic dynamics*.

Let again $n_i(t) = \lfloor 2x_i(t) \rfloor$ for all $i \in V$ and let $n(t) = [n_1(t), \ldots, n_N(t)]^*$. Before finding a recursive equation for n(t), we need to introduce the following random variable. Let

 $T_{all} = \inf \{t : \text{at time t every edge in } \mathcal{E} \text{ has been selected at least once} \}$

 T_{all} is an integer random variable which, since the edges are selected with positive probability, is almost surely finite. Note that, from (6.22), $x_i(t) \in$ $\{a, a + 1/2\}$ for some integer number a, for all $t \ge T_{all}$. This allows us to disregard the evolution before T_{all} and to analyze, for $t > T_{all}$, the symbolic dynamics as follows. For $t \ge T_{all}$, by recalling how the probabilistic quantizer works, we have that

$$q_p[x(t)] = \begin{cases} \frac{n_i(t)}{2} & \text{if } n_i(t) \text{ is even} \\ \begin{bmatrix} \frac{n_i(t)}{2} \end{bmatrix} & \text{with probability } 1/2 \\ \lfloor \frac{n_i(t)}{2} \rfloor & \text{with probability } 1/2 \end{cases} \quad \text{if } n_i(t) \text{ is odd}$$

Let ξ_1 and ξ_2 be two independent Bernoulli random variables with parameter 1/2 and assume that, at time instant t, the edge (i, j) is selected. Then

$$n_i(t+1) = n_j(t+1) = \left\lceil \frac{n_i(t)}{2} \right\rceil + \left\lceil \frac{n_j(t)}{2} \right\rceil - \xi_1 r_{n_i(t)} - \xi_2 r_{n_j(t)}$$
(6.31)

where $r_{n_i(t)}$ and $r_{n_j(t)}$ denote, respectively, the remainders of the euclidean division of $n_i(t)$ over 2 and of $n_j(t)$ over 2. Let us define $g_2 : \mathbb{Z} \times \mathbb{Z} \to \mathbb{Z}$ such that

$$g_2(h,k) = \left\lceil \frac{h}{2} \right\rceil + \left\lceil \frac{k}{2} \right\rceil - \xi_1 r_h - \xi_2 r_k$$

then we have that

$$(n_i(t+1), n_j(t+1)) = (g_2(n_i(t), n_j(t)), g_2(n_i(t), n_j(t))).$$
(6.32)

The following result characterizes the convergence properties of (6.32).

Theorem 6.16 Let n(t) be as above. Consider (6.32). Then almost surely there exists $T_{con} \in \mathbb{N}$ and $\alpha \in \mathbb{Z}$ such that $n_i(t) = 2\alpha$ for all $i \in V$ and for all $t \geq T_{con}$.

Proof: The proof is similar to the proof of Theorem 6.7 and Theorem 6.4, and it is based again on proving the following three facts:

- (i) The evolution process is a Markov process with a finite number of states;
- (ii) The process (6.32) has absorbing states;
- (iii) There is a positive probability of reaching an absorbing state in a finite time, starting for any initial state.

Let us now check them in order. Given n(t) let m(t) and M(t) be defined as in (6.14) and (6.15).

- (i) The Markov property of the process follows from Assumption 6.1 and from (6.32). We show now that the states are finite. Let $h \in \mathbb{Z}$. Then, from the structure of g_2 we have that
 - $g_2(h,h) = h$ if h is even;
 - $h 1 \le g_2(h, h) \le h + 1$ if *h* is odd.

The above two properties imply that $m(0) - r_{m(0)} \leq n_i(t) \leq M(0) + r_{M(0)}$ for all $i \in \mathcal{V}$ and for all $t \geq 0$, where $r_{M(0)}$ and $r_{m(0)}$ denote, respectively, the remainders in the euclidean division of M(0) over 2 and of m(0) over 2. Hence the cardinality of the set of the states is upper bounded by

$$(M(0) + r_{M(0)} + 1 - m(0) + r_{m(0)})^{N}.$$

- (ii) We denote as \mathcal{A} the set of absorbing states for (6.32). The form of g_2 implies that if $n_i(t) \neq n_j(t)$, then $n_i(t+1) = n_j(t+1)$, and if $n_i(t) = n_j(t) = a$ then $n_i(t+1) = n_j(t+1) = a$ with probability 1, if and only if a is even. Thus, as for (6.21), we have that $\mathcal{A} = \{y \in \mathbb{Z}^N : \exists \alpha \in \mathbb{Z} \text{ such that } y = 2\alpha 1 \forall i\}.$
- (iii) Observe that

$$g_2(h,k) = g_1(h,k) - \xi_1 r_h - \xi_2 r_k,$$

where g_1 is the map defining the evolution of (6.21). Hence

$$\mathbb{P}[g_2(h,k) = g_1(h,k)] \ge \frac{1}{4}.$$

This fact, combined with the fact (iii) proved along the proof of Theorem 6.7, ensures that, also for (6.32), there is a positive probability of reaching an absorbing state in a finite time, starting from any initial state.

This proves the thesis.

The above theorem and the previous remarks about T_{all} lead to the following claim about the original system.

Corollary 6.17 Let x(t) evolve following (6.22). Then almost surely there exists $T_{con} \in \mathbb{N}$ and $\alpha \in \mathbb{Z}$ such that $x_i(t) = \alpha$ for all $i \in \mathcal{V}$ and for all $t \geq T_{con}$.

Proof: It follows directly from the definition of $n_i(t) = \lfloor 2x_i(t) \rfloor$.

As for (6.9) we have not been able so far to provide a theoretical estimation of the consensus point to which (6.22) leads the systems from the average of the initial condition.

We limit our analysis to the following simulations. In Figure 6.7 we plot the variable z as previously defined for the globally quantized strategy using deterministic quantizers, i.e., $z = |c - 1/N1^*x(0)|$ where c is such that $\lim_{t\to\infty} x(t) = c1$. z represents the distance from the consensus point to which the globally quantized strategy leads the systems and the average of the initial condition. We plot the value of z for a family of random geometric graphs of increasing size from N = 10 up to N = 80. The initial condition $x_i(0)$ is chosen randomly inside the interval [-100, 100] for all $1 \le i \le N$. Moreover for each N, z is calculated as the mean of 100 trials. In Figure 6.8 we provide a comparison between (6.9) and (6.22). Surprisingly, the globally quantized strategy using probabilistic quantizers, differently from the globally quantized strategy using deterministic quantizers, seems to reach the consensus very close to the average of the initial condition.



Figure 6.7: Behavior of z for a family of random geometric graphs when considering the globally quantized strategy using probabilistic quantizers.



Figure 6.8: Comparison in terms of z between the "deterministic" and the "probabilistic" strategy, for a family of random geometric graphs.

6.5 Conclusions

In this chapter we addressed the quantized consensus problem for the symmetric gossip algorithm. In order to face the effects due to the quantization (both deterministic and probabilistic) we proposed here two updating rules: the globally quantized strategy and the partially quantized strategy. In the former the nodes use only quantized information in order to update their state. In the latter they have access also to exact information regarding their own state.

We have seen that the partially quantized strategy, with both the quantiz-

ers, deterministic and probabilistic, does not reach the consensus in general, but maintain the average of the state at each iteration and drive all the states very close to the average of the initial condition.

On the other hand, we have shown that the globally strategy leads almost surely to a consensus which, however, does not coincide with the average of the initial condition. We have provided some simulations characterizing the distance between the consensus point and the initial average. While using the deterministic quantizer this distance turns out to be not negligible, with the probabilistic quantizer the consensus is reached surprisingly very close to the average of the initial condition.

Providing some theoretical insights on this fact will be the object of future research. Moreover, an another interesting issue to address will be the evaluation of the speed of convergence of the strategies considered in this chapter.

Chapter 7

Quantized average consensus via dynamic coding/decoding schemes

7.1 Introduction

We have seen in Chapter 2 that the main features of the $ideal^1$ average consensus algorithm, both for time-varying version and the time-invariant version, are

- preservation of the average of the state at each iteration
- asymptotic convergence to an agreement that, since the algorithm preserves the average of the state at each iteration, coincides with the initial average.

In a more realistic case, in which the systems can communicate only through digital channels, i.e., they can exchange only symbolic data, the above conditions are in general violated. In the previous two chapters, in presence of quantized communications, we have elaborated strategies that:

• maintain the average of the state at each iteration, but do not converge asymptotically to the average consensus, even if they drive the states of all the systems quite close to it;

¹with *ideal* average consensus algorithm is meant the algorithm in which the agents can exchange perfect information between them

• do not maintain the average of the state at each iteration, but reach asymptotically a consensus which is, in general, different from the average of the initial condition.

The main contribution of this chapter is to introduce a novel quantized strategy that permits both to maintain the initial average and to reach it asymptotically. More precisely, we adapt coding/decoding strategies, that were proposed for the centralized quantized control problems, to the distributed consensus problem. In particular, we present two coding/decoding strategies, one based on the exchange of logarithmically quantized information, the other on a zoom in - zoom out strategy (this latter involves the use of uniform quantizers). We provide analytical and simulative results illustrating the convergence properties of these strategies. In particular we show that the convergence factors depend smoothly on the accuracy parameter of the quantizers used and that, remarkably, the critical quantizer accuracy sufficient to guarantee convergence is independent from the network dimension.

The paper is organized as follows. Section 7.2 briefly reviews the standard average consensus algorithm. In Section 7.3 we present two strategies of coding/decoding of the data throughout reliable digital channels: one based on logarithmic quantizers, the other on uniform quantizers. We analyze the former from a theoretical point in Section 7.4 and Section 7.5. We provide simulations results for the latter in Section 7.6. Finally, we gather our conclusions in Section 7.7.

7.2 Problem Formulation

For the sake of the clarity, we start this section by briefly describing the standard discrete-time consensus algorithm. Assume that we have a set of agents V and a graph \mathcal{G} on V describing the feasible communications among the agents. For each agent $i \in V$ we denote by $x_i(t)$ the estimate of the average of agent i at time t. We have seen in Chapter 2 that standard average consensus algorithms are constructed by choosing a doubly stochastic matrix $P \in \mathbb{R}^{N \times N}$ compatible with \mathcal{G} and assuming that at every times t agent i updates its estimate according to

$$x_i(t+1) = \sum_{j=1}^{N} P_{ij} x_j(t).$$
(7.1)

More compactly we can write

$$x(t+1) = Px(t),$$
 (7.2)

where x(t) is the column vector whose entries $x_i(t)$ represent the agents states.

In our treatment we will restrict to the case in which P is symmetric, i.e., $P^* = P$. Note that a stochastic symmetric matrix P is automatically doubly stochastic.

We have seen in Chapter 2, that, if P is a symmetric stochastic matrix with positive diagonal entries and such that \mathcal{G}_P is connected, then the algorithm (7.2) solves the *average consensus problem*, namely

$$\lim_{t \to +\infty} x(t) = x_{ave} \mathbb{1},$$

where $x_{ave} = 1/N \mathbb{1}^* x(0)$. From now on we will make the following assumption.

Assumption 7.1 *P* is a symmetric stochastic matrix such that $P_{ii} > 0$, for $i \in \{1, ..., N\}$, and \mathcal{G}_P is connected.

Before proceeding we provide the following notational definition. Given the symmetric stochastic matrix P, let $\sigma(P)$ denote the set of eigenvalues of P. We will assume that

$$\sigma(P) = \{1, \lambda_1(P), \ldots, \lambda_{N-1}(P)\},\$$

where $1, \lambda_1(P), \ldots, \lambda_{N-1}(P)$ denote the eigenvalues of P and are such that $\lambda_1(P) \geq \lambda_2(P) \geq \ldots \geq \lambda_{N-1}(P)$. We define

$$\lambda_{\max}(P) = \lambda_1(P),$$

and

$$\lambda_{\min}(P) = \lambda_{N-1}(P).$$

Note that $\max\{|\lambda_1(P)|, |\lambda_{N-1}(P)|\}$ is the essential spectral radius of the matrix P.

Note that the algorithm (7.2) relies upon a crucial assumption: each agent transmits to its neighboring agents the precise value of its state. This implies the exchange of perfect information through the communication network. As in the previous two Chapters , we consider a more realistic case, i.e., we assume that the communication network is constituted only of rate-constrained digital links. As already emphasized, the presence of a rate constraint prevents the agents from having a precise knowledge about the state of the other agents. In fact, through a digital channel, the *i*-th agent can only send to the *j*-th agent symbolic data in a finite or countable alphabet; using only this data, the *j*-th agent can build at most an estimate of the *i*-th agent's state. The main objectives of this Chapter are to understand if there exists some smart way of coding/ decoding information through the digital channels that permits to overcome the forced quantization effects due to the digital channel and to modify the *ideal* standard average consensus algorithm into a scheme which both preserves the average of the state at each iteration and reaches asymptotically the average consensus.

To tackle this problem we take a two step approach. First, we introduce a coding/decoding scheme; each agent uses this scheme to estimate the positions of its neighbors. Second, we consider the standard consensus algorithm where, in place of the exact knowledge of the states of the systems, we substitute estimates calculated according to the proposed coding/decoding scheme.

7.3 Coder/decoder pairs for digital channels

In this section we discuss a general and two specific coder/decoder models for reliable digital channels; we follow the treatment in the survey [91]. We will later adopt this coder/decoder structure to define communication protocols in the robotic network.

Suppose a source wants to communicate to a receiver some time-varying data $x : \mathbb{N} \to \mathbb{R}$ via repeated transmissions at time instants in \mathbb{N} . Each transmission takes place through a digital channel, i.e., messages can only be symbols in a finite or countable set (to be designed). The channel is assumed to be reliable, that is, each transmitted symbol is received without error. A coder/decoder pair for a digital channel is defined by the sets:

- (i) a set Ξ , serving as *state space* for the coder/decoder; a fixed $\xi_0 \in \Xi$ is the *initial coder/decoder state*;
- (ii) a finite or countable set \mathcal{A} , serving as *transmission alphabet*; elements $\alpha \in \mathcal{A}$ are called message;

and by the maps:

- (i) a map $F: \Xi \times \mathcal{A} \to \Xi$, called the *coder/decoder dynamics*;
- (ii) a map $Q: \Xi \times \mathbb{R} \to \mathcal{A}$, being the quantizer function;

(iii) a map $H: \Xi \times \mathcal{A} \to \mathbb{R}$, called the *decoder function*.

The coder computes the symbols to be transmitted according to, for $t \in \mathbb{N}$,

$$\xi(t+1) = F(\xi(t), \alpha(t)), \quad \alpha(t) = Q(\xi(t), x(t)).$$

Correspondingly, the decoder implements, for $t \in \mathbb{N}$,

$$\xi(t+1) = F(\xi(t), \alpha(t)), \quad \hat{x}(t) = H(\xi(t), \alpha(t)).$$

Coder and decoder are jointly initialized at $\xi(0) = \xi_0$. Note that an equivalent representation for the coder is $\xi(t+1) = F(\xi(t), Q(\xi(t), x(t)))$, and $\alpha(t) = Q(\xi(t), x(t))$. In summary, the coder/decoder dynamics is given by

$$\xi(t+1) = F(\xi(t), \alpha(t)),
\alpha(t) = Q(\xi(t), x(t)),
\hat{x}(t) = H(\xi(t), \alpha(t)).$$
(7.3)

In what follows we present two interesting coder/decoder pairs: the logarithmic quantizer strategy and the "zoom in - zoom out" uniform quantizer strategy.

7.3.1 Zoom in - zoom out uniform coder

In this strategy the information transmitted from source to receiver is quantized by a scalar uniform quantizer which can be described as follows. For $L \in \mathbb{N}$, define the *uniform set of quantization levels*

$$S_L = \left\{ -1 + \frac{2\ell - 1}{L} \mid \ell \in \{1, \dots, L\} \right\} \cup \{-1\} \cup \{1\}$$

and the corresponding the uniform quantizer (see Figure 7.1) $\operatorname{unq}_L : \mathbb{R} \to S_L$ by

$$\operatorname{unq}_L(x) = -1 + \frac{2\ell - 1}{L}$$

if $\ell \in \{1, \ldots, L\}$ satisfies $-1 + \frac{2(\ell-1)}{L} \leq x \leq -1 + \frac{2\ell}{L}$, otherwise $\operatorname{unq}_L(x) = 1$ if x > 1 or $\operatorname{unq}_L(x) = -1$ if x < -1.

Note that larger values of the parameter L correspond to more accurate uniform quantizers unq_L . Moreover note that, if we define m to be the number of quantization levels we have that m = L + 2.

For $L \in \mathbb{N}$, $k_{in} \in [0, 1[$, and $k_{out} \in [1, +\infty[$, the zoom in - zoom out uniform coder/decoder has the state space $\Xi = \mathbb{R} \times \mathbb{R}_{>0}$, the initial state $\xi_0 = (0, 1)$, and the alphabet $\mathcal{A} = S_L$. The coder/decoder state is written as $\xi = (\hat{x}_{-1}, f)$ and the coder/decoder dynamics are

$$\hat{x}_{-1}(t+1) = \hat{x}_{-1}(t) + f(t)\alpha(t),$$

$$f(t+1) = \begin{cases} k_{\text{in}} f(t), & \text{if } |\alpha(t)| < 1, \\ k_{\text{out}} f(t), & \text{if } |\alpha(t)| = 1. \end{cases}$$

The quantizer and decoder functions are, respectively,

$$\alpha(t) = \operatorname{unq}_L\left(\frac{x(t) - \hat{x}_{-1}(t)}{f(t)}\right),$$
$$\hat{x}(t) = \hat{x}_{-1}(t) + f(t)\alpha(t).$$

The coder/decoder pair is analyzed as follows. One can observe that $\hat{x}_{-1}(t+1) = \hat{x}(t)$ for $t \in \mathbb{Z}_{\geq 0}$, that is, the first component of the coder/decoder state contains the estimate of the data x. The transmitted messages contain a quantized version of the estimate error $x - \hat{x}_{-1}$ scaled by factor f. Accordingly, the second component of the coder/decoder state f is referred to as the *scaling factor*: it grows when $|x - \hat{x}_{-1}| \geq f$ ("zoom out step") and it decreases when $|x - \hat{x}_{-1}| < f$ ("zoom in step").



Figure 7.1: The uniform quantizer (m = 6).



Figure 7.2: The logarithmic quantizer.

7.3.2 Logarithmic coder

This strategy is presented for example in [53]. Given an accuracy parameter $\delta \in [0, 1]$, define the logarithmic set of quantization levels

$$S_{\delta} = \left\{ \left(\frac{1+\delta}{1-\delta}\right)^{\ell} \right\}_{\ell \in \mathbb{Z}} \cup \{0\} \cup \left\{ - \left(\frac{1+\delta}{1-\delta}\right)^{\ell} \right\}_{\ell \in \mathbb{Z}},\tag{7.4}$$

and the corresponding *logarithmic quantizer* (see Figure 7.2) $\lg q_{\delta} : \mathbb{R} \to S_{\delta}$ by

$$\lg q_{\delta}(x) = \left(\frac{1+\delta}{1-\delta}\right)^{\ell},$$

if $\ell \in \mathbb{Z}$ satisfies $\frac{(1+\delta)^{\ell-1}}{(1-\delta)^{\ell}} \leq x \leq \frac{(1+\delta)^{\ell}}{(1-\delta)^{\ell+1}}$, otherwise $\lg q_{\delta}(x) = 0$ if x = 0 or $\lg q_{\delta}(x) = -\lg q_{\delta}(-x)$ if x < 0.

Note that smaller values of the parameter δ correspond to more accurate logarithmic quantizers $\lg q_{\delta}$. For $\delta \in]0, 1[$, the *logarithmic coder/decoder* is defined by the state space $\Xi = R$, initial state $\xi_0 = 0$, the alphabet $\mathcal{A} = S_{\delta}$, and by the maps

$$\xi(t+1) = \xi(t) + \alpha(t),$$

$$\alpha(t) = \lg q_{\delta}(x(t) - \xi(t)),$$

$$\hat{x}(t) = \xi(t) + \alpha(t).$$
(7.5)

The coder/decoder pair is analyzed as follows. One can observe that $\xi(t+1) = \hat{x}(t)$ for $t \in \mathbb{N}$, that is, the coder/decoder state contains the

estimate of the data x. The transmitted messages contain a quantized version of the estimate error $x - \xi$. The estimate $\hat{x} : N \to \mathbb{R}$ satisfies the recursive relation

$$\hat{x}(t+1) = \hat{x}(t) + \lg_{\delta} (x(t+1) - \hat{x}(t)),$$

with initial condition $\hat{x}(0) = \lg_{\delta}(x(0))$ determined by $\xi(0) = 0$. Finally, define the function $r : \mathbb{R} \to \mathbb{R}$ by $r(y) = \frac{\lg_{\delta}(y) - y}{y}$ for $y \neq 0$ and r(0) = 0. Some elementary calculations show that $|r(y)| \leq \delta$ for all $y \in \mathbb{R}$. Accordingly, if we define the trajectory $\omega : \mathbb{N} \to [-\delta, +\delta]$ by $\omega(t) = r(x(t+1) - \hat{x}(t))$, then we obtain that

$$\hat{x}(t+1) = \hat{x}(t) + (1+\omega(t))\big(x(t+1) - \hat{x}(t)\big).$$
(7.6)

This is called the *multiplicative noise* model for the logarithmic quantizer.

Remark 7.2 Note that, when communicating through digital channels, the use of the logarithmic quantizer described in the above Section, presents an evident drawback with respect to the zoom in- zoom out strategy, due to the fact that the logarithmic set of quantization levels S_{δ} is countable and not finite as the uniform set of quantization levels. This implementation issue could be overcome by truncating the map \lg_{δ} as follows. Let $a, b \in \mathbb{R}$ be such that 0 < a < b; if $a \leq |x| \leq b$ then

$$\begin{split} \lg q_{\delta}(x) &= \operatorname{sgn}(x) \left(\frac{1+\delta}{1-\delta}\right)^{\ell}, \\ \text{where } \ell \in \mathbb{Z} \text{ is such that } \frac{(1+\delta)^{\ell-1}}{(1-\delta)^{\ell}} \leq |x| \leq \frac{(1+\delta)^{\ell}}{(1-\delta)^{\ell+1}}, \text{ otherwise} \\ \lg q_{\delta}(x) &= \begin{cases} 0 & \text{if } |x| < a \\ \operatorname{sgn}(x) \lg q_{\delta}(b) & \text{if } |x| > M \end{cases} \end{split}$$

Again, if m denotes the number of quantization levels, it is possible to see (see [55]) that, for the truncated logarithmic quantizer,

$$m = \frac{2\log C}{\log\frac{1+\delta}{1-\delta}}$$

We will come back on this remark later on.

7.4 Consensus algorithm with exchange of quantized information

We consider now the same algorithm previously illustrated with the assumption that the agents can communicate only through digital channels. Precisely in this Section, we adopt the logarithmic coder/decoder scheme (7.3)

described in Subsection 7.3.2; we analyze the zoom in - zoom out strategy via simulations in Section 7.6.

Here is an informal description of our proposed scheme. We envision that along each communication edge we implement a logarithmic coder/decoder; in other words, each agent transmits through a dynamic encoding scheme to all its neighbors the quantized information regarding its position. Once state estimates of all node's neighbors are available, each node will then implement the average consensus algorithm.

Next, we provide a formal description of the proposed algorithm. Let $P \in \mathbb{R}^{N \times N}$ be a stochastic symmetric matrix with positive diagonal elements and with connected induced graph $\mathcal{G}_{\mathcal{P}}$. Assume there are digital communication channels along all edges of $\mathcal{G}_{\mathcal{P}}$ capable of carrying a countable number of symbols. Pick an accuracy parameter $\delta \in [0, 1[$. The consensus algorithm with dynamic coder/decoder is defined as follows:

- **Processor states:** For each $i \in \{1, \ldots, N\}$, node *i* has a state variable $x_i \in \mathbb{R}$ and state estimates $\hat{x}_j \in \mathbb{R}$ of the states of all neighbors *j* of *i* in $\mathcal{G}_{\mathcal{P}}$. Furthermore, node *i* maintains a copy of \hat{x}_i .
- **Initialization:** The state $x(0) = (x_1(0), \ldots, x_N(0))^* \in \mathbb{R}^N$ is given as part of the problem. All estimates $\hat{x}_j(0)$, for $j \in \{1, \ldots, N\}$, are initialized to 0.
- State iteration: At time $t \in \mathbb{N}$, for each *i*, node *i* performs three actions in the following order:
 - (1) Node i updates its own state by

$$x_i(t) = x_i(t-1) + \sum_{j=1}^{N} P_{ij} \left(\hat{x}_j(t-1) - \hat{x}_i(t-1) \right).$$
 (7.7)

(2) Node i transmits to all its neighbors the symbol

$$\alpha_i(t) = \lg q_\delta(x_i(t) - \hat{x}_i(t-1)).$$

(3) Node i updates its estimates

$$\hat{x}_j(t) = \hat{x}_j(t-1) + \alpha_j(t),$$
(7.8)

for j being equal to all neighbors of i and to i itself.

Before the algorithm analysis, we clarify a few points.

Remark 7.3 Robot *i* and all its neighbors *j* maintain in memory an estimate \hat{x}_i of the state x_i . We denote all these estimates by the same symbol because they are all identical: they are initialized in the same manner and they are updated through the same equation with the same information. On the other hand, it would be possible to adopt distinct quantizer accuracies δ_{ij} for each communication channel (i, j). In such a case then we would have to introduce variables \hat{x}_{ij} that node *i* and *j* would maintain for the estimate of x_i .

Remark 7.4 We could define a different state update equation where each node *i* uses the exact knowledge of its own state x_i instead of the estimate \hat{x}_i , that is, we could adopt

$$x_{i}(t) = x_{i}(t-1) + \sum_{j=1}^{N} P_{ij} \left(\hat{x}_{j}(t-1) - x_{i}(t-1) \right),$$

$$= P_{ii}x_{i}(t-1) + \sum_{j \neq i} P_{ij}\hat{x}_{j}(t-1),$$

(7.9)

instead of equation (7.7). We will discuss the drawback of this choice below.

We now analyze the algorithm. First, we write the closed-loop system in matrix form. Equation (7.7) is written as

$$x(t+1) = x(t) + (P - I)\hat{x}(t).$$
(7.10)

The N-dimensional vector of state estimates $\hat{x} = (\hat{x}_1, \dots, \hat{x}_N)^*$ is updated according to the multiplicative-noise model in equation (7.6). In other words, there exist $\omega_j \colon \mathbb{N} \to [-\delta, +\delta]$, for $j \in \{1, \dots, N\}$, such that

$$\hat{x}_j(t+1) = \hat{x}_j(t) + (1 + \omega_j(t)) \big(x_j(t+1) - \hat{x}_j(t) \big),$$

and, for $\Omega(t) := \text{diag } \{\omega_1(t), \ldots, \omega_N(t)\},\$

$$\hat{x}(t+1) = \hat{x}(t) + (I + \Omega(t)) \big(x(t+1) - \hat{x}(t) \big).$$
(7.11)

Equations (7.10) and (7.11) with multiplicative noise Ω determine the closed-loop system.

Next, we define the estimate error $e = \hat{x} - x \in \mathbb{R}^N$ and rewrite the closeloop system in terms of the quantities x and e. Straightforward calculations show that, for $t \in \mathbb{Z}_{\geq 0}$,

$$\begin{bmatrix} x(t+1)\\ e(t+1) \end{bmatrix} = \begin{bmatrix} I & 0\\ 0 & \Omega(t) \end{bmatrix} \begin{bmatrix} P & P-I\\ P-I & P-2I \end{bmatrix} \begin{bmatrix} x(t)\\ e(t) \end{bmatrix}.$$
 (7.12)

Initial conditions are x(0) and e(0) = -x(0).

Finally, we are ready to state the main properties of our quantized consensus algorithm.

Theorem 7.5 Assume $P \in \mathbb{R}^{N \times N}$ satisfies Assumption 7.1 and define $\bar{\delta} \in \mathbb{R}$ by

$$\bar{\delta} := \frac{1 + \lambda_{\min}(P)}{3 - \lambda_{\min}(P)}.$$
(7.13)

The solution $t \mapsto (x(t), e(t))$ of the consensus algorithm with dynamic coderdecoder has the following two properties:

(i) the state average is maintained constant by the algorithm, that is, defined $x_{ave}(t) = 1/N \mathbb{1}^* x(t)$,

$$x_{ave}(t) = x_{ave}(0)$$

for all $t \in \mathbb{N}$;

(ii) if $0 < \delta < \overline{\delta}$, then the state variables converge to their average value and the estimate error vanishes, that is,

$$\lim_{t \to \infty} x(t) = x_{ave}(0)\mathbb{1}$$

and

$$\lim_{t \to \infty} e(t) = 0.$$

Proof: Observe that

$$\mathbb{1}^* x(t+1) = \mathbb{1}^* P x(t) + \mathbb{1}^* (P - I) e(t)
= \mathbb{1}^* P x(t)$$

where the second equality holds since $\mathbb{1}^*(P-I) = 0$. This proves the first part of the Theorem. The second part follows directly from Theorem 7.11 stated in Section 7.4.1, where we analyze the asymptotic properties of (7.5).

We here consider some remarks and examples.

Remark 7.6 Note that $\bar{\delta}$ is a increasing function on $\lambda_{\min}(P)$ and that $\bar{\delta} = 0$, if $\lambda_{\min}(P) = -1$, and $\bar{\delta} = 1$, if $\lambda_{\min}(P) = 1$ (see Figure 7.3).



Figure 7.3: Behavior of $\overline{\delta}$.

Remark 7.7 The state update in equation (7.9) does not maintain the average. This fact motivates the choice of state update equation (7.7).

Example 7.8 Consider the sequence of circulant matrices $P_N \in \mathbb{R}^{N \times N}$ defined by

$$P_N = \begin{pmatrix} \frac{1}{3} & \frac{1}{3} & 0 & 0 & 0 & \frac{1}{3} \\ \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & 0 & 0 & 0 \\ 0 & \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & 0 & 0 & 0 \\ 0 & \frac{1}{3} & \frac{1}{3} & \frac{1}{3} & 0 & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ \frac{1}{3} & 0 & 0 & 0 & \cdots & 0 & \frac{1}{3} & \frac{1}{3} \end{pmatrix} .$$
(7.14)

For this sequence of symmetric stochastic matrices we know, from Chapter 3, that $\lambda_{\min}(P_N) = \frac{1}{3} - \frac{2}{3} \cos\left(\frac{2\pi}{N} \lfloor \frac{N}{2} \rfloor\right)$. Hence $\lambda_{\min}(P_N) \geq -\frac{1}{3}$, implying therefore that $\overline{\delta} \geq \frac{1}{5}$ for all N. This shows that $\overline{\delta}$ is uniformly bounded away from 0. This is a remarkable property of scalability on the dimension of the network.

Remark 7.9 The fact that the critical accuracy sufficient to guarantee convergence is independent on the network dimension is more general than what seen in the previous example. Indeed, assume that $P_N \in \mathbb{R}^{N \times N}$ is a sequence of matrices of increasing size, where each P_N satisfies Assumption 7.1 and where each P_N has all the diagonal elements greater than a positive real number \bar{p} . Then, by Gershgorin's Theorem we have that $\lambda_{\min}(P_N) \geq -1 + 2\bar{p}$

and hence $\bar{\delta} \geq \frac{\bar{p}}{2-\bar{p}}$ for all N. It follows that the critical accuracy sufficient to guarantee convergence is bounded away from zero uniformly on the dimension of the network.

7.4.1 Convergence analysis

In this section we provide the analysis of the asymptotic properties of system (7.12), by treating it as a LPV system. For the sake of the notational convenience, let us define the matrix, belonging to $\mathbb{R}^{2N \times 2N}$,

$$\mathcal{A}(t) = \begin{bmatrix} I & 0\\ 0 & \Omega(t) \end{bmatrix} \begin{bmatrix} P & P - I\\ P - I & P - 2I \end{bmatrix}.$$
 (7.15)

Consider now the system

$$z(t+1) = \mathcal{A}(t)z(t), \qquad (7.16)$$

where $z(t) \in \mathbb{R}^{2N}$ for all $t \geq 0$ and where z(0) is any vector in \mathbb{R}^{2N} . We start our analysis by rewriting (7.15) in a more suitable way. Let

$$\mathcal{E} = \left\{ \text{diag } \{e_1, \dots, e_N\} \in \mathbb{R}^{N \times N} : e_i \in \{-1, +1\}, i \in \{1, \dots, N\} \right\}.$$

Notice that \mathcal{E} contains 2^N elements. Hence, we can write $\mathcal{E} = \{E_1, \ldots, E_{2^N}\}$, where we are assuming that some suitable way to enumerate the matrices inside \mathcal{E} has been used. We assume that $E_1 = I$. We define now $\mathcal{E}_{\delta} = \{\delta E_1, \ldots, \delta E_{2^N}\}$. Observe that $\Omega(t) \in Co\{\mathcal{E}_{\delta}\}$ for all $t \geq 0$, where $Co\{\mathcal{E}_{\delta}\}$ denotes that convex hull of the set \mathcal{E}_{δ} . By means of the above definitions we can introduce an another set of matrices

$$\mathcal{R} = \left\{ R_i = \begin{bmatrix} I & 0 \\ 0 & \delta E_i \end{bmatrix} \begin{bmatrix} P & P - I \\ P - I & P - 2I \end{bmatrix} : E_i \in \mathcal{E} \right\}.$$
(7.17)

Accordingly to the definition of E_1 we have that

$$R_1 = \begin{bmatrix} I & 0 \\ 0 & \delta I \end{bmatrix} \begin{bmatrix} P & P - I \\ P - I & P - 2I \end{bmatrix}.$$
 (7.18)

The set \mathcal{R} is useful because it is easy to see that the matrix $\mathcal{A}(t)$, belongs to $Co\{\mathcal{R}\}$ for all $t \geq 0$, where $Co\{\mathcal{R}\}$ denote the convex hull of the set \mathcal{R} . In other words, for all $t \geq 0$, there exist $\nu_1(t), \ldots, \nu_{2^N}(t)$ nonnegative real numbers such that $\sum_{i=1}^{2^N} \nu_i(t) = 1$ and

$$\mathcal{A}(t) = \sum_{i=1}^{2^{N}} \nu_{i}(t) R_{i}.$$

We state the following result that will permit us to analyze the system (7.16) by means of Theorem C.2 (see Appendix C).

Lemma 7.10 For $v = [1^* \ 0^*]^*$, we have

$$R_i v = v$$
, and $v^* R_i = v^*$, for all $i \in \{1, \dots, 2^N\}$.

Moreover, for $\overline{\delta}$ as in equation (7.13), the following facts are equivalent:

- (i) 1 is the only eigenvalue of unit magnitude of the matrix R_1 , and all its other eigenvalues are strictly inside the unit disc;
- (*ii*) $0 \leq \delta < \overline{\delta}$.

Proof: The first part of the lemma is easily proved by observing that

$$\begin{bmatrix} I & 0 \\ 0 & \delta E_i \end{bmatrix} \begin{bmatrix} P & P-I \\ P-I & P-2I \end{bmatrix} \begin{bmatrix} 1 \\ 0 \end{bmatrix} = \begin{bmatrix} I & 0 \\ 0 & \delta E_i \end{bmatrix} \begin{bmatrix} 1 \\ 0 \end{bmatrix} = \begin{bmatrix} 1 \\ 0 \end{bmatrix},$$

and

$$\begin{bmatrix} \mathbf{1}^* & \mathbf{0}^* \end{bmatrix} \begin{bmatrix} I & 0\\ 0 & \delta E_i \end{bmatrix} \begin{bmatrix} P & P-I\\ P-I & P-2I \end{bmatrix} = \begin{bmatrix} \mathbf{1}^* & \mathbf{0}^* \end{bmatrix} \begin{bmatrix} P & P-I\\ P-I & P-2I \end{bmatrix}$$
$$= \begin{bmatrix} \mathbf{1}^* & \mathbf{0}^* \end{bmatrix}.$$

Consider now R_1 ; to compute its eigenvalues we calculate

$$\det (sI - R_1) = \det \begin{bmatrix} sI - P & -(P - I) \\ -\delta(P - I) & sI - \delta(P - 2I) \end{bmatrix}$$

Since each block of the above matrix commute with each other block, we have from [124] that

$$\det (sI - R_1) = \det \left[(sI - P)(sI - \delta(P - 2I)) - \delta(P - I)^2 \right] = \det \left[s^2I - s\left(\delta(P - 2I) + P\right) + \delta\left(P^2 - 2P - P^2 - I + 2P\right) \right] = \prod_{i=0}^{N-1} \left[s^2 - \left(\delta(\lambda_i - 2) + \lambda_i\right) s - \delta \right] = \left(s^2 - (1 - \delta)s - \delta \right) \prod_{i=1}^{N-1} \left(s^2 - \left(\delta(\lambda_i - 2) + \lambda_i\right) s - \delta \right).$$

Hence the eigenvalues of R_1 are given by the solution of the following N second order equations

$$s^{2} - (1 - \delta)s - \delta = 0, \qquad (7.19)$$

and

$$s^{2} - (\delta(\lambda_{i} - 2) + \lambda_{i})s - \delta = 0, \qquad i \in \{1, \dots, N - 1\}.$$
(7.20)

The solutions of (7.19) are 1 and $-\delta$. Consider now (7.20). Given *i*, let $s_1^{(i)}$ and $s_2^{(i)}$ denote the two solutions of (7.20). We have that

$$s_1^{(i)} = \frac{\delta(\lambda_i - 2) + \lambda_i - \sqrt{(\delta(\lambda_i - 2) + \lambda_i)^2 + 4\delta}}{2}$$

and

$$s_2^{(i)} = \frac{\delta(\lambda_i - 2) + \lambda_i + \sqrt{(\delta(\lambda_i - 2) + \lambda_i)^2 + 4\delta}}{2}$$

Now we have to analyze the conditions $|s_1^{(i)}| < 1$ and $|s_2^{(i)}| < 1$, for all $i \in \{1, \ldots, N-1\}$. To this purpose, we consider the bilinear transformation of the equation (7.20), i.e., we substitute to s the term $\frac{1+\tilde{s}}{1-\tilde{s}}$. We obtain the new equation

$$(1+\delta)(1-\lambda_i)\tilde{s}^2 + 2(1+\delta)\tilde{s} + 1 + \lambda_i + \delta(\lambda_i - 3) = 0.$$
 (7.21)

Let $\tilde{s}_1^{(i)}$ and $\tilde{s}_2^{(i)}$ denote the two solutions of (7.21). From the property of the bilinear transformation, we have that $|s_1^{(i)}| < 1$ and $|s_2^{(i)}| < 1$ if and only if $\tilde{s}_1^{(i)} < 0$ and $\tilde{s}_2^{(i)} < 0$. Since $1 + \delta > 0$ and $(1 + \delta)(1 - \lambda_i) > 0$ for $i \in \{1, \ldots, N - 1\}$, we obtain, from the Cartesian rule, that $\tilde{s}_1^{(i)} < 0$ and $\tilde{s}_2^{(i)} < 0$ for all $i \in \{1, \ldots, N - 1\}$, if and only if $1 + \lambda_i + \delta(\lambda_i - 3) > 0$ for all $i \in \{1, \ldots, N - 1\}$. This last condition is verified if and only if $\delta < \overline{\delta}$.

We are able now to state the following theorem characterizing the asymptotic stability of the system (7.16).

Theorem 7.11 Consider the system (7.16). The following facts are equivalent:

- (a) $\delta < \bar{\delta}$;
- (b) for each initial condition $z(0) \in \mathbb{R}^{2N}$ and for any sequence $\{\Omega(t)\}_{t=0}^{+\infty}$ with $\Omega(t) \in Co\{\mathcal{E}_{\delta}\}$ for all $t \geq 0$, we have

$$\lim_{t \to +\infty} z(t) = \begin{bmatrix} \alpha \mathbb{1} \\ 0 \end{bmatrix}, \qquad (7.22)$$

for $\alpha = \frac{1}{N} [1^* \ 0^*] z(0).$

Proof: We start by proving that (b) implies (a). To this aim, we consider the sequence $\mathcal{A}(0) = \mathcal{A}(1) = \mathcal{A}(2) = \ldots = R_1$. In this case z(t) is the evolution of an autonomous linear time invariant discrete-time systems with updating matrix R_1 . Therefore, by Lemma 7.10, (7.22) holds true if and only if and only if $\delta < \overline{\delta}$.

We prove now that (a) implies (b). We will show that, for $\delta < \overline{\delta}$, there exists a suitable symmetric matrix $L \in \mathbb{R}^{2N \times 2N}$ satisfying the following three properties

$$L \left[\mathbb{1}^* \ \mathbf{0}^* \right]^* = 0, \tag{7.23}$$

$$z^*Lz > 0,$$
 (7.24)

$$z^*\left(\frac{1}{2}\left(R_i^*LR_j + R_j^*LR_i\right) - L\right)z < 0, \quad \text{for all } R_i, R_j \in \mathcal{R}, \qquad (7.25)$$

 $\forall z \in < [1^* 0^*]^* >^{\perp}$. This fact, together with Lemma 7.10 and Theorem C.2, ensures that fact (a) implies (b). As candidate matrix L we select

$$L = \begin{bmatrix} I - P & 0\\ 0 & \gamma I \end{bmatrix}, \tag{7.26}$$

where γ is a suitable positive scalar to be determined. Observe that the eigenvalues of I - P are 0 and $1 - \lambda_i$ for $i \in \{1, \ldots, N - 1\}$, where it is immediate to see that $1 - \lambda_i > 0$ for $i \in \{1, \ldots, N - 1\}$. Since $\sigma(L) = \sigma(I-P) \cup \sigma(\gamma I)$ it follows that also L has an eigenvalue equal to 0 and all other eigenvalues positive. Moreover, since $L [\mathbb{1}^* \mathbf{0}^*]^* = [((I - P)\mathbb{1})^* \mathbf{0}^*]^* = 0$, we have that the eigenspace associated to the eigenvalue 0 is generated by the vector $[\mathbb{1}^* \mathbf{0}^*]^*$. Hence L satisfies (7.23) and (7.24). Moreover, by the structure of L, it is easy to check that $R_i^* L R_j = R_j^* L R_i$ for all $R_i, R_j \in \mathcal{R}$. Thus, verifying (7.25) is equivalent to verify

$$z^* \left(R_i^* L R_j - L \right) z < 0, \quad \text{for all } R_i, R_j \in \mathcal{R}, \tag{7.27}$$

for any nonzero $z \in < [\mathbb{1}^* \ \mathbb{0}^*]^* >^{\perp}$. We have that

$$R_i^* L R_j - L = R_1 L R_1 - L + Q,$$

where

$$R_1^*LR_1 - L = = \begin{bmatrix} (I-P)^2(\gamma\delta^2 I - I - P) & (I-P)(P(P-I) - \gamma\delta^2(P-2I)) \\ (I-P)(P(P-I) - \gamma\delta^2(P-2I)) & (I-P)^3 + \gamma\delta^2(P-2I)^2 - \gamma I \end{bmatrix}$$

$$\begin{split} Q &= -\gamma \delta^2 \begin{bmatrix} (P-I)K^2(P-I) & (P-I)K^2(P-2I) \\ (P-2I)(P-I) + (P-2I)K^2(P-I) & (P-2I)K^2(P-2I) \end{bmatrix} \\ &= -\gamma \delta^2 \begin{bmatrix} (P-I)K \\ (P-2I)K \end{bmatrix} \left[K(P-I) & K(P-2I) \right], \end{split}$$

with K such that $K^2 = I - E_i E_j^2$. Clearly, $Q = Q^* \leq 0$ and $Q [\mathbb{1}^* \ \mathbf{0}^*]^* = 0$. If (7.27) is satisfied for i = j = 1, then (7.27) holds also for any pair R_i, R_j belonging to \mathcal{R} .

Observe that, by Lemma C.4 (see Appendix C), we immediately have that

$$z^*(R_1^*LR_1 - L)z < 0, \quad \forall z \in < [\mathbb{1}^* \ \mathbf{0}^*]^* >^{\perp}$$
 (7.28)

if we choose

and

$$\gamma = \frac{1 + \lambda_{\min} + \delta^2 \left(\lambda_{\min} - 3\right)}{2\delta^2}$$

7.5 Exponential convergence

The objective of this section is to understand how much the quantization affects the performance of the consensus algorithm. To this aim, by means of a Lyapunov analysis, we will provide a characterization of the asymptotic speed of the convergence toward the consensus of both the ideal algorithm (7.2) and the algorithm (7.12).

We start by introducing some definitions. A function $f : \mathbb{N} \to \mathbb{R}$ converges to 0 exponentially fast if there exist a constant C > 0 and another constant $\xi \in [0, 1[$ such that $|f(t)| \leq C\xi^t$, for all t; the infimum among all numbers $\xi \in [0, 1[$ satisfying the exponential convergence property is called the exponential convergence factor of f. In other words, the exponential convergence factor of f is given by (see Section 2.2.2)

$$\limsup_{t \to \infty} |f(t)|^{\frac{1}{t}}.$$

Consider first the system (7.2). To quantify the speed of convergence of (7.2) toward consensus, we introduce the following variable

$$\bar{x}(t) := x(t) - x_{ave}(0)\mathbb{1},$$

 $^{^2 {\}rm Of}$ note is that $I-E_i E_j$ is a positive semidefinite matrix and hence the matrix K is well-defined.

where $x_{ave}(0) = \frac{1}{N} \mathbb{1}^* x(0)$. Note that the *i*-th component of $\bar{x}(t)$ represents the distance of the state of the *i*-th system from the initial average. Clearly, $\lim_{t\to\infty} x(t) = x_{ave}(0)\mathbb{1}$ if and only if $\lim_{t\to\infty} \bar{x}(t) = 0$. It is easy to see that the variable \bar{x} satisfies the same recursive equation of the variable x, that is,

$$\bar{x}(t+1) = P\bar{x}(t).$$
 (7.29)

Moreover note that $\mathbb{1}^* \bar{x}(t) = 0$, for all $t \ge 0$. We define the exponential convergence factor of $\bar{x}(t)$, for a given initial condition $\bar{x}_0 \in <\mathbb{1} >^{\perp}$, to be

$$\rho(P, \bar{x}_0) := \limsup_{t \to \infty} ||\bar{x}(t)||^{\frac{1}{t}}$$

We can get rid of the initial condition and define the *exponential convergence* factor of the system (7.2) as follows

$$\rho(P) := \sup_{\bar{x}_0 \in <\mathbb{1}>^{\perp}} \rho(P, \bar{x}_0)$$
(7.30)

Consider now the positive semidefinite matrix I - P. Notice that

$$\rho(P, \bar{x}_0) = \limsup_{t \to \infty} (\bar{x}(t)^* (I - P) \bar{x}(t))^{\frac{1}{2t}}$$

and so we can characterize the speed of convergence to 0 of the variable \bar{x} by studying the exponential convergence factor of the Lyapunov function $\bar{x}(t)^*(I-P)\bar{x}(t)$.

Theorem 7.12 Consider (7.29) with $P \in \mathbb{R}^{N \times N}$ satisfing Assumption 7.1. Then the function $t \mapsto (\bar{x}(t)^*(I-P)\bar{x}(t))^{1/2}$, defined along any trajectory $t \mapsto \bar{x}(t)$, converges exponentially fast to 0. Moreover, the factor $\rho(P)$, defined in equation (7.30), satisfies

$$\rho(P) = \max\left\{\lambda_{\max}(P), -\lambda_{\min}(P)\right\}.$$

Proof: Let $\alpha := \max \{\lambda_{\max}^2(P), \lambda_{\min}^2(P)\}$ so that $z^*P^2z \leq \alpha z^*z$ for all $z \in <\mathbb{1} >^{\perp}$ and, in turn,

$$z^*(P(I-P)P)z \le \alpha z^*(I-P)z,$$
 (7.31)

for all $z \in \langle 1 \rangle^{\perp}$. This shows that the map $t \mapsto \bar{x}(t)^*(I-P)\bar{x}(t)$ converges exponentially fast to 0 along any trajectory $t \mapsto \bar{x}(t)$ and that $\rho(P) \leq \sqrt{\alpha}$. Moreover, observe that, if z is equal to the eigenvector corresponding to the eigenvalue defining β , then (7.31) holds true as equality. Then, if \bar{x}_0 is equal to this eigenvector, we obtain a trajectory $t \mapsto \bar{x}(t)$ along which the function $t \mapsto \bar{x}(t)^*(I-P)\bar{x}(t)$ has exponential convergence factor equal to $\sqrt{\alpha}$.

This concludes the analysis of the algorithm (7.2). In the sequel of this section, we provide a similar analysis of the system (7.12). To this aim we consider again the system (7.16), that is

$$z(t+1) = \mathcal{A}(t)z(t), \qquad (7.32)$$

where $z(0) = z_0$ is any vector in \mathbb{R}^{2N} . To perform a Lyapunov analysis of (7.32), it is convenient to introduce the variable

$$\bar{z}(t) = \begin{bmatrix} I - \frac{1}{N} \mathbb{1} \mathbb{1}^* & 0\\ 0 & I \end{bmatrix} z(t).$$

Clearly, condition (b) of Theorem 7.11 holds true if and only if $\lim_{t\to\infty} \bar{z}(t) = 0$. It is straightforward to see that \bar{z} satisfies the same recursive equation of z(t), i.e.,

$$\bar{z}(t+1) = \mathcal{A}(t)\bar{z}(t) \tag{7.33}$$

and that $[\mathbb{1}^* \ \mathbf{0}^*]^* \bar{z}(t) = 0$ for all $t \ge 0$. Consider now the matrix $L \in \mathbb{R}^{2N \times 2N}$, introduced along the proof of Theorem 7.11 and defined as

$$L = \begin{bmatrix} I - P & 0 \\ 0 & \gamma I \end{bmatrix}.$$

For each $\gamma > 0$ define

$$\tilde{\rho}\left(P,\delta,\gamma;\bar{z}_{0},\left\{\mathcal{A}(t)\right\}_{t=0}^{\infty}\right) := \limsup_{t\to\infty} (\bar{z}(t)^{*}L\bar{z}(t))^{\frac{1}{2t}}$$
(7.34)

We can get rid of the initial conditions \bar{z}_0 and the sequences $\{\mathcal{A}(t)\}_{t=0}^{\infty}$ by considering

$$\tilde{\rho}(P,\delta,\gamma) := \sup_{\bar{z}_0\{\mathcal{A}(t)\}_{t=0}^{\infty}} \tilde{\rho}\left(P,\delta,\gamma; \bar{z}_0, \{\mathcal{A}(t)\}_{t=0}^{\infty}\right)$$
(7.35)

where the initial conditions \bar{z}_0 belong to the set of vectors orthogonal to $[\mathbb{1}^* \ \mathbf{0}^*]^*$ and the sequences $\{\mathcal{A}(t)\}_{t=0}^{\infty}$ are such that $\mathcal{A}(t) \in Co\{\mathcal{R}\}$ for all $t \geq 0$. It can be shown that $\tilde{\rho}(P, \delta, \gamma)$ is independent of γ and for this reason we denote it as $\tilde{\rho}(P, \delta)$.

We characterize now $\tilde{\rho}(P, \delta, \gamma)$. To this aim, consider the following semidefinite programming problem

$$\bar{\beta}(P,\delta,\gamma) := \frac{\max \ \beta}{\text{such that } R_1^* L R_1 - L \le -\beta L}$$
(7.36)

We have the following result.

Theorem 7.13 Consider (7.33) with the matrix P satisfing Assumption 7.1. Let $\bar{\delta}$ be defined as in (7.13) and let $\delta \in \mathbb{R}$ be such that $0 \leq \delta < \bar{\delta}$. Moreover let $\gamma \in \mathbb{R}$ be such that $\gamma > 0$, and let $\bar{\beta}(P, \delta, \gamma)$ be defined as in (7.36). Then, the function $t \to (\bar{z}(t)^* L \bar{z}(t))^{1/2}$, defined along any trajectory $t \to \bar{z}(t)$ converges exponentially fast to 0 and the factor $\tilde{\rho}(P, \delta)$, defined in equation (7.35), satisfies

$$\tilde{\rho}(P,\delta) \leq \sqrt{1 - \bar{\beta}(P,\delta,\gamma)}.$$

Proof: We start by recalling, that since $\mathcal{A}(t)$ belongs to $Co(\mathcal{R})$, we can write that $\mathcal{A}(t) = \sum_{i=1}^{2^N} \nu_i(t) R_i$, where $\nu_1(t), \ldots, \nu_{2^N}(t)$ are nonnegative real numbers such that $\sum_{i=1}^{2^N} \nu_i = 1$. Along the proof of Theorem 7.11, we have seen that

$$z^*(R_i^*LR_j - L)z \le z^*(R_1^*LR_1 - L)z < 0,$$

for all $z \in \mathbb{R}^{2N}$ such that $z \in \langle [\mathbb{1}^* \ \mathbf{0}^*]^* \rangle^{\perp}$ and for any pair of matrices R_i, R_j belonging to \mathcal{R} . Hence we have that

$$z^{*}(\mathcal{A}^{*}(t)L\mathcal{A}(t) - L)z = z^{*} \left(\left(\sum_{i=1}^{2^{N}} \nu_{i}(t)R_{i} \right)^{*} L \left(\sum_{j=1}^{2^{N}} \nu_{j}(t)R_{j} \right) - L \right) z$$

$$= z^{*} \left(\sum_{i=1}^{N} \sum_{j=1}^{N} \left(\nu_{i}(t)\nu_{j}(t)R_{i}^{*}LR_{j} - \nu_{i}(t)\nu_{j}(t)L \right) \right) z$$

$$\leq z^{*} \left(\sum_{i=1}^{N} \sum_{j=1}^{N} \nu_{i}(t)\nu_{j}(t)(R_{1}^{*}LR_{1} - L) \right) z$$

$$= z^{*}(R_{1}^{*}LR_{1} - L)z,$$

for all $z \in \mathbb{R}^{2N}$ such that $z \in \langle [\mathbb{1}^* \ \mathbf{0}^*]^* \rangle^{\perp}$. Observe finally that $z^*(R_1^*LR_1 - L)z \leq \bar{\beta}z^*Lz < 0$, from which we can argue that $\bar{z}(t+1)^*L\bar{z}(t+1) \leq (1-\bar{\beta})\bar{z}(t)^*L\bar{z}(t)$ and so the theses follow.

It is worth noting that the above Theorem relates $\tilde{\rho}(P, \delta)$ to the resolution of a LMI [21]. It is well known that the computational effort required by the resolution of a LMI strictly depends on its dimensionality. However, we can observe that Lemma C.3 (see Appendix C) provides an efficient way of solving (7.36), that drastically reduces its computational complexity. Indeed, we have that $\bar{\beta}(P, \delta, \gamma) = \min\{\beta_{\min}^{-1}(\delta, \gamma), \beta_{\max}^{-1}(\delta, \gamma)\}$, where $\beta_{\min}^{-1}(\delta, \gamma), \beta_{\max}^{-1}(\delta, \gamma)$ are defined in Lemma C.3. This means that one has to calculate only the value of the two variables $\beta_{min}^{-}(\delta, \gamma), \beta_{max}^{-}(\delta, \gamma)$ and evaluate the minimum between them. Differently from the method based on the LMI, the complexity of this method is independent of N.

Example 7.14 In this example we consider a connected random geometric graph generated by choosing N = 30 points at random in the unit square, and then placing an edge between each pair of pints at distance less than 0.4. The matrix P is built using the Metropolis weights [149]. In this case we have that $\lambda_{min} = -0.013$ and $\bar{\delta} = 0.327$. In figure 7.4, we plot the behavior of β_{min}^- and β_{max}^- as functions of γ . The value of δ is assumed constant and precisely equal to 0.25.



Figure 7.4: Behavior of $\tilde{\rho}$ as function of γ for P and δ fixed.

In general, assigned the matrix P and the value of the accuracy parameter δ , one could be interested in determining the maximum value of $\bar{\beta}$, as function of γ . Clearly, the best bound on $\tilde{\rho}(P, \delta)$ corresponds to to the maximum value of $\bar{\beta}$, namely

$$\tilde{\rho}(P,\delta) \le \sqrt{1 - \bar{\beta}_{opt}(P,\delta)}$$

where

$$\bar{\beta}_{opt}(P,\delta) := \max_{\gamma>0} \bar{\beta}(P,\delta,\gamma).$$

We illustrate this discussion in the following example.



Figure 7.5: Behavior of $\sqrt{1 - \bar{\beta}_{opt}(P, \delta)}$.

Example 7.15 We consider the same matrix P generated in the previous example. In Figure 7.5, we depict the behavior of $\sqrt{1 - \overline{\beta}_{opt}(P, \delta)}$ as a function of δ . The dotted line represents the value of $\rho(P)$, that is, the convergence factor of the ideal algorithm (7.29). Notice that the convergence factor $\sqrt{1 - \overline{\beta}_{opt}(P, \delta)}$ depends smoothly on the accuracy parameter δ and that

$$\lim_{\delta \to 0} \sqrt{1 - \bar{\beta}_{opt}(P, \delta)} = \rho(P).$$

An interesting characterization of $\tilde{\rho}$ can be provided when considering a family of matrices $\{P_N\}$ of increasing size whose maximum eigenvalue converges to 1. It is worth noting that this situation is encountered in many practical situations [86, 22, 35]. We formalize this situation as follows.

Assumption 7.16 (Vanishing spectral gap) Assume we have a sequence of symmetric stochastic matrices $\{P_N\} \subset \mathbb{R}^{N \times N}$ satisfying Assumption 7.1 and the following conditions

- (i) $\lambda_{\min}(P_N) > c$ for some $c \in [-1, 1[$ and for all $N \in \mathbb{N}$;
- (ii) $\lambda_{\max}(P_N) = 1 \epsilon(N) + o(\epsilon(N))$ as $N \to \infty$, where $\epsilon : \mathbb{N} \to \mathbb{R}$ is a positive function such that $\lim_{N\to\infty} \epsilon(N) = 0$.

According to Theorem 7.12, as $N \to \infty$, we have that $\rho(P_N) = 1 - \epsilon(N) + o(\epsilon(N))$. In considering the quantized version of the consensus algorithm,

together with the sequence $\{P_N\}$, we have also to fix the sequence $\{\delta_N\}$. For simplicity, in the following we will assume that, $\{\delta_N\}$ is a constant sequence, i.e., $\delta_N = \delta$ with suitable δ such that $\delta < \frac{1+c}{3-c}$ which ensures the stability for all N.

Theorem 7.17 Let $\{P_N\} \subset \mathbb{R}^{N \times N}$ be a family of matrices of increasing size satisfying Assumptions 7.1 and 7.16. Let $\delta \in \mathbb{R}$ be such that $\delta < \frac{1+c}{3-c}$. Then, as $N \to \infty$, we have that

$$\tilde{\rho}(P_N,\delta) \le 1 - \left(1 - \frac{1 + c + \delta^2(c - 3)}{4(1 - \delta^2)}\right)\epsilon(N) + o(\epsilon(N)).$$

Proof: We choose

$$\gamma = \frac{1 + c + \delta^2 \left(c - 3 \right)}{2\delta^2}.$$

Consider the polynomial f defined in (C.8). Let $\beta_{min}^{-}(\delta, \gamma, N)$ and $\beta_{max}^{-}(\delta, \gamma, N)$ be as defined in Lemma C.3 relatively to the matrix P_N .

Notice that $f(1, \delta, \gamma, \beta) = \gamma \beta^2 + (\gamma \delta^2 - \gamma) \beta$. Then the equation $f(1, \delta, \gamma, \beta) = 0$ has solutions $\beta = 0$ and $\beta = 1 - \delta^2$. This implies that, since $\lambda_{max}(P_N) \to 1$, then $\beta_{max}^-(\delta, \gamma, N) \to 0$ as $N \to \infty$. This implies that for N big enough we have that

$$\min\{\beta_{\min}^{-}(\delta,\gamma,N),\beta_{\max}^{-}(\delta,\gamma,N)\}=\beta_{\max}^{-}(\delta,\gamma,N)$$

and hence, from Theorem 7.13 and Lemma C.3 (see Appendix C), it follows that for N big enough we have that

$$\tilde{\rho}(P_N, \delta) \le \sqrt{1 - \beta_{max}^-(\delta, \gamma, N)}$$

Let $\lambda_N := \lambda_{max}(P_N)$ and $\beta_N := \beta_{max}^-(\delta, \gamma, N)$ so the we have that $\lambda_N \to 1$ and $\beta_N \to 0$. We know that $f(\lambda_N, \delta, \gamma, \beta_N) = 0$. As $N \to \infty$, from the implicit function theorem, we have that

$$\beta_N = \left[\frac{\frac{\partial}{\partial\lambda}f}{\frac{\partial}{\partial\beta}f}\right]_{|\lambda=1,\beta=0} \epsilon(N) + o(\epsilon(N)).$$

Now notice that

$$\frac{\partial f}{\partial \lambda} = \left(-3(1-\lambda)^2 + 2\gamma\delta^2(\lambda-2)^2 - \gamma^2\delta^2 + 2\gamma\delta)\right)\beta + \gamma^2\delta^2 + (1-\lambda)^2 + \gamma(1+\lambda+\delta^2(\lambda-3)) + (1-\lambda)(\gamma-2\gamma\delta^2 + 2(1-\lambda))$$

and that

$$\frac{\partial f}{\partial \beta} = 2\gamma\beta + (1-\lambda)^3 + \gamma\delta^2(\lambda-2)^2 - \gamma + \gamma(1-\lambda)(\gamma\delta^2 - 1 - \lambda).$$

which lead to

$$\frac{\partial f}{\partial \lambda}_{|\lambda=1,\beta=0} = -(2\gamma - 2\gamma^2 \delta^2 - \gamma \delta^2)$$

and

$$\frac{\partial f}{\partial \beta}_{|\lambda=1,\beta=0} = \gamma \delta^2 - 1.$$

Then

$$\beta_N = \left(2 - \frac{\gamma \delta^2}{1 - \delta^2}\right) \epsilon(N) + o(\epsilon(N)).$$

The thesis follows by expanding in Taylor's series the function $\sqrt{1-\beta_N}$.

Notice that the coefficient in front of $\epsilon(N)$ is negative. Indeed, it can be seen that that coefficient is negative if and only if

$$\delta^2 < \frac{3-c}{1+c}$$

and this is true since we have chosen $\delta < \frac{1+c}{3-c}$ and since $\delta < 1$.

7.6 Numerical simulations

In this section we consider two examples providing some numerical results illustrating the performance respectively of the Zoom in -Zoom out strategy and of the truncated version of the logarithmic quantizer discussed in Remark 7.2.

Example 7.18 In this example we consider a connected random geometric graph generated by choosing N points at random in the unit square, and then placing an edge between each pair of points at distance less than 0.25. We assume that N = 30 and that the initial conditions has been generated randomly inside the interval [-100, 100]. Again, the matrix P is built using the Metropolis weights (see Section 2.2.1). For all the experiments, we set the parameters k_{in} and k_{out} to the values 1/2 and 2 respectively, and initialized the scaling factor f of each agent to the value 50. Moreover we run simulations for two different values of m, m = 5 and m = 10. The results

obtained are reported in Figure 7.6. The variable plotted is the normalized Euclidean norm of the vector $\bar{x}(t) := x(t) - x_{ave}(0)\mathbb{1}$, that is,

$$s(t) = \sqrt{\frac{1}{N} \sum_{i=1}^{N} \bar{x}_i^2(t)}.$$

Note that, as depicted in Figure 7.6, also the zoom in- zoom out uniform coder- decoder strategy seems to be very efficient in achieving the consensus. In particular it is remarkable that this strategy works well even if the uniform quantizer has a low number of quantization levels (m = 5). Finally it is worth observing, that as theoretically proved in the logarithmic coder-decoder strategy, also in this case the performance degrades smoothly as the quantization becomes coarser.



Figure 7.6: Zoom in- zoom out strategy

It is worth noting that some preliminary results regarding the convergence of the zoom in- zoom out strategy are present in [11].

Example 7.19 In this example we consider the same matrix P of the previous example. Moreover we assume again that the initial conditions have been generated randomly inside the interval [-100, 100]. The information exchanged between the systems is quantized by the *truncated* logarithmic quantizer discussed in Remark 7.2. More precisely, we assume that the real numbers a, b introduced in Remark 7.2 are equal respectively to 0.5 and 100. The result obtained is reported in Figure 7.7.



Figure 7.7: Truncated logarithmic quantizer

The variable plotted is

$$d(t) := \|\bar{x}(t)\|_{\infty}.$$

One can see that d(t) does not converge asymptotically to 0. However, at the steady state, d(t) oscillates inside an interval whose amplitude is comparable to 0.5, that is, the lower value at which we have truncated the logarithmic quantizer.

This numerical observations leads to the following consideration. Assume that our goal is to have convergence of the initial states $x_i(0) \in [-M, M]$ to a target configuration $x_i(\infty) \in [\alpha - \epsilon, \alpha + \epsilon]$, where α is a constant depending only on the initial condition x(0) and ϵ describes the desired agreement precision. This is a "practical stability" requirement. In this case the contraction rate is $C := M/\epsilon$. Assume that, as in [75], the exact data transmission are substituted by transmissions of precision ϵ uniformly quantized data. In this framework it is well known [56] that each uniform quantizer needs C different levels and so the transmission of its data needs an alphabet of C different symbols. Assume now that the information is encoded by truncated logarithmic quantizers where $a = \epsilon$ and b = M. We have seen in Remark 7.2 that in such case each logarithmic quantizer needs

$$\frac{2\log C}{\log\frac{1+\delta}{1-\delta}}$$

different symbols. Note that for C sufficiently large, with the logarithmic communications we obtain a significantly improvement in terms of the communication effort required. It will be the subject of future research to analyze

the tradeoff between the steady state of d(t) and the values of the parameters a, b at which we truncate the logarithmic quantizers.

7.7 Conclusions

In this chapter we presented a new approach solving the average consensus problem in presence of only quantized exchanges of information. In particular we considered two strategies, one based on logarithmic quantizers, and the other one based on a zooming in-zooming out strategy. We studied them with theoretical and experimental results proving that using these schemes the average consensus problem can be efficiently solved even if the agents can share only quantized information. Additionally, we show that the convergence factors depend smoothly on the accuracy parameter of the quantized and that, remarkably, that the critical quantizer accuracy sufficient to guarantee convergence is independent from the network dimension. A field of future research will be to look for encoding and decoding methods which are able to solve the average problem also with noisy digital channels.

Chapter 8

Distributed Kalman filtering

8.1 Introduction

The main objective of this chapter deals with a possible application of the consensus ideas to the wide field of the distributed estimation. It is worth mentioning that, since eighties, the problems related to the distributed estimation have attracted intensively the attention of the scientific community producing, along the years, a very rich literature. Nevertheless in the last period, the interest toward this kind of problems is renewing. This is mainly due to the recent technological advances in wireless communication that, combined to the decreasing in cost and size of electronic devices, are promoting the appearance of large inexpensive interconnected systems, each with computational and sensing capabilities. These complex systems of agents, for instance, can be used for monitoring very large scale areas with fine resolution. However, in this application, collecting measurements from distributed wireless sensors nodes at a single location for on-line data processing may not be feasible due to several reasons among which long packet delay (e.g. due to multi-hop transmission) and/or limited bandwidth of the wireless network, due e.g. to energy consumption requirements.

This problem is particularly relevant in wireless ad-hoc sensor networks where information needs to be multi-hopped from one node to another using closer neighbors. Therefore there is a growing need for in-network data processing tools and algorithms that provide high performance in terms of on-line estimation while

(i) reducing the communication load among all sensor nodes,

- (ii) being very robust to sensor node failures or replacements and packet losses, and
- (iii) being suitable for distributed control applications.

As previously said, the literature is very rich of contributions addressing several aspects of distributed estimation. We can classify these works into few large classes based on the modeling adopted: *static vs dynamic* estimation, *distributed vs hierarchical* estimation, and *all-to-all vs multi-hop* communication networks. The distinction between static and dynamic estimation depends on whether the quantities to be estimated are constant or time-varying. The second distinction between hierarchical and distributed depends on whether the global estimate is required to be computed at a specific location on the network or at many sensing locations. The last distinction between all-to-all vs multi-hop depends on whether one node can directly send a message to any other node, or it has to route it through intermediate nodes. The work of this chapter belongs to the class of dynamic distributed estimation under multi-hop communication, while most of the works in the literature focus on different combinations of the three classes mentioned above.

For example, [142] derives conditions under which one can reconstruct the global sufficient statistics from local sufficient statistics; [85] investigates how much information two sensors (say S_1 and S_2) have to transmit regarding their measurements (say y_1 and y_2) in order for a fusion center to be able to evaluate certain functions of the measured data y_1 and y_2 ; this latter paper and [64] introduce the concept of *communication complexity* since computation efficiency depends also on the underlying communication graph, thus shedding some new light on well-known data fusion formulas. These works however, are limited to static hierarchical estimation in rooted-tree communication networks which is a scenario where communication delay is irrelevant. There is also a vast literature dealing with dynamic estimation with all-to-all communication where the main goal is to find the minimal representation of sufficient statistics to reduce either computation load in the central node [78] [144] or to reduce bandwidth requirements [132, 116]. However delay due to multi-hop topologies is not considered in these works.

An interesting approach dealing with distributed estimation and control in a multi-hop setting can be found in [82] where the authors propose to design linear distributed estimators and controllers which use only local information. However, this approach leads to a nonlinear optimization problem affected by well known problems of multiple local minima. Nonetheless, this is
very similar, in spirit, to the approach followed in this paper. In fact, we will focus on distributed estimation of dynamical systems for which sensor nodes are not physically co-located and exchange information only with their neighbors. For example, suppose that we want to track a dynamic quantity that changes according to a random walk, i.e x(t+1) = x(t) + w(t), where w(t) is a zero mean white noise process with covariance q, and we have N sensors that can measure this quantity corrupted by some noise, i.e. $y_i(t) = x(t) + n_i(t)$, where $n_i(t)$ are uncorrelated zero mean white noise processes with same covariance r. If all measurements were instantaneously available to a single location, it is well known from the centralized Kalman filter that the sufficient statistics necessary to reconstruct the optimal estimate would be given by the mean of all measurements, i.e. $\operatorname{mean}(y(t)) := \frac{1}{N} \sum_{i=1}^{N} y_i(t)$. In a multi-hop setting, it is not possible to assume that all measurements are instantaneously available at a specific location, since data from distant nodes, at best, arrives with some delay which depends on the specific network topology. Moreover, in a distributed setting where each sensor node is required to compute a global state estimate, the number of messages from every node to every other node can congest the network.

However, if it was possible to provide an algorithm that computes the mean of a set of numbers only through local communication, then the optimal estimate could be computed at each sensor node as follows:

$$\hat{x}_i(t+1) = (1 - \ell_0) \operatorname{mean}(\hat{x}(t)) + \ell_0 \operatorname{mean}(y(t)) \\ = \operatorname{mean}((1 - \ell_0)\hat{x}(t) + \ell_0 y(t))$$

where \hat{x}_i is the local estimate of i-th sensor. Algorithms able to compute the average of a set of numbers in a distributed way are known as *average consensus algorithms*, whose one popular class, based on linear iterations $z^+ = Pz$, where z is the vector whose entries are the quantities to be averaged and P is a *quasi-doubly stochastic matrix*, has been illustrated in Chapter 2. It is worth recalling that, under some weak connectivity properties, see Section 2.2.1, these algorithms guarantee that $\lim_{m\to\infty} [P^m z]_i = \text{mean}(z)$, i.e. all elements of vector $P^m z$ converge to their initial mean mean(z). Therefore, provided that it is possible to communicate sufficiently fast within two subsequent sensor measurements, i.e. $m \gg 1$, then intuitively we can assume that the following distributed estimation strategy yields the optimal global state estimate:

$$z = (1 - \ell_0)\hat{x}_i(t) + \ell_0 y_i(t)$$
 measur. & predict.
 $\hat{x}_i(t+1) = [P^m z]_i$ consensus

To our's knowledge Olfati-Saber [102] and Spanos et al. [129] were the first ones to propose this two-stage strategy based on computing first the mean of the sensor measurements via consensus algorithms, and then to update and predict the local estimates using the centralized Kalman optimal gains. This approach has been extended to static multivariable systems in [149] and to dynamic multivariable systems in [130][96]. In this context of fast communication, i.e. $m \gg 1$, it is natural to optimize P for fastest convergence rate of P^m , which corresponds to the second largest singular value of P, for which there are already very efficient optimization tools available [147] [148]. The assumption $m \gg 1$ is reasonable in applications for which communication is inexpensive as compared to sensing. This is the case, for example, in rendezvous control or coordination of mobile sensors where moving and sensing is energetically more expensive than communicating. However, there are many other important applications in which the number m of messages exchanged per sampling time per node needs to be small, as required in battery-powered wireless sensor networks. Therefore the assumption that $[P^m z]_i \approx \text{mean}(z)$ is not valid. In this context, for example, it is not clear whether maximizing the rate of convergence of P is the best strategy. Moreover, also the optimal gain ℓ becomes a function of the matrix P and the number of exchanged messages m; this will unlikely coincide with the optimal centralized Kalman gain, which is the strategy proposed in all the aforementioned papers [102][129][130][96][149].

Recently, Alriksson at al. [5] and Speranzon et al. [131], considered the case m = 1, i.e. sensors are allowed to communicate only once between sampling instants. In particular, in [5], the authors consider a general MIMO scenario where the matrix P = P(t) and the gain $\ell = \ell_i(t)$ (W and K, respectively, in their terminology) are selected at each time step in order to minimize the estimation error covariance of each sensor for the next time step, with the only constraint to maintain the estimate unbiased. In [131] the authors consider a single update equation $\hat{x}(t+1) = K\hat{x}(t) + Hy(t)$, and similarly to [5], they minimize the sum of all covariance errors at each time step, but differently they simultaneous optimize the gains K and H by enforcing stability of the matrix K. Both these iterative estimation algorithms seem to converge in numerical simulations and provide good performance, but no proof of stability and global optimality nor insights about the effect of connectivity of the underlying graph on the overall performance were given.

It is worth noting that the algorithms based on stochastic matrices represent just one class among all the average consensus algorithms [52] [134]. For example, there has been recent work (see e.g. [40, 143]) on *finite-time* consensus algorithms. These approaches guarantee that consensus can be reached in finite time; however these schemes are nonlinear and hence more difficult to analyze. Moreover, they are limited to the continuous time framework, making them less attractive in the presence of communication constraints. Another class of average consensus algorithms includes iterative distributed gradient descent algorithms based on Lagrange multipliers [15], which have been recently proposed by Skizas et al. [121] in the context of static distributed estimation. The authors proved that these algorithms converge to the centralized optimal estimator in the ideal scenario of fast communication and that they maintain good performance even under quantization, nongaussian noise and small number of consensus iterations.

In this chapter, we want to study the interaction between the consensus matrix Q, the number of messages per sampling time m, and the gain ℓ . With respect with the aforementioned works, we consider a simpler scenario with a scalar state which can be measured by N identical and independent sensors, a setup which still captures some of the most important features of the problem. In fact, also in this simple setup the joint optimization of Qand ℓ is not convex, as discussed in Section 8.5. Our goal is to provide better insights about the problem of distributed estimation using consensus matrices, rather then posing it as a black-box optimization algorithm. Therefore, we explore some important regimes, namely fast communication $m \to \infty$, "small" measurement noise $(r/q \to 0)$ and "small" process noise $(q/r \to 0)$.

This analysis provides useful guidelines for choosing the local filter gain ℓ and the consensus matrix Q also for more general scenarios. As a side result of our analysis, we also see that the standard recipe of choosing Q optimizing the second largest eigenvalue is not necessarily the best thing to do; similarly choosing the centralized optimal gain ℓ_c is not necessarily the optimal strategy.

Finally we provide some numerical examples to clarify the proposed analytical results.

8.2 Problem formulation

Consider a set V of N sensor nodes which are labeled i = 1, 2, ..., N. These sensors can communicate over a network modeled as a direct graph $\mathcal{G} = (V, E)$, where the edge (i, j) is in E if and only if the node i can transmit its information to the node j. We assume that the graph \mathcal{G} is time-invariant. A physical process with state $x \in \mathbb{R}$ evolves according to the continuous-time system

$$\dot{x}(t) = v(t) \tag{8.1}$$

where v(t) is a continuous-time white noise¹ of zero mean and intensity $q \ge 0$, that is $\mathbb{E}[q(t)q(s)] = q\delta_{ts}$, where δ_{hk} is the Kronecker delta. The initial condition is also assumed to be a random variable with mean x_0 and variance σ . Each sensor take measurements of the physical process according to the equation

$$y_i(kT) = x(kT) + n_i(kT), \qquad k \in \mathbb{Z}, \tag{8.2}$$

where T is the sampling time. Note that $y_i \in \mathbb{R}$, $\forall i$. We denote $y(kT) = [y_1(kT), \ldots, y_N(kT)]^*$ and $n(kT) = [n_1(kT), \ldots, n_N(kT)]^*$. Moreover the noise processes $n_i(kT) \in \mathbb{R}$ are such that $\mathbb{E}[n(kT)] = 0$, $E[n(kT)n(hT)] = rI\delta_{hk}$. Note also that (8.2) can be rewritten in the following vector form

$$y(kT) = x(kT)\mathbb{1} + n(kT).$$
 (8.3)

From now on, without loss of generality, we shall assume that T = 1. Suppose also that, between each pair of subsequent measurement update indexes k and k+1, each node exchanges m messages; we assume that these transmissions take place at the following times $k+\delta, k+2\delta, \ldots, k+(m-1)\delta, k+m\delta$, where $\delta = \frac{1}{m}$. Note that $k+m\delta = k+1$.

We shall denote with $\hat{x}_i(k + h\delta|k)$ the estimate, at node *i*, of the state *x* at time $k + h\delta$ given measurements up to time *k*. In compact form we shall also denote with $\hat{x} := [\hat{x}_1, \ldots, \hat{x}_N]^*$ the vector of estimates throughout the network; more precisely, making the dependence upon time explicit:

$$\hat{x} \left(k + h\delta | k \right) := \begin{bmatrix} \hat{x}_1 \left(k + h\delta | k \right) \\ \vdots \\ \hat{x}_N \left(k + h\delta | k \right) \end{bmatrix}.$$

We assume that these estimates are updated, for $k \ge 0$, according to the following rule

$$\begin{cases} \hat{x}(k|k) = (1 - \ell(k)) \hat{x}(k|k - 1) + \ell(k)y(k) \\ \hat{x}(k + h\delta|k) = P(k,h)\hat{x}(k + (h - 1)\delta|k), \end{cases}$$
(8.4)

where h = 1, ..., m and where P(k, h) is a suitable matrix compatible with the communication graph and where $0 < \ell(k) < 1, \forall k \ge 0$. From now on we shall use constant "gains" $\ell(k)$ and P(k, h), i.e. $\ell(k) = \ell$ and P(k, h) = P; furthermore, we shall also assume the the recursions (8.4) are initialized by

$$\hat{x}(0|-1) := x_0. \tag{8.5}$$

¹We recall that what is commonly referred to as "continuous time white noise" can be thought of as the "derivative" of a Wiener process which, unfortunately, is nowhere differentiable. More rigorously x(t) is a Wiener process.

A natural request is also that \hat{x}_i be an unbiased estimator for all *i*'s and for all times, i.e. $\mathbb{E}[\hat{x}_i(k+h\delta|k)] = x_0, \forall i, \forall k \ge 0, \forall h \in [1, m]$. This leads to the condition

$$P1 = 1.$$
 (8.6)

In fact, from the update rule (8.4) it follows that

$$x_0 \mathbb{1} = \mathbb{E}\left[\hat{x}\left(k + (h+1)\delta|k\right)\right] = \mathbb{E}\left[P\hat{x}\left(k + h\delta|k\right)\right] = x_0 P \mathbb{1}.$$

For $x_0 \neq 0$, the equality $x_0 P \mathbb{1} = x_0 \mathbb{1}$ implies (8.6). Hence it turns out that P has to be a quasi-stochastic matrix, which we shall assume from now on. Now we define the new variable $\tilde{x} (k + h\delta|k) = x (k + h\delta) \mathbb{1} - \hat{x} (k + h\delta|k)$ which represents the estimation error. In order to analyze the structure of the recursive equations that $\tilde{x} (k + h\delta|k)$ satisfies, it is convenient to discretize (8.1) in the following way

$$x(k + (h+1)\delta) = x(k + h\delta) + w(k + h\delta)$$

where

$$w(k+h\delta) = \int_{k+h\delta}^{k+(h+1)\delta} v(\tau)d\tau$$

Note that w is a discrete time white noise with mean zero and variance q/m, i.e. $\mathbb{E}[w(k+h\delta)] = 0$ and $\mathbb{E}[w^2(k+h\delta)] = \frac{q}{m}$. By straightforward calculations, for h = 0, we get

$$\tilde{x}(k|k) = (1-\ell)\,\tilde{x}(k|k-1) - \ell\,n(k)$$

and, for $1 \le h \le m$,

$$\tilde{x}\left(k+h\delta|k\right) = P^{h}\tilde{x}\left(k|k\right) + \left(\sum_{i=0}^{h-1} w\left(k+i\delta\right)\right)\mathbb{1}$$

We shall be concerned with the *second order* properties of the error, represented by the covariance matrices

$$\Sigma (k + h\delta|k) = \mathbb{E} \left[\tilde{x} \left(k + h\delta|k \right) \tilde{x} \left(k + h\delta|k \right)^* \right],$$

defined for $0 \le h \le m$. One can show that $\Sigma(k + h\delta|k)$ satisfies, for h = 0,

$$\Sigma(k|k) = (1-\ell)^2 \Sigma(k|k-1) + \ell^2 r I$$
(8.7)

and, for h = m,

$$\Sigma(k+1|k) = P^m \Sigma(k|k) (P^m)^* + q \mathbb{1} \mathbb{1}^*.$$
(8.8)

Plugging (8.7) into (8.8) we obtain the recursive equation

$$\Sigma(k+1|k) = (1-\ell)^2 P^m \Sigma(k|k-1) (P^*)^m + \ell^2 r P^m (P^*)^m + q \mathbb{1}\mathbb{1}^*$$
(8.9)

while, inserting (8.8) into (8.7) evaluated at the index k + 1 we get:

$$\Sigma(k+1|k+1) = (1-\ell)^2 P^m \Sigma(k|k) \left(P^*\right)^m + (1-\ell)^2 q \mathbb{1}\mathbb{1}^* + \ell^2 r I. \quad (8.10)$$

From (8.5) it follows that the error covariance $\Sigma(0|-1)$ satisfies $\Sigma(0|-1) = \sigma \mathbb{1}\mathbb{1}^*$; similarly $\Sigma(0|0)$ is given by $\Sigma(0|0) = (1-\ell)^2 \sigma \mathbb{1}\mathbb{1}^* + \ell^2 r I$. Iterating the update rule (8.9), starting from the initial condition $\Sigma(0|-1)$, we obtain

$$\Sigma(k|k-1) = (1-\ell)^{2k} P^{km} \Sigma(0|-1) (P^*)^{km} + \ell^2 r \sum_{i=0}^{k-1} (1-\ell)^{2i} P^{(i+1)m} (P^*)^{(i+1)m} + q \sum_{i=0}^{k-1} (1-\ell)^{2i} \mathbb{1}\mathbb{1}^* ;$$

similarly, iterating (8.10) with initial condition $\Sigma(0|0)$, we get

$$\begin{split} \Sigma(k|k) &= (1-\ell)^{2k} P^{km} \Sigma(0|0) \left(P^*\right)^{km} + q \sum_{i=0}^{k-1} (1-\ell)^{2i+2} \mathbb{1} \mathbb{1}^* + \\ &+ \ell^2 r \sum_{i=0}^{k-1} (1-\ell)^{2i} P^{im} \left(P^*\right)^{im} \end{split}$$

In this chapter we shall be concerned with the asymptotic $(k \to \infty)$ behavior of the error covariance. Hence we consider the limits

$$\lim_{k \to \infty} \Sigma(k|k-1) = r\ell^2 \sum_{i=0}^{\infty} (1-\ell)^{2i} P^{(i+1)m} \left(P^*\right)^{(i+1)m} + q \frac{1}{1-(1-\ell)^2} \mathbb{1}\mathbb{1}^*$$

and

$$\lim_{k \to \infty} \Sigma(k|k) = q \frac{(1-\ell)^2}{1-(1-\ell)^2} \mathbb{1}\mathbb{1}^* + r\ell^2 \sum_{i=0}^{\infty} (1-\ell)^{2i} P^{im} \left(P^*\right)^{im}$$

and define the cost functions²

$$J_1(P,\ell;m,r,q) = \operatorname{tr}\left\{\lim_{k \to \infty} \Sigma(k+1|k)\right\}$$

²In the remainder of the paper, when there is no risk of confusion, we might drop some arguments of the cost (e.g. denote $J_1(P, \ell)$ rather than $J_1(P, \ell; m, r, q)$).

and

$$J_2(P,\ell;m,r,q) = \operatorname{tr}\left\{\lim_{k \to \infty} \Sigma(k|k)\right\}$$

Note that some restrictions on P need to be imposed to ensure that the limits converge. From now given a quasi-stochastic matrix, we will denote its essential spectral by $\rho(P)^3$ If $\rho(P) \leq 1$ and $0 < \ell < 1$, then convergence is guaranteed. In the following of the chapter we will restrict to parameters which satisfy these constraints.

The costs defined above lead to the formulation of the following minimization problem:

Problem 8.1 Given a graph \mathcal{G} and a nonnegative integer m, find a real $\ell \in (0,1)$ and a matrix $P \in \mathcal{P}$, where \mathcal{P} is the set of the quasi-stochastic matrices compatible with the graph \mathcal{G} , minimizing J_1 or J_2 .

In the sequel the set \mathcal{P} may be further restricted while always being compatible with the topology of the communication network.

Remark 8.2 In the sequel we will consider only J_1 . The reason will be clear in the next sections where the minimization on J_1 will permit us to retrieve, for some particular cases, the results already known in the literature regarding Kalman filtering. For the sake of simplicity, we will denote this cost function simply by J in place of J_1 . Hence

$$J = r\ell^2 \operatorname{tr} \left\{ \sum_{i=0}^{\infty} (1-\ell)^{2i} P^{(i+1)m} \left(P^*\right)^{(i+1)m} \right\} + q \frac{1}{1-(1-\ell)^2} N \qquad (8.11)$$

Remark 8.3 Let us denote with λ_i , i = 0, ..., N - 1 the eigenvalues⁴ of P; since P is quasi-stochastic, we can set, without loss of generality, $\lambda_0 = 1$, i.e. $\sigma(P) = \{1, \lambda_1, \lambda_2, ..., \lambda_{N-1}\}$. Note that, if P is a normal matrix, i.e. $PP^* = P^*P$, then tr $\{P^{im}(P^*)^{im}\} = \sum_{h=0}^{n-1} |\lambda_h|^{2im}$; it is hence easy to see that formula (8.11) can be rewritten as follows:

$$J = \frac{r\ell^2 + qN}{1 - (1 - \ell)^2} + r\ell^2 \sum_{i=1}^{N-1} \frac{|\lambda_i|^{2m}}{1 - (1 - \ell)^2 |\lambda_i|^{2m}}$$
(8.12)

³In Chapter 3 we denote the essential spectral by ρ_{ess} . Since there is no risk of confusion we use here the simpler notation ρ .

⁴Multiple eigenvalues are counted as many times as their algebraic multiplicity.

Also note that, if P is normal and quasi-stochastic, then it is also quasidoubly stochastic. Relevant subclasses of normal matrices are, for instance, Abelian Cayley matrices [10], circulant matrices and symmetric matrices.

The previous remark suggests that the following assumption is both useful and reasonable.

Assumption 8.4 The set \mathcal{P} in problem 8.1 is the subset of normal quasistochastic matrices with $\rho(P) \leq 1$ which are compatible with the graph \mathcal{G} .

Example 8.5 Assume that \mathcal{P} is the set of the circulant stochastic matrices of the form

$$P_k = \begin{bmatrix} 1-k & k & 0 & 0 & \cdots & 0 & 0 \\ 0 & 1-k & k & 0 & \cdots & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ k & 0 & 0 & 0 & \cdots & 0 & 1-k \end{bmatrix}$$

where $k \in [0, 1]$ and assume that m = 1. Let

$$P^{opt}, \ \ell^{opt} \in \underset{\ell \in (0,1); P \in \mathcal{P}}{\operatorname{arg min}} J(P,\ell;m,r,q)$$
(8.13)

It is well known in the literature [46] that the eigenvalues of P can be expressed in the following form $\lambda_h = 1 - k + k e^{j\frac{2\pi}{N}h}$, $0 \le h \le N - 1$. Notice that $\left|1 - k + k e^{j\frac{2\pi}{N}h}\right| \ge \left|1/2 + 1/2e^{j\frac{2\pi}{N}h}\right|$, $\forall h : 0 \le h \le N - 1$. Hence it follows that

$$P_{\frac{1}{2}} = \mathop{\mathrm{arg\,min}}_{P \in \mathcal{P}} \ J(P,\ell;1,r,q)$$

for all $\ell \in (0,1)$, i.e. $P^{opt} = P_{\frac{1}{2}}$. In order to calculate the optimal gain we have to solve

$$\underset{\ell \in (0,1)}{\operatorname{arg min}} \frac{qN}{1 - (1 - \ell)^2} + r\ell^2 \sum_{h=0}^{N-1} \frac{|\lambda_h|^2}{1 - (1 - \ell)^2 |\lambda_h|^2}$$
(8.14)

where $\lambda_h = \frac{1}{2} + \frac{1}{2}e^{j\frac{2\pi}{N}h}$. Unfortunately it is not possible to give, in general, a closed form expression for the optimizing gain ℓ^{opt} , which has to be found using numerical search techniques.

However a simple expression for (8.14) can be obtained when the number of sensors N goes to infinity, which allows to study the behavior of the optimal ℓ for large networks. To this purpose it is convenient to define the normalized cost

$$\bar{J}_N := \frac{J}{N} = \frac{q}{1 - (1 - \ell)^2} + \frac{r\ell^2}{N} \sum_{h=0}^{N-1} \frac{|\lambda_h|^2}{1 - (1 - \ell)^2 |\lambda_h|^2}$$
$$= \frac{q}{1 - (1 - \ell)^2} + \frac{r\ell^2}{N} \tilde{J}_N$$

where the last equation defines \tilde{J}_N . Define also the function $f: \mathbb{C} \to \mathbb{C}$

$$f(z) = \frac{1}{2} + \frac{1}{2}z.$$

and note that $\lambda_h = f\left(e^{j\frac{2\pi}{N}h}\right)$. As $N \to \infty$, \tilde{J}_N converges to

$$\lim_{N \to \infty} \tilde{J}_N = \tilde{J}_\infty = \frac{1}{2\pi} \int_0^{2\pi} \frac{|f(e^{j\phi})|}{1 - (1 - \ell)^2 |f(e^{j\phi})|^2} d\phi$$
$$= \frac{1}{2\pi j} \oint_\gamma \frac{z^{-1}(2 + z + z^{-1})}{4 - (1 - \ell)^2 (2 + z + z^{-1})} dz$$

where γ is the unit circle. It is straightforward to see that the poles of $\frac{z^{-1}(2+z+z^{-1})}{4-(1-\ell)^2(2+z+z^{-1})}$ inside γ are $z_1 = \frac{2-(1-\ell)^2}{(1-\ell)^2} - \sqrt{\left(\frac{2-(1-\ell)^2}{(1-\ell)^2}\right)^2 - 1}$, and $z_2 = 0$; the integral can be computed explicitly using the residue theorem yielding:

$$\tilde{J}_{\infty} = \frac{1}{(1-\ell)^2} \left(\frac{1}{\sqrt{1-(1-\ell)^2}} - 1 \right)$$

Hence

$$\lim_{N \to \infty} \bar{J}_N\left(P_{\frac{1}{2}},\ell;1,r,q\right) = \frac{q}{1-(1-\ell)^2} + \frac{r\ell^2}{(1-\ell)^2}\left(\frac{1}{\sqrt{1-(1-\ell)^2}} - 1\right).$$

value of ℓ^{opt} can be found by minimizing the above limiting expression for \bar{J}_N .

8.3 Optimal consensus matrix P for fixed gain ℓ

In this section we assume that the estimation gain ℓ is fixed, and thus consider the optimization problem:

$$P^{opt}(\ell;m) = \underset{P \in \mathcal{P}}{\operatorname{arg min}} J(P,\ell;m).$$
(8.15)

Although the analysis of this problem is quite hard in general, a detailed study can be carried out in some interesting situations. In particular in the following we will restrict to the three special cases:

- the communication graph \mathcal{G} is undirected
- the sensors can communicate arbitrarily fast within two subsequent measurements, i.e., $m \to \infty$
- the estimation gain ℓ is sufficiently large, i.e. $\ell \to 1$; this intuitively corresponds to the situation in which the variance of the measurement noise is negligible with respect to the variance of the process, i.e. $\frac{r}{q} \approx 0$.

Before proceeding, we observe that

$$\min_{P \in \mathcal{P}} J = \frac{r\ell^2 + qN}{1 - (1 - \ell)^2} + \min_{P \in \mathcal{P}} r\ell^2 \sum_{j=1}^{N-1} \frac{|\lambda_j|^{2m}}{1 - (1 - \ell)^2 |\lambda_j|^{2m}}$$

and hence only the second term on the right hand side play a role in the optimization. We can therefore restrict to consider only this latter quantity which, for convenience of notation, we denote as

$$S(P,\ell;m) = \sum_{j=1}^{N-1} \frac{|\lambda_j|^{2m}}{1 - (1-\ell)^2 |\lambda_j|^{2m}}$$
(8.16)

8.3.1 Undirected communication graph G

We start by observing that, imposing that a matrix P is compatible with a graph is a convex constraint. Observe moreover that, since we assume that \mathcal{P} is a subset of the normal matrices, for all $P \in \mathcal{P}$ we have that $\|P\|_2 = \max_i |\lambda_i|$ and so also the condition $\rho(Q) \leq 1$ is a convex constraint as well.

Unfortunately, matrix normality is not a convex constraint in general. This difficult can be overcome when the graph is undirected as the lemma below shows. Indeed note that, if the communication graph \mathcal{G} is undirected, $P \in \mathcal{P}$ implies that also $P^* \in \mathcal{P}$. For any $P \in \mathcal{P}$, consider its symmetric part $P_{sym} := (P + P^*)/2$. Clearly, P_{sym} is normal and it is compatible with \mathcal{G} , therefore $P_{sym} \in \mathcal{P}$. The following lemma provides an interesting comparison between $J(P, \ell; m)$ and $J(P_{sym}, \ell; m)$ showing that the former is always greater or at most equal to the latter.

Lemma 8.6 Let P be any matrix in \mathcal{P} and let P_{sym} be defined as above. Then

$$J(P_{sym}, \ell; m) \le J(P, \ell; m).$$

Proof: Let λ_i be any eigenvalue of P. Then $Re\{\lambda_i\}$ is an eigenvalue of P_{sym} , where $Re\{\lambda_i\}$ denotes the real part of λ_i . Clearly $|Re\{\lambda_i\}| \leq |\lambda_i|$ which implies that

$$\frac{|Re\left\{\lambda_i\right\}|^{2m}}{1-(1-\ell)^2|Re\left\{\lambda_i\right\}|^{2m}} \leq \frac{|\lambda_i|^{2m}}{1-(1-\ell)^2|\lambda_i|^{2m}}$$

Therefore, from (8.16), it follows that $S(P_{sym}, \ell; m) \leq S(P, \ell; m)$; hence also $J(P_{sym}, \ell; m) \leq J(P, \ell; m)$ holds true.

Remark 8.7 It is important to note that normality plays a fundamental role in the previous lemma which cannot be generalized to quasi-stochastic matrices P. In fact, one can find a non-normal P for which the symmetric part P_{sym} gives a larger cost index.

An immediate consequence of Lemma 8.6 is that, when the communication graph is undirected, the minimum of the cost function J is reached by symmetric matrices. Thus, if \mathcal{P}_{sym} is the subset of \mathcal{P} containing the symmetric matrices, i.e. $\mathcal{P}_{sym} = \{P \in \mathcal{P} : P = P^*\}$, solving (8.15) is equivalent to solve

$$\underset{P \in \mathcal{P}_{sym}}{\operatorname{arg min}} J(P, \ell; m).$$
(8.17)

The following result provides a powerful characterization of (8.17) which has important implications when it comes to performing optimization.

Theorem 8.8 Let \mathcal{P}_{sym} be as above. Then the cost function $J(P, \ell; m)$ defined on \mathcal{P}_{sym} is a convex function.

Proof: Consider the function $f : \mathcal{B} \subseteq \mathbb{R}^n \to \mathbb{R}$ defined as

$$f(x) = \sum_{i=1}^{n} \frac{x_i^{2m}}{1 - \alpha x_i^{2m}},$$

where $m \in \mathbb{N}$, $0 < \alpha < 1$, $x = [x_1, x_2, \dots, x_n]^T$ and where $\mathcal{B} = \{x \in \mathbb{R}^n : |x_i| \le 1\}$. It is easy to verify that the function f is convex and symmetric, i.e. it is invariant to any permutation of the vector entries x_i . Hence, it follows from the theory of convex spectral functions that also J is a convex function [18].

Theorem 8.8 states that (8.17) is a convex problem thus implying that the solution of (8.17) can be performed efficiently by suitable numerical algorithms. In fact, Xiao et al. [148] adopted this strategy to optimize similar performance costs over symmetric stochastic matrices.

8.3.2 Fast communication $(m \to \infty)$

We have seen in Chapter 2 that the speed of consensus algorithms is governed by the essential spectral radius of the consensus matrix.

Indeed, the essential spectral radius plays also an important role in distributed estimation provided communication is "sufficiently fast". In fact the following theorem shows that optimizing the essential spectral radius improves the performance provided that the number of message exchanges is larger than a specified bound. However, when comparing the performance of two consensus matrices P_1 and P_2 , this bound is a function of $\rho(P_1)$ and $\rho(P_2)$ as explained in the following theorem.

Theorem 8.9 Let P_1 and P_2 be two matrices such that $\rho(P_1) > \rho(P_2)$. Then there exists \overline{m} (depending only on $\rho(P_1) - \rho(P_2)$) such that

$$J(P_1, \ell; m) > J(P_2, \ell; m), \qquad \forall m > \bar{m}.$$

Proof: Let $\rho_1 = \rho(P_1)$, $\rho_2 = \rho(P_2)$ and $\epsilon = \rho(P_1) - \rho(P_2) = \rho_1 - \rho_2$. Observe that

$$S(P_1, \ell; m) \ge \frac{\rho_1^{2m}}{1 - (1 - \ell)^2 \rho_1^{2m}}$$

> $\frac{(\rho_2 + \epsilon)^{2m}}{1 - (1 - \ell)^2 \rho_2^{2m}}$

and that

$$S(P_2, \ell; m) \le N \frac{\rho_2^{2m}}{1 - (1 - \ell)^2 \rho_2^{2m}}.$$

Hence, if $(\rho_2 + \epsilon)^{2m} > N\rho_2^{2m}$, we have that $S(P_1, \ell; m) > S(P_2, \ell; m)$. Straightforward calculations show that this last condition is satisfied if and only if $m > \frac{\log N}{2\log\left(1+\frac{\epsilon}{\rho_2}\right)}$. Note that $\frac{\log N}{2\log(1+\epsilon)} \ge \frac{\log N}{2\log\left(1+\frac{\epsilon}{\rho_2}\right)}$. Therefore by letting $\bar{m} = \left\lceil \frac{\log N}{2 \log(1+\epsilon)} \right\rceil$ the statement of the theorem follows. Note that \bar{m} depends only on ϵ .

8.3.3 Large gain $(\ell \rightarrow 1)$

In this section we shall instead consider the number of message exchanges m as fixed and assume the gain ℓ is "large"; this, as we shall also see in Section 8.5.2, is what happens when the measurement noise is small as compared to process noise, a situation frequently encountered in practice. It is remarkable that in this case the Frobenius norm is instead the "right" way to compare consensus matrices. This is made precise by the following theorem.

Theorem 8.10 Let P_1, P_2 be two matrices such that $||P_1^m||_F > ||P_2^m||_F$. Then there exists $\bar{\ell}$ (depending only on $||P_1^m||_F - ||P_2^m||_F$) such that

$$J(P_1,\ell;m) > J(P_2,\ell;m), \qquad \forall \,\ell > \bar{\ell}. \tag{8.18}$$

Proof: Let $f(\ell, m) = S(P_1, \ell; m) - S(P_2, \ell; m)$ and $\epsilon = ||P_1^m||_F - ||P_2^m||_F$. We have that $f(\ell, m)$ can be written in the following way

$$f(\ell,m) = \sum_{i=0}^{\infty} \left(\|P_1^{(i+1)m}\|_F^2 - \|P_2^{(i+1)m}\|_F^2 \right) (1-\ell)^{2i} = \sum_{i=0}^{\infty} \alpha_i (1-\ell)^{2i}$$

where $\alpha_i = \|P_1^{(i+1)m}\|_F^2 - \|P_2^{(i+1)m}\|_F^2$. It is straightforward to see that $\alpha_0 > \epsilon^2$ and that, for $i \ge 1$, $\alpha_i \ge -(N-1)\rho^{2m(i+1)}$, where $\rho = \rho(P_2)$. Hence, it follows that

$$f(\ell, m) > \epsilon^2 - (N-1) \sum_{i=1}^{\infty} \rho^{2m(i+1)} (1-\ell)^{2i}$$
$$> \epsilon^2 - \frac{(1-\ell)^2 (N-1)}{1-(1-\ell)^2}$$

Clearly, if the last term is positive, then also $f(\ell, m) > 0, \forall m$. This condition is satisfied if and only if

$$\ell \ge 1 - \sqrt{\frac{\epsilon^2}{N - 1 + \epsilon^2}} = \bar{\ell}.$$

Therefore, $S(P_1, \ell; m) - S(P_2, \ell; m) > 0$, $\forall \ell > \overline{\ell}$ and consequently $J(P_1, \ell; m) > J(P_2, \ell; m)$, thus proving the claim of the theorem.

Remark 8.11 At first sight, Theorem 8.9 and Theorem 8.10 seem in contradiction. However, this can be explained by observing that $||P^m||_F^2 = 1 + \alpha \rho^{2m}(P) + o(\rho^{2m}(P))$, where α is the algebraic multiplicity of the essential spectral radius of P. Therefore, for large m, minimizing the Frobenius norm of P^m or the essential spectral radius of P is almost equivalent.

8.4 Optimal gain ℓ for fixed consensus matrix P

In this section we assume that the consensus matrix P is fixed. Hence the problem we want to solve is the following

$$\ell^{opt} \underset{\ell \in (0,1)}{\operatorname{arg min}} J(P,\ell;m)$$
(8.19)

The previous optimization problem is convex in ℓ . This fact can be easily checked by observing that the cost functional J can be written as sum of functions of the form:

$$g(\ell) = \frac{x\ell^2}{1 - x(1 - \ell)^2}, \quad h(\ell) = \frac{x}{1 - x(1 - \ell)^2}, \quad x \in [0, 1]$$

which are convex in $\ell \in (0, 1)$.

Given any $P \in \mathcal{P}$ we define

$$J^{opt}(m, r, q|P) \stackrel{\triangle}{=} \min_{\ell} J(P, \ell; m, r, q)$$
(8.20)

$$\ell^{opt}(m, r, q|P) \in \arg\min_{\ell} J(P, \ell; m, r, q)$$
(8.21)

Note that $J^{opt}(m, r, q|P) = J(P, \ell^{opt}(m, r, q|P); m, r, q)$ and that $P^{opt}(m, r, q) \in \arg \min_P J^{opt}(m, r, q|P)$. In the sequel, without risk of confusion, we shall omit arguments which are kept fixed.

Convexity of J allows easy computations of $\ell^{opt}(m|P)$. In the remaining of this section we shall see that the sequence $\{\ell^{opt}(m|P)\}_{m=0}^{\infty}$ is monotonically non-decreasing in m. Moreover, it is bounded below and above by ℓ_d^{opt} and ℓ_c^{opt} , which are the optimal gains minimizing J respectively when P = I and when $P = \frac{1}{N} \mathbb{1}\mathbb{1}^*$, namely

$$\ell_d^{opt} = \underset{\ell \in (0,1)}{\operatorname{arg min}} J(I,\ell;m,r,q), \ \ell_c^{opt} = \underset{\ell \in (0,1)}{\operatorname{arg min}} J\left(\frac{1}{N}\mathbb{1}\mathbb{1}^*,\ell;m,r,q\right)$$

Note that P = I and $P = \frac{1}{N} \mathbb{1}\mathbb{1}^*$ represent the two extreme cases in modeling the flow of information between the sensors. Indeed, P = I corresponds to the situation in which the sensors do not communicate; in such a case there are N Kalman filters running separately (the subscript "d" in ℓ_d^{opt} means decentralized, i.e. no communication). In the other case, instead, we have that the underlying communication graph is complete and this means that each sensor has full knowledge about the estimates of all the other sensors (the subscript "c" in ℓ_c^{opt} means centralized, i.e. full communication). Since $I^m = I$ and $(1/N\mathbb{1}\mathbb{1}^*)^m = 1/N\mathbb{1}\mathbb{1}^*$ for all $m \geq 1$, we have that ℓ_c^{opt} and ℓ_d^{opt} are independent on m. The following proposition characterizes precisely ℓ_d^{opt} .

Proposition 8.12 Let ℓ_d^{opt} and ℓ_c^{opt} be as above. Then

$$\ell_d^{opt} = \frac{-q + \sqrt{q^2 + 4qr}}{2r}, \ \ell_c^{opt} = \frac{-q + \sqrt{q^2 + 4q\bar{r}}}{2\bar{r}}$$

where $\bar{r} = \frac{r}{N}$.

Proof: The proof follows from standard results known in Kalman filtering.

The following theorem shows also that ℓ_d^{opt} and ℓ_c^{opt} play the role of, respectively, lower and upper bounds for $\ell^{opt}(m|P)$. Indeed, a stronger result can be obtained, which characterizes the sequence $\{\ell^{opt}(m|P)\}_{m=0}^{\infty}$.

Theorem 8.13 Let $P \in \mathcal{P}$. Let $\ell^{opt}(m|P)$ be defined as above. Then the following chain of inequalities holds true

$$\ell_d^{opt} = \ell^{opt}(0|P) \le \ell^{opt}(1|P) \le \dots \le \ell^{opt}(m|P) \le \le \ell^{opt}(m+1|P) \le \dots \le \ell^{opt}(\infty|P) \le \ell_c^{opt}.$$

The achieved (optimal) cost satisfies the reversed chain of inequalities:

$$J\left(P,\ell_d^{opt};0\right) \ge J\left(P,\ell^{opt}(1|P);1\right) \ge \ldots \ge J\left(P,\ell^{opt}(m|P);m\right) \ge$$
$$> J\left(P,\ell^{opt}(m+1|P);m+1\right) \ldots \ge J\left(P,\ell_c^{opt};\infty\right)$$

Moreover $\ell^{opt}(\infty|P) = \ell_c^{opt}$ if and only if $\rho(P) < 1$.

Proof: Note that the equality $\ell^{opt}(0|P) = \ell_d^{opt}$ follows directly from the fact that $P^0 = I$. We prove now that $\ell^{opt}(m|P) \leq \ell^{opt}(m+1|P)$. We start

by calculating the partial derivative

$$\frac{\partial J(P,\ell;m)}{\partial \ell} = 2\sum_{i=1}^{N-1} \frac{r\ell |\lambda_i|^{2m} - r\ell(1-\ell)|\lambda_i|^{4m}}{\left[1 - (1-\ell)^2 |\lambda_i|^{2m}\right]^2} - 2\frac{Nq(1-\ell) - r\ell^2}{\left[1 - (1-\ell)^2\right]^2}$$

Consider now the functions $p: [0,1] \times [0,\infty) \to \mathbb{R}$ defined as

$$p(x,m) = \frac{x^{2m} - (1-\ell)x^{4m}}{\left(1 - (1-\ell)^2 x^{2m}\right)^2}.$$

We have that

$$\frac{\partial p}{\partial m} = 2x^{2m} \frac{1 - (1 - \ell^2) x^{2m}}{\left[1 - (1 - \ell)^2 x^{2m}\right]^3} \log x.$$

It is easy to check that $\frac{\partial p}{\partial m} < 0 \ \forall \ell \in (0,1)$ and $\forall x \in [0,1]$. This implies that

$$\begin{aligned} \frac{\partial J(P,\ell;m)}{\partial \ell} &- \frac{\partial J(P,\ell;m+1)}{\partial \ell} = \\ &= 2r\ell \sum_{i=0}^{N-1} \frac{|\lambda_i|^{2m} - (1-\ell)|\lambda_i|^{4m}}{[1-(1-\ell)^2|\lambda_i|^{2m}]^2} - 2r\ell \sum_{i=0}^{N-1} \frac{|\lambda_i|^{2(m+1)} - (1-\ell)|\lambda_i|^{4(m+1)}}{[1-(1-\ell)^2|\lambda_i|^{2(m+1)}]^2} \\ &\ge 0. \end{aligned}$$

Since we already know that for all m the function $J(P, \ell; m)$ is convex in $\ell \in (0, 1)$, it follows that $\ell^{opt}(m+1|P) > \ell^{opt}(m|P)$. In order to show that $\ell^{opt}(m|P) \leq \ell_c^{opt}$, we remark that the matrix $\frac{1}{N} \mathbb{1} \mathbb{1}^*$ has an eigenvalue equal to 1 and N-1 eigenvalues equal to 0. This implies that, for all m,

$$\frac{\partial J\left(\frac{1}{N}\mathbb{1}\mathbb{1}^*,\ell;m\right)}{\partial \ell} = -2\frac{Nq(1-\ell)-r\ell^2}{\left[1-(1-\ell)^2\right]^2}.$$

Thus we have that

$$\frac{\partial J(P,\ell;m)}{\partial \ell} - \frac{\partial J\left(\frac{1}{N}\mathbb{1}\mathbb{1}^*,\ell;m+1\right)}{\partial \ell} = 2r\ell \sum_{i=1}^{N-1} \frac{|\lambda_i|^{2m} - (1-\ell)|\lambda_i|^{4m}}{\left[1 - (1-\ell)^2|\lambda_i|^{2m}\right]^2} > 0$$

where the last inequality follows from the fact that $x - \alpha x^2 > 0, \forall x \in (0, 1)$ and $\forall \alpha \in (0, 1)$. From this last inequality $\ell^{opt}(m|P) < \ell_c^{opt}$ follows.

We prove now $\ell^{opt}(m|P) = \ell_c^{opt}$ if and only if $\rho(P) < 1$. If $\rho(P) < 1$, then $\lim_{k\to\infty} P^k = \frac{1}{N} \mathbb{1}\mathbb{1}^*$. Conversely, if $\rho(P) = 1$, then there exists λ_i , $1 \leq i \leq N-1$ such that $|\lambda_i| = 1$ implying that

$$\frac{\partial J(P,\ell;m)}{\partial \ell} - \frac{\partial J\left(\frac{1}{N}\mathbb{1}\mathbb{1}^*,\ell;m\right)}{\partial \ell} \ge \frac{2r\ell^2}{\left[1-(1-\ell)^2\right]^2} > 0.$$

Thus $\ell^{opt}(\infty|P) < \ell_c^{opt}$.

The chain of inequalities on the cost functionals follows from the fact that $J(P, \ell; m)$ is a decreasing function on m.

The following two lemmas provide some interesting properties of $J^{opt}(m, r, q|P)$ as a function of m and as a function of r, q. They will be useful in the next Section.

Lemma 8.14 Let $P, \overline{P} \in \mathcal{P}$ be such that $\rho(\overline{P}) + \epsilon \leq \rho(P)$, where $\epsilon > 0$. Then there exists \overline{m}_{ϵ} , depending only on ϵ , such that the cost function (8.20) satisfies

$$J^{opt}(m|P) > J^{opt}(m|\bar{P}), \qquad \forall m > \bar{m}_{\epsilon}.$$
(8.22)

Proof: Let $\rho = \rho(P)$, $\bar{\rho} = \rho(\bar{P})$ and λ_i and $\bar{\lambda}_i$ be the eigenvalues of P and \bar{P} , respectively. Observe that

$$\sum_{j=1}^{N-1} \frac{|\lambda_j|^{2m}}{1 - (1-\ell)^2 |\lambda_j|^{2m}} \ge \frac{\rho^{2m}}{1 - (1-\ell)^2 \rho^{2m}} \ge \frac{(\bar{\rho}+\epsilon)^{2m}}{1 - (1-\ell)^2 \bar{\rho}^{2m}}$$

and that

$$\sum_{j=1}^{N-1} \frac{|\bar{\lambda}_j|^{2m}}{1 - (1-\ell)^2 |\bar{\lambda}_j|^{2m}} \le N \frac{\bar{\rho}^{2m}}{1 - (1-\ell)^2 \bar{\rho}^{2m}}.$$

Hence, if $(\bar{\rho} + \epsilon)^{2m} > N\bar{\rho}^{2m}$, we have that

$$\sum_{j=1}^{N-1} \frac{|\lambda_j|^{2m}}{1 - (1-\ell)^2 |\lambda_j|^{2m}} > \sum_{j=1}^{N-1} \frac{|\bar{\lambda}_j|^{2m}}{1 - (1-\ell)^2 |\bar{\lambda}_j|^{2m}}$$
(8.23)

Straightforward calculations show there exists \bar{m}_{ϵ} depending only on ϵ such that $(\bar{\rho} + \epsilon)^{2m} > N\bar{\rho}^{2m}$ for all $m \geq \bar{m}_{\epsilon}$ so ensuring that Eqn. (8.23) holds. This implies that $J(P, \ell; m) > J(\bar{P}, \ell; m)$ for all $m > \bar{m}_{\epsilon}$. The lemma follows from the fact that

$$\begin{aligned} J^{opt}(m|\bar{P}) &= J(\bar{P}, \ell^{opt}(m|\bar{P}); m) \leq J(\bar{P}, \ell^{opt}(m|P); m) < \\ &< J(P, \ell^{opt}(m|P); m) = J^{opt}(m|P) \end{aligned}$$

Lemma 8.15 Let $P, \overline{P} \in \mathcal{P}$ and assume that $\|\overline{P}^m\|_F + \epsilon \leq \|P^m\|_F$, where $\epsilon > 0$. Then there exists δ_{ϵ} depending only on ϵ such that the cost function (8.20) satisfies

$$J^{opt}(r,q|P) > J^{opt}(r,q|\bar{P}), \qquad \forall r/q < \delta_{\epsilon}.$$

Proof: Let

$$f(\ell,m) := \sum_{j=1}^{N-1} \frac{|\lambda_j|^{2m}}{1 - (1-\ell)^2 |\lambda_j|^{2m}} - \sum_{j=1}^{N-1} \frac{|\bar{\lambda}_j|^{2m}}{1 - (1-\ell)^2 |\bar{\lambda}_j|^{2m}}$$

where λ_i and $\bar{\lambda}_i$ are the eigenvalues of P and \bar{P} , respectively. We have that $f(\ell, m)$ can be written in the following way

$$f(\ell,m) = \sum_{\substack{i=0\\\infty}}^{\infty} \left(\|P^{(i+1)m}\|_F^2 - \|\bar{P}^{(i+1)m}\|_F^2 \right) (1-\ell)^{2i}$$
$$= \sum_{\substack{i=0\\i=0}}^{\infty} \alpha_i (1-\ell)^{2i}$$

where $\alpha_i = \|P^{(i+1)m}\|_F^2 - \|\bar{P}^{(i+1)m}\|_F^2$. It is straightforward to see that $\alpha_0 > \epsilon^2$. Also since all eigenvalues of P^k and \bar{P}^k are inside the unit circle for all k then $\|P^{(i+1)m}\|_F \ge 1$ and $\|\bar{P}^{(i+1)m}\|_F \le N$ for $i \ge 1$, therefore $\alpha_i \ge -(N-1)$. Hence, it follows that

$$f(\ell,m) > \epsilon^2 - (N-1) \sum_{i=1}^{\infty} (1-\ell)^{2i} = \epsilon^2 - \frac{(1-\ell)^2 (N-1)}{1 - (1-\ell)^2}$$

Clearly, if the last term is non-negative, then also $f(\ell, m) > 0, \forall m$. This condition is satisfied if and only if

$$\ell \ge 1 - \sqrt{\frac{\epsilon^2}{N - 1 + \epsilon^2}} =: \ell_{\epsilon}.$$

Therefore, we have that $J(P, \ell; r, q) > J(\bar{P}, \ell; r, q)$ for all $\ell \ge \ell_{\epsilon}$. Now observe that if $\frac{r}{q} \to 0$ then $\ell_d^{opt} \to 1$. Since $\ell_{\epsilon} < 1$, then there exists $\delta_{\epsilon} > 0$ such that if $r/q < \delta_{\epsilon}$ then $\ell_d^{opt} > \ell_{\epsilon}$ and so, by Theorem 8.13, $\ell^{opt}(r, q|P) > \ell_{\epsilon}$. By the previous arguments this implies that

$$J^{opt}(r,q|P) = J(P,\ell^{opt}(r,q|P);r,q)$$

> $J(\bar{P},\ell^{opt}(r,q|P);r,q)$
 $\geq J(\bar{P},\ell^{opt}(r,q|\bar{P});r,q)$
= $J^{opt}(r,q|\bar{P})$

which completes the proof.

e

8.5 Joint optimization of P and ℓ : special cases

We have shown in the previous two sections that the cost functional J is a convex function, both in $P \in \mathcal{P}_{sym}$ for ℓ fixed and in $\ell \in (0, 1)$ for P fixed. Unfortunately, as simple examples demonstrate, J is not a convex function jointly in ℓ and $P \in \mathcal{P}_{sym}$. Therefore, the joint minimization of J

$$(P^{opt}(m, r, q), \ell^{opt}(m, r, q)) \in \arg\min_{\ell \in (0,1); P \in \mathcal{P}_{sym}} J(P, \ell; m, r, q)$$
(8.24)

results to be quite hard in general. Nevertheless, an analytical characterization is possible when restricting to some "asymptotic cases" on the values of m, r and q. In particular we will consider the following situations:

- the sensors can communicate arbitrarily fast within two subsequent measurements, i.e., $m \to \infty$
- $\frac{r}{q} \approx 0$, i.e. the variance of the measurement noise is negligible with respect to the variance of the process
- $\frac{q}{r} \approx 0$, i.e. the variance of the process is negligible with respect to the variance of the measurement noise

First note that $P^{opt}(m, r, q)$ and $\ell^{opt}(m, r, q)$ are indeed only functions of m and r/q. In the sequel, without risk of confusion, we shall omit arguments which are kept fixed.

8.5.1 Fast communication $(m \to \infty)$

The results of this section parallel those of section 8.3.2. Indeed it will be shown that when $m \to \infty$, optimizing P for "fast convergence", i.e. minimizing the essential spectral radius is the "right" thing to do. Moreover, as expected, the optimal gain converges to the centralized gain ℓ_c^{opt} .

Theorem 8.16 Let $P^{opt}(m)$, $\ell^{opt}(m)$ be a solution of (8.24). Then

$$\lim_{m \to \infty} \rho(P^{opt}(m)) = \min_{P \in \mathcal{P}} \rho(P).$$

and

$$\lim_{m \to \infty} \ell^{opt}(m) = \ell_c^{opt}.$$

Moreover, if arg $\min_{P \in \mathcal{P}} \rho(P)$ is a singleton, then also

$$\lim_{m \to \infty} P^{opt}(m) = \arg\min_{P \in \mathcal{P}} \rho(P)$$

Proof: Consider a sequence $\{\rho(P^{opt}(m))\}_{m=0}^{\infty}$. Let \overline{P} be any matrix minimizing the essential spectral radius inside \mathcal{P} . We shall now prove that

$$\lim_{m \to \infty} \rho\left(P^{opt}(m)\right) = \rho(\bar{P}) \tag{8.25}$$

For notational convenience, we denote along the proof $\rho(P^{opt}(m))$ by ρ_m and $\rho(\bar{P})$ by $\bar{\rho}$. Assume by contradiction that $\lim_{m\to\infty}\rho_m \neq \bar{\rho}$. This means that there exists $\epsilon > 0$ and a sequence of integers $m_1 < m_2 < m_3 < \ldots$, such that $\rho_{m_i} \geq \bar{\rho} + \epsilon$, $\forall i \in \mathbb{N}$ and consequently that $\inf \{\rho_{m_i}\} \geq \bar{\rho} + \epsilon$. This implies, by Lemma 8.14, that there exist \bar{m}_{ϵ} , depending only on ϵ , such that $J^{opt}(m|P^{opt}(m_i)) > J^{opt}(m|\bar{P}), \forall i \in \mathbb{N}$ and $\forall m > \bar{m}_{\epsilon}$. Therefore, if \bar{i} is such that $m_{\bar{i}} > \bar{m}_{\epsilon}$, we get

$$J^{opt}(m_i|P^{opt}(m_i)) > J^{opt}(m_i|\bar{P}), \ \forall i \ge \bar{i},$$

contradicting the fact that $P^{opt}(m_i) \in \arg \min_P J^{opt}(m_i|P)$ and thus proving Equation (8.25).

Consider now any sequence of integers $\tilde{m}_1 < \tilde{m}_2 < \tilde{m}_3 < \tilde{m}_4 \dots$ such that $\lim_{i\to\infty} P^{opt}(\tilde{m}_i) = \tilde{P}$ where \tilde{P} is a suitable matrix inside \mathcal{P} . It follows, by the continuity of ρ that $\lim_{i\to\infty} \rho(P^{opt}(\tilde{m}_i)) = \rho(\tilde{P})$. Clearly $\rho(\tilde{P}) = \bar{\rho}$. Suppose now that arg $\min_{P \in \mathcal{P}} \rho(P)$ is a singleton and call \bar{P} the unique element in this set. It follows that $\tilde{P} = \bar{P}$, thus implying, from the compactness of \mathcal{P} , that

$$\lim_{m \to \infty} P^{opt}(m) = \arg\min_{P \in \mathcal{P}} \rho(P).$$

Consider now a sequence $\{\ell^{opt}(m)\}_{m=0}^{\infty}$ and let $\tilde{\rho}$ be such that $\bar{\rho} < \tilde{\rho} < 1$. Let us introduce a matrix \tilde{P} such that $\sigma(\tilde{P}) = \{1, \tilde{\rho}, \dots, \tilde{\rho}\}$, that is \tilde{P} has N-1eigenvalues equal to $\tilde{\rho}$. It follows that

$$J\left(\tilde{P},\ell;m\right) = \frac{r\ell^2 + qN}{1 - (1-\ell)^2} + (N-1)\frac{r\ell^2\tilde{\rho}^{2m}}{1 - (1-\ell)^2\tilde{\rho}^{2m}}$$

and, by recalling the expression of $\frac{\partial J(P,\ell;m)}{\partial \ell}$, that

$$\frac{\partial J\left(\tilde{P},\ell;m\right)}{\partial \ell} = 2r\ell(N-1)\frac{\tilde{\rho}_m^{2m} - (1-\ell)\tilde{\rho}_m^{4m}}{\left[1 - (1-\ell)^2\tilde{\rho}_m^{2m}\right]} - 2\frac{Nq(1-\ell) - r\ell^2}{\left[1 - (1-\ell)^2\right]^2}$$

Let now ϵ be any real such that $0 < \epsilon < \tilde{\rho} - \bar{\rho}$. We have just proved that there exists a positive integer \bar{m}_{ϵ} such that $\forall m > \bar{m}_{\epsilon}$ we have that $\rho_m < \tilde{\rho} - \epsilon$. Consider now the function $f : [0, 1] \times (0, 1) \to \mathbb{R}$ defined as

$$f(x,\alpha) = \frac{x^{2m} - \alpha x^{4m}}{(1 - \alpha^2 x^{2m})^2},$$

where $m \in \mathbb{N}$. By straightforward calculations we have that

$$\frac{\partial f(x,\alpha)}{\partial x} = \frac{2mx^{2m}\left(1 + \alpha^2 x^{2m} - 2\alpha x^{2m}\right)}{x\left(1 - \alpha^2 x^{2m}\right)^3}.$$

It is possible to show that $\frac{\partial f(x,\alpha)}{\partial x} > 0$, $\forall x \in (0,1], \forall \alpha \in (0,1)$. Since, by the definition of essential spectral radius, we have that all the eigenvalues of $P^{opt}(m)$ different from 1, are, in absolute value, smaller than ρ_m and therefore of $\tilde{\rho}$, it follows that

$$\frac{\partial J\left(P(m),\ell;m\right)}{\partial \ell} \le \frac{\partial J\left(\tilde{P},\ell;m\right)}{\partial \ell}, \qquad \forall \ell \in (0,1).$$
(8.26)

Let now $\tilde{\ell}^{opt}(m) = \arg \min_{\ell \in (0,1)} J\left(\tilde{P}, \ell; m\right)$. From Theorem 8.13 we have that $\lim_{m \to \infty} \tilde{\ell}^{opt}(m) = \ell_c^{opt}$. On the other hand, by the convexity of J on ℓ and by (8.26) it follows that $\tilde{\ell}^{opt}(m) \leq \ell^{opt}(m), \forall m > \bar{m}_{\epsilon}$. Therefore $\lim_{m \to \infty} \ell^{opt}(m) = \ell_c^{opt}$.

8.5.2 Small measurement noise $(r/q \rightarrow 0)$

In this subsection we treat the case in which the variance of the measurement noise is negligible with respect of the variance of the process, that is $r/q \rightarrow 0$. This parallels the case analyzed in Section 8.3.3. Also here, as in Section 8.3.3, it is the Frobenius norm of P which plays a crucial role.

Indeed, while in Section 8.3.3 the gain was assumed to be large, here ℓ is shown to converge to 1 as r/q goes to zero.

Theorem 8.17 Let $P^{opt}(r,q)$, $\ell^{opt}(r,q)$ be a solution of Equation (8.24) and let

$$\bar{P} \in \underset{P \in \mathcal{P}}{\operatorname{arg min}} \|P^m\|_F.$$

Then

$$\lim_{r/q \to 0} \|P^{opt}(r,q)^m\|_F = \|\bar{P}^m\|_F.$$

Moreover

$$\ell^{opt}(r,q) = 1 - \frac{\|P\|_F^2}{N} \frac{r}{q} + o(r/q).$$

In addition, if $\arg \min_{P \in \mathcal{P}} \|P^m\|_F$ is a singleton also

$$\lim_{r/q \to 0} P^{opt}(r,q) = \underset{P \in \mathcal{P}}{\arg\min} \|P^m\|_F$$

holds.

Proof: Let $J^{opt}(r, q|P)$ and $\ell^{opt}(r, q|P)$ be quantities defined in (8.20) and (8.21). We shall first prove that

$$\lim_{r/q \to 0} \| (P^{opt}(r,q))^m \|_F = \| \bar{P}^m \|_F.$$
(8.27)

Assume by contradiction that $\lim_{r/q\to 0} \| (P^{opt}(r,q))^m \|_F \neq \|\bar{P}^m\|_F$. This means that there exists $\epsilon > 0$ and a sequence $\frac{r_1}{q_1} > \frac{r_2}{q_2} > \frac{r_3}{q_3} > \ldots$, such that $\| (P^{opt}(r_i, q_i))^m \|_F \geq \|\bar{P}^m\|_F + \epsilon, \forall i \in \mathbb{N}$. By Lemma 8.15 this implies that there exists δ_{ϵ} , depending only on ϵ , such that

$$J^{opt}(r,q|P^{opt}(r_i/q_i)) > J^{opt}(r,q|\bar{P}),$$

for each pair r, q such that $\frac{r}{q} < \delta_{\epsilon}$ and $\forall i \in \mathbb{N}$. Therefore, if \overline{i} is such that $\frac{r_{\overline{i}}}{q_{\overline{i}}} < \delta_{\epsilon}$, we get

$$J^{opt}(r_i, q_i | P^{opt}(r_i, q_i)) > J^{opt}(r_i, q_i | \bar{P}), \quad \forall i \ge \bar{i},$$

which yields a contradiction and hence proves (8.27).

Consider now any sequence $\frac{r_{i_1}}{q_{i_1}} > \frac{r_{i_2}}{q_{i_2}} > \frac{r_{i_3}}{q_{i_3}} > \dots$, such that

$$\lim_{h \to \infty} P^{opt}(r_{i_h}, q_{i_h}) = \tilde{P},$$

where \tilde{P} is a suitable matrix inside \mathcal{P} . It follows that

$$\lim_{h \to \infty} \| \left(P^{opt}(r_{i_h}, q_{i_h}) \right)^m \|_F = \| \tilde{P}^m \|_F.$$

Clearly $\|\tilde{P}^m\|_F = \|\bar{P}^m\|_F$. Suppose now that $\arg\min_{P\in\mathcal{P}}\|P^m\|_F$ is a singleton and call \bar{P} the unique element in this set. It follows that $\tilde{P} = \bar{P}$, thus implying, for the compactness of \mathcal{P} , that

$$\lim_{r/q \to 0} P^{opt}(r,q) = \bar{P} = \underset{P \in \mathcal{P}}{\operatorname{arg min}} \|P^m\|_F$$

Consider now $\ell^{opt}(r,q)$. Preliminarily we observe that $\lim_{r/q\to 0} \ell_d^{opt} = 1$. Since by Theorem 8.13 we have that $\ell^{opt}(r,q|P) \geq \ell_d^{opt}$, it follows that also $\lim_{r/q\to 0} \ell^{opt}(r,q|P) = 1$ and $\lim_{r/q\to 0} \ell^{opt}(r,q) = 1$. Using the implicit function theorem it follows that $\ell^{opt}(r,q|P)$ is differentiable around r/q = 0, i.e.

$$\ell^{opt}(r,q|P) = 1 + \alpha(P)\frac{r}{q} + o\left(\frac{r}{q}\right)$$

where the coefficient $\alpha(P)$ is given by

$$\alpha(P) = -\left[\left(\frac{\partial^2 J}{\partial \ell^2}\right)^{-1} \frac{\partial^2 J}{\partial \ell \partial r/q}\right]_{\ell=1, P, r/q=0} = -\frac{\|P^m\|_F^2}{N}$$

Since $\ell^{opt}(r,q) = \ell^{opt}(r,q|P^{opt}(r,q))$, then $\ell^{opt}(r,q) = 1 + \alpha(P^{opt}(r,q))r/q + o(r/q)$. Using the fact that $\lim_{r/q\to 0} \|P^{opt}(r,q)\|_F = \|\bar{P}\|_F$, then

$$\ell^{opt}(r,q) = 1 + \alpha(\bar{P})r/q + o(r/q).$$

Remark 8.18 It is interesting to observe that the communication graph \mathcal{G} determines the coefficient of the first order expansion of ℓ in r/q. Note that, in the extreme cases $P^{opt} = I$ (no communication, decentralized estimation) and $P^{opt} = \frac{11^*}{N}$ (communication graph fully connected, centralized estimation) one recovers, respectively, $\ell_d^{opt} = 1 - \frac{r}{q} + o(\frac{r}{q})$ and $\ell_c^{opt} = 1 - \frac{1}{N}\frac{r}{q} + o(\frac{r}{q})$, which can also be easily obtained from the standard expressions of ℓ_d^{opt} and ℓ_c^{opt} .

Remark 8.19 Observe that $||P^m||_F^2 = 1 + \mu \rho^{2m}(P) + o(\rho^{2m}(P))$, where μ is the algebraic multiplicity⁵ of the second largest eigenvalue. Therefore, for large m, minimizing the Frobenius norm of P^m is equivalent to minimizing the essential spectral radius of P (up to $o(\rho^{2m}(P))$ terms).

Remark 8.20 It is interesting to observe that the communication graph \mathcal{G} determines the coefficient of the first order expansion of ℓ in r/q. Note that, in the extreme cases $P^{opt} = I$ (no communication, decentralized estimation) and $P^{opt} = \frac{11^*}{N}$ (communication graph fully connected, centralized estimation) one recovers, respectively, $\ell_d^{opt} = 1 - \frac{r}{q} + o(\frac{r}{q})$ and $\ell_c^{opt} = 1 - \frac{1}{N}\frac{r}{q} + o(\frac{r}{q})$, which can also be easily obtained from the standard expressions of ℓ_d^{opt} and ℓ_c^{opt} .

⁵We assume here $\rho(P) < 1$.

8.5.3 High measurement noise $(q/r \rightarrow 0)$

Similarly to the previous section, we now consider the other limiting case for $q/r \approx 0$. The following result holds.

Theorem 8.21 Let $P^{opt}(r,q)$, $\ell^{opt}(r,q)$ be defined as above and denote with p(P) the number of eigenvalues of P on the unit circle. Then

$$\lim_{q/r \to 0} p(P^{opt}(r,q)) = \min_{P \in \mathcal{P}} p(P) =: p^{opt}.$$

Moreover

$$\ell^{opt}(r,q) = \sqrt{\frac{N}{p^{opt}}}\sqrt{\frac{q}{r}} + o\left(\sqrt{q/r}\right).$$

Proof: We start by observing that $\lim_{q/r\to 0} \ell_d^{opt} = 0$ and $\lim_{q/r\to 0} \ell_c^{opt} = 0$. More precisely it is possible to see $\ell_d^{opt} = \sqrt{\frac{q}{r}} + o\left(\frac{q}{r}\right)$ and $\ell_c^{opt} = \sqrt{N}\sqrt{\frac{q}{r}} + o\left(\frac{q}{r}\right)$. From the previous observation and from Theorem 8.13 it follows that

$$\lim_{q/r \to 0} \ell^{opt}(r, q|P) = 0.$$
(8.28)

Assume now by contradiction that $\lim_{q/r\to 0} p\left(P^{opt}(r,q)\right) \neq p^{opt}$. This means that there exists a sequence $\frac{q_1}{r_1} > \frac{q_2}{r_2} > \frac{q_3}{r_3} > \ldots$, such that $p\left(P^{opt}(r_i,q_i)\right) \geq p^{opt} + 1, \forall i \in \mathbb{N}$. Let us rewrite J in the following way

$$\begin{split} J(P,\ell;r,q) &= \frac{qN + r\ell^2 p(P)}{1 - (1-\ell)^2} + r\ell^2 \sum_{i:|\lambda_i| \neq 1} \frac{|\lambda_i|^{2m}}{1 - (1-\ell)^2 |\lambda_i|^{2m}} \\ &= \frac{qN + r\ell^2 p(P)}{1 - (1-\ell)^2} + \frac{r\ell^2}{1 - (1-\ell)^2} f(\ell) \end{split}$$

where

$$f(\ell) = \left(1 - (1 - \ell)^2\right) \sum_{i:|\lambda_i| \neq 1} \frac{|\lambda_i|^{2m}}{1 - (1 - \ell)^2 |\lambda_i|^{2m}}$$

Note that $\lim_{\ell \to 0} f(\ell) = 0$. Hence

$$J(P,\ell;r,q) = \frac{qN + r\ell^2 p(P)}{1 - (1-\ell)^2} + o\left(\frac{r\ell^2}{1 - (1-\ell)^2}\right)$$
(8.29)

Let now \overline{P} be such that $p(\overline{P}) = p^{opt}$. From (8.29) it follows that

$$J(P,\ell;r,q) - J(\bar{P},\ell;r,q) = (p(P) - p^{opt}) \frac{r\ell^2}{1 - (1-\ell)^2} + o\left(\frac{r\ell^2}{1 - (1-\ell)^2}\right).$$

Equation (8.28) implies that also $\lim_{i\to\infty} \ell^{opt}(r_i, q_i|P) = 0$ so that for *i* large enough

$$J(P^{opt}(q_i, r_i), \ell^{opt}(q_i, r_i); r_i, q_i) > J(\bar{P}, \ell^{opt}(q_i/r_i); r_i, q_i)$$

which yields a contradiction.

Consider now $\ell^{opt}(r,q)$. Let us first compute the partial derivative

$$\frac{\partial J(P^{opt}(r,q),\ell;r,q)}{\partial \ell} = \frac{r\left(2p(P^{opt}(r,q))\ell^2 - 2\gamma^2 N + 2\gamma^2 N\ell\right)}{(1-(1-\ell)^2)} + \frac{r\left(+2\ell^2 f(\ell) + 2\ell^3 \dot{f}(\ell) - \ell^4 \dot{f}(\ell)\right)}{(1-(1-\ell)^2)}$$

where $\gamma = \sqrt{\frac{q}{r}}$. Let now $F(\gamma, \ell) = 2p(P^{opt}(r, q))\ell^2 - 2\gamma^2 N + 2\gamma^2 N l + 2\ell^2 f(\ell) + 2\ell^3 \dot{f}(\ell) - \ell^4 \dot{f}(\ell)$ and consider the equation $F(\gamma, \ell) = 0$. We adopt an argument similar to the proof of the implicit function theorem. By applying the Taylor's expansion around (0, 0) we get

$$F(\gamma, \ell) = -4N\gamma^2 + 4p(P^{opt}(r, q))\ell^2 + o\left(\|\gamma^2\| + \|\ell^2\|\right)$$

and equating this expression to 0 we obtain that $\ell^{opt}(r,q) = \ell^{opt}(r,q|P^{opt}(r,q))$ satisfies

$$\ell^{opt}(r,q) = \gamma \sqrt{\frac{N}{p(P^{opt}(r,q))}} + o(\gamma)$$
$$= \sqrt{\frac{N}{p(P^{opt}(r,q))}} \sqrt{\frac{q}{r}} + o\left(\sqrt{\frac{q}{r}}\right)$$

Therefore, using an argument similar to the one adopted in the proof of Theorem 8.17, it follows that

$$\ell^{opt}(r,q) = \sqrt{\frac{N}{p^{opt}}} \sqrt{\frac{q}{r}} + o\left(\sqrt{\frac{q}{r}}\right)$$

which concludes the proof.

Remark 8.22 Also in this case the constant in the first order expansion of the optimal gain as a function of q/r depends on the communication graph; the extreme cases are, respectively, $\ell_c^{opt}(r,q) = \sqrt{N}\sqrt{q/r} + o(\sqrt{q/r})$ and $\ell_d^{opt}(r,q) = \sqrt{q/r} + o(\sqrt{q/r})$. In fact, if sensors cannot communicate, then necessarily $P^{opt} = I$, therefore $\ell^{opt}(r,q) = \ell_d^{opt}(r,q)$, while if the communication graph is fully connected, then $P^{opt} = \frac{1}{N}\mathbb{1}\mathbb{1}^*$, therefore $\ell^{opt}(r,q) = \ell_c^{opt}(q/r)$. The previous theorem shows also that for $q \ll r$ the optimizing P, while being consistent with the communication graph, has to minimize the number of unitary eigenvalues.

8.6 Illustrative examples

In this section we provide some examples illustrating the approach proposed in this paper. In particular in Example 8.23 we solve analytically the minimization problem formulated in Section 8.3. In Example 8.24 and in Example 8.25 we provide a numerical comparison between the approach presented here and the method proposed in [5]. In Example 8.27, inspired by [90], we propose a new scheme for updating the estimates which seems to be very promising in order to improve the performance. In particular we allow the sensors to keep in memory also the previously received estimates, and not only the current one, and to use them to build an updated estimate. More precisely the estimate $\hat{x}(t+(h+1)\delta)$ will be a suitable weighted combination of the current estimate $\hat{x}(t+h\delta)$ and of the previous estimate $\hat{x}(t+(h-1)\delta)$. It has been shown by Muthukrihnan et al. [90] in the context of load balancing, and by Cao et al. [28] in the context of gossip algorithms, that the use of memory permits to speed up convergence and to improve performance. We show here, by means of simulative results, that for the set of matrices considered in Example 8.23 and 8.24 the presence of one level of memory permits to reach better performance in terms of the cost function J.

Example 8.23 Assume that Q is the set of the circulant stochastic matrices of the form

$$P_{k} = \begin{bmatrix} 1 - 2k & k & 0 & 0 \cdots & 0 & k \\ k & 1 - 2k & k & 0 \cdots & 0 & 0 \\ 0 & k & 1 - 2k & k \cdots & 0 & 0 \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ k & 0 & 0 & 0 \cdots & k & 1 - 2k \end{bmatrix}$$
(8.30)

where k is a real in $\left[0, \frac{1}{2}\right]$ and assume that m = 1. We want to solve the minimization problem considered in Section 8.3, i.e.

$$P^{opt} = \underset{P \in \mathcal{P}}{\arg\min} \ J = \underset{P \in \mathcal{P}}{\arg\min} \ \sum_{i=1}^{N-1} \frac{|\lambda_i|^{2m}}{1 - (1-\ell)^2 |\lambda_i|^{2m}}.$$
 (8.31)

Unfortunately, even if \mathcal{P} exhibits a particular structure, it is not possible to give, in general, a closed form expression for the optimizing matrix P^{opt} ,

which has to be found using numerical search techniques.

However, similarly to Example 8.5, a more detailed study of (8.31) can be provided when the number of sensors N goes to infinity. To this purpose it is convenient to consider the normalized cost:

$$\bar{J}_N := \frac{1}{N} \sum_{i=0}^{N-1} \frac{|\lambda_i|^2}{1 - (1-\ell)^2 |\lambda_i|^2}$$

and the function $f:\mathbb{C}\to\mathbb{C}$

$$f(z) = 1 - 2k + kz + kz^{-1}.$$

The eigenvalues of \mathcal{P}_k can be expressed as $\lambda_h = 1 - 2k + ke^{j\frac{2\pi}{N}h} + ke^{-j\frac{2\pi}{N}h}$ (see e.g. [46]). Note that $\lambda_h = f\left(e^{j\frac{2\pi}{N}h}\right)$. As the number of agents grows, the normalized cost J_N converges to:

$$\lim_{N \to \infty} J_N = \bar{J}_\infty = \frac{1}{2\pi} \int_0^{2\pi} \frac{|f(e^{j\phi})|^2}{1 - (1 - \ell)^2 |f(e^{j\phi})|^2} d\phi = \frac{1}{2\pi j} \oint_\gamma \frac{z^{-1} f^2(z)}{1 - (1 - \ell)^2 f^2(z)} dz$$

where γ is the unit circle. It is straightforward to see that the poles of $\frac{z^{-1}f^2(z)}{1-(1-\ell)^2f^2(z)}$ inside γ are

$$z_1 = \frac{\ell + 2(1-\ell)k - \sqrt{\ell^2 + 4\ell(1-\ell)k}}{2(1-\ell)k},$$
$$z_2 = \frac{\ell - 2 - 2(1-\ell)k + \sqrt{(2-\ell)^2 + 4(1-\ell)(2-\ell)k}}{2(1-\ell)k}$$

and

$$z_3 = 0.$$

The integral can be computed explicitly using the residue theorem yielding:

$$\bar{J}_{\infty} = -\frac{1}{(1-\ell)^2} + \frac{1}{2(1-\ell)^2} \left(\frac{1}{\sqrt{\ell^2 + 4\ell(1-\ell)k}} - \frac{1}{\sqrt{(2-\ell)^2 - 4(1-\ell)(2-\ell)k}} \right)$$

By equating the first derivative of \bar{J}_{∞} to zero, we get that P^{opt} tends to the matrix $P_{k^{opt}}$ for $N \to \infty$, where

$$k^{opt} = \frac{2 - 2\ell + \sqrt[3]{(2 - \ell)\ell^2} - \sqrt[3]{\ell(2 - \ell)^2}}{4(1 - \ell)}.$$

Example 8.24 In [5] the authors analyze a general MIMO scenario where the gain $\ell = \ell(t)$ (K in their terminology) and the consensus matrix P = P(t)are time varying matrices which are chosen recursively at each time step. In order to compare the results in [5] with our approach we assume that the averaging matrix W in [5] corresponds to performing m consensus iterations using the matrix P, i.e. $W = P^m$. In [5] the gain ℓ is chosen to minimize the estimation error covariance of the local estimators (i.e. in a decentralized fashion) and it is different for each sensor. Moreover the consensus matrix P is chosen so that the estimation error covariance of the local estimators is minimized after consensus (weighted averaging in [5])⁶.



Figure 8.1: Comparison between $J_1(P_1^{opt}(m), \ell_1^{opt}(m), m)$, and $J_1^r(m)$ (*left*). Comparison between $J_2(P_2^{opt}(m), \ell_2^{opt}(m), m)$, and $J_2^r(m)$ (*right*).

In the simulation reported in this example, we assume that N = 100, q = 1 and r = 1. Moreover we assume that \mathcal{P} is the same set of circulant matrices defined in the previous example. We consider the minimization of both J_1 and J_2 , with J_1 , J_2 defined as in the Section II. We use the following notational conventions: $P_1^{opt}(m)$, $\ell_1^{opt}(m)$ and $P_2^{opt}(m)$, $\ell_2^{opt}(m)$ are the optimal consensus matrices and the optimal gains respectively for J_1 and J_2 obtained by solving numerically the problem formulated at the end of Section II, given by:

$$\begin{pmatrix} P_1^{opt}(m), \ \ell_1^{opt}(m) \end{pmatrix} \in \arg \min_{\ell \in (0,1), P \in \mathcal{P}} J_1(P, \ell; m, r, q) \\ \begin{pmatrix} P_2^{opt}(m), \ \ell_2^{opt}(m) \end{pmatrix} \in \arg \min_{\ell \in (0,1), P \in \mathcal{P}} J_2(P, \ell; m, r, q)$$

As mentioned earlier, in [5] the optimal gain and the optimal consensus matrix are found *recursively* at each time step t. The fact that the consensus matrix is of the form (8.30) implies that the gain is the same for all the

⁶Technically, the approach proposed in [5] was applied only for the case m = 1.

sensors: we denote it by $\ell^r(t,m)$. Moreover $P^r(t,m)$ represents the optimal consensus matrix (the superscript "r" means *recursively*). The asymptotic values (in t) of $\ell^r(t,m)$ and $P^r(t,m)$ are denoted respectively by $\ell^r(m)$ and $P^r(m)$, i.e. $\lim_{t\to\infty} \ell^r(t,m) = \ell^r(m)$ and $\lim_{t\to\infty} P^r(t,m) = P^r(m)$. Similarly, we indicate by $J_1^r(m)$ and $J_2^r(m)$ the asymptotic cost values⁷ to which J_1 and J_2 converge using the method proposed in [5]. Finally, since the set \mathcal{P} is parameterized by k, we shall identify a matrix P_k with the corresponding value of the parameter k. Hence, we use $k_1^{opt}(m)$, $k_2^{opt}(m)$ and $k^r(m)$ in lieu of $P_1^{opt}(m)$, $P_2^{opt}(m)$ and $P^r(m)$. Simulations for m ranging in the interval [1, 15] are shown in Figures 8.1 and 8.2.



Figure 8.2: Optimal gains *(left)* and consensus matrices *(right)* as a function of number of exchanged messages m. Optimal consensus matrices P are parameterized by k. In the right panel we also report the values $k(\rho^{opt})$, which minimizes the spectral radius $\rho(P)$ and $k(||P||_F^{opt})$ which minimizes the Frobenius norm $||P||_F$.

From left panel of Figure 8.2 we can see that the iterative (local) optimization proposed in [5] converges to values of the parameters $\ell^r(m)$ which are different from the optimal values obtained minimizing the asymptotic cost proposed in this paper. Indeed, the recursive approach seems to give a worse performance than the approach proposed in this paper both in terms of the performance costs J_1 and J_2 , as shown in Figure 8.1. Moreover, for large m, the gain $\ell_1^{opt}(m)$ converges to the optimal centralized gain ℓ_c^{opt} , as shown in Section 8.4.

Note also from right panel of Figure 8.2 that the optimal consensus parameter k of the matrix P_k for small m is close to the value that minimizes the Frobenius norm $||P||_F$, while for large m it converges to the value that minimizes the spectral radius $\rho(P)$, as shown in Section 8.3.

⁷There is no proof of convergence in [5]; however this is observed experimentally.

Example 8.25 In this simulation we consider a strongly connected random geometric graph generated by choosing N points at random in the unit square, and then placing an edge between each pair of points at distance less than 0.3. We assume that N = 30, q = 1 and r = 1. It is worth noticing that in this case differently from the previous one, the gains, calculated by the method proposed in [5] recursively at each instant time t, are in general different for each sensor. Hence we report here only the comparison between the cost functions. We run simulations for m ranging in the interval [1, 10]. The results obtained are depicted in Figure 8.3. The notation is the same used previously. Note that $\ell_1^{opt}(m), P_1^{opt}(m)$ and $\ell_2^{opt}(m), P_2^{opt}(m)$ yield again



Figure 8.3: Comparison between $J_1(P_1^{opt}(m), \ell_1^{opt}(m), m)$, $J_1(P_2^{opt}(m), \ell_2^{opt}(m), m)$ and $J_1^r(m)$ (left). Comparison between $J_2(P_2^{opt}(m), \ell_2^{opt}(m), m)$, $J_2(P_1^{opt}(m), \ell_1^{opt}(m), m)$ and $J_2^r(m)$ (right).

values of J_1 and J_2 which are better respectively than J_1^r and J_2^r .

Remark 8.26 It is worth remarking that the optimization strategy proposed in [5] gives worse performance in terms of the asymptotic values for both costs J_1 and J_2 in both the previous examples. This is somewhat to be expected since the former approach recursively optimize the cost at the next time step, not the steady state cost. This recursive strategy gives the optimal steady state performance only for the centralized scenario, i.e. when the communication graph is fully connected, as well known from any standard textbook on optimal estimation and Kalman filtering [36]. When the graph is not fully connected, this strategy is not guaranteed to converge to the optimal value, as indeed shown in the previous numerical examples.

Example 8.27 Inspired by the work of Muthukrihnan et al. [90], in this example we propose the following scheme of updating the estimate $\hat{x}(k+h\delta|k)$

which adds a memory buffer,

$$\hat{x}(k + (h+1)\delta|k) = P\hat{x}(k + h\delta|k) \quad \text{if } h = 0,$$

and

$$\hat{x}(k+(h+1)\delta|k) = \nu P\hat{x}(k+h\delta|k) + (1-\nu)\hat{x}(k+(h-1)\delta|k) \quad \text{if } h = 1, \dots, m-1$$
(8.32)

where ν is a weighting parameter. It has been proved in [90], that, when we are dealing with a consensus algorithm, if $\nu \notin [0,2]$ then (8.32) does not yield the convergence. Let $\tilde{x} (k + h\delta|k) = x(k + h\delta)\mathbb{1} - \hat{x}(k + h\delta|k)$ denote again the estimation error and let us introduce, for $1 \leq h \leq m$, the following quantities

$$z(k+h\delta|k) = \begin{bmatrix} \tilde{x}(k+h\delta|k)\\ \tilde{x}(k+(h-1)\delta|k) \end{bmatrix}$$

and

$$\Sigma(k+h\delta|k) = \mathbb{E}\left[z(k+h\delta|k)z^*(k+h\delta|k)\right] = \begin{bmatrix}\Sigma_{11}(k+h\delta|k) \ \Sigma_{12}(k+h\delta|k)\\ \Sigma_{21}(k+h\delta|k) \ \Sigma_{22}(k+h\delta|k)\end{bmatrix}.$$

where

$$\Sigma_{11}(k+h\delta|k) = \mathbb{E}[\tilde{x} (k+h\delta|k) \tilde{x}^* (k+h\delta|k)]$$

$$\Sigma_{12}(k+h\delta|k) = \mathbb{E}[\tilde{x} (k+h\delta|k) \tilde{x}^* (k+(h-1)\delta|k)]$$

$$\Sigma_{21}(k+h\delta|k) = \mathbb{E}[\tilde{x} (k+(h-1)\delta|k) \tilde{x}^* (k+h\delta|k)]$$

$$\Sigma_{22}(k+h\delta|k) = \mathbb{E}[\tilde{x} (k+(h-1)\delta|k) \tilde{x}^* (k+(h-1)\delta|k)]$$

Note that $\Sigma_{12}(k+h\delta|k) = \Sigma_{21}^*(k+h\delta|k)$. It is possible to see that the above quantities satisfy the following recursive equations

$$\begin{split} \Sigma_{11} \left(k + (h+1)\delta \right) &= \nu^2 P \Sigma_{11} \left(k + h\delta \right) P^* + 2\nu (1-\nu) \frac{q}{m} \mathbb{1} \mathbb{1}^* + \frac{q}{m} \mathbb{1} \mathbb{1}^* + \\ &+ \nu (1-\nu) \left[P \Sigma_{12} \left(k + h\delta \right) + \Sigma_{21} \left(k + h\delta \right) P^* \right] + \\ &+ (1-\nu)^2 \Sigma_{22} \left(k + h\delta \right) + (1-\nu)^2 \frac{q}{m} \mathbb{1} \mathbb{1}^* \\ \Sigma_{12} \left(k + (h+1)\delta \right) &= \nu P \Sigma_{11} \left(k + h\delta \right) + (1-\nu) \Sigma_{21} \left(k + h\delta \right) + (1-\nu) \frac{q}{m} \mathbb{1} \mathbb{1}^* \\ \Sigma_{22} \left(k + (h+1)\delta | k \right) &= \Sigma_{11} \left(k + h\delta \right) \end{split}$$

In our simulation we assume that N = 50, m = 5, q = 1, r = 1 and that $P = P_{\frac{1}{3}}$ where $P_{\frac{1}{3}}$ is as the matrix introduced in Example 8.23. Moreover $\ell = \ell^{opt}$ where

$$\ell^{opt} = \underset{\ell \in (0,1)}{\arg \min} J\left(P_{\frac{1}{3}}, \ell; 5, 1, 1\right) \approx 0.879.$$

| Section 8.3 | undirected graph | $m \to \infty$ | $\ell \to 1$ |
|------------------------|--|--|--|
| Fixed ℓ | $P^{opt} \in \text{symmetric}$ Section 8.3.1 | $P^{opt} \in \underset{P \in \mathcal{P}}{\arg\min} \rho(P)$ Section 8.3.2 | $P^{opt} \in \underset{P \in \mathcal{P}}{\operatorname{arg min}} \ P^m\ _F$ Section 8.3.3 |
| u | 1 | | |
| Section 8.4 | m = 0 | $0 < m < \infty$ | $m \to +\infty$ P primitive |
| Fixed P | $\ell^{opt}(P,0) = \ell^{opt}_d$ | $\ell_d^{opt} < \ell^{opt}(P, m) \le \ell^{opt}(P, m+1) < \ell_c^{opt}$ | $\ell^{opt}(P,\infty) = \ell_c^{opt}$ |
| | Theorem 8.13 | Theorem 8.13 | Theorem 8.13 |
| Section 8.5 | $m \to \infty$ | r/q ightarrow 0 | $r/q \rightarrow \infty$ |
| Optimal ℓ and P | $\begin{array}{c} P^{opt} \in \arg\min_{\substack{P \in \mathcal{P} \\ \ell^{opt} \rightarrow \ell_c^{opt} \\ \text{Section 8.5.1}} \end{array}$ | $P^{opt} \in \underset{\substack{P \in \mathcal{P} \\ P \in \mathcal{P}}}{\operatorname{arg min}} \ P^{m}\ _{F},$ $\ell^{opt} = 1 - \frac{\ \bar{P}^{m}\ _{F}^{2}}{N} \frac{r}{q} + o\left(\frac{r}{q}\right)$ Section 8.5.2 | $P^{opt} \in \arg\min p(P),$ $P \in \mathcal{P}$ $\ell^{opt} = \sqrt{\frac{N}{p^{opt}}} \sqrt{\frac{q}{r}} + o\left(\sqrt{\frac{q}{r}}\right)$ Section 8.5.3 |

Table 8.1: Summary of the results

Clearly, if $\nu = 1$, then $\lim_{k \to +\infty} \operatorname{tr} (\Sigma_{11}(k+1|k)) = J\left(P_{\frac{1}{3}}, \ell^{opt}; 5, 1, 1\right)$, which is the cost function introduced in Equation (8.12) for which no memory is added. In Figure 8.4 we depict the behavior of $\lim_{k \to +\infty} \operatorname{tr} (\Sigma_{11}(k+1|k))$ for ν ranging in [0, 2]. It is remarkable to note that the minimum is reached when $\nu \approx 1.44$, meaning that the presence of memory can improve the performance.



Figure 8.4: Steady state error tr $(\Sigma_{11}(k+1|k))$ of distributed optimal estimation with memory as a function of weighting parameter ν . Figure also indicates optimal cost in the absence of memory which corresponds to $\nu = 1$.

8.7 Conclusions

In this chapter we have studied a prototypical problem of distributed estimation for sensor networks; the state of a scalar linear system is estimated via a two stage procedure which consists in (i) a standard (and decentralized) Kalman-like update and (ii) information propagation using consensus strategies. To this purpose two design parameters, i.e. the Kalman gain ℓ and the consensus matrix P have to be designed. This choice is made by optimizing the steady state prediction (or estimation) error. We have discussed, under specific circumstances, the behavior of the "optimal" parameters. This is summarized in table of Figure 8.6.

Although these results have been obtained for simple scenarios where the state is scalar and all sensors are equal, they provide useful guidelines for choosing the local filter gain ℓ and the consensus matrix P also for more general scenarios. In particular, we showed that the common practise of finding algorithms that minimize the spectral radius for the consensus matrix P is not necessarily the optimal strategy in the context of optimal estimation of time-varying signals. In fact, we showed that depending on some specific regimes, it is more convenient to optimize the Frobenius norm $||P^m||_F$ or the sum of the unitary eigenvalues p(P). Moreover, as discussed in Section V, we showed that the joint optimization of P and ℓ is not convex even in our simple setup. We also compared our approach with the recursive optimization proposed by Alriksson et al. [5], showing also that their strategy based on minimization of the estimation error at the next time step, fails to minimize the steady state cost (see Figures 8.1 and 8.3).

Many research avenues still deserve to be explored. The most promising one is the use of memory to improve performance. As shown in Example 8.27 in the previous section, memory can considerably improve estimation performance. However, the impact of memory length on performance and optimization algorithms for the weighting parameters are still open problems. Another important aspect is the extension of the results in this paper to the multivariable case where the gains for the sensors can be different. Finally, it is fundamental to find provably optimal strategies to simultaneously optimize the consensus matrix P and the update gains ℓ , since the steady state cost function is non-convex in these parameters.

Chapter 9

Conclusions

The consensus problem and the more specific average consensus problem represent the first step toward a comprehensive understanding of how, in complex networked systems, the dynamics of the individual systems can give rise to a group behavior. This is mainly motivated by two reasons. The first is that these problems are the simplest problems we can formulate in the new emerging area of the cooperative control, the second is related to the wide range of applications of the consensus algorithms in information processing in sensor networks, in load balancing and in Multi-Vehicle Cooperative Control.

In this dissertation we focused on the average consensus problem by providing some theoretical developments that, now, we briefly summarize.

In Chapter 3, we provided a mathematical characterization of the intuitive idea that the amount of information exchanged by the systems in the consensus algorithm has to influence the rate of convergence towards the asymptotic agreement. We did this for a class of graphs exhibiting particular symmetries: the Cayley graphs defined on Abelian groups. We showed that, if we impose symmetries in the communication network, and thus also in the consensus matrix, and we keep the number of incoming arcs in each node bounded, the convergence rate degrades as the number of systems increases.

In Chapter 4 we focused on *randomized consensus algorithms*. We reviewed the concepts of *probabilistic consensus* and *average probabilistic consensus*. In the first part of the chapter we introduced two random strategies, that illustrate a remarkable property of the randomized consensus algorithms, i.e., they allow to achieve better performance than deterministic ones with comparable complexity. In the second part of the chapter we considered the symmetric gossip algorithm, which is particularly suitable to model the

communication network in applications to sensors, peer-to-peer and *ad hoc* networks. We provided an interesting characterization of this algorithm when considering the Cayley graphs.

In Chapters 5, 6, 7, we analyzed the more realistic and practical situation in which the communication network between the systems is constituted of only rate-constrained digital links. We assumed that the nodes quantize their information before transmitting it. In particular, we considered two types of quantizers, well-known in the literature: the *deterministic* uniform quantizer and the *probabilistic* uniform quantizer.

In Chapter 5 we focused on the time-invariant situation by providing a simple and effective adaptation of the standard average consensus algorithm which does not converge to an asymptotic agreement but is able to preserve the average of states and to drive the systems reasonably near to the consensus. We studied this scheme by means of a worst-case model and a probabilistic model showing favorable convergence properties and providing performance bounds for the limit points of the iterates generated.

In Chapter 6 we analyzed the effects of the quantization on the symmetric gossip algorithm. In order to overcome these effects, we introduced two updating rule, the *globally quantized* strategy and the *partially quantized* strategy. In the former the nodes use only quantized information to update their state, instead in the latter they have access also to exact information regarding their own state. We saw that the partially quantized strategy, with both the quantizers, deterministic and probabilistic, do not reach the consensus in general, but maintain the average of the state at each iteration and drive all the states very close to the average of the initial condition. On the other hand, we showed that the globally strategy leads almost surely to a consensus which, however, do not coincide with the average of the initial condition. We provided some simulations characterizing the distance between the consensus point and the initial average. While using the deterministic quantizer this distance turns out to be not negligible, with the probabilistic quantizer the consensus is surprisingly reached very close to the average of the initial condition.

In Chapter 7 we introduced coding/decoding schemes that leads to a quantized strategy that permits both to maintain the initial average and to reach it asymptotically. More precisely we adapted coding/decoding stchemes, that have been proposed for centralized quantized control problems, to the distributed consensus problem. In particular, we presented two coding/decoding strategies, one based on the exchange of logarithmically quantized information, the other on a zoom in- zoom out strategy which involves the use of uniform quantizers. We provided analytical and numerical results illustrat-
ing the convergence properties of these strategies. In particular we showed that the convergence factors depend smoothly on the accuracy parameter of the quantizers used and that, remarkably, that the critical quantizer accuracy sufficient to guarantee convergence is independent from the network dimension.

Finally in Chapter 8 we considered a possible application of the consensus ideas to a prototypical problem of distributed estimation for sensor networks. More precisely we analyzed a two-stage procedure, estimating the state of a scalar linear system, which consists in (i) a standard (and decentralized) Kalman-like update and (ii) information propagation using consensus strategies. In this procedure two design parameters, i.e. the Kalman gain ℓ and the consensus matrix P have to be designed. This choice is made by optimizing the steady state prediction (or estimation) error. We discussed, under specific circumstances, the behavior of the "optimal" parameters.

Many research avenues still deserve to be explored. In particular, regarding the randomized consensus algorithms,

- to provide an analysis of randomized consensus algorithms for wide classes of graphs;
- to provide a detailed studied of the rate of convergence for the symmetric gossip algorithms for the Cayley graphs and for the random geometric graphs;

regarding the quantized consensus,

- to explain why and in which sense the probabilistic analysis seems to capture the main features of the quantized strategies also when the information is quantized by means of deterministic quantizers;
- to evaluate the speed of convergence of the *partially quantized* strategy and of the *globally quantized* strategy;
- to carry out a deeper analysis on the *globally* quantized strategy using probabilistic quantizers; in particular, to estimate, in some probabilistic way, the distance of the consensus point to which this strategy leads the systems from the average of the initial condition;
- to extend the analysis, carried out in Chapter 7, to the case of a general quasi-doubly stochastic matrix *P*;

- to analyze, from a theoretical point of view, the zoom-in/ zoom-out strategy;
- to look for encoding and decoding methods which are able to solve the average consensus problem also with noisy digital channel;

finally, regarding the distributed Kalman filtering,

- to explore the use of memory to improve the performance of the twostep procedure introduced in Chapter 8;
- to find provably optimal strategies to simultaneously optimize the consensus matrix P and the update gains ℓ , since the state cost function is non-convex in these parameter.

Appendix A

Harmonic analysis on finite group

A.1 Introduction

The main objective of this appendix is that of reviewing some concepts on harmonic analysis on finite groups. The presentation is mainly based on [14].

A.2 Finite Abelian groups

We start by recalling the definition of a finite Abelian group.

Definition A.1 A finite Abelian group G of order N, is a set G of cardinality |G| = N, closed under a binary operation +, such that the following axioms are satisfied¹

(i) (Associativity) For all $g, h, \ell \in G$, we have that

$$(g+h) + \ell = g + (h+\ell).$$

(ii) (Commutativity) The binary operation + is commutative, i.e. for any $g, h \in G$ we have that

$$g + h = h + g.$$

 $^{^1\}mathrm{In}$ the following we will use |.| to denote both the cardinality of a set and the absolute value operator.

(iii) (Neutral element) There exists a neutral element $e \in G$ such that for all $g \in G$

g + e = g.

In the following, by convention, we will indicate with 0 such element.

(iv) (Inverse element) Corresponding to each $g \in G$ there is an element $g' \in G$ such that

$$g + g' = 0.$$

In the following we will indicate such element with -g.

From now on when we refer to a group we will implicitly consider finite Abelian groups. We will assume the operation to be addition if not otherwise stated.

Example A.2 Let us consider $\mathbb{Z}_N = \{0, 1, ..., N-1\}$. If we consider as binary operation the addition modulo N, all the previous axioms are satisfied, and thus it represents a finite Abelian group.

Let G and H be two Abelian groups. The two groups, in general, need not to be finite. An homomorphism is a map $\phi: G \to H$ satisfying

$$\phi(g+h) = \phi(g) + \phi(h)$$

with $g \in G$ and $h \in H$. An homomorphism that is bijective is called *i*somorphism. Two Abelian groups are isomorphic if there is an homomorphism between them. Isomorphic groups are regarded as "equal" from a structural or group-theoretic point of view, even though their elements might be quite different kinds of objects. In the following we will write $G \cong H$ to denote that G is isomorphic to H.

Example A.3 Let us consider the finite additive group $\mathbb{Z}_4 = \{0, 1, 2, 3\}$ and the binary group $B = \{00, 01, 10, 11\}$ with operation the 2-bits binary sum. If we consider the map

$$\phi: G \to H: g \mapsto \phi(g) = binary_2(g)$$

where $binary_2(g)$ is the binary representation in two bits of the integer g, then we see immediately that it is an isomorphism. Thus $G \cong H$.

Given a finite group G, a subgroup H of G is a subset of G that also forms a group under the same binary operation +.

Example A.4 Let us consider \mathbb{Z}_8 , and let $H = \{0, 4\} \subset \mathbb{Z}_8$. It is easy to see that H is also a group with respect to the addition modulo eight.

Given a finite group G let us consider a subset S of a group. Then $\langle S \rangle$, the subgroup generated by S, is the smallest subgroup of G containing every element of S. Equivalently, $\langle S \rangle$ is the subgroup of all elements of G that can be expressed as the finite product of elements in S and their inverses.

Example A.5 Let us consider the group

 $\mathbb{Z}_3 \times \mathbb{Z}_3 = \{(0,0), (0,1), (0,2), (1,0), (1,1), (1,2), (2,0), (2,1), (2,2)\}.$

Let us consider $S = \{(0,0)\}$. It is clear that $\langle S \rangle = S$ is a subgroup of G. Let us consider $S = \{(0,0), (0,1)\}$ then we see that $S \subset \langle S \rangle = \{(0,0), (0,1), (0,2)\}$ is the smallest subgroup of G containing S.

If $G = \langle S \rangle$, then we say S generates G; and the elements in S are called generators or group generators.

Example A.6 Let us consider again group $\mathbb{Z}_3 \times \mathbb{Z}_3$. Then we see that if we choose $S = \{(0,0), (0,1), (1,0)\}$ then $\langle S \rangle = G$.

Definition A.7 A finite Abelian group G is called a cyclic group if its elements are all of the form kg for $k \in \mathbb{Z}$ for some fixed $g \in G$. The element g is called the generator of the group G, and we will write that $\langle g \rangle = G$.

It turns out that a finite cyclic group is a group generated by a single element. In a finite cyclic group of order N, the generator satisfies Ng = 0, and N is the smallest positive integer with this property, and N is called the order of the generator. Thus the order of the generator is equal to the order of the group (even if the sense of the word "order" is different). It is then easy to see that any two finite cyclic groups of the same order are isomorphic.

Example A.8 Let us consider the group \mathbb{Z}_4 . Then we have that

 $<1>=<3>=\mathbb{Z}_4.$

Notice that $\langle 2 \rangle = \{0, 2\}$ which, clearly, is not \mathbb{Z}_4 . The group \mathbb{Z}_4 is called the cyclic group of the integers modulo four.

The direct sum $G \oplus H$ of two Abelian groups G and H is the set of all ordered pairs (g, h), with $g \in G$ and $h \in H$. If we define the following addition operation

$$(g_1, h_1) + (g_2, h_2) = (g_1 + g_2, h_1 + h_2)$$

then it is easy to see that $G \oplus H$ is an Abelian group whose neutral element is (0,0) and the inverse of (g,h) is (-g,-h). The definition of direct sum is easily extended to more than two Abelian groups. We have the following fundamental result.

Theorem A.9 [Fundamental theorem of finite Abelian groups] Let G be a finite Abelian group. Then there exists cyclic groups $\mathbb{Z}_{q_1}, \mathbb{Z}_{q_2}, \ldots, \mathbb{Z}_{q_r}$ of orders $q_1, q_2, \ldots, q_r > 1$, respectively, where the q_i are prime powers, for $1 \leq i \leq r$, such that

$$G \cong \mathbb{Z}_{q_1} \oplus \mathbb{Z}_{q_2} \oplus \cdots \oplus \mathbb{Z}_{q_r}.$$

Thus G is isomorphic to the direct sum of cyclic groups.

Example A.10 Let us consider the group $\mathbb{Z}_6 = \{0, 1, 2, 3, 4, 5\}$. Using the previous theorem we have that

$$\mathbb{Z}_6 = \mathbb{Z}_2 \oplus \mathbb{Z}_3.$$

Indeed, let us consider the following map

$$\phi: \mathbb{Z}_2 \oplus \mathbb{Z}_3 \to : (g,h) \mapsto (3g+2h) \mod 6,$$

with $g \in \mathbb{Z}_2$ and $h \in \mathbb{Z}_3$. It is easy to see that ϕ is an isomorphism, and thus $\mathbb{Z}_6 \cong \mathbb{Z}_2 \oplus \mathbb{Z}_3$.

A.3 Group characters

Let G a finite Abelian group of order N. We can define the character of the group G as follows.

Definition A.11 A character of G is a homomorphism $\chi : G \to \mathbb{C}^* = \mathbb{C} - \{0\}$, which maps G to the non-zero multiplicative group of complex numbers.

Since χ is a homomorphism then we have that

$$\chi(g+h) = \chi(g)\chi(h), \qquad g,h \in G.$$

In particular, we have that

$$\chi(g)^N = \chi(Ng) = \chi(0) = 1, \ g \in G,$$

and so the values of χ are the N^{th} roots of the unity. Notice moreover that

$$\chi(-g) = \chi(g)^{-1} = \chi(\overline{g})$$

where the bar indicates the complex conjugate. The character defined by

$$\chi(g) = 1, \quad \forall g \in G,$$

is called the trivial (or principal) character of the group G. All the others are called non-trivial characters.

Example A.12 Consider the group \mathbb{Z}_N . We have that the characters of \mathbb{Z}_N are

$$\chi_{\ell}: g \mapsto \exp\left(i2\pi \frac{\ell}{N}g\right),$$

with $\ell = 0, 1, \ldots, N-1$ and $g \in \mathbb{Z}_N$. It is easy to see that the trivial character of group \mathbb{Z}_N is $\chi_0(g)$.

The characters of a finite Abelian group have many properties. We summarize here some important facts.

Proposition A.13 For any nontrivial character χ of G,

$$\sum_{g \in G} \chi(g) = 0.$$

Proof: Let χ be a non-trivial character of G and let $h \in G$ be such that $\chi(h) \neq 1$. Let $L = \sum_{g \in G} \chi(g)$, then we have

$$\chi(h)L = \sum_{g \in G} \chi(h)\chi(g) = \sum_{g \in G} \chi(h+g) = L.$$

We then have

$$L(\chi(h) - 1) = 0$$

which implies L = 0 since $\chi(h) \neq 0$. This concludes the proof.

Let us now denote \hat{G} the set of all characters. Let us define the following operation between two characters, χ and ϕ , of a group G

$$(\chi\phi)(g) = \chi(g)\phi(g). \tag{A.1}$$

It is easy to see that this set forms an Abelian group under the operation defined by (A.1). The conjugation represent the inversion in the group \hat{G} . The group of the characters \hat{G} is called the *d*ual group of *G*.

Proposition A.14 Let χ and ϕ be two characters of the group G. Then we have

$$\sum_{g \in G} \chi(g)\bar{\phi}(g) = \begin{cases} N & \text{if } \chi = \phi \\ 0 & \text{otherwise} \end{cases}$$

Proof: If $\chi = \phi$ then this follows from the fact that $\bar{\chi}(g) = \chi(g)^{-1}$. If $\chi \neq \phi$, then $\chi \bar{\phi}$ is a nontrivial character of G and using Proposition A.13 the result follows.

Example A.15 Consider the group \mathbb{Z}_N . The characters of the group are

$$\chi_\ell = \exp\left(i2\pi\frac{\ell}{N}g\right)$$

 $\ell = 0, \ldots, N-1$ and $g \in \mathbb{Z}_N$. Then the dual group of \mathbb{Z}_N is

$$\mathbb{Z}_N = \left\{ \chi_0, \ldots, \chi_{N-1} \right\}.$$

Lemma A.16 $\hat{\mathbb{Z}}_N \cong \mathbb{Z}_N$.

Proof: Since \mathbb{Z}_N is a cyclic group then $\langle 1 \rangle = \mathbb{Z}_N$. For the characters of \mathbb{Z}_N we have the form

$$\chi_{\ell}(k) = \exp\left(i2\pi\frac{\ell}{N}k\right)$$

where $k \in \mathbb{Z}, \ell = 0, \ldots, N - 1$. But this shows that χ_1 is a generator of \hat{Z}_N . Since two cyclic groups of the same order are isomorphic then we can conclude.

Lemma A.17 If the group G is expressed as direct sum, namely $G = G_1 \oplus G_2$ and $\phi_i : G_i \to \mathbb{C}^x$ is a character of G_i , i = 1, 2, then $\chi = \phi_1 \oplus \chi_2$, defined as

$$\chi(g_1, g_2) = \phi_1(g_1)\phi_2(g_2), \tag{A.2}$$

is a character of G. Moreover, all characters of G are of this form. Thus we have that

$$\hat{G} \cong \hat{G}_1 \oplus \hat{G}_2.$$

Proof: It is easy to show that χ is a character of G since it defines an homomorphism between G and $\mathbb{C}^x \times \mathbb{C}^x$ and

$$\chi(g+h) = \chi((g_1, g_2) + (h_1, h_2))$$

= $\chi(g_1 + h_1, g_2 + h_2)$
= $\phi(g_1 + h_1)\phi(g_2 + h_2)$
= $\phi(g_1)\phi(h_1)\phi(g_2)\phi(h_2)$
= $\phi(g_1 + g_2)\phi(h_1 + h_2)$
= $\phi(g)\phi(h)$

where we used the fact that ϕ_i is a character of G_i . It is clear that $\hat{G}_1 \oplus \hat{G}_2 \to \hat{G}$ defined by (A.2) is injective. We need to show then that if we consider $\chi(\tilde{g}_1, g_2) = \chi(g_1, g_2)$ then $\phi_1(\tilde{g}_1) = \phi_1(g_1)$. It follows that

$$0 = \chi(\tilde{g}_1, g_2) - \chi(g_1, g_2) = \phi_1(\tilde{g}_1)\phi_2(g_2) - \phi_1(g_1)\phi_2(g_2) = \phi_2(g_2)\Big(\phi_1(\tilde{g}_1) - \phi_1(g_1)\Big)$$

Since $\phi_2(g_2) = 0$ then it follows that $\phi_1(\tilde{g}_1) = \phi_1(g_1)$ as we wanted. Let us now consider $\chi \in \hat{G}$. Then the restriction $\phi_i = \chi|_{G_i}$ is a character of G_i , and it is easy to verify that $\chi = \phi_1 \oplus \phi_2$.

Theorem A.18 For arbitrary finite Abelian groups $\hat{G} \cong G$.

Proof: From Theorem A.9 we have that $G \cong \mathbb{Z}_{q_1} \oplus \cdots \oplus \mathbb{Z}_{q_r}$. We know that $\hat{\mathbb{Z}}_{q_i} \cong \mathbb{Z}_{q_i}$ by Lemma A.16. From Lemma A.17 follows that the direct sum $\hat{G} \cong \hat{\mathbb{Z}}_{N_1} \oplus \cdots \oplus \hat{\mathbb{Z}}_{N_k} \cong G$.

Let \mathbb{C}^G denote the space of functions $f : G \to \mathbb{C}$. This represents an *N*-dimensional linear space over \mathbb{C} . We introduce an inner product over this space

$$\langle f_1, f_2 \rangle = \frac{1}{N} \sum_{g \in G} f_1(g) f_2(\overline{g}), \qquad f_1, f_2 \in \mathbb{C}^G.$$

We then have the following result.

Theorem A.19 The elements of the set \hat{G} forms an orthonormal basis in \mathbb{C}^{G} .

Proof: Orthogonality follows directly from Lemma A.17. Completeness follows from the fact the $G \cong \hat{G}$, which implies $|\hat{G}| = N = \dim \mathbb{C}^G$.

Let $\chi_0, \ldots, \chi_{N-1}$ be the characters of $G = \{g_0, \ldots, g_{N-1}\}$. We can then consider the following matrix

$$[C]_{ij} = \chi_i(g_j)$$

which is called the character table of G.

A.4 Fourier transform on groups

We now introduce the Fourier transform of functions defined on Abelian group G.

Definition A.20 Let $f : G \to \mathbb{C}$ be any function. We define the Fourier transform $\hat{f} : \hat{G} \to \mathbb{C}$ of f by

$$\hat{f}(\chi) = \sum_{g \in G} f(g)\bar{\chi g}, \qquad \chi \in \hat{G}.$$
(A.3)

Example A.21 Let us consider again the group \mathbb{Z}_N . The characters, as we have seen before, are given by

$$\chi_{\ell}(g) = e^{i\frac{2\pi}{N}\ell g}, \quad g \in \mathbb{Z}_N, \quad \ell = 0, \dots, N-1.$$

The correspondence $\ell \to \chi_{\ell}$ yields an explicit isomorphism between \mathbb{Z}_N and $\hat{\mathbb{Z}}_N$. Given any function $f : \mathbb{Z}_N \to \mathbb{C}$, its Fourier transform is given by

$$\hat{f}(\chi_{\ell}) = \sum_{g=0}^{N-1} f(g) e^{-i\frac{2\pi}{N}\ell g}.$$

This transformation is easily inverted, and we define the inverse Fourier transform as follows

$$f = \frac{1}{N} \sum_{\chi \in \hat{G}} \hat{f}(\chi) \chi(g), \qquad g \in G.$$
(A.4)

An important fact about the Fourier transform is that it is an isometry with respect to a (suitably normalized) L_2 -norm. Indeed we have the following result.

Theorem A.22 (Plancherel's formula) For any $f_1, f_2 \in \mathbb{C}^G$,

$$\langle \hat{f}_1, \hat{f}_2 \rangle = N \langle f_1, f_2 \rangle.$$

Proof: Let us define the following vectors

$$f_1 = (f_1(g_0), \dots, f_1(g_{N-1}))$$

$$f_2 = (f_2(g_0), \dots, f_2(g_{N-1}))$$

and

$$\hat{f}_1 = (\hat{f}_1(\chi_0), \dots, \hat{f}_1(\chi_{N-1}))$$
$$\hat{f}_2 = (\hat{f}_2(\chi_0), \dots, \hat{f}_2(\chi_{N-1})).$$

Let C be the character table of G. Then we have that $\hat{f}_1 = f_1 C$ and $\hat{f}_2 = f_2 C$, and thus

$$\langle \hat{f}_1, \hat{f}_2 \rangle = \frac{1}{N} f_1 C C^* f_2^* = f_1 f_2^* = N \langle f_1, f_2 \rangle,$$

where we used the fact that $CC^* = NI$, and where the star indicates the transpose and complex conjugation operator.

Corollary A.23 (Parceval's formula) For any $f \in \mathbb{C}^{G}$,

$$<\hat{f}, \hat{f}> = N < f, f >$$
.

Appendix B

Graph Theory

B.1 Graph Theory

In this appendix we recall some notation and concepts on directed and undirected graphs (the reader can further refer to textbooks on graph theory such as [65] or [49]).

A directed graph is a pair (V, \mathcal{E}) where $V = \{1, \ldots, N\}$ is a finite nonempty node set and $\mathcal{E} \subseteq \mathcal{V} \times \mathcal{V}$ is an edge set of ordered pairs of nodes, called edges. The notation (i, j) means an edge getting out from the node *i* and getting in the node *j*. For the edge (i, j) *i* is the *parent node* and *j* is the *child node*. The self-loops are denoted by (i, i).

The adjacency matrix A is a $\{0, 1\}$ -valued square matrix indexed by the elements in V defined by letting $A_{ij} = 1$ if and only $(i, j) \in \mathcal{E}$. It is worth noting that, in this thesis we admit the presence of self-loops in the graph \mathcal{G} and hence if $(i, i) \in \mathcal{E}$ then $A_{ii} = 1$; often in the literature the self-loops are not encountered in the adjacency matrix.

In contrast to a direct graph, the pairs of nodes in an *undirected graph* are unordered; an undirected graph can be viewed as a special case of a directed graph, where an edge (i, j) in the undirected graph corresponds to edges (i, j) and (j, i). Clearly in the case of an undirected graph the adjacency matrix is symmetric.

Define the in-degree of a vertex j as $indeg(j) := \sum_i A_{ij}$ and the outdegree of a vertex i as $outdeg(i) := \sum_j A_{ij}$. A graph is called in-regular (out-regular) of degree k if each vertex has in-degree (out-degree) equal to k.

Given a direct graph \mathcal{G} , a *directed graph* consists of a sequence of vertices

 $i_1 i_2 \dots i_r$ such that $(i_\ell, i_{\ell+1}) \in \mathcal{E}$ for every $\ell = 1, \dots, r-1$; i_1 (resp. i_r) is said to be the initial (resp. terminal) vertex of the path. An *undirected* graph in an undirected graph is defined analogously.

In a directed graph, a *cycle* is a directed path in which the initial and the terminal vertices coincide. A vertex i is said to be connected to a vertex j if there exists a path with initial vertex i and terminal vertex j.

A directed graph is said to be *connected* if, given any pair of vertices i and j, either i is connected to j or j is connected to i. A directed graph is said to be *strongly connected* if, given any pair of vertices i and j, i is connected to j. An undirected graph is *connected* if there is an undirected path between every pair of distinct nodes. Note that, for an undirected graph, the notions of to be connected or strongly connected coincide.

A *direct tree* is a directed graph in which every node has exactly one parent except for one node, called the *root*, which has no parent and which has a direct path to every other node. Note that a directed tree has no cycle because every edge is oriented away from the root. In undirected graphs, a *tree* is a graph in which every pair of nodes is connected by exactly one undirected path.

A subgraph (V_s, \mathcal{E}_s) of (V, \mathcal{E}) is a graph such that $V_s \subseteq V$ and $\mathcal{E}_s \subseteq \cap(V_s \times V_s)$. A directed spanning tree (V_s, \mathcal{E}_s) of the directed graph (V, \mathcal{E}) is a subgraph of (V, \mathcal{E}) such that (V_s, \mathcal{E}_s) is a directed tree and $V_s = V$. An undirected spanning tree of an undirected graph is defined analogously. Note that in undirected graphs, the existence of an undirected spanning tree is equivalent to being connected, while in directed graphs is a weaker condition than being strongly connected.

Given a direct graph $\mathcal{G} = (V, E)$, a node $k \in V$ is said to be connected to a node $\ell \in V \{k\}$ if there is a path from k to ℓ in the graph with respects the orientation of the arcs. Note that the directed graph $\mathcal{G}(V, \mathcal{E})$ has a direct spanning tree if and only if (V, \mathcal{E}) has at least one node connected to all the other nodes. Finally, given a sequence of direct graphs $\mathcal{G}(t)(V, E(t)), t \in \mathbb{N}$, a node $k \in V$ is said to be connected to a node $\ell \in V \{k\}$ across an interval $I \subseteq \mathbb{N}$ if k is connected to ℓ for the directed graph $(V, \cup_{t \in I} E(t))$.

Finally, given the graphs $\mathcal{G}_1 = (V, \mathcal{E}_1)$ and $\mathcal{G}_2 = (V, \mathcal{E}_2)$, the union of \mathcal{G}_1 and \mathcal{G}_2 is defined by

$$\mathcal{G}_1 \cup \mathcal{G}_2 = (V, \mathcal{E}_1 \cup \mathcal{E}_2).$$

Appendix C

Some useful algebraic results

C.1 Stability of discrete time linear parameter varying (LPV) systems

Given $A_1, \ldots, A_k \in \mathbb{R}^{n \times n}$, we let $\{A(t)\}_{t=0}^{+\infty} \subset \operatorname{Co}\{A_1, \ldots, A_k\}$ denote a sequence of matrices taking values in the convex hull of $\{A_1, \ldots, A_k\}$. We consider the dynamical system

$$x(t+1) = A(t)x(t).$$
 (C.1)

The following result is an extension to the discrete-time system (C.1) of the classical result stated in [21], establishing a sufficient condition for the stability of continuous-time LPV systems.

Theorem C.1 (Common Lyapunov function) For $A_1, \ldots, A_k \in \mathbb{R}^{n \times n}$, if there exists a symmetric matrix $P \in \mathbb{R}^{n \times n}$ such that

$$\frac{A_i^* P A_j + A_j^* P A_i}{2} - P < 0, \quad \text{for all } i, j \in \{1, \dots, k\},\$$

then, for all initial conditions $x(0) \in \mathbb{R}^n$ and sequences

 $\{A(t)\}_{t=0}^{+\infty} \subset \operatorname{Co}\{A_1, \dots, A_k\},\$

the solution to (C.1) satisfies

$$\lim_{t \to +\infty} x(t) = 0.$$

The objective of this Subsection is to generalize this classic result as follows.

Theorem C.2 (Common Lyapunov function for convergence to eigenspace) Assume that 1 is a simple eigenvalue with left and right eigenvector $v \in \mathbb{R}^n$ for each matrix $A_1, \ldots, A_k \in \mathbb{R}^{n \times n}$. If there exists a symmetric matrix $P \in \mathbb{R}^{n \times n}$ satisfying, for all nonzero $z \notin \operatorname{span}\{v\}$,

$$Pv = 0, \tag{C.2}$$

$$z^* P z > 0, \tag{C.3}$$

and

$$z^* \left(\frac{A_i^* P A_j + A_j^* P A_i}{2} - P\right) z < 0, \quad \text{for all } i, j \in \{1, \dots, k\},$$
(C.4)

then, for all initial conditions $x(0) \in \mathbb{R}^n$ and sequences

$$\{A(t)\}_{t=0}^{+\infty} \subset \operatorname{Co}\{A_1, \dots, A_k\},\$$

the solution to (C.1) satisfies

$$\lim_{t \to +\infty} x(t) = \alpha v, \quad \alpha = \frac{1}{\|v\|^2} v^* x(0).$$

Proof: Because v is a left and right eigenvector with eigenvalue 1, we have

$$A_i v = v$$
, and $v^* A_i = v^*$, for $i \in \{1, \dots, k\}$. (C.5)

Consider the following decomposition

$$x(t) = x_{\text{ave}}(t) v + x_{\perp}(t),$$

where $x_{\perp} \perp v$ and where $x_{\text{ave}}(t) = \frac{1}{\|v\|^2} v^* x(t) \in \mathbb{R}$. Straightforward calculations show that x_{ave} satisfies the recursive relation

$$x_{\text{ave}}(t+1) = x_{\text{ave}}(t) + \frac{1}{\|v\|^2} v^* A(t) x_{\perp}(t) = x_{\text{ave}}(t),$$

where in the last equality we have used the facts that $v^*A(t) = v^*$ and $v^*x_{\perp} = 0$. Hence, $x_{\text{ave}}(t) = x_{\text{ave}}(0) = (1/||v||^2)v^*x(0)$, for all t. Now, let $v_1 = v$ and consider a basis $\{v_1, v_2, \ldots, v_n\}$ of \mathbb{R}^n with the orthogonality property $v_1 \perp v_i$ for all $i \in \{2, \ldots, n\}$. Let $T = [v_1, \ldots, v_n] \in \mathbb{R}^{n \times n}$. Let $\bar{x} =$

 $T^{-1}x = [\tilde{x}_1, \ldots, \tilde{x}_n]^*$ and let $\tilde{x} = [\tilde{x}_2, \ldots, \tilde{x}_n]^* \in \mathbb{R}^{n-1}$. By the hypothesis (C.2) and the assumption (C.5) we have that

$$T^{-1}PT = \begin{bmatrix} 0 & 0\\ 0 & \tilde{P} \end{bmatrix},$$
$$T^{-1}A_iT = \begin{bmatrix} 1 & 0\\ 0 & \tilde{A}_i \end{bmatrix},$$
$$T^{-1}A_jT = \begin{bmatrix} 1 & 0\\ 0 & \tilde{A}_j \end{bmatrix},$$

and

where
$$\tilde{P}$$
, \tilde{A}_i , $\tilde{A}_j \in \mathbb{R}^{(n-1)\times(n-1)}$ and, by hypothesis (C.3) $\tilde{P} > 0$. It follows that

$$\frac{1}{2} \left(\begin{bmatrix} 1 & 0 \\ 0 & \tilde{A}_i^* \end{bmatrix} \begin{bmatrix} 0 & 0 \\ 0 & \tilde{P} \end{bmatrix} \begin{bmatrix} 1 & 0 \\ 0 & \tilde{A}_j \end{bmatrix} + \begin{bmatrix} 1 & 0 \\ 0 & \tilde{A}_j^* \end{bmatrix} \begin{bmatrix} 0 & 0 \\ 0 & \tilde{P} \end{bmatrix} \begin{bmatrix} 1 & 0 \\ 0 & \tilde{A}_i \end{bmatrix} \right) - \begin{bmatrix} 0 & 0 \\ 0 & \tilde{P} \end{bmatrix} \\ = \begin{bmatrix} 0 & 0 \\ 0 & \frac{1}{2} \left(\tilde{A}_i^* \tilde{P} \tilde{A}_j + \tilde{A}_j^* \tilde{P} \tilde{A}_i \right) - \tilde{P} \end{bmatrix},$$

where, by hypothesis (C.4), $\frac{1}{2} \left(\tilde{A}_i^* \tilde{P} \tilde{A}_j + \tilde{A}_j^* \tilde{P} \tilde{A}_i \right) - \tilde{P} < 0$, for all $i, j \in \{1, \ldots, k\}$. Hence, Theorem C.1 implies $\lim_{t \to +\infty} \tilde{x}(t) = 0$ and, in turn, $\lim_{t \to +\infty} x_{\text{ave}}(t) = (1/\|v\|^2)v^*x(0)$.

C.2 Solvability of a Lyapunov equation

The proof of some results contained in Chapter 7 are based on the solvability of the following Lyapunov equation

$$z^*(R_1^*LR_1 - (1 - \beta)L)z < 0, \quad \forall z \in < [1^* \ 0^*]^* >^{\perp}$$
 (C.6)

where

$$L = \begin{bmatrix} I - P & 0 \\ 0 & \gamma I \end{bmatrix}, \qquad R_1 = \begin{bmatrix} I & 0 \\ 0 & \delta I \end{bmatrix} \begin{bmatrix} P & P - I \\ P - I & P - 2I \end{bmatrix}, \qquad (C.7)$$

 $P \in \mathbb{R}^{N \times N}$ satisfies Assumption 7.1 (pag. 153), $0 < \delta < 1$ and $\gamma > 0$.

The following lemma helps to determine for what parameters γ, β, δ the Lyapunov inequality (C.6) holds.

Lemma C.3 Let λ_{min} and λ_{max} be the minimum and the maximum eigenvalue in $\sigma(P) \setminus \{1\}$, respectively. Define the polynomial

$$f(\lambda, \delta, \gamma, \beta) := \gamma \beta^{2} + \left((1-\lambda)^{3} + \gamma \delta^{2} (\lambda-2)^{2} - \gamma + \gamma (1-\lambda) (\gamma \delta^{2} - 1 - \lambda) \right) \beta$$
$$+ (1-\lambda) \left(-\gamma^{2} \delta^{2} + \gamma (1+\lambda+\delta^{2} (\lambda-3)) - (1-\lambda)^{2} \right)$$
(C.8)

Then (C.6) holds true if and only if

$$\beta < \min\left\{\beta_{\min}^{-}(\delta,\gamma), \beta_{\max}^{-}(\delta,\gamma)\right\},\tag{C.9}$$

where $\beta_{\min}^{-}(\delta, \gamma)$ and $\beta_{\max}^{-}(\delta, \gamma)$ are the minimum real roots of $f(\lambda_{\min}, \delta, \gamma, \beta)$ and of $f(\lambda_{\max}, \delta, \gamma, \beta)$ as polynomials in β .

Proof: We start by observing that

$$R_{1}^{*}LR_{1} - (1 - \beta)L = \begin{bmatrix} (I - P)^{2}(\gamma\delta^{2}I - I - P) + \beta(I - P) & (I - P)(P(P - I) - \gamma\delta^{2}(P - 2I)) \\ (I - P)(P(P - I) - \gamma\delta^{2}(P - 2I)) & (I - P)^{3} + \gamma\delta^{2}(P - 2I)^{2} - \gamma I + \gamma\beta I \end{bmatrix}.$$

Note that

$$\left(R_1^*LR_1 - (1-\beta)L\right)\begin{bmatrix}\mathbf{1}\\\mathbf{0}\end{bmatrix} = 0$$

and hence showing (C.6) is equivalent to show that the symmetric matrix $-R_1^*LR_1 + (1 - \beta)L$ has all positive eigenvalues except one, which is zero and has multiplicity one. If we define the polynomials $q_{11}(\lambda), q_{22}(\lambda), q_{12}(\lambda)$ as follows

$$q_{11}(\lambda) := (1-\lambda)^2 (\gamma \delta^2 - 1 - \lambda) + \beta (1-\lambda)$$

$$q_{22}(\lambda) := (1-\lambda)^3 + \gamma \delta^2 (\lambda - 2)^2 - \gamma + \gamma \beta$$

$$q_{12}(\lambda) := (1-\lambda) (\lambda (\lambda - 1) - \gamma \delta^2 (\lambda - 2))$$

we can write

$$R_1^* L R_1 - (1 - \beta) L = \begin{bmatrix} q_{11}(P) & q_{12}(P) \\ q_{12}(P) & q_{22}(P) \end{bmatrix},$$

To compute the eigenvalues of $-R_1^*LR_1 + (1 - \beta)L$ we consider its characteristic polynomial. Using the same arguments used in the proof of Lemma 7.10 we can argue that

$$\det(sI + R_1^*LR_1 - (1 - \beta)L) =$$

= $\prod_{i=0}^{N-1} \left[s^2 + (q_{11}(\lambda_i) + q_{22}(\lambda_i))s + (q_{11}(\lambda_i)q_{22}(\lambda_i) - q_{12}(\lambda_i)^2) \right]$

where $\lambda_0 = 1, \lambda_1, \ldots, \lambda_{N-1}$ denote the eigenvalues of P. Observe now that, for i = 0 the polynomial in the previous product is

$$s(s - \gamma\delta^2 + \gamma - \gamma\beta)$$

which gives one zero eigevalue and another eigenvalue equal to $\gamma(1 - \delta^2 - \beta)$. We can argue that, since this must positive, then we have this first constraint

$$\beta < 1 - \delta^2 \tag{C.10}$$

Moreover all the roots of the other polynomials for i = 1, ..., N - 1 must be all positive. Observe that $s^2 + (q_{11}(\lambda_i) + q_{22}(\lambda_i))s + (q_{11}(\lambda_i)q_{22}(\lambda_i) - q_{12}(\lambda_i)^2)$ can be seen as the characteristic polynomial of the 2×2 matrix

$$\begin{bmatrix} -q_{11}(\lambda_i) & -q_{12}(\lambda_i) \\ -q_{12}(\lambda_i) & -q_{22}(\lambda_i) \end{bmatrix}$$

and, to impose that the roots of this polynomial are positive is equivalent to impose that such a matrix is positive definite and so that

$$-q_{11}(\lambda_i) > 0 \qquad q_{11}(\lambda_i)q_{22}(\lambda_i) - q_{12}^2(\lambda_i) > 0.$$

Therefore, together with condition (C.10), we have other 2N - 2 conditions. Some of these conditions are superfluous. We start from condition $-q_{11}(\lambda_i) > 0$. Observe that, since $1 - \lambda_i > 0$ for all $i \in \{1, \ldots, N - 1\}$, then $q_{11}(\lambda_i) < 0$ for all $i \in \{1, \ldots, N - 1\}$ if and only if $\beta + \gamma \delta^2(1 - \lambda_i) - 1 + \lambda_i^2 < 0$ for all $i \in \{1, \ldots, N - 1\}$ and this happens if and only if

$$\beta < 1 - \lambda_{\min}^2 - \gamma \delta^2 (1 - \lambda_{\min}) \tag{C.11}$$

$$\beta < 1 - \lambda_{max}^2 - \gamma \delta^2 (1 - \lambda_{max}) \tag{C.12}$$

Notice now that

$$q_{11}(\lambda)q_{22}(\lambda) - q_{12}^2(\lambda) = (1-\lambda)f(\lambda)$$

where $f(\lambda) = f(\lambda, \delta, \gamma, \beta)$ is defined in (C.8). Notice that

$$\frac{\partial^2 f}{\partial \lambda^2} = (1 - \beta)(6\lambda - 6 - 2\gamma(1 + \delta^2))$$

which is negative for $\lambda < 1$. This implies that $f(\lambda)$, is a concave function in λ , for $\lambda \in [-1, 1]$ and so $q_{11}(\lambda_i)q_{22}(\lambda_i) - q_{12}^2(\lambda_i) > 0$ for all $i = 1, \ldots, N-1$ if and only if

$$f(\lambda_{\min}, \delta, \gamma, \beta) > 0 \tag{C.13}$$

$$f(\lambda_{max}, \delta, \gamma, \beta) > 0. \tag{C.14}$$

At this point we have that (C.6) holds true if and only if conditions (C.10), (C.11), (C.12), (C.13) and (C.14) hold true. Consider the condition $f(\lambda_{min}, \delta, \gamma, \beta) > 0$. Observe that, if $\beta = 1 - \lambda_{min}^2 - \gamma \delta^2 (1 - \lambda_{min})$, then $q_{11} = 0$ and so

$$f(\lambda_{\min}, \delta, \gamma, \beta)_{|\beta=1-\lambda_{\min}^2 - \gamma \delta^2(1-\lambda_{\min})} = \frac{-q_{12}^2}{1-\lambda} < 0.$$

We can argue that $f(\lambda_{min}, \delta, \gamma, \beta)$ is a convex parabola in β which has always two real roots $\beta_{min}^{-}(\delta, \gamma)$ and $\beta_{min}^{+}(\delta, \gamma)$ which satisfy

$$\beta_{\min}^{-}(\delta,\gamma) < 1 - \lambda_{\min}^{2} - \gamma \delta^{2}(1 - \lambda_{\min}) < \beta_{\min}^{+}(\delta,\gamma)$$
(C.15)

and morever $f(\lambda_{min}, \delta, \gamma, \beta) > 0$ if and only if

$$\beta < \beta_{\min}^{-}(\delta, \gamma) \quad \text{or} \quad \beta > \beta_{\min}^{+}(\delta, \gamma).$$
 (C.16)

This implies that conditions (C.11) and (C.13) hold if and only if $\beta < \beta_{\min}^{-}(\delta, \gamma)$. Reasoning similarly for the condition $f(\lambda_{\max}, \delta, \gamma, \beta) > 0$ we obtain that conditions (C.12) and (C.14) hold if and only if $\beta < \beta_{\max}^{-}(\delta, \gamma)$.

We prove finally that condition (C.10) is superfluous which would give the thesis. To prove this observe that

$$f(\lambda_{\min}, \delta, \gamma, \beta)_{|\beta=0} =$$

= $(1 - \lambda_{\min})[-\lambda_{\min}^2 + (2 + \gamma\delta^2 + \gamma)\lambda_{\min} - (1 + 3\gamma\delta^2 + \gamma^2\delta^2 - \gamma)]$

and

$$f(\lambda_{\min}, \delta, \gamma, \beta)_{|\beta=1-\delta^2} = \\ = \delta^2 (1 - \lambda_{\min}) [-\lambda_{\min}^2 + (2 + \gamma \delta^2 + \gamma) \lambda_{\min} - (1 + 3\gamma \delta^2 + \gamma^2 \delta^2 - \gamma)]$$

This implies that we can have three cases

- 1. We have $f(\lambda_{min}, \delta, \gamma, \beta)_{|\beta=0} = 0$. In this case we have that $\beta_{min}^{-}(\delta, \gamma) = 0$ and $\beta_{min}^{+}(\delta, \gamma) = 1 \delta^{2}$.
- 2. We have $f(\lambda_{min}, \delta, \gamma, \beta)_{|\beta=0} < 0$. In this case we have that $\beta_{min}^{-}(\delta, \gamma) < 0 < 1 \delta^2 < \beta_{min}^{+}(\delta, \gamma);$
- 3. We have $f(\lambda_{\min}, \delta, \gamma, \beta)_{|\beta=0} > 0$. In this case we may have three situations:

1a.
$$0 < \beta_{\min}^{-}(\delta, \gamma) \leq \beta_{\min}^{+}(\delta, \gamma) < 1 - \delta^{2};$$

2b. $0 < 1 - \delta^{2} < \beta_{\min}^{-}(\gamma) \leq \beta_{\min}^{+}(\delta, \gamma);$
3c. $\beta_{\min}^{-}(\delta, \gamma) \leq \beta_{\min}^{+}(\delta, \gamma) < 0 < 1 - \delta^{2}.$

However the cases 2b and 2c cannot occur since the $\beta_{\min}^-(\delta, \gamma)$ is a continuous function of γ , while in these two cases the value of this function would pass from 0 to $1 - \delta^2$ in a neighbor of the γ 's such that $f(\lambda_{\min}, \delta, \gamma, \beta)_{|\beta=0} = 0$ Notice finally that in all the cases which can occur we have that $\beta_{\min}^-(\delta, \gamma) \leq 1 - \delta^2$.

We provide now a consequence of the previous result.

Lemma C.4 Assume the same assumptions of the previous lemma hold. Let $\bar{\delta}$ be defined as in (7.13) and $\delta \in \mathbb{R}$ be such that $0 \leq \delta < \bar{\delta}$. If we let

$$\bar{\gamma} := \frac{1 + \lambda_{\min} + \delta^2(\lambda_{\min} - 3)}{2\delta^2}$$

Then $\bar{\gamma} > 0$ satisfies

$$z^*(R_1^*LR_1 - L)z < 0, \quad \forall z \in < [1^* \ 0^*]^* >^{\perp}, \quad (C.17)$$

Proof: Notice first that $0 \leq \delta < \bar{\delta}$ implies that $\bar{\gamma} > 0$. By the previous lemma, (C.17) holds true if and only if $\beta = 0$ is an admissible solution of (C.9) and this happens if and only if both $\beta_{\min}^{-}(\delta, \bar{\gamma}) > 0$ and $\beta_{\max}^{-}(\delta, \bar{\gamma}) > 0$. Notice now that $f(\lambda, \delta, \gamma, \beta)$ can be written as follows

$$f(\lambda,\delta,\gamma,\beta) = \gamma\beta^2 + [(1-\lambda)p(\lambda,\delta,\gamma) - \gamma(1-\delta^2)]\beta - (1-\lambda)p(\lambda,\delta,\gamma)$$

where

$$p(\lambda, \delta, \gamma) = \delta^2 \gamma^2 - [1 + \lambda + \delta^2 (\lambda - 3)] \gamma + (1 - \lambda)^2$$

Notice moreover that, $\beta_{\min}^{-}(\delta,\bar{\gamma}) > 0$ if and only if $(1 - \lambda_{\min})p(\lambda_{\min}, \delta, \bar{\gamma}) - \bar{\gamma}(1 - \delta^2) < 0$ and $(1 - \lambda_{\min})p(\lambda_{\min}, \delta, \bar{\gamma}) < 0$ and these two conditions occurs if and only if $p(\lambda_{\min}, \delta, \bar{\gamma}) < 0$. Similarly we can see that $\beta_{\max}^{-}(\delta, \bar{\gamma}) > 0$ if and only if $p(\lambda_{\max}, \delta, \bar{\gamma}) < 0$. Notice now that, since

$$\frac{\partial p}{\partial \lambda} = -\gamma - \gamma \delta^2 - 2(1 - \lambda),$$

is negative for $\lambda < 1$, then $p(\lambda_{max}, \delta, \bar{\gamma}) < 0$ is implies by $p(\lambda_{min}, \delta, \bar{\gamma}) < 0$ which is the only condition we need to prove. Notice now that $\bar{\gamma}$ is the minimizer of $p(\lambda_{min}, \delta, \gamma)$ as a function of γ . Therefore $p(\lambda_{max}, \delta, \bar{\gamma}) < 0$ if and only if the discriminant is positive, namely if and only if $(1 + \lambda_{min} + \delta^2(\lambda_{min} - 3))^2 - 4\delta^2(1 - \lambda_{min})^2 > 0$. Observe that this last inequality holds true if and only if

$$(3 - \lambda_{min})^2 \delta^4 - 2(5 - 2\lambda_{min} + \lambda_{min}^2) \delta^2 + (1 + \lambda_{min})^2 > 0.$$
 (C.18)

Consider the equation $(3 - \lambda_{min})^2 x^2 - 2(5 - \lambda_{min} + \lambda_{min}^2)x + (1 + \lambda_{min})^2 = 0.$ The solutions of this equation are $x_1 = 1$ and $x_2 = \left(\frac{1 + \lambda_{min}}{3 - \lambda_{min}}\right)^2$. Since $\lambda_{min} < 1$ we have that $x_2 < 1$. Hence, since $(3 - \lambda_{min})^2 x^2 - 2(5 - \lambda_{min} + \lambda_{min}^2)x + (1 + \lambda_{min})^2 > 0$ is a convex parabola, we have that $(3 - \lambda_{min})^2 x^2 - 2(5 - \lambda_{min} + \lambda_{min}^2)x + (1 + \lambda_{min})^2 > 0$ if and only if $x < x_2$ and $x > x_1$. It follows that, if $\delta^2 < \left(\frac{1 + \lambda_{min}}{3 - \lambda_{min}}\right)^2$, i.e., if $\delta < \overline{\delta}$, then (C.18) is satisfied.

Bibliography

- I. Akyildiz, W. Su, Y. Sankarasubramniam, and E. Cayirci. A survey on sensor networks. *IEEE Communications Magazine*, pages 102–114, 2002.
- [2] R. Albert and A. L. Barabasi. Statistical mechanics of complex networks. *Review Modern Physics*, 74(1):47–97, 2002.
- [3] N. Alon. Eigenvalues and expanders. *Combinatorica*, 6(2):83–86, 1986.
- [4] N. Alon and Y. Roichman. Random Cayley graphs and expanders. Random Structures and Algorithms, 5:271–284, 1994.
- [5] P. Alriksson and A. Rantzer. Distributed Kalman filtering using weighted averaging. In Proceedings of the 17th International Symposium on Mathematical Theory of Networks and Systems, Kyoto, Japan, 2006.
- [6] M. Amin and P. F. Schewe. Preventing blackouts. Scientific American, pages 60–67, 2007.
- [7] D. Angeli and P.-A. Bliman. Stability of leaderless discrete-time multi-agent systems. *Mathematics of Control, Signals and Systems*, 18(4):293–322, 2006.
- [8] T. C. Aysal, M. Coates, and M. Rabbat. Distributed average consensus using probabilistic quantization. In *IEEE Workshop on Statistical Signal Processing*, pages 640–644, Maddison, Wisconsin, August 2007.
- [9] T. C. Aysal, M. Coates, and M. Rabbat. Rates of convergence of distributed average consensus using probabilistic quantization. In *Allerton Conference on Communication, Control, and Computing*, 2007.
- [10] L. Babai. Spectra of Cayley graphs. Journal of Combinatorial Theory, Series B, 27:180–189, 1979.

- [11] G. Baldan. An efficient quantization algorithm for solving average consensus problems. Master Thesis, University of Padua, 2008.
- [12] G. W. Barlow. Hexagonal territories. Animal Behavior, 22:876–878, 1974.
- [13] D. Bauso, L. Giarré, and R. Pesenti. Nonlinear protocols for optimal distributed consensus in networks of dynamic agents. Systems & Control Letters, 55(11):918–928, 2006.
- [14] E. Behrends. Introduction to Markov Chains (with Special Emphasis on Rapid Mixing). Vieweg Verlag, 1999.
- [15] D. Bertsekas and J. Tsitsiklis. Parallel and Distributed Computation: Numerical Methods. 1999.
- [16] V. D. Blondel, J. M. Hendrickx, A. Olshevsky, and J. N. Tsitsiklis. Convergence in multiagent coordination, consensus, and flocking. In *IEEE Conf. on Decision and Control and European Control Conference*, pages 2996–3000, Seville, Spain, December 2005.
- [17] S. Boinski and A.-F. Campbell. Use of trill vocalizations to coordinate troop movement among whitefaced capuchins-a 2nd field-test. *Behaviour*, 132(11), 1995.
- [18] J.-M. Borwein and A.-S. Lewis. Convex Analysis and Nonlinear Optimazation. CMS Books in Mathematics. Canadian Mathematical Society, 2000.
- [19] S. Boyd, P. Diaconis, P. Parrilo, and L. Xiao. Symmetry analysis of reversible Markov chains. Technical report, Department of Statistics, Stanford University, 2003.
- [20] S. Boyd, P. Diaconis, and L. Xiao. Fastest mixing Markov chain on a graph. SIAM Review, 46:667–689, 2004.
- [21] S. Boyd, L. El Ghaoui, E. Feron, and V. Balakrishnan. *Linear Matrix Inequalities in System and Control Theory*, volume 15 of *Studies in Applied Mathematics*. SIAM, Philadelphia, Pennsylvania, 1994.
- [22] S. Boyd, A. Ghosh, B. Prabhakar, and D. Shah. Gossip algorithms: Design, analysis, and application. In *IEEE Conference on Computer Communications (INFOCOM)*, pages 1653–1664, March 2005.

- [23] S. Boyd, A. Ghosh, B. Prabhakar, and D. Shah. Mixing times for random walks on geometric random graphs. In SIAM ANALCO, 2005.
- [24] S. Boyd, A. Ghosh, B. Prabhakar, and D. Shah. Randomized gossip algorithms. *IEEE Transactions on Information Theory*, 52(6):2508– 2530, 2006.
- [25] F. Bullo, J. Cortes, and S. Martinez. Distributed Control of Robotic Networks. Available at http://coordinationbook.info, 2008.
- [26] M. Cao. Multi-Agent Formations and Sensor Networks. PhD thesis, Yale University, August 2007.
- [27] M. Cao, A. S. Morse, and B. D. O. Anderson. Reaching a consensus in a dynamically changing environment - convergence rates, measurement delays and asynchronous events. *SIAM Journal on Control and Optimization*, 47(2):601–623, 2008.
- [28] M. Cao, D. A. Spielman, and E. M. Yeh. Accelerated gossip algorithms for distributed computation. In 44th Annual Allerton Conference on Communication, Control, and Computation., 2006.
- [29] R. Carli and F. Bullo. Quantized coordination algorithms for rendezvous and deployment. December 2007. Submitted. Available electronically at http://motion.mee.ucsb.edu/papers.
- [30] R. Carli, F. Bullo, and S. Zampieri. Quantized average consensus via dynamic coding/decoding schemes. In *IEEE Conf. on Decision and Control*, Cancun, Mexico, December 2008. Submitted.
- [31] R. Carli, A. Chiuso, L. Schenato, and S. Zampieri. Distributed kalman filtering based on consensus strategies. *IEEE J. on Selected Areas of Communications*, 26(4):622–633, 2008.
- [32] R. Carli, A. Chiuso, L. Schenato, and S. Zampieri. A PI consensus controller for networked clocks synchronization. In *IFAC Congress*, Seoul, Corea, July 2008.
- [33] R. Carli, F. Fagnani, P. Frasca, T. Taylor, and S. Zampieri. Average consensus on networks with transmission noise or quantization. In *European Control Conference*, pages 1852–1857.
- [34] R. Carli, F. Fagnani, P. Frasca, and S. Zampieri. A probabilistic analysis of the average consensus algorithm with quantized communication. In *IFAC Congress*, Seoul, Corea, July 2008.

- [35] R. Carli, F. Fagnani, A. Speranzon, and S. Zampieri. Communication constraints in the average consensus problem. *Automatica*, 44(3):671– 684, 2008.
- [36] G. Chen, G. Chen, and S. Hsu. *Linear Stochastic Control Systems*. CRC Press, 1995.
- [37] R. Cogburn. On products of random stochastic matrices. Random matrices and their applications volume 50 of Contemp. Math., pages 199–213, 1986.
- [38] L. Conradt and T. J. Roper. Group decession-making in animals. Nature, 421(6919):155–158, 2003.
- [39] J. Cortés. Distributed algorithms for reaching consensus on arbitrary functions. *Automatica*, October 2006. Submitted.
- [40] J. Cortés. Finite-time convergent gradient flows with applications to network consensus. Automatica, 42(11):1993–2000, 2006.
- [41] I. D. Couzin, J. Krause, N. R. Franks, and S. A. Levin. Effective leadership and decision-making in animal groups on the move. *Nature*, 433(7025):513–516, 2005.
- [42] F. Cucker and S. Smale. Emergent behavior in flocks. *IEEE Transac*tions on Automatic Control, 52(5):852–862, 2007.
- [43] D. Cvetovic, M. Doob, and H. Sachs. Spectra of graphs. Academic Press, 1980.
- [44] G. Cybenko. Dynamic load balancing for distributed memory multiprocessors. Journal of Parallel and Distributed Computing, 7:279–301, 1989.
- [45] R. D'Andrea and G. E. Dullerud. Distributed control design for spatially interconnected systems. *IEEE Transactions on Automatic Control*, 48(9):1478–1495, 2003.
- [46] P. J. Davis. Circulant matrices. A Wiley-Interscience Publication, Pure and Applied Mathematics. John Wiley & Sons, New York-Chichester-Brisbane, 1979.
- [47] J. C. Delvenne, R. Carli, and S. Zampieri. Quick strategies in a consensus problem. In *IEEE Conf. on Decision and Control*, New Orleans, USA, December 2007.

- [48] P. Diaconis. Group Representations in Probability and Statistics, volume 11 of IMS Lecture Notes - Monograph Series, S. S. Gupta (ed.). Institute of Mathematical Statistics, Hawyard Ca, 1988.
- [49] R. Diestel. Graph Theory, volume 173 of Graduate Texts in Mathematics. Springer-Verlag, Heidelberg, third edition edition, 2005.
- [50] S. N. Dorogovtsev and J. F. F. Mendes. Evolution of networks. from biological nets to the internet and www. Technical report.
- [51] A. S. A. Dymakis and M. Wainright. Geographic gossip: efficient aggregation on sensor networks. Technical report.
- [52] A. Bhaya E. Kaszkurewicz and D. Siljak. On the convergence of parallel asyncronous block-iterative computations. *Linear Algebra and its Applications*, 131, 1990.
- [53] N. Elia and S. K. Mitter. Stabilization of linear systems with limited information. *IEEE Transactions on Automatic Control*, 46(9):1384– 1400, 2001.
- [54] Robert Elsässer, Burkhard Monien, and Stefan Schamberger. Distributing unit size workload packages in heterogeneous networks. J. Graph Algorithms Appl., 10(1):51–68 (electronic), 2006.
- [55] F. Fagnani and S. Zampieri. Performance evaluations of quantized stabilizers. In *IEEE Conf. on Decision and Control*, pages 1897–1901, Maui, HI, December 2003.
- [56] F. Fagnani and S. Zampieri. Quantized stabilization of linear systems: Complexity versus performances. *IEEE Transactions on Automatic Control*, 49(9):1534–1548, 2004.
- [57] F. Fagnani and S. Zampieri. Average consensus with packet drop communication. 2008. Special issue on "Control and Optimization in Cooperative Networks." Accepted.
- [58] F. Fagnani and S. Zampieri. Randomized consensus algorithms over large scale networks. *IEEE J. on Selected Areas of Communications*, 26(4):634–649, 2008.
- [59] L. Fang and P. J. Antsaklis. Information consensus of asynchronous discrete-time multi-agent systems. In *American Control Conference*, pages 1883–1888, Portland, OR, 2005.

- [60] G. Ferrari-Trecate, A. Buffa, and M. Gati. Analysis of coordination in multi-agent systems through partial difference equations. *IEEE Transactions on Automatic Control*, 51(6):1058–1063, 2006.
- [61] P. Frasca, R. Carli, F. Fagnani, and S. Zampieri. Average consensus on networks with quantized communication. *Submitted to International Journal of Robust and Nonlinear Control*, 2008.
- [62] J. Friedman. A proof of Alon's second eigenvalue conjecture. Accepted to the Memoirs of the A.M.S., 2004.
- [63] F. R. Gantmacher. The theory of matrices. New York: Chelsea, 1959.
- [64] A. Giridhar and P. Kumar. Towards a theory of in-network computation in wireless sensor networks. *IEEE Communications Magazine*, 44(4):98–107, 2006.
- [65] C. Godsil and G. Royle. Algebraic Graph Theory. Springer, 2001.
- [66] S. Gueron and S. A. Levin. Self-organization of front patterns in large wildebeest herds. J. Theoretical Biol., 165(4):541–552, 1993.
- [67] P. Gupta and P.R. Kumar. The capacity of wireless networks. Information Theory, IEEE Transactions on, 46(2):388–404, Mar 2000.
- [68] V. Guruswani. Rapidly mixing markov chains: A comparison of techniques. Technical report.
- [69] Frank Harary, John P. Hayes, and Horng-Jyh Wu. A survey of the theory of hypercube graphs. *Computational Mathematical Applications*, 15(4):277–289, 1988.
- [70] Y. Hatano and M. Mesbahi. Agreement over random networks. *IEEE Transactions on Automatic Control*, 50:1867–1872, 2005.
- [71] J. M. Hendrickx. Graphs and Networks for the Analysis of Autonomous Agent Systems. PhD thesis, Université Catholique de Louvain, Belgium, February 2008.
- [72] B. Hohlt, L. Doverty, and E. Brewer. Flexible power scheduling for sensor networks. In In IEEE and ACM International Symposium on Information Processing in Sensor Networks (IPSN'04), April 2004.

- [73] Y. Igarashi, T. Hatanaka, M. Fujita, and M. W. Spong. Passivitybased 3D attitude coordination: Convergence and connectivity. In *IEEE Conf. on Decision and Control*, pages 2558–2565, New Orleans, LA, December 2007.
- [74] A. Jadbabaie, J. Lin, and A. S. Morse. Coordination of groups of mobile autonomous agents using nearest neighbor rules. *IEEE Transactions* on Automatic Control, 48(6):988–1001, 2003.
- [75] A. Kashyap, T. Başar, and R. Srikant. Quantized consensus. Automatica, 43(7):1192–1203, 2007.
- [76] A. D. D. Kempe and J. Gehrke. Gossip-based computation of aggregate information. In Proceedings of the 44th IEEE Symposium on Foundations of Computer Science, 2003.
- [77] H. J. Landau and A. M. Odlyzko. Bounds for eigenvalues of certain stochastic matrices. *Linear Algebra and its Applications*, 38:5–15, 1981.
- [78] B. Levy, D. Castanon, G. Verghese, and A. Willsky. A scattering framework for decentralized estimation problems. *Automatica*, 19(4):373– 384, 1983.
- [79] Z. Lin. Coupled Dynamic Systems: From Structure Towards Stability and Stabilizability. PhD thesis, University of Toronto, Canada, September 2005.
- [80] Z. Lin, M. Broucke, and B. Francis. Local control strategies for groups of mobile autonomous agents. *IEEE Transactions on Automatic Control*, 49(4):622–629, 2004.
- [81] Z. Lin, B. Francis, and M. Maggiore. State agreement for continuoustime coupled nonlinear systems. SIAM Journal on Control and Optimization, 46(1):288–307, 2007.
- [82] D. Looze, P. Houpt, N. Sandell Jr, and M. Athans. On decentralized estimation and control with application to freeway ramp metering. *IEEE Transactions on Automatic Control*, 23(2):268–275, 1978.
- [83] J. Lorenz. Repeated Averaging and Bounded Confidence Modeling, Analysis and Simulation of Continuous Opinion Dynamics. PhD thesis, University of Bremen, Germany, February 2007. Available electronically at http://nbn-resolving.de/urn:nbn:de:gbv:46-diss000106688.

- [84] J. Lorenz and D. A. Lorenz. On conditions for convergence to consensus, March 2008. Available electronically at http://arxiv.org/abs/0803.2211.
- [85] Z. Luo and J. Tsitsiklis. Data fusion with minimal communication. *IEEE Transactions on Information Theory*, 40(5):1551–1563, 1994.
- [86] S. Martínez, F. Bullo, J. Cortés, and E. Frazzoli. On synchronous robotic networks – Part I: Models, tasks and complexity. *IEEE Transactions on Automatic Control*, 52(12):2199–2213, 2007.
- [87] M. B. Miller and B. L. Bassler. Quorum sensing in bacteria. Ann. Rev. Microbiol., 55:165–199, 2001.
- [88] L. Moreau. Stability of multi-agent systems with time-dependent communication links. *IEEE Transactions on Automatic Control*, 50(2):169– 182, 2005.
- [89] M. R. Murty. Ramanujan graphs. Journal of Ramanujan Mathematical Society, 18(1):1–20, 2003.
- [90] S. Muthukrishnan, B. Ghosh, and Schultz M. H. First and second order diffusive methods for rapid, coarse, distributed load balancing. *Theory* of Computing Systems, 31:331–354, 1998.
- [91] G. N. Nair, F. Fagnani, S. Zampieri, and R. J. Evans. Feedback control under data rate constraints: An overview. *IEEE Proceedings*, 95(1):108–137, 2007.
- [92] A. Nedic, A. Olshevsky, A. Ozdaglar, and J. N. Tsitsiklis. On distributed averaging algorithms and quantization effects. Technical Report 2778, LIDS-MIT, Nov. 2007.
- [93] J. R. Norris. Markov Chains. Cambridge University Press, Cambridge, UK, 1997.
- [94] S. Oh, L. Schenato, and S. Sastry. A hierarchical multiple-target tracking algorithm for sensor networks. In *IEEE International Conference* on Robotics and Automation, pages 2197–2202, Barcelona, Spain, April 2005.
- [95] A. Okubo. Dynamical aspects of animal grouping: Swarms, schools, flocks and herds. Adv. Biophys., 22:1–94, 1986.

- [96] R. Olfati-Saber. Distributed kalman filter with embedded consensus filters. In Proceedings of the 44th IEEE Conference on Decision and Control, and European Control Conference, pages 8179–8184, December 2005.
- [97] R. Olfati-Saber. Ultrafast consensus in small-world networks. In American Control Conference, 2005.
- [98] R. Olfati-Saber. Flocking for multi-agent dynamic systems: Algorithms and theory. *IEEE Transactions on Automatic Control*, 51(3):401–420, 2006.
- [99] R. Olfati-Saber, J. A. Fax, and R. M. Murray. Consensus and cooperation in networked multi-agent systems. *IEEE Proceedings*, 95(1):215– 233, 2007.
- [100] R. Olfati-Saber, E. Franco, E. Frazzoli, and J. S. Shamma. Belief consensus and distributed hypothesis testing in sensor networks. In P.J. Antsaklis and P. Tabuada, editors, Network Embedded Sensing and Control. (Proceedings of NESC'05 Worskhop), volume 331 of Lecture Notes in Control and Information Sciences, pages 169–182. Springer Verlag, New York, 2006.
- [101] R. Olfati-Saber and R. M. Murray. Consensus problems in networks of agents with switching topology and time-delays. *IEEE Transactions* on Automatic Control, 49(9):1520–1533, 2004.
- [102] R. Olfati-Saber and J. Shamma. Consensus filters for sensor networks and distributed sensor fusion. In *In Proceedings of the 45th IEEE Conference on Decision and Control*, page 6698, 2005.
- [103] A. Olshevsky and J. N. Tsitsiklis. Convergence rates in distributed consensus and averaging. In *IEEE Conf. on Decision and Control*, pages 3387–3392, San Diego, CA, December 2006.
- [104] A. Olshevsky and J. N. Tsitsiklis. Convergence speed in distributed consensus and averaging. 2007. Special issue on "Control and Optimization in Cooperative Networks." Submitted.
- [105] J. K. Parrish, S. V. Viscido, and D. Grunbaum. Self-organized fish schools: An examination of emergent properties. *Biol. Bull.*, 202(3):296–305, 2002.

- [106] S. Patterson, B. Bamieh, and A. El Abbadi. Distributed average consensus with stochastic communication failures. In *IEEE Conf. on Deci*sion and Control, pages 4215–4220, New Orleans, LA, December 2007.
- [107] S. Patterson, B. Bamieh, and A. El Abbadi. Distributed average consensus with stochastic communication failures. *IEEE Transactions on Automatic Control*, 2008. Submitted.
- [108] G. Picci and T. Taylor. Almost sure convergence of random gossip algorithms. In *IEEE Conf. on Decision and Control*, pages 282–287, New Orleans, LA, December 2007.
- [109] M. Porfiri and D. J. Stilwell. Consensus seeking over random weighted directed graphs. *IEEE Transactions on Automatic Control*, 52(9):1767– 1773, 2007.
- [110] R.Carli, F. Fagnani, A. Speranzon, and S. Zampieri. Communication constraints in coordinated consensus problem. In *American Control Conference*, Minneapolis, USA, June 2006.
- [111] B. Recht and R. D'Andrea. Distributed control of systems over discrete groups. *IEEE Transactions on Automatic Control*, 49(9):1446–1452, 2004.
- [112] W. Ren and R. W. Beard. Consensus seeking in multi-agent systems under dynamically changing interaction topologies. *IEEE Transactions* on Automatic Control, 50(5):655–661, 2005.
- [113] W. Ren and R. W. Beard. Distributed Consensus in Multi-vehicle Cooperative Control. Springer, 2007.
- [114] W. Ren, R. W. Beard, and E. M. Atkins. Information consensus in multivehicle cooperative control: Collective group behavior through local interaction. *IEEE Control Systems Magazine*, 27(2):71–82, 2007.
- [115] W. Ren, R. W. Beard, and T. W. McLain. Coordination variables and consensus building in multiple vehicle systems. In V. Kumar, N. E. Leonard, and A. S. Morse, editors, *Cooperative Control*, volume 309 of *Lecture Notes in Control and Information Sciences*, pages 171–188. Springer Verlag, 2005.
- [116] A. Ribeiro, G. Giannakis, and S. Roumeliotis. SOI-KF: Distributed Kalman filtering with low-cost communications using the sign of innovations. *IEEE Transactions on Signal Processing*, 54(12):4782–4795, 2006.

- [117] L. Saloff-Coste. Random walks on finite groups. In Harry Kesten, editor, *Encyclopaedia of Mathematical Sciences*, pages 263–346. Springer, 2004.
- [118] A. Sarlette and R. Sepulchre. Consensus optimization on manifolds. 2007. Special issue on "Control and Optimization in Cooperative Networks." Submitted.
- [119] L. Scardovi, A. Sarlette, and R. Sepulchre. Synchronization and balancing on the N-torus. Systems & Control Letters, 56(5):335–341, 2007.
- [120] L. Schenato and G. Gamba. A distributed consensus protocol for clock synchronization in wireless sensor network. In *IEEE Conf. on Decision* and Control, New Orleans, USA, December 2007.
- [121] I. Schizas, A. Ribeiro, and G. Giannakis. Consensus in ad hoc WSNs with noisy links - Part I: Distributed estimation of deterministic signals. *IEEE Transactions on Signal Processing*, 56(1):350–364, 2007.
- [122] T. D. Seeley and S. C. Buhrman. Group decision-making in swarms of honey bees. *Behav. Ecology Sociobiol.*, 45(1):19–31, 1999.
- [123] E. Seneta, editor. Non-negative Matrices and Markov Chains. Springer Verlag, New York, 2 edition, 1981.
- [124] J.R. Silvester. Determinants of block matrices. *The Mathematical Gazette*, 84(501):460–467, 2000.
- [125] O. Simeone and U. Spagnolini. Distributed time synchronization in wireless sensor networks with coupled discrete-time oscillators. *EURASIP Journal on wireless sensor networks with coupled discretetime oscillators*, 2007.
- [126] A. R. Sinclair. The African Buffalo, A Study of Resource Limitation of Population. Chicago, IL: University of Chicago Press, 1977.
- [127] S. L. Smith, M. E. Broucke, and B. A. Francis. A hierarchical cyclic pursuit scheme for vehicle networks. *Automatica*, 41(6):1045–1053, 2005.
- [128] N. P. Sokolov. Quasispectral properties of p-dimensional matrices. Ukrainin Mathematical Journal, 19(4):437–448, 1967.
- [129] D. P. Spanos and R. M. Murray. Distributed sensor fusion using dynamic consensus. In In Proceedings of the 16th IFAC World Congress, 2005.

- [130] D. P. Spanos, R. Olfati-Saber, and R. M. Murray. Distributed kalman filtering in sensor networks with quantifiable performance. In *Proc*cedings of the Information Processing for Sensor Networks (IPSN'05), pages 133–139, 2005.
- [131] A. Speranzon, C. Fischione, and K. Johansson. Distributed and collaborative estimation over wireless sensor networks. In *Proceedings of* the IEEE Conference on Decision and Control (CDC'06), pages 1025– 1030, December 2006.
- [132] J. Speyer. Computation and transmission requirements for a decentralized linear-quadratic-gaussian control problem. *IEEE Transactions* on Automatic Control, 24(2):266–269, 1979.
- [133] K. J. Stewart and A. H. Harcourt. Gorillas vocalizations during rest periods-signal of impending departure. *Behaviour*, 130(1-2):29–40, 1994.
- [134] Y. Su, A. Bhaya, E. Kaszkurewicz, and V. Kozyakin. Further results on convergence of asyncronous linear iterations. *Linear Algebra and its Applications*, 281, 1998.
- [135] S. Sundaram and C. N. Hadjicostis. Distributed function calculation and consensus using linear iterative strategies. *IEEE Journal on Selected Areas in Communications*, 26(4):650–660, 2008.
- [136] R. Szewczyk, E. Osterweil, J. Polastre, M. Hamilton, A. M. Mainwaring, and D. Estrin. Habitat monitoring with sensor networks. *Communication of the ACM*, 46:34–40, 2004.
- [137] A. Tahbaz-Salehi and A. Jadbabaie. Consensus over random networks. *IEEE Transactions on Automatic Control*, 53:791 – 795, 2008.
- [138] J. N. Tsitsiklis. Problems in Decentralized Decision Making and Computation. PhD thesis, MIT, November 1984. Technical Report LIDS-TH-1424, Laboratory for Information and Decision Systems.
- [139] J. N. Tsitsiklis, D.P. Bertsekas, and M.Athans. Distributed asynchronous deterministic and stochastic gradient optimization algorithms. *IEEE Transactions on Automatic Control*, 31(9):803–812, 1986.
- [140] A. Valette. Graphes de Ramanujan. Astérisque, 245:247–276, 1997.

- [141] T. Vicsek, A. Czirók, E. Ben-Jacob, I. Cohen, and O. Shochet. Novel type of phase transition in a system of self-driven particles. *Physical Review Letters*, 75(6-7):1226–1229, 1995.
- [142] R. Viswanathan. A note on distributed estimation and sufficiency. *IEEE Transactions on Information Theory*, 39(5):1765–1767, 1993.
- [143] L. Wang and F. Xiao. Finite-time consensus problem for networks of dynamic agents. Technical Report arXiv:math/0701724v1, University of Peking, China, 2007.
- [144] J. Wolfe and J. Speyers. A low-power filtering scheme for distributed sensor networks. In In Proceedings of the 42th IEEE Conference on Decision and Control, page 6325, 2003.
- [145] C. W. Wu. Synchronization and convergence of linear dynamics in random directed networks. *IEEE Transactions on Automatic Control*, 51(7):1207–1210, 2006.
- [146] J.-J. Xiao and Z.-Q. Lou. Decentralized estimation in an inhomogeneous sensing environment. *IEEE Transactions on Information Theory*, 51(10):3564–3575, 2005.
- [147] L. Xiao and S. Boyd. Fast linear iterations for distributed averaging. Systems & Control Letters, 53(1):65–78, 2004.
- [148] L. Xiao, S. Boyd, and S.-J. Kim. Distributed average consensus with least-mean-square deviation. Journal of Parallel and Distributed Computing, 67(1):33-46, 2007.
- [149] L. Xiao, S. Boyd, and S. Lall. A scheme for asynchronous distributed sensor fusion based on average consensus. In *International Conference* on *Information Processing in Sensor Networks (IPSN'05)*, pages 63–70, Los Angeles, CA, April 2005.
- [150] M. E. Yildiz and A. Scaglione. Differential nested lattice encoding for consensus problems. In Symposium on Information Processing of Sensor Networks (IPSN), pages 89–98, Cambridge, MA, April 2007.
- [151] M. Zhu and S. Martínez. Dynamic average consensus on synchronous communication networks. In *American Control Conference*, Seattle, WA, July 2008. To appear.

[152] M. Zhu and S. Martínez. On the convergence time of distributed quantized averaging algorithms. In *IEEE Conf. on Decision and Control*, Cancun, Mexico, December 2008. Submitted.